i

TECHNICAL MANUAL

No. 11-5805-652-14

HEADQUARTERS DEPARTMENT OF **THE ARMY** WASHINGTON, DC, 31 October 1974

Operator's, Organizational, Direct Support and General Support Maintenance Manual

CENTRAL OFFICE TELEPHONE, ELECTRONIC AN/FTC-37(V)1 AND AN/FTC-37(V)2 (TE-400 EPABX)

	•	Paragraph	Page
PART A.	INTRODUCTION	0 1	
Section I.	General		
	Scope	1-1	vii
	Indexes of publications	1-2	vii
	Porms and records	1-5	vii
п	Reporting of errors	1-4	
11.	Description and use	15	vii
	Pulpose and use	- 1-J 5	vii
	Difference between models	1_7	vii
	Items comprising an operable equipment 1-8	8''	vii
PART ONE	SUPPLEMENT EPARX INSTALLATION AND MAINTENANCE DATA	-	
Section I			
Section 1.	General	_ 1.0	1-1
	Equipment description	2.0	1-1
	Teat equipment	_ 3.0	1-1
	Installation drawings	- 4.0	1-1
	Packaging data	_ 5.0	1-1
	Preparation for reshipment	6.0	1 - 1
II.	Pre-Instalation	1.0	2 1
	General	- 1.0	2-1
	Equipment capabilities and standard configuration	- 2.0	2-1
111.	Cabinet familiarization	1.0	2 1
	Preliminary installation procedure	- 1.0	2 1
	System cabinet description	3.0	2 1
117	Principal items locations	5.0	3-1
1 V.	Power-on connections and tests	.10	4-1
	Power/ground connections	- 2.0	4-1
	Power-off measurements	- 30	4-1
V.	Power-on adjustments	5.0	• •
	General	1.0	5-1
	Initial adjustment	2.0	5-1
XII.	General operational description		
	General	1.0	12-1
	(200-line) dc voltage generation	2.0	12-1
	(100-line) dc voltage generation	3.0	12-5
3/111	Three volt dc supplies	4.0	12-9
XIII.	Trouble detection and repair	1.0	10.1
	Totalayatam failure	1.0	13-1
	Portial system failure	2.0	13-1
DART TWO	SUDDI FWENT CONSOLE ODED ATION	5.0	13-1
TART TWO.	Console operation	- 1	1
	Multiattendant operation	2	4
THREE	BASIC EPABX INSTALLATION AND MAINTENANCE DATA		•
Section I.	scope		
	General	1.0	1-1
	Satellite drawings	2.0	1-1

Terms 10 2-1 Prinataliais planning 3.0 2-5 III. Cohinet familiarization 1.0 3-1 System cabinet(s) description 2.0 3-1 System cabinet(s) description 2.0 3-1 System cabinet(s) description 3.0 3-1 Sub-rack drawing number 4.0 3-2 Visual 4.0 3-2 Visual 3.0 4-1 Powerform connections 3.0 4-2 Intercabinet (cables 3.0 4-2 Intercabinet (cables) 6.0 4.0 Previnced line(s) 4.0 4.0 A-B leak connections 6.0 4.10 Previnced line(s) 9.0 4.33 PFCT connections with previned cabling 10.0 4.40 Unassigned night answer previned cabling 10.0 4.41 Other fastures strapping 13.0 4.41 Tunkite line loog adjustnet trisslor 13.0 4.41 Other fastures strapping 13.0 4.41 Other fastures strapping 10.0 5.1	Section II	Preinetellation	Paragraph	Page
Bujonent capabilities and studard configurations 2.0 2.5 III. Cabinet familiarization 3.0 2.5 III. Cabinet familiarization 1.0 3.1 System cabinet(s) description 2.0 3.1 System cabinet(s) description 2.0 3.1 Sub-rack drawing number 4.0 3.2 IV. Prevenfor connections and tests 1.0 3.1 Proverfor connections 3.0 3.1 Convertion downer cables 4.0 4.1 Powerfor connections 5.0 4.10 Prewind line(s) tip-fing connections 5.0 4.10 Prewind line(s) tip-fing connections 5.0 4.10 Prewind line(s) tip-fing connections 7.0 4.35 Prewind line(s) tip-fing connections 8.0 4.35 A B leads connection prewind cabling 10.0 4.41 Unassigned right answer prewind cabling 10.0 4.44 Tumk/tic line top-fing on distribution risitor 10.0 4.41 Tumk/tic line tops adjustment risitor 10.0 4.41 Tumk/tic line tops adjustment risitor 10.0 4.41	beetion n	UTICIAI	- 1.0	2 1
Preinstabilizion planning 3.0 2-5 III. Cabinet familiarizzation procedure 1.0 3-1 System cabine(3) description 2.0 3-1 Principal items locations 3.0 3-1 System cabine(3) description 4.0 3-2 Vincenzk drawing number 4.0 3-2 Vincenzk drawing number 1.0 4-1 Consoles) cable(3) connections 2.0 4-1 Consoles) cable(3) connections 3.0 4-1 Consoles) cable(3) connections 5.0 4-10 Prewired ine(4) rig connections 6.0 4-10 Prewired ine(5) rig connections 7.0 4-35 Prewired runksite lines tip-ring connections 7.0 4-35 Prewired runksite lines tip-ring connections 8.0 4-35 Installer wired cabling 9.0 4-35 PFCT connections with prewired cabling 10.0 4-40 Ubits address connection-prewired cabling 10.0 4-40 Other fautures strapping 10.0 4-41 Class of service assignment/cquipping 13.0 4-41 Tunk/tie lines loop		Equipment capabilities and standard configurations	2.0	2-4
11. Colore translarization 1.0 3-1 System cabinet(s) description 2.0 3-1 Principal items to cations 3.0 3-1 Sub-rack flarwing number 3.0 3-1 Sub-rack flarwing number 3.0 3-1 Principal items torins and tests 1.0 4-1 Power for connections and tests 3.0 4-1 Power formal connections 3.0 4-2 Intercabinet cables 4.0 4-2 Intercabinet cables 5.0 4-10 Prewired link(s) tip-fing connections 5.0 4-10 Prewired link(s) tip-fing connections 7.0 4-35 Prewired link(s) tip-fing connections 8.0 4-35 Installer wired cabling 10.0 4-4 Previred link induce cabling 10.0 4-4 Previred link induce cabling 10.0 4-4 Previred link induce cabling 10.0 4-4 Previred runks/site adapts 10.0 4-4 Previred might answer prewired cabling 10.0 4-4 Prunk is drassignnent/resistor 10.0	111	Preinstallation planning	3.0	2-5
System colume(s) description 10 3.1 Principal items locations and tests 40 3.2 IV. Power-off connections and tests 10 4.1 Power-off connections and tests 10 4.1 Prince connocitions and tests 10 4.1 Prover-off connections and tests 3.0 4.2 Intercalmet cables 4.0 4.1 Console(s) cable(a) connections 5.0 4.0 Prevince intrackite lines tip-ing connections 6.0 4.10 Prevince trunckite lines tip-ing connections 7.0 4.33 Prevince PFCT connections with previned cabling 9.0 4.35 PFCT connections with previned cabling 10.0 4.40 Other features strapping 10.0 4.41 Common control circuits tests 3.0 5.5 Junct (31604) und feature(s) circuit tests 3.0 5.6 Intriation adjustment test docation 7.0 5.0 Gener	111	Cabinet familiarization	1.0	2 1
Pineipal items locations 5.0 3-1 Sub-rack drawing number 40 3-2 IV. Power-off connections number 1.0 4-1 Power/off connections 2.0 4-1 Console(s) cable(a) connections 3.0 4.2 Intercabinet cables 4.0 4.6 Main distribution frame (MDP) layout and assignment (typical) 5.0 4-10 Prewired link(s) tip-ring connections 7.0 4-35 Prewired runk(s) lines up ring connections 7.0 4-35 Prewired runk(s) lines up ring connections 9.0 4-35 Installer wired cabling 0.0 4-40 Unassigned night answer prewired cabling 1.0 4-41 Other features strapping 1.0 4-41 Other features strapping 1.0 4-41 Console of service assignment/equipping 1.0 4-41 Tunk/tie lines loop adjustment resistor 1.0 4-41 Statilic equipment cabling 1.0 4-41 Console operation 1.0 5-1 Initial adjustments 1.0 5-1 Connecol operation		System cabinet(s) description	- 2 0	3-1 3-1
Sub-rick drawing number 4.0 3-2 IV. Power-off connections and tests 1.0 4-1 Power/ground connections 2.0 4-1 Console(s) cable(a) connections 3.0 4-2 Intercabinet cables 4.0 4.6 Main distribution frame (MDP) layout and assignment (typical) 5.0 4-10 Prewired Ine(s) tip-ring connections 6.0 4-10 Prewired Ine(s) tip-ring connections 7.0 4-35 Prewired Ine(s) tip-ring connections 8.0 4-35 Installer wired cabling 10.0 4-40 Unassigned night answer prewired cabling 10.0 4-41 Other features strapping 10.0 4-41 Other features strapping 15.0 4-41 Tunk/tie lines top adjustment resistor 15.0 4-41 Statellite cquipment cabling 10.0 4-56 General 1.0 5-1 Initial adjustments 2.0 5-1 General 1.0 5-1 Initial adjustments 2.0 5-1 Common control circuits tests 3.0 5-5 <td></td> <td>Principal items locations</td> <td>3.0</td> <td>3-1</td>		Principal items locations	3.0	3-1
IV. Power-off connections 1.0 4-1 Power/ground connections 3.0 4-2 Console(s) cable(a) connections 3.0 4-2 Intercabinet cables 4.0 4-6 Main distribution frame (MDP) layout and assignment (typical) 5.0 4-10 Prewired link(s) tip-ring connections 7.0 4-35 Prewired runks/tile lines (ip-ring connections connecting cable and trunk E-M and the system cabling 0.0 4-35 Previred runks/tile lines (ip-ring connection previned cabling 0.0 4-36 Previred runks/tile cabling 0.0 4-35 Previred runks/tile cabling 0.0 4-40 Unassigned night answer previred cabling 1.0 4-41 Other features strapping 15.0 4-41 Class of service asignment/equipping 15.0 4-41 Statellic equipment cabling 1.0 5-1 Initial adjustments 2.0 5-5 Common control circuits tests 5.0 5-7 Feature circuits test 6.0 5-7 Trunk via light read-out test 9.0 5-12 Comono control circuits tests 5.0 </td <td></td> <td>Sub-rack drawing number</td> <td>- 4.0</td> <td>3-2</td>		Sub-rack drawing number	- 4.0	3-2
Visual1.04-1Power/ground connections2.04-1Console(s) cable(a) connections3.04-2Intercabinet cables4.04.0Main distribution frame (MDP) layout and assignment (typical)5.04-10Prewired Inc(s) tip-ring connections6.04-10Prewired Inc(s) tip-ring connections8.04-35Prewired PFCT and miscellaneous connecting cable and trunk E-M and4.3A-B leads connections9.04-35PFCT connections with prewired cabling10.04-40Unassigned might answer prewined cabling10.04-41Other features strapping13.0441Class of service assignment/cquipping14.04-41Tunk/tie lines loop adjustment resistor15.0441Satellite cquipment cabling10.04-41V. Power-on adjustments2.05-1General1.05-1Initial adjustments2.05-1Common control circuit stest5.05-7Feature circuit (316024) test7.05-1Initial adjustments7.05-1Initial adjustments10.05-1Console operation8.05-12Intract (316024) test7.05-13Console operation8.05-12Tunk allotter (316034) test7.05-13Console operation (316057) test10.05-13Attendant tigt read-out test10.05-13Tunk (allotter (16034) test10.05-1	IV	. Power-off connections and tests		
Power ground connections104-1Console(s) cable(a) connections3.04-2Intercabinet cables4.04.0Main distribution frame (MDF) layout and assignment (typical)5.04.10Prewired trunks/telines up-ing connections6.04.10Prewired trunks/telines up-ing connections6.04.35Prewired PFCT and miscellaneous connecting cable and trunk E-M and8.04.35Prewired PFCT connections with prewired cabling0.04.40Unassigned night answer prewired cabling11.04.41Prinkit address connection-prewired cabling11.04.41Other features strapping13.04.41Class of service assignment/equipping15.04.41Trunkrite lines loop adjustment resistor15.04.41Statilite equipment cabling10.05-1Common control circuits tests3.05-5Junctrol (316040) and feature(s) circuit tests5.05-7Frature circuits test5.05-7Trunk Allotter (316031) test		Visual	- 1.0	4-1
Consoleta) Contections4.0Intercabine cables4.0Main distribution frame (MDF) layout and assignment (typical)5.04.10Prewired Irtuckytie lines tip-ring connecting cable and trunk E-M andA.B leads connections8.0A.B leads connections with prewired cabling10.04.41Unassigned night answer previned cabling10.04.42Unassigned night answer previned cabling10.04.43PFCT connections with prewired cabling10.04.41Public address connection-prewired cabling10.04.41Class of service assignment/equipping10.04.41Tunkrite lines loop adjustment resistor15.04.41Stabilitie equipment cabling10.04.41V. Power-on adjustments2.0General1.01.05.1Initial adjustments5.0General1.01.05.1Initial adjustments5.0Console operation5.0Console operation5.07.7Trunk allotter (316034) test7.05.9Console operation10.05.13Attendant line test7.05.13Attendant line test7.05.13Attendant line test7.05.13Attendant line test (31607) test7.05.13Attendant line test (31607)7.141.07.157.16 <td></td> <td>Power/ground connections</td> <td>- 2.0</td> <td>4 - 1</td>		Power/ground connections	- 2.0	4 - 1
Main distribution frame (MDF) layout and assignment (typical)5.04.10Prewired trunk/tie lines tip-ring connections604.10Prewired trunk/tie lines tip-ring connections604.10AB leads connections704.35Prewired trunk/tie lines tip-ring9.04.33Previred trunk/tie lines tip-ring9.04.33Previred trunk/tie lines tip-ring9.04.33Previred cabling10.04.40Unassigned right answer previred cabling10.04.41Connection-previred cabling11.04.41Chass of service assignment/equipping13.04.41Class of service assignment/equipping15.04.41Trunk/tie lines loop adjustment resistor15.04.41Power-ond measurements17.04.41Verwer-on adjustments10.05-1Initial adjustments2.05-1Connon control circuits tests3.05-5Juctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05-7Trunk allotter (316031) test9.05-10Trunk/tie line test9.05-12Toll resistor (316031) test11.05-13Attendant line test10.05-12Toll resistor (31607) test10.05-13Attendant line test10.05-13Trunk/tie line test10.05-13Attendant line test10.05-13Trunk test (316051) test procedure20.05-		Unisole(s) cables	3.0	4-2
Pewired line(i) tip-ing connections 60 4.16 Prewired line(i) tip-ing connections 60 4.35 Prewired PPCT and miscellaneous connecting cable and trunk E-M and 4.35 A-B leads connections 90 4.33 PPCT connections with prewired cabling 100 4.40 Unassigned night answer prewired cabling 100 4.41 Public address connection-prewired cabling 100 4.41 Public address connection-prewired cabling 100 4.41 Class of service assignment/equipping 14.0 4.41 Tunk/tie lines loop adjustment resistor 15.0 4.41 Satellite equipment cabling 10 4.41 Power-oft maximents 2.0 5-1 Common control circuit setst 3.0 5-5 Junctor (316024) test 6.0 5-7 Feature (circuits test) 6.0 5-7 Tunk allotter (circuit sets) 5.0 5-7 Tunk allotter (circuit sets) 5.0 5-7 Trunk allotter (circuit sets) 5.0 5-7 Trunk allotter (circuit sets) 0.0 5-12 Tunk all		Main distribution frame (MDF) layout and assignment (typical)	4.0	4-0
Prewired PFCT and miscellaneous connecting cable and trunk E-M and A-B leads connections with prewired cabling 9.0 4.35 PFCT connections with prewired cabling 9.0 4.35 PFCT connections with prewired cabling 10.0 4.40 Public address connections prewired cabling 10.0 4.41 Other features strapping 13.0 4.41 Class of service assignment/equipping 13.0 4.41 Trunk/tic lines loop adjustment resistor 15.0 4.41 Trunk/tic lines loop adjustment resistor 10.0 4.14 Power-off measurements 10.0 5.1 Initial adjustments 2.0 5-1 Cormon control circuits tests 3.0 5-5 Junctor (316024) test 5.0 5.7 Feature circuit (316040) and feature(s) circuit tests 5.0 5.7 Feature circuits test 9.0 5.10 Trunk/tie line test 0.0 5.12 Tunk/tie line test 0.0 5.12 Tunk dalotter (316031) test 10.0 5.12 Tunk dalotter (316031) test 10.0 5.12 Tunk/tie line test (316067) test 10.0<		Prewired line(s) tip-ring connections	6.0	4-10
Prewired PFCT and miscellaneous connecting cable and trunk E-M and A-B leads connections8.04.35A-B leads connections9.04.35PFCT connections with prewired cabling10.04.40Unassigned night answer prewired cable11.04.41Public address connection-prewired cable12.04.41Other features strapping13.04.41Class of service assignment/equipping14.04.41Trunk/tie lines loop adjustment resistor15.04.41Satellite equipment cabling16.04.41Power-off10.05.1Initial adjustments10.05.1General1.05.1Common control circuits tests3.05.5Junctor (316024) test5.05.7Feature circuit (16034) test5.05.7Trunk (16 line test9.05.10Trunk (16 line test9.05.10Trunk (16 line test10.05.12Toll resistor (316057) test10.05.12Attendant11.05.13Attendant11.05.13Power failure cucturing test10.05.12Toll resistor (316057) test10.05.13Console operation5.05.10Trunk (16051 and 316075)1.06.1City trunk test (16061 and 316075)1.06.1City trunk test (16061 and 316075)1.06.1The test (316051) test2.07.1Pre-cutover initial installation2.07.1<		Prewired trunks/tie lines tip-ring connections	7.0	4-35
A-B leads connections8.04.35Installer vired cabling10.04.40Unassigned night answer prewired cabling10.04.40Unbic address connection-prewired cabling11.04.41Pbbic address connection-prewired cabling12.04.41Other features strapping13.04.41Class of service assignment/cquipping14.04.41Trunk/tie lines loop adjustment resistor15.04.41Statilite equipment cabling16.04.41Power-on adjustments17.04.41Power-on adjustments2.05-1Initial adjustments2.05-1Common control circuits tests3.05-5Junctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05-7Feature circuit(316047) test7.05-9Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (316047) test11.05-13Attendant digit read-out test11.05-13Attendant digit read-out test13.05-13Trunk cuber (316031) test13.05-13Trunk cuber (316051) test13.05-13Power of failure cut-through test13.05-13Trunk test (316061 and 316075)10.06-1Tie test (316051 and 316075)10.06-1Tie test (316051 and 316075)10.06-1Cut over procedures2.06-2Direct inward dial and tie		Prewired PFCT and miscellaneous connecting cable and trunk E-M and	-	
Installer wird cabling		A-B leads connections	8.0	4-35
PrC1 connections with prewred cabing		Installer wired cabling	- 9.0	4-35
Public address connection-prewired cable 120 441 Other features strapping 130 441 Class of service assignment/equipping 150 441 Trunk/tie lines loop adjustment resistor 150 441 Satellite equipment cabling 160 441 Power-off measurements 170 441 V. Power-on adjustments 20 5-1 Common control circuits tests 20 5-1 Common control circuits tests 50 5-7 Trunk allotter (316024) test 60 5-7 Trunk allotter (316034) test 60 5-7 Trunk allotter (316034) test 90 5-10 Trunk/tie line test 90 5-12 Toll resistor (316067) test 100 5-12 Attendant line test 13.0 5-13 Trunk/tie line test (316067) test 13.0 5-13 Power failure cut-through test 13.0 5-13 Trunk conference accessing (316077) test 16.0 5-13 Power failure cut-through test procedure 5.0 6-5 Campon-busy (316031) test procedure		Unassigned night answer prewired cabling	- 10.0 - 11.0	4-40
Other features strapping13.04-41Class of service assignment/equipping14.04-41Trunk/tic lines loop adjustment resistor15.0441Satellite equipment cabling16.04-41Poweroffmeasurements10.05-1Initial adjustments2.05-1Common control circuits tests3.05-5Juncor (316024) test5.05-7Feature circuit (316044) test5.05-7Feature circuits test5.05-7Trunk allotter (316034) test7.05-9Console operation7.05-9Console operation7.05-9Tunk/tic line test9.05-10Tunk/tic line test9.05-12Toll resistor (316067) test11.05-13Campon-busy (316031) test11.05-13City tunk test (3160617) test13.05-13Power failure cut-through test16.05-13VI. Test procedures10.05-13City tunk test (316061 and 316075)106-1Tre tirct (316057) test		Public address connection-prewired cable	- 12.0	4-41
Class of service asignment/equipping14.04.41Trunk/tie lines loop adjustment resistor15.04.41Satellite equipment cabling16.04.41Power-on adjustments17.04.41V. Power-on adjustments2.05-1General2.05-5Lone circuit (316024) test2.05-6Line circuit (316040) and feature(s) circuit tests5.05.7Feature circuits test5.05.7Trunk allotter (316034) test7.05-9Console operation8.05-10Trunk (316077) test10.05-12Toll resistor (3160677) test10.05-13Attendant ligit read-out test11.05-13Trunk conference accessing (316077) test14.05-13Test procedures15.05-13City trunk test (316061) test rest test5.05-13Tie line test (316061) and 316075)06-1Tie line test (316061) and 316075)1.06-1Tie line test (316061) and 316075)1.06-1Tie line test (316061) and 316075)2.06-2Direct inward dial and tie trunk test procedure5.06-6VII. Cut-over procedures1.07-1Pre-cutover-initial installation2.07-1Pre-cutover initial adjustments of Section V completely, use Section VI as required3.0Repeat all test and procedures outlines in paragraphs 1.0, 2.0, and 4.0 of section VII3.0VII. System expansion procedures-130097 TE-400 cabinet(s)2.0<		Other features strapping	- 13.0	4-41
Trunk/tie lines loop adjustment resistor15.04.41Satellite equipment cabling17.04.41Power-offmeasurements17.04.41V. Power-on adjustments1.05-1General2.05-1Initial adjustments3.05-5Common control circuits tests3.05-5Junctor (316024) test5.05-7Feature circuits tests6.05-7Feature circuits tests6.05-7Console operation8.05-10Trunk allotter (316034) test9.05-12Toll resistor (316067) test10.05-12Toll resistor (316067) test10.05-13Campon-busy (316031) test12.05-13Attendant line test13.05-13Power failure cut-through test16.05-13Power failure cut-through test2.06-2Direct inwal dial and tie trunk test procedure2.06-6VI. Test procedures1.07-1City trunk test (316067) test1.06-13Toll resistrict (316067) test general1.07-1Pre-cutover fuelecommunication system in current operation)3.07-1Pre-cutover initial installation2.07-1Pre-cutover initial installation in system in current operation)3.07-1Pre-cutover initial installation in system in current operation)3.08-1Repeat all test and procedures outlines in paragraphs 1.0, 2.0, and 4.06-68-1VII. Sys		Class of service assignment/equipping	- 14.0	4-41
Satellite equipment cabling16.04-41Power-off measurements17.04-41V. Power-on adjustments1.05-1General1.05-1Initial adjustments2.05-1Common control circuits tests3.05-5Junctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05.7Feature circuits test7.05-9Console operation8.05-10Trunk / tile test9.05-12Toll resistor (316067) test11.05-13Campon-busy (316031) test11.05-13Attendant line test13.05-13Trunk conference accessing (316077) test14.05-13Restricted station-class of service test15.05-13VI. Test procedures1.06-110Cit y trunk test (316061 and 316075)1.06-1Tie line test (316067) test2.06-2Direct inward dial and tie trunk test procedureCity trunk test (316061 and 316075)1.06-1Tie line test (316057) test1.07-1Pre-cutover initial installation2.06-2Direct inward dial and tie trunk test procedureGeneral1.07-1Pre-cutover (telecommunication system in current operation)3.07-1Post-cutover1.07-1Pre-cutover (telecommunication system in current operation)-2.0Repeat all test and<		Trunk/tie lines loop adjustment resistor	15.0	4-41
Power-offmeasurements17.04-41V. Power-on adjustments1.05-1Initial adjustments2.05-1Common control circuits tests3.05-5Junctor (3)6024) test4.05-6Line circuit (3)6040) and feature(s) circuit tests5.05-7Feature circuits test6.05-7Trunk allotter (3)16034) test7.05-9Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (3)607) test10.05-13Attendant digit read-out test11.05-13Campon-busy (3)16031) test13.05-13Attendant linetest15.05-13Power failure cut-through test16.05-13VI. Test procedures10.06-1City trunk test (3)16051) and 316060)-2.06-2Direct inward dial and tie trunk test procedure-3.06-3Catu-over procedures1.07-17-1Pre-cutover initial installation2.07-1Pre-cutover (telecommunication system in current operation)3.07-1Precutover initial installation2.07-1Pre-cutover (telecommunication system in current operation)3.08-1Repeat all test and adjustments of Section V completely.3.08-1It is recommended that the installer remain in close proximity with the system for 2/hours after it is replaced in service5.08-1		Satellite equipment cabling	- 16.0	4-41
V. Power-on adjustments1.05-1General2.05-1Common control circuits tests3.05-5Junctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05.7Feature circuits test6.05-7Freature circuits test6.05-7Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (316067) test10.05-12Attendant ligit read-out test11.05-13Campon-busy (316031) test11.05-13Attendant ligit read-out test13.05-13Power failure cut-through test16.05-13VI. Test procedures16.05-13VI. Test procedures10.06-1Tie line test (316051 and 316060)2.06-2Direct inward dial and tie trunk test procedure-3.06-3Toll restrictor (316067) test-4.06-5Campon-busy (316031) test procedure3.06-3Toll restrictor (316067) test4.06-5Campon-busy (316031) test procedure0.06-1Tie line test (316051 and 316060)0.06-3Toll restrictor (316067) test3.06-36-3Campon-busy (316031) test procedure3.06-3Toll restrictor (316067) test<	V	Power-off measurements	17.0	4-41
Initial adjustments 2.0 5-1 Common control circuits tests 3.0 5-5 Junctor (316024) test 5.0 5.7 Feature circuit (316040) and feature(s) circuit tests 5.0 5.7 Trunk allotter (316034) test 6.0 5-1 Trunk allotter (316037) test 6.0 5-12 Toll resistor (316067) test 9.0 5-12 Toll resistor (316067) test 10.0 5-13 Campon-busy (316031) test - 12.0 5-13 Attendant line test - 12.0 5-13 Attendant line test - 12.0 5-13 City trunk conference accessing (316077) test 15.0 5-13 Power failure cut-through test 16.0 5-13 VI. Test procedures - - 3.0 6-3 Campon-busy (316051) test - - - 0 6-5 Campon-busy (316051) test - - - 0 6-5 City trunk test (316061 and 316075) 1.0 6-1 6-6 6 5 6 Campon-busy (316031) test procedu	v	General	1.0	5-1
Common control circuits tests5.055Junctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05.7Feature circuits test6.05.7Trunk allotter (316034) test7.05.9Console operation8.05.10Trunk/tie line test9.05.12Toll resistor (316067) test10.05.13Attendant digit read-out test11.05.13Campon-busy (316031) test11.05.13Trunk conference accesing (316077) test14.05.13Power failure cut-through test16.05.13VI. Test procedures16.05.13City trunk test (316061 and 316075)1.06-1Tie line test (316067) test		Initial adjustments	2.0	5-1
Junctor (316024) test4.05-6Line circuit (316040) and feature(s) circuit tests5.05-7Feature circuits test6.05-7Trunk allotter (316034) test7.05-9Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (316067) test9.05-13Camp-on-busy (316031) test11.05-13Camp-on-busy (316031) test11.05-13Trunk cline test11.05-13Restricted station-class of service test15.05-13Power failure cut-through test16.05-13VI. Test procedures10.06-1City trunk test (316061) and 316075)106-1Tie line test (316057) test-2.06-2Direct inward dial and tie trunk test procedure3.06-3Toll restrictor (316067) test4.06 - 5Campon-busy (316031) test procedure3.06-3Toll restrictor (316067) test4.06 - 5Campon-busy (316031) test procedure5.06-6VII. Cut-over procedures107-17-1Pre-cutover (telecommunication system in current operation)-3.07-1Post-cutover		Common control circuits tests	- 3.0	5-5
Line circuit (316040) and feature(s) circuit tests5.05.7Feature circuits test6.05.7Trunk allotter (316034) test7.05.9Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (316067) test10.05-13Attendant digit read-out test11.05-13Campon-busy (316031) test12.05-13Attendant line test13.05-13Restricted station-class of service test14.05-13Power failure cut-through test16.05-13VI. Test procedures10.06-1City trunk test (316051 and 316075)1.06-1Tie line test (316051 itest procedure		Junctor (316024) test	- 4.0	5-6
Feature circuits test6.05-7Trunk allotter (316034) test7.05-9Console operation8.05-10Trunk/tie line test9.05-12Toll resistor (316067) test10.05-13Attendant line test11.05-13Camp-on-busy (316031) test11.05-13Trunk conference accessing (316077) test14.05-13Restricted station-class of service test15.05-13Power failure cut-through test16.05-13VI. Test procedures10.06-1City trunk test (316061 and 316075)1.06-1Direct inward dial and tie trunk test procedure-3.06-3Toll restrictor (316067) test-4.06 - 5Camp-on-busy (316031) test procedure-5.06-6VII. Cut-over procedures-1.07-1Pre-cutover-initial installation3.07-1Pre-cutover-initial installation-2.07-1Pre-cutover (telecommunication system in current operation)3.07-1Pretiminary2.08-1Repeat all tests and adjustments of Section V completely. Use Section VI as required-3.08-1Repeat all tests and procedures outlines in paragraphs 1.0, 2.0, and 4.0 of section VII		Line circuit (316040) and feature(s) circuit tests	_ 5.0	5-7
I'unk allotter (\$16034) test7.05-10Console operation		Feature circuits test	6.0	5-7
Console Operation 8.0 5-12 Trunk/tile line test 9.0 5-12 Toll resistor (316067) test 11.0 5-13 Attendant digit read-out test 11.0 5-13 Campon-busy (316031) test 12.0 5-13 Attendant line test 13.0 5-13 Attendant line test 13.0 5-13 Restricted station-class of service test 15.0 5-13 Power failure cut-through test 16.0 5-13 VI. Test procedures 16.0 5-13 City trunk test (316061 and 316075) 10 6-1 Tie line test (316051 and 316075) 2.0 6-2 Direct inward dial and tie trunk test procedure		Concolo operation	/.0	5-10
Toll resistor (316067) test		Trunk/tie line test	9.0	5-12
Attendant digit read-out test 11.0 5-13 Camp-on-busy (316031) test - 12.0 5-13 Attendant line test - 12.0 5-13 Attendant line test - 13.0 5-13 Trunk conference accessing (316077) test - 14.0 5-13 Restricted station-class of service test 15.0 5-13 Power failure cut-through test 16.0 5-13 VI. Test procedures 10 6-1 6-1 Tie line test (316051 and 316060) - - 2.0 6-2 Direct inward dial and tie trunk test procedure - - 3.0 6-3 Toll restrictor (316067) test - - 6-6 6-5 Camp-on-busy (316031) test procedure - - 5.0 6-6 VII. Cut-over procedures 10 7-1 7-1 7-1 Pre-cutover-initial installation - - 3.0 7-1 Precutover-initial installation - 2.0 7-1 Pre-cutover (telecommunication system in current ope		Toll resistor (316067) test	10.0	5-12
Camp-on-busy (316031) test		Attendant digit read-out test	11.0	5-13
Attendantlinetest13.05-13Trunk conference accessing (316077) test		Camp-on-busy (316031) test	12.0	5-13
Trunk conference accessing (31607/) test		Attendant line test	13.0	5-13
Power failure cut-through test 15.0 5-13 VI. Test procedures 16.0 5-13 City trunk test (316061 and 316075) 1.0 6-1 Tie line test (316051 and 316075) 2.0 6-2 Direct inward dial and tie trunk test procedure - - 2.0 6-2 Direct inward dial and tie trunk test procedure - - 4.0 6 - 5 Camp-on-busy (316031) test procedure - - 5.0 6-6 VII. Cut-over procedures - - 5.0 6-6 VII. Cut-over procedures - - - 6-7 General - - - 0 7-1 Pre-cutover (telecommunication system in current operation) - 7-1 7-1 Post-cutover - - - 0 7-1 Post-cutover from in-use cabinet(s) _ - 2.0 8-1 Repeat all tests and adjustments of Section V completely. 0 8-1 8-1 Repeat all test and procedures outlines in paragraphs 1.0, 2.0 8-1 It is recommended that the installer remain in close prox		Trunk conference accessing (3160//) test	- 14.0	5-13
VI. Test procedures 1.0 6-1 City trunk test (316061 and 316075) 1.0 6-1 Tie line test (316051 and 316060) 2.0 6-2 Direct inward dial and tie trunk test procedure - - 3.0 6-3 Toll restrictor (316067) test - - 4.0 6 - 5 Camp-on-busy (316031) test procedure - - 5.0 6-6 VII. Cut-over procedures - - 5.0 6-6 General - - 3.0 7-1 Pre-cutover (telecommunication system in current operation) - 3.0 7-1 Pre-cutover (telecommunication system in current operation) - - 3.0 7-1 Pre-cutover (telecommunication system in current operation) - - 3.0 7-1 VIII. System expansion procedures-130097 TE-400 cabinet(s) - - 2.0 8-1 Repeat all tests and adjustments of Section V completely. - - 3.0 8-1 Use Section VI as required - - - 3.0 8-1 Repeat all test and procedures outlines in paragraphs 1.0,		Power failure cut through test	16.0	5-13
City trunk test (316061 and 316075)1.06-1Tie line test (316051 and 316060)	VI	. Test procedures	10.0	0 10
Tie line test (316051 and 316060)6-2Direct inward dial and tie trunk test procedure		City trunk test (316061 and 316075)	1.0	6-1
Direct inward dial and tie trunk test procedure		Tie line test (316051 and 316060)	- 2.0	6-2
10ll restrictor (316067) test		Direct inward dial and tie trunk test procedure	- 3.0	6-3
VII. Cut-over procedures		Toll restrictor (31606/) test	4.0	6 - 5
General	VII	Cut_over procedures	- 5.0	0-0
Pre-cutover-initial installation 2.0 7-1 Pre-cutover (telecommunication system in current operation) 3.0 7-1 Pre-cutover (telecommunication system in current operation) 4.0 7-1 VIII. System expansion procedures-130097 TE-400 cabinet(s) 1.0 8-1 Preliminary 1.0 8-1 Remove power from in-use cabinet(s) 2.0 8-1 Repeat all tests and adjustments of Section V completely. 3.0 8-1 Use Section VI as required 9000000000000000000000000000000000000	¥ 11	General	1.0	7-1
Pre-cutover (telecommunication system in current operation)		Pre-cutover-initial installation	2.0	7-1
Post-cutover 4.0 7-1 VIII. System expansion procedures-130097 TE-400 cabinet(s) 1.0 8-1 Preliminary 1.0 8-1 Remove power from in-use cabinet(s) 2.0 8-1 Repeat all tests and adjustments of Section V completely. 3.0 8-1 Use Section VI as required		Pre-cutover (telecommunication system in current operation),	- 3.0	7-1
VIII. System expansion procedures-130097 TE-400 cabinet(s) 1.0 8-1 Preliminary 1.0 8-1 Remove power from in-use cabinet(s) 2.0 8-1 Repeat all tests and adjustments of Section V completely. 2.0 8-1 Use Section VI as required		Post-cutover	4.0	7-1
Preliminary 1.0 8-1 Remove power from in-use cabinet(s) 2.0 8-1 Repeat all tests and adjustments of Section V completely. 2.0 8-1 Use Section VI as required	VIII	System expansion procedures-130097 TE-400 cabinet(s)	1.0	0 1
Repeat all tests and adjustments of Section V completely. 3.0 8-1 Repeat all test and procedures outlines in paragraphs 1.0, 2.0, and 4.0 of section VI 4.0 8-1 It is recommended that the installer remain in close proximity with the system for 72 hours after it is replaced in service 5.0 8-1		Preliminary	1.0	0-1 8-1
Use Section VI as required		Repeat all tests and adjustments of Section V completely	2.0	0-1
Repeatalltestandproceduresoutlinesinparagraphs1.0,2.0, and 4.0 of section VII4.08-1It is recommended that the installer remain in close proximity with the system for 72 hours after it is replaced in service5.08-1		Use Section VI as required	3.0	8-1
2.0, and 4.0 of section VII 4.0 8-1 It is recommended that the installer remain in close proximity with the system for 72 hours after it is replaced in service 5.0 8-1		Repeat all test and procedures outlines in paragraphs 1.0,		
It is recommended that the installer remain in close proximity with the system for 72 hours after it is replaced in service 5.0 8-1		2.0, and 4.0 of section VII	4.0	8-1
		It is recommended that the installer remain in close proximity with the system for 72 hours after it is replaced in service	5.0	8-1

		Paragraph	Page
Section IV	Decemination of achimet assembling 320138		
Section IA.	Description of capinet assembles betrev	1.0	9-1
	General	_ 2.0	9-1
Х.	Deinted circuit heard functional description		
	General	1.0	10-1
	Index of assemblies	2.0	10-1
	Assemblies general description	3.0	10-2
XI.	Printed circuit board interchangeability		
	General	1.0	
	Table 11-1 notes	2.0	11-/
XII.	General operational description	1.0	12.1
	General	1.0	12-1
	lime control mark generator	2.0	12-1
	Termination (station to station)	4.0	12-3
	Termination (station-to-station)	4.0	12-4
	PNPN matrix	5.0	12-5
	Dc voltage generation	60	12-5
	Three-volt dc supplies	7.0	12-7
	Attendant override	8.0	12-10
	Group hunt	9.0	12-10
	Single digit access	10.0	12-10
	Auxiliary junctor applique circuit (316079)	11.0	12-11
	Executive override (EOR)	13.0	12-11
	Trunk-tie line seizure	- 14.0	12-12
	Two-way city trunks	15.0	12-15
	Iwo-way DID loop tie trunk (loop signaling) and E&M tie trunk (E&M	16.0	12-19
	Two way tig line 21606()68 and 216051(68) Loop signaling and E&M	10.0	12 17
	signaling	17.0	12-24
	Consoleoperation	18.0	12-25
XIII.	Trouble detection and repair	10.0	
	Routine maintenance	1.0	13-1
	Total system failure	2.0	13-1
	Major alarms	3.0	13-2
	Minor alarms	4.0	13-2
	Station troubles	5.0	13-3
	Console troubles	. 0.0	13-4
VIV	Miscellaneous	. 7.0	15-0
AIV. VV	Clossery of terms		
PART FOUR	KEV SERVICE LINITS (LISED WITH AN/ETC_37(V)) AND AN/ETC_37(V)?)		
CHAPTER 1	K-501 SERIES KEY SERVICE UNITS		
Section 1.	General description		
	Introduction	- 1	4
	Features	- 2	4
	System components	- 3	4
	Description of key telephone system components	4	5
2	Operating instructions	3	5
2.	Installation	1	8
	Mounting	2	8
	Wiring	3	8
	Fauinment	- 4	9
	Power supply	5	9
	Installation	6	12
3.	Circuit diagrams	-	21
4.	Maintenance		
	General	- 1	30
	Maintenance aids	- 2	31
_	Relay maintenance	- 3	31
5.	Assembly and test	1	22
	Applicable decuments	1	33 22
	Application documents	- 2	33
	Ouality assurance provisions	4	36
	Zamit, assumed highlight		

iii

CHAPTER	Preparation for delivery Notes	Paragraph 5 6	Page 35 35
	Identification Application Installation Maintenance Replacement parts V 5040 KEN SERVICE UNITS	1.0 2.0 3.0 - 5.0 - 6.0	1 2 2 11 11
	3. K-584B AND K-584C KEY SERVICE UNITS Identification Installation Maintenance of K-584 type panels Replacement parts 4. K400(D)962 KEY TELEPHONE UNIT (KTU) CENTRAL OFFICE OR PBX LINE	1.0 3.0 - 4.0 - 5.0	2 3 17 17
	CIRCUIT FOR K-1A2 KEY TELEPHONE SYSTEM General	$-\frac{1}{2}{3}$	1 1 3
Part F	MULTI-KEY TELEPHONES General information Installation Expansion from Mine to 11-line capacity Explanation of code series in table I Disassembly and reassembly	$ \begin{array}{c} 1 \\ 2 \\ 3 \\ 4 \\ 5 \end{array} $	350.01 350.01 350.04 350.04 350.05
CHAPTER	37(V)2) 1. SPEAKERPHONE N-3 TYPE INSTALLATION AND MAINTENANCE General Installation Maintenance General purpose of circuit Circuit description Description of transistorized control unit Vario-lossers Power relay Reason for reissue 2. NE-3 Type Speakerphone System Control Units and Transmitter Units	$ \begin{array}{c} 1 \\ 2 \\ 3 \\ 4 \\ - 5 \\ 6 \\ - 7 \\ - 8 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3 \\ 3$	1 2 3 3 5 6 7
APPENDIX	Connections A. REFERENCES		A-1
Section	B. MAINTENANCE ALLOCATION I. Introduction II. Maintenance allocation chart		B-1 B-3

IDENTIFYING TECHNICAL PUBLICATION SHEET

I. IDENTIFICATION DATA

PURPOSE: This technical publication i's issued for the purpose of identifying an authorized commercial manual for Army use and for providing supplemental technical information hereto.

MANUFACTURER: ITT Telecommunications 2107 Swift Drive **Oak** Brook, Illinois 60521

Supplied underEquipment:Contract No.:Electronic TelephoneDAAB07-71-c-007Central OfficeDAAB07-72-C-0116AN/FTC-37(V)2DAAB07-72-C-0116

Electronic Telephone DAAB07-72-C-0103 Central Office AN/FTC-37(V)1

- AN/FIC-3/(V)I
- TITLE: ITT Telecommunications Electronic Private Automatic Branch Exchange Installation Maintenance Manual

II. SUPPLEMENTAL DATA

1. LIST OF AFFECTED PAGES IN BASIC MANUAL.

Section	No.	Page No.
Ι		1-1
ΙI		2-1, 2-2, 2-3, 2-4
III		3-1, 3-2, 3-6, 3-9
IV		4-1, 4-31
V		5-1, 5-2, 5-5, 5-10, 5-11, 5-12
XII		12-7, 12-9, 12-10
XIII		13-1. 13-2

a. SUPPLEMENTARY INFORMATION. The information contained in the above identified commercial manual is supplemented as follows:

- a. Introduction/Scope
- b. Pre-Installation
- c. Cabinet Familiarization
- d. Power-off Connection and Tests
- e. Power-On Adjustment
- f. General Operational Description
- p. Trouble Detection and Repair
- h. Console Operation

PART A

INTRODUCTION

Section I. GENERAL

1-1. Scope

a. This manual contains procedures for operating, installing and making Elecionic Telephone Central Office AN/FTC-37(V)1

(100-line Electronic Private Automatic Branch Exchange (EPABX) system) and AN/FTC [37(V)2 (200-line EPABX system). These procedures include operation, installation, and maintenance instructions for the K-500 Series Key Service Units (KSUs) and for the NE-3 speakerphone system, used with AN/FTC-37(V)1 and AN/FTC-37(V) 2. This manual also contains operating procedures for Control Monitor C-9263/FTC (Attendant's Console), used with AN/FTC-37(V)1, and Control Monitor (C-8958/FTC, the attendant's console used with AN/FTC-37(V)2.

b. For a listing of applicable publications referred to in this manual, refer to appendix A, References.

c. Refer to appendix B for the Maintenance Allocation Chart (MAC).

d. For identification of the component parts comprising the equipments covered in this manual, refer to TM 11-5805-652-24P, Repair Parts and Special Tools List (RPSTL).

1-2. Indexes of Publications

a. DA Pam 310-4. Refer to the latest issue of DA Pam 310-4 to determine whether there are new editions, changes, or additional publications pertaining to the equipment.

1-5. Purpose and Use

Refer to supplement section I, page 1-1, paragraph 1.0. Refer also to the two pages preceding page 1-1. Thes pages, unnumbered, are titled "Identifying Technical Publication Sheet" and "Introduction," **and are located in part** *ONE* **of** this manual (as indicated in the table of contents). **b.** DA Pam 310-7. Refer to DA Pam 310-7 to determine whether there are modification work orders (MWOs) pertaining to the equipment.

1-3. Forms and Records

a. Reports of Maintenance and Unsatisfactory **Equipment.** Maintenance forms, records, and reports which are to be used by maintenance personnel at all maintenance levels are listed in and prescribed by TM 38-750.

b. Report of Packaging and Handling Deficiencies. Fill out and forward DD Form 6 (Report of Packaging and Handling Deficiencies) as prescribed in AR 700-58/NAVSUP PUB 378/ AFR 71-4/MCO P4030.29, and DSAR 4145.8.

C. Discrepancy in Shipment Report (DIS-REP) (SF 361). Fill out and forward Discrepancy in Shipment Report (DISREP) (SF 361). as prescribed in AR 55-38/NAVSUPINST 4610.33/AFM 75-18/MCO P4610.19A, and DSAR 4500.15.

1-4. Reporting of Errors

Report of errors, omissions, and recommendations for improving this publication by the individual user is encouraged. Reports should be submitted on DA Form 2028, Recommended Changes to Publications and Blank Forms, and forwarded direct to Commander, US Army Electronics Command, ATTN: AMSEL-MA-CW, Fort Monmouth, NJ 07703.

1-6. Description

Section II. DESCRIPTION AND DATA

Refer to supplement section I, page 1-1, paragraph 2.0 (part ONE).

1-7. Difference Between Models

Refer to supplement section I, page 1-1, paragraph 2.0 (part ONE). Refer also to paragraph 1-3 below.

1-8. Items Comprising an Operable Equipm e n t

Refer to supplement section I, page 1-2, table 1-1, (part ONE), for a description of major equipments comprising the AN/FTC-37(V)1 and the AN/FTC-37(V)2. The AN/FTC-37(V)1 and the AN/FTC-37(V)2 and the major equipments comprising them are further described below. The used-on codes employed for identifying and

sociating these equipments are defined **as fol**-lows:

A-AN/FTC-37(V)1 B-AN/FTC-37(V)2 C-OA-8721/FTC37(V) D-OA-8658/FTC-37(V) E-OA-8659/FTC-37(V) F-OP-83/FTC-37(V) G-OP-71/FTC-37(V)

EQN Normanalatura		Mfr's	Used-On Code							
FSN	Qty Nomenciature (es)		Part No	A	B	С	D	E	F	G
5805-155-8029	1	Central Office, Telephone, Electronic AN/ETC-37(V)1	6287302							
5805-134-8419	1	Central Office, Telephone, Electronic AN/FTC-37(V)2	6287266							
5805-001-4071	1	Line Circuit and Matrix Switching Group OA-8721/ FTC-37(V)	130203-101 (modified)	x						
5805-134-8412	1	Line Circuit and Matrix Switching Group OA-8658/ FTC-37(V)	130097-1		x					
5805-136-9014	1	Line Circuit and Matrix Switching Group OA-8659/ FTC-37(V)	130097-1		x					
5805-155-8030	1	Power Supply Group OP-83/ FTC-37(V)	6287294	x						
5805-134-8415	1	Power Supply Group OP-71/ FTC-37(V)	6287263		x					
5805-155-8041	1	Control Monitor C-9263/FTC	316199(2)122	x						
5805-134-8414	1	Control Monitor C-8958/FIC	316199(2)122		x					
5830-139-1054		Power Supply Assembly	616300(1)122	x	x				х	х
5830-139-1055	1	Power Supply Assembly	616300(2)122	x	x				x	x
5830-139-1056	1	Power Supply Assembly	316300(3)122	x	x				х	x
5830-162-4449	10	Power Supply Assembly	310300(4)122	X	x				х	x
3830-138-4832 5830 150 5060	10	Line Circuit Assembly	310040(1)122			x	x	x		
5050-159-5909 5020 120 4020	10	Originate Control and Group	510040(2)122			х	х	x		
3030-130-4029	2	Generator Assembly	010029(1)122	1		x	х			
5805 147 5506	2	Originate Control and Group	316020(2)122							
3003-147-3390	2	Generator Assembly	510029(2)122					x		
5895-005-2463	3	Register Assembly	316302(1)122	1		~	v			
5805-160-8948	1	Group Hunt Assembly	316045(1)122			Ŷ	Ŷ	Â		
5805-160-8948	3	Group Hunt Assembly	316045(1)122			A	Â	v		
5839-138-4831	2	System Alloter Assembly	316039(1)122			x	x	î		
5805-150-1754	2	System Allotter Assembly	316039(3)122	1				x		
5805-168-4928	12	Junctor Assembly	316024(1)122	ł		x	x	x		
5830-138-4836	1	Override Attendant Assembly	316062(3)122			x	x			
5830-150-1747	1	Override Attendant Assembly	316062(4)122					х		
None Assigned	1	Miscellaneous Panel Assembly	130046-5			x				
	1		(modified)			:				
None Assigned	1	Miscellaneous Panel Assembly	130046-1				x			
New Assigned	1	Misseller and Denel Assembles	(modified)							
Non Assigned	1	Miscellaneous Panel Assembly	130046-2					x		
5830-138-4834	1	Class of Service Assembly	(modified) 316043(2)122							1
5830-153-9422	1	Class of Service Assembly	310043(2)122			x	x	x		1
5805-148-0674	2	Primary-Secondary Matrix	316065(16)122							
5005 170 00/ 1	2	Assembly	510005(10)122			×	x	x		
None Assigned	1	Tertiary-Quarternary Matrix Assembly	316041(11)122			x				
5805-148-0673	1	Tertiary-Quarternary Matrix Assembly	316041(12)122				x			

_

FSN Qt; (es		Novensleture	M6-2-	Used-On Code						
		Asiaticity	Part No	A	B	С	D	E	F	G
5805-148-0673	1	Tertiary-Quarternary Matrix Assembly	316049(12)122							
5830-139-1058	1	5V Supply and Alarms, Assembly	316301(1)122			x	x	x	x	
5830-139-1040	1	Main Alarm Voltage Check Assembly	316069(2)122			x	x		x	
5830-139-1041	1	voltage check Supply Assembly	316070(2)122			x	x		x	
5830-138-4838	1	System Main Alarm Assembly	316068(2)122			x	x			
5830-138-4830	2	Tone Supply Assembly	316037(1)122			х	х			
5830-138-4835	1	Public Address Assembly	316047(1)122			x	x			
5895-009-1113	1	Conference Assembly	316077(1)122			x	х			
None Assigned	1	Attendant% Control Assembly	316035(2)122			x	х			
5805-147-5595	1	Attendant% Trunk and Allotter Assembly	316073(2)122			x	x			
5830-150-1748	1	Attendant's Trunk Assembly	316099(2)122			х	х			
None Assigned	6	City Trunks Assembly	316075(4)122			х				
None Assigned	1	Trunk Allotter Assembly	316034(2)122			х				
None Assigned 1 Unassinged Night Answering Assembly		316036(6)122			x					
None Assigned 1 Assigned Night Answering Assembly		316042(1)122			x					
None Assigned	1	Series Voltage Regulator Assembly	316141(3)122			x				
None Assigned	1	Miscellaneous Panel Assembly	6287303			x				
None Assigned	1	Miscellaneous Panel Assembly	6287264				х			
5805-169-5017	1	Power Supply Assembly	316300(6)122					x	x	
5805-169-5018	1	Power Supply Assembly	316300(6)122					x	x	
5805-162-4444	1	Power Supply Assembly	316300(7)122					x	x	
5805-460-8820	1	Power Supply Assembly	361300(8)122			L		x	x	

_

PART ONE

SUPPLEMENT

EPABX INSTALLATION AND MAINTENANCE DATA

INTRODUCTION

PURPOSE OF SUPPLEMENT.

This supplement extends coverage of the ITT Telecommunications TE-400 EPABX Installation and Maintenance commercial manual. The purpose of the supplementary material provided IS twofold. First, it supplies additional material needed to expand the basic commercial manual. Secondly, it provides material to cover the No-Break Power modifications that have been incorporated into Electronic Telephone Central **Office** AN/FTC-37(V)2 (200-line **EPABX sys**tern) and Electronic Telephone Central Office AN/ FTC-37(V)1 (100-line EPABX system).

This supplement consists of two parts: Part 1, Supplement EPABX Installation and Maintenance Data, contains both additional data to supplement the basic manual and data covering the No-Break Power Modification; Part 2, Console Operation, contains additional data on console operation. Refer to Repair Parts and Special Tools List for component Parts Identification information.

USE OF SUPPLEMENT.

The 200-line **EPABX** system is similar to the 100-line EPABX system. This supplement covers both systems. Certain paragraphs in this supplement are applicable to the 200-line and/or 100-line EPABX

systems. The paragraph text is prefixed with system identification (200-line) 200-line EPABX system or (100-line) 100-line EPABX system as applicable. The **designation** assigned for a numbered paragraph will also apply to all sub-paragraphs unless otherwise indicated. The absence of the prefix designates that the paragraph is applicable to both 200-line and 100-line EPABX systems.

The contents of this supplement should be examined by all **users of EPABX** equipment to familiarize themselves with the supplemental data provided and also to ascertain what material in the basic manual has been augmented. The following sections in the basic manual have been supplemented:

Section I	scope		
Section II	Pre-Installation		
Section III	Cabinet Familiarization		
Section IV	Power-Off Connection and Test		
Section V	Power-On Adjustment		
Section XII	General Operational Description		
Section XIII	Trouble Detection and Repair		

Supplement

Section I CONTENTS

1.0	General.	Page 1 - 1
2.0	Equipment Description	1 - 1
3.0	Test Equipment	1-1
4.0	Installation Drawings	1 - 1
5.0	Packaging Data	1 - 1
6.0	Preparation For Reshipment · · · · · · · · · · · · · · · · · · ·	1 - 1

Tables

1-1	Official Nomenclature to Common Name	1-2
1-2	Test Equipment	1-2

Supplement Section I SCOPE

1.0 GENERAL

The installation and maintenance information provided herein supplements the installation and maintenance data provided in the basic EPABX commercial manual. In addition, material is provided to cover the No-Break Power modification. Table 1-1 lists the components of the EPABX, the official nomenclature, common name and manufacturer's part number.

2.0 EQUIPMENT DESCRIPTION

The components that make up the various configuration of EPABX are listed in table I-I. The differences between for example, a 100-line EPABX and a 200-line EPABX is basically in the No-Break Power system. Differences within the functional areas of the EPABX can be found within the appropriate EPABX configuration satellite drawings and in the basic commercial manual. The differences between the 200-line and 100-line EPABX No-Break Power system is discussed below:

(200 Line) The 200-line EPABX system with No-Break Power modification consists of three cabinets. Cabinet No. 1 is a basic unit with minor changes resulting from the No-Break Power modification. The changes involve the modification of the Miscellaneous Panel Assembly and the addition of an audible alarm to signify when a system main alarm occurs. Cabinet No. 2 is a basic unit with minor changes also made to the Miscellaneous Panel Assembly. Cabinet No. 3 contains the No-Break Power supplies. The No-Break Power modification, as described in this supplement, is not the same option described in the basic manual; that is, battery standby power is not provided. The No-Break Power modification consists of the addition of a third cabinet housing a duplicate set of power supplies as provided in cabinets 1 and 2. The standby power supplies are connected through control circuitry to the power bus bars as are the primary power supplies. If a primary power supply fails, control circuitry will automatically remove the defective supply from the bus bar and the standby power supply will assume the full load with no break in service.

(100 Line) The 100-line EPABX system with No-Break Power modification is similar in design to the 200-line EPABX system except that the 100-line EPABX No-Break power supply cabinet contains one set of power supplies. The power supplies used in both cabinets are identical to the ones used in the 200line EPABX system. The 100-line EPABX functions are all located in the Line Circuit and Matrix Switching Group OA-8721/FTC-37(V) (Cabinet 1). The changes made to this cabinet are the same as those &scribed for the 200-line EPABX. If a primary power supply in cabinet 1 fails, control circuitry will automatically *remove* the defective supply from the bus bar and the standby power supply in the No-Break cabinet will assume the full load with no break in service.

3.0 TEST EQUIPMENT

 Table 1-2 lists the test equipment required for direct support maintenance.

4.0 INSTALLATION DRAWINGS

Installation drawings supplied with the equipment are referenced in this supplement. Consult the Installation Specifications drawing, which is a part of the installation drawing package supplied with your equipment, for a list, by title and number,

with your equipment, for a list, by title and number, of all the installation drawings required. A basic complement of the types of drawings normally supplied in this installation drawing package are as follows:

- (a) Location and Plant Plan
- (b) Equipment Layout and Installation Details
- (c) Cable Carrier Layout and Installation Details
- (d) Cabling Diagram
- (e) Distribution Frame Layout
- (f) Wire Run List
- (g) Consolidated List of Materials

A specific installation drawing will be referenced by drawing type only. Consult the Installation Specifications & awing list for the drawing number applicable to your facility.

5.0 PACKAGING DATA

The EPABX is packaged in accordance with specification MIL-P-116, Method 11. The equipment cabinets are packaged in individual woo&n shipping crates. Corners and sharp edges of cabinets are covered with cushioning material to protect the vaporproof barrier.

6.0 PREPARATION FOR RESHIPMENT

Repackaging of equipment for shipment or limited storage will normally be performed at a packaging facility or by a repackaging team. Should emergency packaging be required, select the materials from those listed in SB 38-100, Preservation, Packaging, and Packing Materials, Supplies, and Equipment Used by the Army. Package the equipment in accordance with the original packaging, so far as possible, or with the available materials.

Official Nomenclature	Common Name	Mfr. Part No.	Cabinet No.
Central Office, Telephone, Electronic AN/FTC-37(V)2 consisting of:	200-Line EPABX	6287266	
Line Circuit and Matrix Switching Group OA-8658/FTC-37(V)	Matrix Switching Group	130097-1	1
Line Circuit and Matrix Switching Group OA-8659/FTC-37(V)	Switching Group	130097-1	2
Power Supply Group OP-71/FTC- 37(V)	No-Break Cabinet	6287263	3
Control-Monitor C-8958/FTC	Console	316199(2)122	N/A
Central Office, Telephone, Electronic AN/FTC-37(V)1 consisting of:	100-Line EPABX	6287302	
Line Circuit and Matrix Switching Group OA-8721/FTC-37(V)	Matrix Switching Group	130203-101 (Modified)	1
Power Supply Group OP-83/FTC- 37(V)	No-Break Cabinet	6287294	2
Control-Monitor C-9263/FTC	Console	316199(1)122	N/A

Table 1-1. Official Nomenclature to Common Name

Table 1-2. Test Equipment

Test Equipment	Mfr. Part No.	Nomenclature or FSN
Multimeter	Simpson 260	AN/USM-210
Oscilloscope	HP120B	6625-016-1874
Card Extender	130587-1	
Card Extender	130587-2	
Card Extender	130587-3	
Card Extender	130680-1	
Hand Test Telephone	Automatic Elec. L-9066-CF	
Cord and Plug Assy.	Automatic Elec. D-543142-A	

Section II

C O N T E N T S

1.0	General.		2-1
	1.1	AC Input Per No-Break Cabinet	2-1
	1.2	AC Power Protection Per No-Break Cabinet	2-1
	1.3	External AC Cabling Per Bay	2-1
	1.4	External DC Power Requirements	2-1
	1.5	Grounding	2-1
2.0	Equip	ment Capabilities and Standard Configuration	2-1
	2.1	(200 line) Power Equipment.	2-1
	2.2	(100 line) Power Equipment.	2 - 1

Section II

PRE-INSTALLATION

1.0 GENERAL

- 1.1
 AC Input Per No-Break Cabinet

 (200 Line, Cabinet No. 3)
 (a) 115VAC, 60 Hz, 30 Amp

 (100 Line, Cabinet No. 2)
 (b) 220VAC, 50 Hz, 15 Amp
- 1.2 AC Power Protection Per No-Break Cabinet
 - (200 Line, Cabinet No. 3) (a) 30 Amp circuit breaker per 115VAC input
 - (100 Line, Cabinet No. 2) (b) 15 Amp circuit breaker per 220VAC input
- 1.3 External AC Cabling Per Bay

Refer to **appropriate** Installation **drawing**.

- 1.4 External DC Power Requirements Not applicable.
- 1.5 Grounding

Refer to Section III, **Fig.** 3-1, in basic manual **and** appropriate Installation drawing.

2.0 EQUIPMENT CAPABILITIES AND STANDARD CONFIGURATION

2.1 (200 Line) Power Equipment

The power supply equipment provided consists of a primary and a No-Break back-up secondary arrangement.

2.1.1 Primary Power Supply (316300)

The primary power supply equipment consists of four individual power modules mounted vertically at the extreme left side of cabinets 1 and 2 at levels D-G. The power modules in cabinet 1 are cross-connected to modules of similar output in cabinet 2. The power modules in cabinet 1 are the master units while those in cabinet 2 are the slave units. All voltage adjustments are made in the master units. The combined outputs of master and slave modules are applied to common bus bars in cabinets 1 and 2 through two power relays. Refer to System Power and Miscellaneous Frame Circuit (Mod) drawings.

2.1.2 Ho-Break Power Supply (316300)

The No-Break back-up power **supply equi**pment consists of eight individual power **modules** mounted in cabinet 3. Four modules(master units) are mounted vertically at the left at levels D-G while four modules (slave units) are mounted vertically at the right also at levels D-G. The modules at the left are cross-connected to the modules at the right with similar outputs. The combined outputs of master and slave modules are applied to commonbus bars in cabinets 1 and 2 through two power relays located in cabinet 3. Refer to System Power and Miscellaneous Frame Circuit (Mod) drawings.

2.2 (100 Line) Power Equipment

The power supply equipment provided consists of a primary and a No-Break back-up secondary arrangement.

2.2.1 Primary Power Supply (316300)

The primary power supply equipment is the same as described in paragraph 2.1.1 and is also located in cabinet 1. The difference being that the power supply outputs are cross-connected to modules of similar outputs in the No-Break cabinet. The outputs of cabinet 1 modules are applied to a common bus bar in cabinet 1 through a power relay. Refer to System Power and Miscellaneous Frame Circuit (Mod) drawings.

2.2.2 No-Break Power Supply (316300)

The No-Break back-up power supply equipment consists of four individual power modules mounted in the No-Break cabinet (cabinet 2). The four modules are mounted vertically at the left at levels D-G. The outputs of the modules are applied to common bus bars in cabinet 1 through a power relay located in cabinet 2. Refer to System Power and Miscellaneous Frame Circuit (Mod) drawings.

Section III CONTENTS

Page

1.0	Preliminar	y Installation Procedure	3-1
2.0	System	Cabinet Description	3-1
	2.1	200-Line Cabinets No. I and 2 and 100-Line Cabinet No. 1 • • • • •	3-1
	2.2	No-Break Power cabinet · · · · · · · · · · · · · · · · · · ·	3-1
3.0	Principa	al Items Locations	3-1
	3.1	System Test Panel. • • • • • • • • • • • • • • • • • • •	3-1
	3.2	No-Break Power Cabinet, AC Power Source, DC Power Supplies and Blower Locations	3-1
	3.3	(200 Line) Miscellaneous Panel Assembly	3-1
	3.4	(100 Line) Miscellaneous Panel Assembly.	3-6
	3.5	No-Break Cabinet Horizontal and Vertical Bus Bars	3-6
	3.6	No-Break Cabinet Printed Circuit Board Locations	3-6

FIGURES

3-1	200-Line EPABX No-Break Power Configuration	3-2
3-2	100-Line EPABX No-Break Power Configuration	3-3
3-3	200-Line EPABX, Cabinet 1 Miscellaneous Panel Assembly.	3-4
3-4	200-Line EPABX, Cabinet 2 Miscellaneous Panel Assembly	3-5
3-5	200-Line EPABX, No-Break Cabinet Miscellaneous Panel Assembly.	3-7
3-6	100-Line EPABX, Cabinet 1 Miscellaneous Panel Assembly	3-8
3-7	100-Line BPABX, No-Break Cabinet Miscellaneous Panel Assembly	3-9

Section III

1.0 PRELIMINARY INSTALLATION PROCEDURE

Place equipment in the assigned 1.1 applicable place. Refer to Fig. 3-1 in the basic manual and applicable Equipment Layout and Installation Detail drawings.

Unpack the equipment and inventory. Use applicable Consolidated List of Materials drawing to check items.

2.0 SYSTEM CABINET DESCRIPTION

The first and second cabinet of the 200-line EPABX system and the first cabinet of the 100-line EPABX system is described in the basic manual.

Figure 3-1 illustrates the No-Break Power configura-tion for the 200-line EPABX cabinet No. 3. Figure 3-2 illustrates the No-Break Power configuration for the 100-line EPABX cabinet No. 2. Only levels C through G of these cabinets are used. Level C is designated Attendant Control Sub-Back. However, this level contains only three printed circuit boards in the No-Break Power configuration. The DC power supplies are equipped on levels D through G.

3.0 PRINCIPAL ITEMS LOCATIONS

System Test Panel 3.1

The No-Break Power cabinets are equipped with a modified version of the System Test Panel illustrated in Fig. 3-8 of the basic manual. The No-Break cabinets have only a voltage check pushbutton (labeled V. C. RESET).

3.2	No-Break Power Cabinet, AC
	Power Source, DC Power Supplies
	and Blower Locations

Remove rear door and face rear. Refer to No-Break Power Cabinet assembly drawing.

> The blower input and filter input 3.2.1 are located at the bottom left.

NOTE: Blockage of blower input will remove power from the system.

The AC power transformers and 3.2.2 their associated capacitors are positioned at the bottom right.

3.2.3 The AC power cord access entrance is located between the blower input and AC power transformer.

3.3 (200 Line) Miscellaneous Panel Assembly (Fig. 3-3, 3-4, and 3-5)

Facing the rear of a cabinet, the miscellaneous panel assembly is on the inside right.

Letters A, B, C etc., have been assigned to individual **items of** the panel assembly. The items are identified below:

- A AC Circuit Breaker for
- Cabinets 1 and 2 B - VC Transformer Fuse, 5VDC Supply, Blower Fuses, and 3 VDC Power Relay
- C Spare Fuses D Voltage Check Transformer
- E Power Relay F Ring Generator
- G Ring Generator Suppression Assembly, and Diodes CRA3, CRA4
- H PFCT1, PFCT2, Auxiliary Relay, and RL1 J - MC1 through MC5 (5VDC
- Supply) K MB1 (5VDC Supply Rectifier), and Diodes CR5A, CR5B
- Q1, MR1, and MCB (5VDC
- M TE Resistor Assembly and RV1, RV2, SCR1, RV21, **RV22**
- N TE Resistor Assembly PCB
- O Miscellaneous Panel Assembly, Connector Block
- 3.3.2 Cabinet 2 Miscellaneous Panel Assembly (Fig. 3-4)

Letters A, B, C etc., have been assigned to individual items of the panel assembly. The items are identified below:

- A Not Used B VC Transformer Fuse, 5VDC
 - Supply, and Blower Fuses
- C Spare Fuses D - Voltage Check Transformer
- E Power Relay
- F Ring Generator



Figure 3-1. 200-Line EPABX No-Break Power Configuration.

3 - 2



Figure 3-2. 100-Line EPABX No-Break Power Configuration



Figure 3-3. 200-Line EPABX, Cabinet 1 Miscellaneous Panel Assembly.



Figure 3-4. 200-Line EPABX, Cabinet 2 Miscellaneous Panel Assembly.

G - Ring Generator Suppression

- Assembly H- PFCT1, PFCT2, 3VDC Power Relay, aid RL1
- J Not Used K- Diode CR5C L Q1, MR1, and MCB (5VDC Supply)
- Y- Not Used
- N- Not Used O Miscellaneous Panel Assembly, Connector Block
- cabinet 3 Miscellaneous Panel 3.3.3 Assembly (Fig. 3-5)

Letters A, B, C, etc., have been assigned to indivi-dual items of the panel assembly. The items are identified below:

- A AC Circuit Breaker B VC Transformer Fuse, 5VDC Supply, and Blower Fuses
- C Spare Fuses D Voltage Check Transformer E Power Relay F Power Relay

- G Diodes CRA1, CRA2 H Relay No. 2 J MC1 through MC5 (5VDC
- S u p p l v K MB1 (5VDC Supply Rectifier) L Q1, MR1, and MCB (5VDC
- M MR2, CR5D, CR5E, CR5F,
- SCR1 O2
- Miscellaneous Panel Assembly, Connector Block
 Diode CRB1 0
- Ρ
- 3.4 (100 Line) Miscellaneous Panel Assembly (Fig. 3-6 and 3-1)

Facing the rear of a cabinet, the miscellaneous panel assembly is on the inside right.

> Cabinet 1 Miscellaneous Panel 3.4.1Assembly (Fig. 3-6)

Letters A, B, C, etc., have been assigned to indivi-dual items of the panel assembly. The items are identified below:

- A AC Circuit Breaker for Cabinet 1
- VC Transformer Fuse, 5VDC Supply, Blower Fuses, and 3VDC Power Relay
- C Spare Fuses

D - Voltage Check Transformer

- E Power Relay

- F- Ring Generator G- Diodes CRA3, CRA4 H PFCT1, PFCT2, Auxillary Relay, and RL1 - MC1 through MC5 (5VDC
- I
- **K** MB1 (5VDC Supply Rectifier) and Diodes CR5A, CR5B L Q1, MR1, and MCB(5VDC
- Supply) M- TE Resistor Assembly and RV1, RV2, SCR1, RV21, **RV22**
- N TE Resistor Assembly PCB O Miscellaneous Panel Assem-
- bly, Connector Block
- 3.4.2 Cabinet 2 Miscellaneous Panel Assembly (Fig. 3-7)

Letters A, B, C, etc., have been assigned to indivi-dual items of the panel assembly. The items are identified below:

A - AC Circuit Breaker B - VC Transformer Fuse, 5VDC Supply, and Blower Fuses C- Spare Fuses D - Voltage Check Transformer E - Power Relay F - Diodes CRA1, CRA2 G - Relay No. 2 (AUK) H - MC1 through MC5 (5VDC) J - MB1 (5VDC Supply Rectifier) K - Q1, IR1, and MCB(5VDC - CR5D CR5E, SCR1 M - Miscellaneous Panel Assembly, Connector Block N - Diode CRB1

No-Break Cabinet Horizontal and Vertical Bus Bars 3.5

Refer to basic manual for description of the No-Break Cabinet horizontal and vertical bus bars.

No-Break Cabinet Printed Circuit 3.6 Board Locations

For location of PCB's on 200-line No-Break cabinet, see Fig. 3-1, on 100-line No-Break cabinet, see Fig. 3-2, **Each** PCB is inserted into the appropriate female connector and each has a distinct plug arrangement that prevents inadvertent insertron of the PCB into an incorrect position.



Figure 3-5. ZOO-Line EPABX, No-Break Cabinet Miscellaneous Panel Assembly



Figure 3-6. 100-Line EPABX, Cabinet 1 Miscellaneous Panel Assembly.



3-9

S u p p l e m e n t

Section IV

CONTENTS

Page

1.0	General	4-1
2.0	Power/Ground Connections	4-1
3.0	Power-Off Measurements	4-1

Supplement

Section IV

POWER-OFF CONNECTIONS AND TEST

1.0 GENERAL

Refer to the material in the basic manual and the Installation Drawings listed in Section I of this supplement for equipment interface information.

2.0 POWER/GROUND CONNECTIONS

No internal AC power input and power multiple cable connections are necessary.

These connections have already been accomplished **Perform external ground table connection as** described in the applicable Installation Specificat^{ion} drawing.

3.0 POWER-OFF MEASUREMENTS

Perform power-off measurements as described in basic manual.

Supplement Section V CONTENTS

Page

1.0	Gen	e r a l	5-1
2.0	Initial A	Adjustments	5-1
	2.1	AC Power On	5-1
	2.2	Blower Operation	5-1
	2.3	DC Power Supplies Access	5-1
	2.4	DC Power Supplies Adjustments.	5-1
	2.5	+3VDC and -3VDC Adjustments	5-1
	2.6	(260 Line) 5VDC Power Supply Adjustment	5-1
	2.7	(100 Line) 5VDC Power Supply Adjustment	5-1
	2.8	Power Removal	5-1
	2.9	Reapplying AC Power	5-1
	2.10	Voltage Check and +5VDC Adjustments	5-1
	2.11	±3VDC Alarm Adjustments	5-3
	2.12	AC Power On-Off Test	5-4

Supplement

Section V

POWER-ON ADJUSTMENTS

1.0 GENERAL

Refer to material in basic manual and material in this section to accomplish **powe** adjustments. Figure 5-1, 5-2 and 5-3 referenced throughout this section can be found in the basic manual

2.0 INITIAL ADJUSTMENTS

2.1 AC Power On

Apply AC power by:

- (a) Connect AC power cords to AC source
- Position AC circuit breakers (b) in cabinet 1 and No-Break cabinet to ON.

2.2 Blower Operation

Insure that blowers are in operation by checking each air output (lower front).

DC Power Supplies Access 2.3

Access to the DC power supplies (for adjustment) is achieved by lifting the ventilation assembly up and swinging assembly to left.

DC Power Supplies Adjustments 2.4

Refer to Fig. 5-2 for test point (TP) and adjustment locations. (On the 200-line (TP) and adjustment locations. (On the 200-line EPABX system only, the master power supplies in cabinet 1 and No-Break cabinet are adjustable; slave power supplies in cabinet 2 are not adjustable. On the 100-line EPABX system, the power supplies in cabinet 1 and the No-Break cabinet are adjustable.) Using a calibrated 20,000 ohm per volt VOM, adjust the cabinet power supplies (-18VDC, +18.5VDC, +36 VDC, and -36VDC power supplies). Repeat the ad-justment procedure for the power supplies located in the No-Break cabinet. Any convenient chassis point may be used for VOM ground connectron. Place VOM on appropriate DC scale for accurate reading. Reon appropriate DC scale for accurate reading. Replace DC power supply ventilation assembly to closed position.

- 2.5
- +3VDC and -3VDC Adjustment

Adjust the +3VDC and -3VDC supplies in each of the cabinets. Refer to Fig. 5-2 for test point (TP) and Adjustment locations.

2.6 (200 Line) 5VDC Power Supply Adjustment

2.6.1 Refer 1 & Fig. 3-3 supplement section III. Insert cabinet 1 +5 VDC supply and alarm PCB into its connector. Con nect VOM positive lead to **bottom terminal of power** resistor 1R1. Connect VOM negative lead to chassis ground. Adjust 5VDC power supply for +5VDC by adjusting the potentiometer located on the bottom front of the +5VDC supply and alarm PCB.

2.6.2 Refer to Fig. 3-5 of supplement Section III. Insert cabinet 3 +5VDC supply and alarm PCB into its connector. Repeat adjustment procedure outlined in paragraph 261

2.7 (106 Line) 5VDC Power Supply Adjustment

2.7.1 Refer to Fig. 3-6 of supplement Section III and repeat the proce-dure given in paragraph 2.6.1 for cabinet 1.

2.7.2 Refer to Fig. 3-7 of supplement Section III and repeat the proce-dure given in paragraph 2.6.1 for the 100-line No-Break cabinet.

2.8 **Power Removal**

2.8.1 Remove AC power <u>before</u> insert-ing PCB's by placing AC circuit breakers in cabinet 1 and No-Break cabinet to OFF.

Insert the following PCB's in cab-inet 1 and No-Break cabinet: 2.8.2

(a) Voltage check reference 316070()122
(b) Voltage check main alarm

316069()122 (c) System main alarm (cabinet 1 only) 316068(2)122

2.9 **Reapplying AC Power**

2.10

Return AC power by placing AC circuit breakers in cabinet 1 and 2.9.1 No-Break cabinet to ON.

> Voltage Check and +5VDC Adjustment

The power relay $(PR)_{\mbox{may}}$ or may not operate when the AC power is reapplied. The alarm lamps on the

316069 PDB may or may not be "on." Location of **alarm** lamps are shown on Fig. 5-1 and 5-3.

2.10.1 PR relay operates-Lamps on 316069 and Hi-Lo lamps on 316301 are extinguished.

NOTE: If PR does not operate, **proceed to paragraph** 2.10.2

2.10.1.1 Using a calibrated VOM, measure the outputs of the Voltage Check PCB (316670). Each output may be measured from the test point at the front of the PCB. Each TPL identified, All voltages have a ±16% tolerance.

2.10.1.2 Main Alarm Voltage Check - Retfer to Fig. 5-2 and 5-3. Place AC circuit breaker in No-Break cabinet to OFF.

NOTE: The power supplies in the No-Break cabinet must be disabled to permit the high and low adjustments of cabinet 1 power supplies.

Hi-Lo Voltage Detection Adjustments

- (a) Connect a calibrated DC voltmeter (use appropriate scale) to the test jack of the +36 volt power supply.
- (b) Adjust the +36 volt power supply for an output of 39.6 volts.
- (c) Press the voltage check reset pushbutton while rotating the +36 volt high "potentiometer adjust" (second from bottom of main alarm voltage check PCB) clockwise until the top lamp is extinguished.
- (d) Slowly rotate the "potentiometer adjust" counterclockwise until the associated lamp comes "on"
- (e) Return the +36 volt power supply output to +36 volts.
- (f) Press the voltage check reset pushbutton to extinguish the lamp.
- (g) Adjust the +36 volt power supply for an output of 32.4 volts.
- (h) Press the voltage check reset pushbutton while rotating the +36 volt low "potentiometer adjust" (top of PCB)

counterclockwise until the associated lamp is extinguished.

- (i) Slowly rotate the "potentiometer adjust" clockwise until the lamp comes "on, "
- (j) Return the +36 volt power supply output to +36 volts.
- (k) Press the voltage check reset pushbutton to extinguish the lamp.
- (1) Repeat Steps (a) through (k) using the -36 volt power supply, the -36 volt low potentiometer and lamp (third from top of board), and the -36 volt high potentiometer and lamp (fourth from bottom of board).
- (m) Connect the DC voltmeter to the test jack of the +18.5 volt power supply.
- (n) Adjust the +18.5 volt power supply for an output of 20.4 volts.
- (o) Press the voltage check reset pushbutton while rotating the +18.5 volt high "potentiometer adjust" (first from bottom of board) clockwise until the associated lamp is extinguished.
- (p) Slowly rotate the "potentiometer adjust" counterclockwise until the lamp comes 'on."
- (g) Return the +18.5 volt power supply output to +18.3 volts.
- (r) Press the voltage check- reset pushbutton to extinguish the lamp.
- (s) Adjust the +18.5 volt power supply for an output of 16.6 volts.
- (t) Press the voltage check reset pushbutton while rotating the +18.5 volt low "potentiometer adjust" (fourth from top of board) counterclockwise until the associated lamp is extinguished.
- (u) Slowly rotate the "potentiometer adjust" clockwise until the lamp comes "on."

- (v) Return the +18.5 volt power supply output to +18.5 volts.
- (w) Press the voltage check reset pushbutton to extinguish the lamp.
- (x) Repeat steps (m) through (w) using the -18 volt power supply, the -18 volt low potentiometer and lamp (secand form top of board), and the -16 volt high potentiometer and lamp (third from bottom of board). All lamps should not be extinguished and the power relay should remain operated.

2.10.1.3 +5VDC Hi-Lo Adjustments -Refer to paragraph 2.6 for 200-line EPABX system, or paragraph 2-7 for 100line EPABX system.

Low Voltage Adjustment

- (a) Adjust +5VDC Supply to +4.2 volts. The PR relay "drops out" and "sets up" an alarm condition causing the +5 volt alarm lamp to turn "'on."
- (b) Readjust voltage to +5 volts.
- (c) Depress VC reset button. The PR relay operates and the 5 volt alarm lamp goes "Out."

High Voltage Adjustment

- (a) Adjust +5VDC supply to +5.8 volts. The PR relay "drops out" and "sets up" an alarm condition causing the 5 volt alarm lamp to turn "on."
- (b) Readjust voltage to +5 volts.
- (c) Depress VC reset button. The PR relay operates and the 5 volt alarm lamp goes "Out."
- 2.10.2 Power relay does not operate lamps on 316069 and/or 316301

remain "on."

2.10.2.1 Perform the following steps if the alarm lamp(s) on the 316069 PCB

- remain "on."
- (a) Refer to Fig. 5.2 and repeat all steps of paragraphs 2.4, 2.5 and 2.9.

- (b) **Refer** to Fig. 5-3 and **repeat** all **steps of paragraphs** 2.10.1.1, 2.10.1.2, and 2.10.1.3
- (c) If alarm lamp(s) remain "on" refer to Section XII, paragraph 7.0, and Section XIII, paragraph 2.1 in basic manual:

2.10.2.2 Perform the following steps if alarm lamp(s) of 316301 PCB

remain "on."

(3 Perform step 2.10.1.1

- (b) Follow procedure outlined in paragraph 2.6 for 200-line EPABX system or paragraph 2.7 for 100-line EPABX system.
- (c) Repeat all steps in paragraph 2.10.1.3

2.10.2.3 Place **AC** circuit breakers in No-Break cabinet to ON. Place **AC** circuit breakers in cabinet 1 to OFF.

NOTE: The power supplies in cabinet 1 must he disabled to permit the high and low adjustments of No-Break cabinet power supplies.

2.10.2.4 Perform the adjustment procedures outlined in paragraphs 2.10.1.2 through 2.10.2.2 in No-Break cabinet.

- 2.11 ±3VDC Alarm Adjustments
 - 2.11.1 +3VDC Adjustments
 - (a) Adjust +3VDC output to 4.0 VDC. Alarm lamp(s) should go "on."
 - (b) Adjust +3VDC output to exactly +3.0VDC. Alarm lamps should go "out."
 - (c) Adjust +3VDC output to +2.0 VDC, Alarm lamp should go "on."
 - (d) Readjust +3VDC output to exactly +3.0VDC.
 - 2.11.2 -3VDC Adjustments
 - (a) Adjust -3VDC output to -4.0 VDC. Alarm lamp(s) should go "on.'
 - (b) Adjust -3VDC output to exactly -3.0VDC. Alarm lamps(s) should go "out."

- (c) Adjust -3VDC output to -2.0 VDC. Alarm Lamps should go "on."
- Readjust -3VDC output to exactly -3.0VDC. (d)
- 2.11.3 **Repeat all steps of paragraphs 2.11.1** and 2.11.2 for all equipped

cabinets.

2.12 AC Power On-Off Test

Place AC circuit breakers in cab-inets 1 and No-Break cabinet to OFF position. Wait 1 minute then place circuit breakers to ON position. The PR relay should operate within 3 to 5 seconds and all alarm lamps on 316069 and 31.6301 PCB's should go "out." Check PR relays in each equipped cabinet to insure they have operated.

Supplement Section XII CONTENTS

Page

1.0	General	1 2 - 1
2.0	(200 Line) DC Voltage Generation	. 12-1
3.0	(100 Line) DC Voltage Generation	. 12-5
4.0	Three Volt DC Supplies	. 12-9

FIGURES

12-1	Block Diagram of DC Voltage Generation with No-Break Modification (200-Line EPABX System).	12-3
12-2	Block Diagram of DC Voltage Generation with No-Break Modification (100-Line EPABX System).	12-7

Supplement

Section XII

GENERAL OPERATIONAL DESCRIPTION

1.0 GENERAL

The 200-line EPABX No-Break modification provides an additional cabinet containing **an additional** set of power supplies and control circuitry as housed in cabinets 1 and 2. Paragraphs 2.0 through 2-15 cover this modification. The 100line **EPABX** No-Break modification is similar in design to the 200-line EPABX. The 100-line No-Break modification is covered in paragraphs 3.0 through 3.15. The 3 volt dc supplies used in both systems are identical in operation. Paragraphs 4.0 through 4.5 cover the dc supply operations.

2.0 (200 Line) DC VOLTAGE GENERATION - SEE FIG. 12-2.

2.1 DC voltage generation description for primary and no-break back-up circuits are essentially similar. The AC input is provided to a stepdown transformer located in each cabipet. A consistent voltage type transformer is used, thus allowing for large line fluctuations. Each **cabi**net has its own separate rectifier and series regula**ting** (SR) circuits. Cabinets 1 and 3 have the above, plus the control circuits. The control output regulates all power supplies by controlling the series regulator circuits. Thus, the outputs of all power supplies of a type in cabinets 1 and 2 and those in **cabinet 3 are** paralleled for multibay operation. Thus, most adjustments are made in cabinets 1 and 3. The ± 3 VDC supplies in cabinet 2 are adjustable in that cabinet.

2.2 Each power supply (-18, +18.5, +36, -36VDC) has its separate AC input from the transformer secondary to its rectifier assembly located within the power supply.

2.3 The ± 3 VDC supply voltages are derived from the ± 36 , -36VDC power supplies and are adjusted from the front of each ± 36 , -36VDC supply.

2.4 The voltage check reference transformer provides AC outputs to the 316070 voltage check reference circuit that allows the 316070 to supply highly regulated reference voltages of ± 18 , ± 36 , ± 56 to the 316069 main alarm and voltage check circuit.

2.5 The output of the -18, +1835, -36, +36VDC supplies are also supplied to the main alarm and voltage check circuit. The voltages are compared, and if within $\pm 10\%$, a ground IS provided to the auxiliary relay via the auxiliary relay power signal (APS). With the auxiliary relay operated, a ground is supplied to each power relay (PR) of each cabinet. The PR is operated and dc voltages from power supplies in cabinets 1/2 and 3 are applied to common bus bars in cabinets 1, 2 and 3. Each PR has four C type contact sets. In the idle state, a load is provided. Each contact set is composed of (one each) -18, +18.5, -36, +36VDC.

2.6 In the event of a failure of a power supply in the primary dc power system, the following world occur. The out-oftolerance condition would be detected by the main alarm voltage **check** circuit. This results in the loss of the **APS** ground signal to the auxiliary relay causing the relay to deenergiae. The now open contacts of the auxiliary relay cause relays PR1 and PR2 to deenergize. The & energized state of PR1 and PR2 remove all primary power supplies (-18, +18.5, -36, -3, +36, and +3VDC) from the cabinet bus bars. Service is maintained by the no-break dc power system which is still connected to the bus bars.

2.7 A failure of a supply in the **no**break dc power system would result in removal of all its supplies from the bus bars while the primary dc power system maintains operation.

2.8 As shown in Fig. 12-1, the APS lead for the primary dc power system is routed through the +5VDC supply and alarm circuit and blower switches in cabinets 1 and 2. The failure of a blower in cabinet 1 or 2 will result in the removal of both primary and no-break dc power from the bus bars.

2.9 A failure of the blower in cabinet 3 will **cause only** removal of nobreak dc power from the bus bars. Primary dc power will not be affected and service will be maintained.

2.10 If a +5VDC power supply and alarm circuit should fail **or** is misadjusted, the APS ground circuit will open. The auxiliary and PR relays will deenergize causing removal of the respective **dc** power system from the bus bars. The other dc power system will, however, continue to maintain EPABX operation.

2.11 The main alarm voltage check circuit has **adjustment** and alarm lamps corresponding to each power supply. The **ad**justments are made initially and normally it is **not necessary to readjust them.**

2.12 A manual reset pushbutton (V. C. RESET) is provided **in cabinets** 1



Figure 12-1. Block Diagram of DC Voltage Generation with No-Break Modification (200-Line EPABX System)

and 3 to reset **the** main alarm voltage check circuit. III the event the voltage deviation was temporary, **such as** a surge in **AC** input, the circuit can be reset and normal operation resumed

2.13 The attendant's console contains an ALARM RESET pushbutton which permits the console operator to reset the main alarm voltage check circuit from a remote location.

2.14 A BLOWER ALARM lamp is provided on the top of cabinets 1, 2 and 3. This lamp will signify that blower operation is faulty. The cause could be insufficient air flow due to a dirty filter or blockage, a defective blower vane switch, inoperative blower motor or an open blower fuse. The SYSTEM ALARM lamps on cabinet 1 will also go on when a blower fault occurs. Note that an inoperative blower in cabinet 1 or 2 will remove all power (both primary and no-break) from the bus bars. However. an inoperative blower in cabinet 3 will not interrupt EPABX operation. The SYSTEM ALARM and BLOWER ALARM lamps in cabinet 3 will go on as will the SYSTEM ALARM lamp in cabinet 1.

2.15 An overvoltage or undervoltage condition or permanent breakdown in a -18, +18.5, -36, +36 or +5VDC power supply in the primary dc power system (cabinet 1/2) go on. A similar condition in the no-break dc power system (cabinet 3) will cause the SYSTEM ALARM lamps on cabinets 1 and 3 to go on.

3.0 (100 Line) DC VOLTAGE GENERATION -SEE FIG. 12-2

3.1 DC voltage generation description for primary and no-break back-up circuits are essentially similar. The AC input is provided to a stepdown transformer located in each cabinet. A consistent voltage type transformer is used, thus allowing for large line functions. Each cabinet has its own separate rectifier and series regulating (SR) circuits. Cabinets 1 and 2 have the above, plus the control circuits. The control output regulates all power supplies by controlling the series regulator circuits. Thus, the outputs of the power supplies in cabinets 1 and 2 are paralled for multibay operation. The power supply adjustments are made in cabinets 1 and 2.

3.2 Each power supply (-18, +18.5, +36, -36VDC) has its separate AC input from the transformer secondary to its rectifier assembly located within the power supply.

3.3 The ±3VDC supply voltages are derived from the +36. -36VDC power supplies and are adjusted from the front of <u>each</u> +36, -36VDC supply.

3.4 The voltage check reference transformer provides AC outputs to the 316070 voltage check reference circuit that allows the 316070 to supply highly regulated reference voltages of ±18, ±36, ±56 to the 316069 main alarm and voltage check circuit. 3.5 The output of the -18, +18.5, -36, +36VDC supplies are also supplied to the main alarm and voltage check circuit. The voltages are compared, and if within ±10%, a ground is provided to the auxiliary relay via the auxiliary relay power signal (APS). With the auxiliary relay operated, a ground is supplied to the power relay (PR) of each cabinet. The PR is operated and dc voltages from power supplies in cabinets 1 and 2 are applied to common bus bars in cabinets 1 and 2. Each PR has four C type contact sets. In the idle state a load is provided. Each contact set is composed of (one each) -18, +18.5, -36, +36VBC.

In the event of a failure of **a power**

supply in the primary dc power system, the following would **occur. The** out-oftolerance condition would be detected by the main **alarm** voltage check circuit. This results in the loss of the APS ground signal to the auxiliary relay causing **the relay to deenergize**. **The now** open contacts of the auxiliary relay cause relay PR1 **to** deenergize. The deenergized state of PR1 removes all primary power supplies (-18, +18.5, -36, -3, +36, and +3VBC) from the cabinet bus bars. Service is maintained by the nobreak dc **power system which** IS still **connected to the** bus bars.

3.7 A failure of a supply in the nobreak dc power system would result in removal of all its supplies from the bus bars while the primary dc power system maintains operation.

3.8

3.6

As shown in Fig. 12-1, the APS

system is routed through the +5VDC supply and alarm circuit and blower switch in cabinet 1. The failure of a blower in cabinet 1 will result in the removal of both primary and no-break dc power from the bus bars.

3.9 A failure of the blower in cabinet 2 will cause only removal of **no-break dc power from the bus bars.** Primary dc power will not be affected and service will be maintained.

3.10 If a +5VDC power supply and alarm circuit should fail or is misadjusted, the APS ground circuit will open. The auxiliary and PR relays will deenergize causing removal of the respective dc power system from the bus bars. The other dc power system will, however, continue to maintain EPABX operation,

3.11 The main alarm voltage check circuit has adjustment and alarm lamps corresponding to each power supply. The adjustments are made initially and normally it is not necessary to readjust them.

3-12 A manual reset pushbutton (V. C. RESET) is provided in cabinets 1 and 2 to reset the main alarm voltage check circuit. In the event the voltage deviation was temporary, such as a surge in AC input, the circuit can be reset and normal operation resumed.

12-5



Figure 12-2. Block Diagram of DC Voltage Generation with No-Break Modification (100-Line EPABX System).

12-7
3.13. The attendant's console contains an ALARM RESET pushbutton

uch permits the console operator to reset the main arm voltage check circuit from a remote location.

A BLOWER ALARM lamp is pro-3.14 vided on the top of cabinets 1 and 2. This lamp will signify that blower operation is faulty. The cause could be insufficient air flow duc to a dirty filter or blockage, a defective blower vane switch, inoperative blower motor or an open blower fuse. The SYSTEM ALARM lamps on cabinet 1 will also go on when a blower fault occurs. Note that an inoperative blower in cabinet 1 will remove all power (both primary and no-break) from the bus baas. However, an inoperative blower in cabinet 2 will not in-terrupt EPABX operation. The SYSTEM ALARM and BLOWER ALARM lamps in cabinet 2 will go on as will the SYSTEM ALARM lamp in cabinet 1.

3-15 An overvoltage or undervoltage condition or permanent breakdown 86, +36 or +5VDC power supply in a -18. +18.5, -36, in the primary dc power system will cause the SYS-TEM ALARM lamps on cabinet 1 to go on. A similar condition in the no-break dc power system will cause the SYSTEM ALARM lamps on cabinets 1 and 2 to go an.

4.0 THREE VOLT DC SUPPLIES

4.1The basic functions of the ±3VDC supplies is to provide clamp voltage to the lines, junctors, and all other circuits capable of originating or terminating a call.

- Failure of the -3VDC supply caually results in an inability to ring a line.
 - Failure of the +3VDC supply usually results in an inability to

terminate a call. 4.4

4.2

4.3

are on line

In the event that a switchover from the primary dc power system to

the no-break power system is necessary, the -3V and +3V bus bar connections are also transferred to the no-break -3V and +3V power **supplies**.

4.5 The fault of a -3V or +3V power 4.5 Ine taut of a -3V or +3V power supply does not result in a dc power system switchover. The fault must have occurred in anyone of the -18, +18.5, -36, +36 or +5VDC power supplies. A -3V or +3V power supply fault condition by itself will, however, cause the SYSTEM ALARM lamp(s) to go on. During normal operation, -3V and +3V power supplies in cabinet 1 are on line

TM 11-5805-653-14

Supplement Section XIII CONTENTS

Page 1.0 13-1 General 13-1 1.1 1.2 13-1 1.3 Visual Inspection 13-1 1.4 13-1 2.0Total System Failure 13-1 3.0 Partial System Failure 13-1

Supplement

Section XIII

TROUBLE DETECTION AND REPAIR

1.0 ROUTINE MAINTENANCE

1.1 General

Routine maintenance of a solid state switching is greatly simplified in comparison to a mechanical system.

Periodic Inspection and 1.2 Maintenance

Visual inspection and routine maintenance checks should be made quarterly -- four times per year.

1.3 Visual Inspection

- (a) Check cleanliness of cabinet assembly location area.
- Inspect and clean cabinet **(b)** blower filters. Follow instructions stamped on filter.
- **Check** cables from distribu-tion *frame* to EPABX equip-(c) ment cabinets for damage and snug fit on cable connectors.
- (d) Check equipment cabinets for damage.
- Vacuum interior of cabinets (e) using non-metallic nozzle.
- Check cabinets for disturbance (**f**) of PCB's.
- Electrical Maintenance
 - (a) Refer to supplement Section V. Perform all steps of para-graphs 2.4, 2.5 and 2.6 for the 200-line EPABX system or paragraph 2.7 for the 100-line EPABX system.
 - (b) Refer to Section V in basic manual. Perform all steps of paragraphs 3.2, 3.3, 3.4 and 4.0
 - (cl From a TJP dial trunk assess **digits** and observe **trunk** allot-ter "stepping" through all *four* allott groups.

TOTAL SYSTEM FAILURE 2.0

- 2.1 With the addition of no-break power capability, the possibility of total system failure from breakdown of **a** dc power supply is eliminated. However, certain conditions can still cause complete system failure.
- 2.1.1 No alarms complete system failure. Refer to Section III, Fig. 3-1 in basic manual and supplement Section III, Fig. 3-1, for 200-line **EPABX** No-Break cabinet or Fig. 3.2 for 100-line EPABX No-Break cabinet.
 - 2.1.2 Check the following for **defects**:
 - (a) AC circuit breakers
 - (b) AC input
 - Voltage check transformer (c) fuse, Refer to supplement Section III, Fig. 3-2, 3-3 and 3-4 for 200-line EPABX system or Fig. 3-5 and 3-6 for 100-line EPABX system.
 - Blower motor (d)
 - Blower vane switch (e)
 - Blower motor fuse (f)
- 3.0 PARTIAL SYSTEM FAILURE

3.1

A partial system failure involves the removal from the dc voltage bus bars of a dc power system, either primary or no-break. Use the following procedure to locate the source of the malfunction.

- Main alarm and voltage check (316069) PCB is alarmed. 3.1.1
 - (a) Check DC power supplies -18, +18.5, -36, and +36VDC. Refer to supplement Section V, paragraph 2.4. If power supply output voltages are correct, proceed to step(b). If power supply voltages are out of tolerance and will not educe to proceed to step (c) adjust, proceed to step (e).

1.4

- (b) Check outputs of 316070 voltage check reference supply. Refer to supplement Section V, paragraph 2. 10, 1. 1, and Fig. 5-1 in basic manual.
- (c) If 316070 output voltages are incorrect, replace 316070 PCB and recheck. Main alarm and voltage check 316069 may need adjustment. See paragraphs 2, 10, 1, 2 and 2, 10, 1, 3 of supplement Section V.
- (d) If 316(70 output voltages are correct, perform all steps of paragraphs 2.10.1.2 and 2.10.1.3 of supplement Section V. If alarm lamps remain on and DC power is not restored (PR relay operated),perform all steps of paragraph 17.0 (power-off measurements) in Section IV of

basic manual. Replace PCB 316069 and repeat all steps of paragraphs 2.10.1.2 and 2.10.1.3 in supplement Section V.

- (e) Remove AC power and replace defective supply.
- (f) Make all adjustments on 316069 PCB as described i.: paragraphs 2, 10, 1, 2 and 2, 10, 1, 3 of supplement Section V.
- PCB 316301 (5VDC Supply and Alarms) Alarm Lamps On.

Replace PCB 316301 and **follow instructions outlined in para**graphs 2.6 and 2.10.13 of supplement Section v for 200-line EPABX systems or paragraphs 2.7 and 2.10.13 of supplement Section V *for* 100-line **EPABX system.**

3.1.3 A failure of the ±3VDC power supply will cause **a** major alarm. Test **and/or** adjust 3VDC power supply as **described** in supplement section v, paragraphs 2.4 and 2.5.

3.1.2

PART TWO

SUPPLEMENT: CONSOLE OPERATION

GENERAL

The attendant's console is primarily used to answer listed directory number calls and extend them to the appropriate station. The attendant may be required to assist in extending outgoing traffic, act as information operator and handle other special services. These services may include extension of restricted stations, setting up conferences paging, and other feature calls. Additionally the attendant monitors the systems alarm conditions

PACKAGING

The console IS a desk mounted unit housing the indicators and control required to perform the attendant's functions. The top of the console is a slanted surface engineered for both visibility and accessibilty. (See Figure 1) The control cables enter vu slots in the rear and handset jack is equipped on either side. The cabinet contains dual function push-buttons acting as both illuminous indicators and keys. Other apparatus consists of a standard telephone dial, key set, miscellaneous supervisory and control lamps and keys. The basic console IS equipped with twenty-four trunk keys and a one hundred busy tamp field. Should the systems requirements exceed twenty-four trunk appearances the busy lamp field IS omitted allowing the console to house a maximum of forty-two trunk keys One console IS capable of performing all functions required of a forty-two trunk - tour hundred line system: however, should the attendants traffic exceed the ability of one operator, additional positions can be supplied.

1. CONSOLE OPERATION

The following IS presented to aid the Attendant. 11 Incoming Trunk Call (City Trunks)

111 The incoming call flashes the associated trunk button (TB) 120 IPM bright and sounds the console buzzer

112 The Attendant (ATTD) answers by depressing the flashing trunk button. The trunk button changes to steady bright, the buzzer turns "off", and the talk city (TC) lamp comes "on" steady bright

1.1.3 The Attendant determines the station number desired and key punches the number on the console station number pad.

1.1.4 If the station is net busy, it will ring. Upon answer, the TC lamp will go out and the talk extension (TE) lamp will turn "on" steady bright The Attendant announces the call and, if call is accepted, depresses the release bar. The city trunk party and the station will be connected and the TE lamp will turn "off". The trunk button (TB) will turn "on" dim steady.

1.1.5 The Attendant may depress the release bar if the station is not busy and the Attendant does not wish to announce the call. Ring back tone (RBT) will be transmitted to the city party. The TB lamp will flash at a dim 60 IPM. The TC lamp will go "out". Upon answer by the station, the TB lamp will turn "on" dim steady.

Ine 1B lamp will hash at a dim 60 IPM. The 1C lamp will go "out". Upon answer by the station, the TB lamp will turn "on" dim steady. 1.1.6 To release from a mis-key, busy, or ringing station prior to depressing the release bar, depress the release extension key and re-key.

- 1.1.7 If the station IS busy the Attendant may: a Place the trunk in a hold condition by depressing the hold bar. The trunk busy (TB) lamp will flash a wink signal (.9 seconds on, .1 seconds off).
- b Place the call on camp-on-busy (COB).
- 1.1.8 Camp-On-Busy Operation

a After key punching the desired number and receiving busy tone, depress the COB button.

NOTE Group hunt master numbers may not be camped on unless all associated slave numbers are busy.

b Depress the release bar. The trunk button lamp will flash 120 IPM dim.

c Two station numbers may be placed into COB at the same time. When the second number IS camped on, the COB button will turn "on" bright steady and no further station numbers will be accepted.

d. The COB circuit will attempt to ring the called number every five seconds.

e. The trunk button lamp will turn dim steady when the camped-on station has answered. NOTE: If a trunk call has been answered, and then placed in hold, the trunk must be reseized (by depressing the trunk button), and then depressing the release extension button before it can be placed into Camp-On-Busy operation

1.1.9 Attendant Recall

a. The trunk button lamp will flash 120 IPM bright when the EPABX station connected to the city trunk recalls the Attendant.

b. Depress the trunk button. The TB lamp will turn bright steady.

c. The talk city tamp will not turn "on". d. The Attendant may split the trunk manually by depressing the talk extension or talk city button, or the Attendant may participate in a three-way conversation.

e. If it is desirable to transfer the call, instruct the EPABX station to "hang-up". Depress the talk city button and proceed as though it were an incoming call.

f. If the call has been lost and there is a recall, depress the TC button and proceed as though it were an incoming call

1.2 Incoming Trunk Call (DID with Attendant Access)

1.2.1 The associated trunk button lamp will turn "on" bright steady when the trunk is seized and will flash a dim 60 IPM indicating a call is ringing. The trunk button lamp will turn dim steady indicating an answer.

1.2.2 If the Attendant IS recalled, the call may be treated as an Incoming city trunk call (except: Camp-On-Busy is not accessible.

1.3 Outgoing Call

1.3.1 Any EPABX station may reach an outside

party by dialing the Attendant (digit 0) and requesting a trunk. This action defeats the classof-service (COS) restriction feature when the city number is dialed by the Attendant.

a. The Attendant's trunk button lamp (Op 1 or 2) will flash 120 IPM bright and the console buzzer will sound. The Attendant will answer by depressing the trunk button. The trunk button lamp will turn bright steady.

b. The Attendant may, if desirable, depress an appropriate city trunk button and release from the call by depressing the release bar.

c. The EPABX station will receive city dial tone and may dial the number directly.

d. The Attendant, if time and charges (T and C) are desired, depresses the hold bar instead of the release bar. The trunk button lamp will flash "wink" upon completion of the call, indicating that the call has been completed and that T and C can now be obtained.

e. If desirable, the Attendant may place the statron's call by dialing the number directly, using an appropriate trunk button. Upon answer, the Attendant may release from the call or proceed as in Step (d) above.

f. If requested, the Attendant may book the call and place it at a later time. The Attendant must release the station connected to the Attendant's trunk.

1.3.2 Placing Outgoing Call from Console

a. The Attendant may seize an outgoing trunk by depressing an appropriate trunk button and dialing the desired number.

b If a mis-dial occurs, the city may be released and reseized by depressing the release city (RC) button.

c After the called number has answered, the Attendant may transfer it to an EPABX station in the same manner as an Incoming Call NOTE The Attendant must not extend a booked call to an EPABX station until the distant party has answered

1.3.3 An outgoing call from a station will turn

the associated trunk button "on" dim steady.

1.4 Attendance Line

1.4.1 The Attendant may originate calls to all EPABX stations and features by:

a. Depressing the attd. line button (MB). The ALB lamp will turn "on" and dial tone wilt he received.

b. Using the rotary dial and dialing the desired number(s).

c. The Attendant must depress release bar to release from call.

1.5 Attendant Override (Busy Verification)

1.5.1 The Attendant Override feature may be employed by:

a Depressing attendant override button (AOB). The AOB lamp will turn "on" bright steady.

b. Key punching the desired station number. (The busy station will receive a burst of dial tone to announce the override).

c. Depressing release bar to release.

1.6 Unassigned Night Answer (UNA)

Upon departure from the console, the Attendant turns the UNA key ON. All incoming calls will be answered through the UNA circuit.

1.7 Assigned Night Answer (ANA)

1.7.1 Same as UNA

1.8 Attendant Controlled Conference

1.8.1 Any extension user can call the Attendant via an attendant trunk to request a conference with a maximum of five other extensions and one trunk party

a. After acknowledging the conference request and obtaining the list of conferees, depress the CONF key. The attendant trunk will go dark and the conference lamp will light bright steady, indicating the conference circuit has been seized and the attendant and the extension have been transferred into the conference circuit.

b. Disconnect from the conference circuit by operating the release button.

c. To call an extension conferee, operate attendant trunk #2 (will light steady) and key punch the desired station number.

d. When the extension answers, advise him of the conference. To add the extension into the conference depress the conference key-

e. Repeat the procedure in Steps (b), (c), and (d) for the other extension Conferees.

f. To add an outgoing trunk to the conference, operate an idle outgoing trunk key and dial the desired number.

g. When the outs& party answers, advise him of the conference; to add the outside party, key punch number assigned as trunk conference assessing number (attendant transferred into conference to announce trunk party) and then the release button to disconnect from the conference (trunk party added to conference when attendant releases).

h. If an incoming trunk party requests a conference, obtain the list of the conferees and place trunk on hold.

i. Repeat the procedure in Steps (b), (c), and (d) for the other extension conferees. At least one station must be in the conference before the trunk party is added.

j. Re-enter the incoming trunk and add the incoming party by key punching the assigned number (see Step (g) above).

k. To release the trunk from the conference, station user in the conference must dial the digit 2 or greater. If the conferee wants to contact the attendant he must go "on hook" and then dial "0". The attendant may then return him to the conference by operating the conference key.

3

1.9 Alarms

 $\underline{1.9.1}$ Should a minor alarm occur, the alarm lamp will turn on dim.

 $1.9.2\,$ Should a major alarm occur, the lamp will turn on bright, and buzzer will sound.

 $1.9.3\,$ In either case, the Attendant should depress the Alarm Reset Button for approximately five seconds. If the Alarm is extinguished, the

fault bas been cleared, and requires no trouble call. If the atarm remains, or returns frequently, a trouble report should be made.

1.10 Off-Normal

The Off-Normal lamp should turn "on" when the first digit is key punched, and turn off when the third digit is key punched.

Trunk	Lamp	Signals	on	ITT	Console
	TRUNK	LAMP	SIC	GNAL	S

CALL CONDITION	TYPE TRUNK	TRUNK LAMP
Incoming CO Trunk	Central Office	Flashes bright 120 IPM. Buzzer operates.
Trunk Answered (by attendant)	All Trunks	Bright Steady
Station Ringing	2-Way Central Office DID	Flashes dim 60 IPM
Request for Transfer	2-Way Central Office DID DOD	Flashes bright 120 IPM
Camp-On Established	2-Way Central Off ice	Flashes dim 120 IPM
Station Answers	2-Way Central Off ice DID	Dim Steady
Trunk on Hold	All Trunks	Dim Wink
30 Second Extended Ringing Attendant Recall	DID	Flashes bright 120 IPM. Buzzer operates.
Extension Dials '0"	Attendant Trunk	Flashes bright 120 IPM. Buzzer operates.

2. MULTI-ATTENDANT OPERATION

2.1 All incoming calls will be split (odd and even) between two consoles.

2.2 All calls may be transferred from one console to the other by turning the transfer key on the console from which the call IS to be transferred.



16 --- ANA Assigned Night Answer

Figure 1. Attendance Console, Location of controls and Indicators

5

ltem	Name	Function
1	LINE BUSY LAMP FIELD 00 through 99	Lamp lights to indicate a busy line. May not be included on certain consoles.
2	TRUNK Pushbuttons with integral lamp (labeled with trunk identifying number)	Used to answer or initiate trunk calls. Push- button lamp flashes bright at 120 IPM to indicate incoming call. Lights steady bright to indicate attendant connected to trunk. Flashes dim 120 IPM to indicate line being accessed by Comp On Busy. Lights steady dim to indicate line connected to trunk. Flashes dim 60 IPM to indicate station is ringing.
3	BUSY LAMP	Lights to indicate busy line condition when line accessed through attendant's KEY SET is busy.
4	KEY SET Pushbuttons labeled 1 through 0	Ten pushbutton KEY SET used by attend- ant to access extensions from City Trunks, Override, or Conference.
5	RLS CITY Release City	Releases City party from trunk without re- leasing trunk from attendant.
6	RLS EXT Release Extension	Releases Extension from trunk without re- leasing trunk from attendant.
7	RELEASE	Releases attendant from a trunk or extension or both.
8	HOLD	Used to place a Trunk Call in the Hold condition. Causes Trunk lamp to wink dim.
9	TC Talk City	Used to talk to the city trunk only (extension cannot hear).
10	TE Talk Extension	Used to talk to the extension only (city party cannot hear).
11	COB Camp On Busy	Used when accessing a busy extension. Checks the busy condition of the extension exery six seconds until it becomes idle, then causes it to ring.
12	TELEPHONE DIAL	Used by the attendant when making a Trunk Call or when using the attendant's line to call a feature or extension.
13	OFF NORMAL	Lights to indicate an incomplete number is contained in the attendant's pushbutton digit stores.

Table 1. Functions and Use of Attendant's Controls

ltem	Name	Function
14	ALARM	Lights dim to indicate a minor alarm in the switching equipment. Lights bright to in- dicate a major alarm in the switching equipment.
15	ALM RES Alarm Reset	Used by attendant to reset the alarm indi- cation caused by a temporary malfunction
16	ANA Assigned Night Answer	Switches trunk answer supervision from the attendant to specific extensions.
17	UNA Unausigned Night Answer	Switches trunk answer supervision to any extension; also prepares an indicating de vice to be activited when an incoming cal is received.
18	ATT TRANS Attendant's Transfer	Switches all attendant's functions to another Attendant's Console.
19	ATTD LINE Attendant's Line	Connects the attendant to a specific line circuit so that the attendant can utilize the switching system as would any other ex- tension.
20	ATTD OR Attendant's Override	Provides a cut through feature for the atten- dant to override any extension connected to a trunk, Junctor, or Tieline.
21	OP1 Operator's "0" Trunk	Connects attendant to extension accessing the attendant's trunk circuit #1.
- 22	: ' OP2	Connects attendant to extension accessing the attendant's trunk circuit #2
23	OP3	Future
24	TONE LEVEL	Controls the level of attendant's buzzer Positions are LO-OFF-Hi.
25	CONFERENCE BUSY	Utilized for conference
26	OPERATOR'S JACKS	The jacks one on the left and one on the right of the console) provide for connection of the operator's headset and or handset to Attendant's Console
27 •	HDST HNST	Provides selection for Headset - Handse operation
28	UNASSIGNED	F

Table 1. Function and Use of Attendant's Controls (continued)

7

TM 11-5805-652-14

PART THREE BASIC INSTALLATION AND MAINTENANCE DATA

Section I

CONTENTS

Page

1.0	General		1-1
	1.1	Installation Coverage	1-1
	1.2	Maintenance Coverage	1-1
	1.3	Personnel Involved	1-1
	1.4	Installation and Maintenance Practices	1-1
	1.5	Tables	1-1
	1.6	Tools Required	1-1
	1.7	Test Equipment Required	1-1
	1.8	Terminology	1-1
2.0	Satellite	Drawings	1-1

Section I

SCOPE

The purpose of this manual is to provide sufficient Information and instructions for the installation and maintenance of the TE-400 EPABX system by personnel who have only a minimal background in telecommunications installation and elec-tronic circuit techniques.

CAUTION: Failure to adhere strictly to the sequential installation (Sections I through VIII) procedures outlined in this manual may result in major damage to the system componentry.

Sections I through VIII comprise the 1.1 Installation portion of this manual.

Sections IX through XIV are devoted to the maintenance and repair of the 1.2 TE-400 EPABX.

- In the maintenance portion of this 1.2.1 manual, repair is limited to:
 - Replacement of a printed circuit (a) board (PCB). Location and removal of short (b)
 - circuits. (c) Replacing wires.

 - (d) Replacing fuses.
 - Replacing plug-in lamps. (e)
 - The persons involved should be familiar with:
 - **Band** tools
 - (a) Banu 10015 (b) Wire wrapping
 - (c) Unwrapping
 - (d) Splicing

1.3

- (e) Soldering
- Volt-Ohm-Milliammeter (20,000 (**f**) ohms per volt)

Although it IS not specifically stated throughout this document, all installa-1.4 tion/repair should be performed using "good tele-phone practices" This is especially important in the wrapping of wire, pulling cables, etc.

Numerous tables are provided for the 1.5 installer to complete and maintain These tables are necessary for the installation. For record purposes, these tables should be retained in a secure place

The fallowing tools should be available:

- Insulated screw driver (a)
- Wire wrap (24 gauge, 22 gauge, **(b)** 20 gauge tip)
- Unwrapper (c)
- Soldering iron (80 watt) (d)
- (e) Level (f)
- 6" Adjustable wrench 5" Straight slot (g)
- (ĥ) Flats
- Wire strippers (i)
- (j)
- Long nose pliers (4') K-500 Instrument to be connect-(k) ed with test plug (plug supplied).
- Eight K-500 instruments to be **(l)** connected to the MDF vertical terminals for test. (Instruments may be re-used if not damaged.)

One of the following pieces of test equipment is required. Both are recommended:

- (a) Oscilloscope
- (b) Audio Frequency Generator

The following test equipment is required:

(a) Volt-Ohm-Milliammeter (20,000 ohms per volt) that IS accurately calibrated.

Terminology - All terms relating to 1.8 the manual are explained on first usage. Section XV (EPABX Designations) is also included for reference

2.0 SATELLITE DRAWINGS

- The following drawings are used and supplied in conjunction with this mâñual:
 - TE-400 System Assembly Dwg (a) 130097
 - TE-400 Strapping Dwg 329972 TE-400 Strapping Dwg. 130117 (b) (c)
 - Equipment Specifications (**d**)
 - Sub-Rack Dwgs. 130070 through 130080 (130080 supplied only if (e)
 - utilized)
 - (**f**) System Block Diagram 316023

2.1

1.7.1

1.7

Section II

CONTENTS

			Page
1.0	General	1	2-1
	1.1	AC Input Per Bay	2-1
	1.2	AC Power Protection Per Bay	2-2
	1.3	External AC Cabling Per Bay	2-1
	1.4	External DC Power Requirements	2-1
	1.5	Grounding	2-1
	1.6	System Circuit(s) Protection	2-1
	1.7	Main Distribution Frame Protection	2-1
	1.8	Ring Generator	2-1
	1.9	Dial Pulse Requirements	2-1
	1.10	Call Progress Tones	2-1
	1.11	Line Loop and Instruction Considerations	2-1
	1.12	Instrument Combinations	2-1
	1.13	Representative Figures on Cabling	2-2
	1.14	Trunk/Tie Line Summary	2-2
	1.15	Attendant Console Description	2-2
	1.16	MDF Location	2-4
	1.17	Floor Space	2-4
2.0	Equipr	nent Capabilities and Standard Configurations	2-4
	2.1	PCB Functional Description	2-4
	2.2	Attendant Console	2-4
	2.3	Power Requirements	2-4
	2.4	Power Failure Cut-Through	2-4
	2.5	Satellite Equipment	2-5
	2.6	EPABX Input and Output Circuits	2-5
	2.7	Environmental	2-5
3.0	Preinst	tallation Planning	2-5
	3.1	Equipment Layout	2-5
	3.2	Equipment Specifications	2-5
	3.3	Trunk/Tie Lines Assignments	2-6
	3.4	Station/Line Assignments	2-6

CONTENTS (Cont'd)

TABLES

2-1	Instrument Combinations	2-2
2-2	Cabling	2-2
2-3	Trunk Information	2-3
2-4	EPABX Input and Output Circuits	2-5
2-5	Trunk/Tie Lines Assignments	2-9, 2-11
2-6	Line and Feature Assignments	2 - 1 3 through 2 - 1 9

FIGURES

2-1	Attendant Console	2-3	3
-----	-------------------	-----	---

Section II

PRE-INSTALLATION

1.0 General

- 1.1 AC Input Per Bay
 - (a) 115VAC, 60 Hz, 30 amps, or
 - (b) 115VAC, 50 Hz, 30 amps, or (c) 220VAC, 50 Hz, 15 amps
- 1.2 AC Power Protection Per Bay
 - (a) 30 amp circuit breaker per **115VAC** input 15 amp circuit breaker per **(b)** 220VAC input.
- External AC Cabling Per Bay 1.3

Refer to Section III, Fig. 3-1. (a)

1.4 External DC Power Requirements.

Those system equipped with the fol-lowing features will require an external 48VDC source.

- Direct inward dial (DID) trunks (a) or tie lines
- Standby power (b)

Grounding 1.5

- Earth ground of 1 ohm or less is (a) required. Refer to Section III, Pig. 3-1 for connection.
- Multibay bonding is supplied. Refer to Section III, Fig. 3-1. Refer to Section III, Fig. 3-1 for (b)
- (c) AC power input grounding.

System Circuit(s) Protection 1.6

- (a) (b)
- (c)
- 018 amp, **2AG** fuse (d)
- Voltage Check Reference Supply 1 amp, **2AG** fuse Ring Generator . . . (e)
- 0.5 amp, 2AG fuse Tone Supply Outputs 0.5 amp, **2AG** fuse (f)
- All Other Circuits . . . (g)
- Electronic voltage check circuit (316069, 316070, and 316301)

Main Distribution Frame(MDF) 1.7 Protection

Any circuit(a) terminated on the MBF with external cabling that may be exposed to <u>HIGH VOLTAGE</u> or

HIGH AMPERAGE should be protected with heat coils or carbons.

Ring Generator 1.8

(a) 30 Hz, 5 Watt, one per bay (internally mounted), or

(b) 20 Hz, 5 Watt, one per buy for systems utilizing 50 Hz, AC power (externally mounted).

1.9 **Dial Pulse Requirements**

- 8 to 12 ips (impulses per sec-(a)
- ond), normally 10 ips. 62.5% break (±5% of total per-iod) normal pulsing **from** sub-(h)

scriber's instrument may be distorted by the line conditions.

Call Progress Tones 1.10

- Dial Tone (DT) 600 Hz (a)
- Busy Tone (BT) (b) 600 Hz interrupted
- Ring Tone (RT). (c) (0.5 sec. off, 0.5 sec. on)600 Hz modulated by 12 ips (interrupted 1 sec. on, 3 sec.
- Ring Back Tone (RBT) (d) Same as RT

Line Loop and Instrument 1.11 Considerations

- The recommended instrument(s) (a)
- to be attached to the line is the
- ITT K-500 series, or equivalent.

NOTE: Employing other types may result in degradation of the line performance.

- (b) The maximum allowable number of bridged ringers per line is two.
- (c) The minimum leakage between Tip and Ring (T and R) is 15,000 ohms.
- The minimum leakage between T or Rand ground is 15,000 ohms. (d)
- All ringers are bridged high im-(e) pedance ringers.
- Table 2-1 shows the possible instru-ment combinations that are recom-1.12 mended to equip per line per Station.

NOTE: A line refers to an output from the TE-400 line circuit. A station is a line equipped with instrument(s) connected via T-R cabling to an EPABX line circuit.

Combination	A	в	с	D	E	F	G		
1	1						X_		
2	1	1					X		
3	1		1.			L	X		
4	2		2			<u>x</u>			
5	2			2		L	X		
6			L		1		X		
 Column Explanatory for Table 2-1 A - Standard Instrument with high impedance ringer. B - Instrument with high impedance bridged ringer and bridged neon lamp. The lamp to have a minimum of 15,000 ohms series resistance. C - Instrument with bridged neon lamp. Lamp to have a minimum of 15,000 ohms series resistance. D - Instrument with neon lamp connected to ground. Lamp to have a minimum of 15,000 ohms series resistance. E - Key System Line Input. The TE-400 is compatible with all key systems in manufacture today. 									

Table 2-1. Instrument Combinations

- F DC Loop Resistance (DCR) 600 ohms maximum (Instruments included)
- DC Loop Resistance (DCR) -1200 ohms maximum (instru-G ments included)
- 1.13 Table 2-2 is provided to show representative figures on cabling
- Trunk/Tie tine Summary 1.14
- 1.14.1 Table 2-3 IS presented to provide specific trunk Information.
- 1.14.2 Column Explanatory for Table 2-3
 - A External 48VDC IS required
 - B Type of DCR loop resistor required Type of DCR loop resistor re-
 - с quired D - Type of DCR loop resistor re-
 - quired
 - E Type of DCR loop resistor required External terminating equipment
 - F is required
 - G Loop Start Trunk IS signaled via ring voltage on the R lead The trunk signals via a loop closure

- H Ground Start Trunk is signaled via a ground on the R lead, and signals via ground on the T lead.
- J Loop Trunk/tie lines are signaled in and out by loop open/ closure on T and R leads. K - E&M - Trunks/tie lines are
 - signaled in via the E lead and out via the M lead.
- NOTE: All other columns are selfexplanatory.
- Attendant Console Description. See 1.15 Fig. 2-1
- 1.15.1 Trunk appearances:
 - 24 with 100 line busy lamp field (a) (BLF)
 - (b) 42 maximum (BLF not equipped)

1.15.2 Rotary Dial used for:

- Calls outside system. (a)
- Calls from attendant line to (b) EPABX stations or features.
- 1.15.3 Key Pad used for:
 - Incoming trunk calls to EPABX (a) stations
 - Placing stations and/or trunks (b) into conference.
 - (c) Attendant override.

Table 2-2. Cabling

Cable	Length (Kılo Feet)	DCR (ohms)	DB Loss at 1000 Hz
#26 NL	12 7	1050	6.88
#26 NL	11 08	920	60
#26 NL	5.54	460	3.0
#26 H88	12 4	1050	4.23
#26 H88	8.8	743	3.0
#24 NT.	18	935	7.97
#24 NI.	13.5	703	6.0
#24 NL	6.75	352	3.0
101 100		1050	
#24 H88	19.8	1050	4.50
#24 H88	13	690	30
#22 NL	18	575	6.12
#22 NL	17 6	563	60
#22 NL	8.8	282	3.0
#19 NL	18		4 34
#10 1100	50.0	1050	4.66
#19 H88	37.5	675	3.00

TM 11-5805-652-14 ISSUE 1

	A	B	С	D	F		F		G	H	J	K	L	M	N	0	P	Q	R	s	T	U	V
				1 60			e Are		Su	pv	& S	ig			Ope	rat	ion	al	Fea	tur	es		
Group Identity	External 48 VDC	TR 2	MR 5	MR 6	MR 7	Direction	External Terminating Equipment	Required	Loop Start	Grd Start	Loop	E&M	Att Access	Transfer	Override	Toll Restrict	Msg. Reg.	UNA	ANA	2nd Dial Tone	Rev. Batt	Intercept	Tandem
CITY TRUNK						INC			X				X	X	X			X	X				
316061 () 122		X				OTG			X				X	X	X	X	X						
CITY TRUNK						INC				X			X	X	X			X	X				
316075 () 122		X				OTG			_	X			X	X	X	X	X				_		
DID TRUNK			ŀ			INC					X		X	X	X				_		X	X	
316089 () 122	X		X	X	X	OTG		_			X		X	X	X	X	X						
DID TRUNK			l			INC					X			X	X						X	X	
316056 () 122	X		X	X	X	OTG					X			X	X	X	X		_				
TIE TRUNK	1			Į		INC					X		X		X								
316054 () 122	X		X	X	X	OTG					X		X		X								
TIE TRUNK						INC					X				X					X			
316090 () 122	X		X	X	X	OTG					X				X								
TIE TRUNK						INC		_			X		X		X					X			
316091 () 122	X		X	X	X	OTG					X		X		Х								
TIE TRUNK						INC					X		X	X	X					X			
316092 () 122	Х		X	X	X	OTG					X		X	X	X								
TIE TRUNK						INC					X			X	X					X			
316093 () 122	X		X	X	X	OTG					X			X	X								
DID TRUNK						INC	X					X	X	X	X							X	
316094 () 122	X		X	X	X	OTG	X					X	X	X	X	X							
DID TRUNK						INC	X					X		X	X	X						X	
316095 () 122	X		X	X	X	OTG	X					X		X	X								
TIE TRUNK						INC	X					X			X					X			
316096 () 122	X		X	X	X	OTG	X					X			X								
TIE LINE						INC]			X				X								X
316060 () 122	X		X	X	X	OTG					X				X								X
TIE LINE						INC	X					X			X								X
316051 () 122	X		X	X	Х	OTG	X					X			X								X

Table 2-3. Trunk Information



Fig. 2-3. Attendant Console.

1.15.4 Location from EPABX equipment bay:

	(a) M (b) <u>C</u>	laximum DCR - 2 2able Gauge 24 22 19	26 ohms <u>Feet</u> 500 800 1625		
1.15.5	Dimens Width 2 4 "	sions <u>Depth</u> 12''	Height 5-1/2"		

1.16 MDF Locations

- (a) Non-Critical
- (b) The cable length between the

MDF and EPABX TE-400 bay should be taken into consideration when calculating loop DC resistance.

1.17 Floor Space

The EPABX equipment cabinet for a 100-line system requires a cabinet 77.25" high, 37" wide, and 26.5" deep. Multiple 100-line Increments require cabinets of the same size.

Lift-off doors are provided at the front and **red** of the cabinet. Space between the front or the rear of the cabinet and other objects should be adequate for easy passage of a person (minimum of two feet). Terminal strapping fields are accessible from the rear of the cabinet. Access to the top of the bay is only required for plugging in the connectors of interconnecting cables.

1.17.1 Floor Loading

The equipment cabinet weight is approximately 700 pounds (including printed circuit boards). Four leveling feet, approximately 1-1/2" in diameter, are placed at the four corners of the base.

2.0 EQUIPMENT CAPABILITIES AND STANDARD CONFIGURATIONS

The TE-400 system is arranged to provided growth from 25 to 100 lines, in increments of five lines, within each cabinet. A maximum of four cabinets, providing a total capacity of 400 lines, may be installed in each TE-400 system. Refer to Dwg. 130097 for configurations and PCB positions Refer to Section III, paragraph 2.0.

2.1 Refer to Section X for functional description of printed circuit boards.

2.2 Attendant Console (316028)

Refer to Section X, paragraph 3.3.

2.3 Power Equipment

The power supply assembly provided

with the system is available in two arrangements, regular standby.

2.3.1 Regular Power Supply (316300)

The regular power supply consists of four individual power modules mounted at the extreme left side of the cabinet at levels D-G. The power supply modules of like voltages will be crossconnected to common bus bars in multicabinet installations. Refer to Fig. 3-1, Section III.

2.3.2 Standby Power Supply

Standby power units may be provides on an optional basis as **a** replacement for the **regular** power modules. The standby power units will:

> (a) Provide "no break" power derived from a local battery sup-

rived from a local battery supply, in the event of failure of the 110V 50-60 cycle commercial power.

- (b) Each standby power unit will have the capacity of serving one 100-line EPABX system.
- (c) A maximum of two standby power units will be housed in each standby power cabinet.
- (d) The standby power cabinet will have the same height and depth dimensions as the apparatus cabinet.
- (e) Each standby power cabinet must be located adjacent to the

apparatus cabinet(s) that it is serving. That is, in the case of one standby power cabinet serving two apparatus cabinets, the standby power cabinet must be located between the two apparatus cabinets.

2.4 Power Failure Cut-Through (PFCT)

Provision is made for the extension of central office service direct to preassigned PABX extensions in case of power failure, subject to the following definitions and exceptions.

- 2.4.1 Power failure is defined as a loss of system DC power.
- 2.4.2 Capacity

A maximum of three EPABX stations may cut-through to an equivalent number of trunks for each 100-line cabinet.

2.4.3 Loop Start City Trunks

Loop Start City Trunks - In the case where the trunks cut-through to the EPABX stations are loop start city trunks, normal incoming and outgoing service will be provided between the central office and the EPABX stations.

2.4.4 Ground Start City Trunks

In the case where the trunks cutthrough to the EPABX stations are ground start city trunks, the station instruments must be equipped with a ground source under control of a push button. This button must be depressed to obtain a central office dial tone when originating a call. Incoming service will be provided in a normal manner.

2.4.5 DID Trunks (Loop Signaling)

In the case where the trunks cutthrough to the EPADX stations are direct inward dial trunks, only outgoing service will be provided.

2.4.6 DID Trunks (E&M Signaling)

The power failure cut -through feature will not function for either incoming or outgoing service if the DID trunks are arranged for **E & M** signaling.

2.5 Satellite Equipment

The following equipment may be required upon installation. The required equipment will depend on the features incorporated in the system.

2.5.1 A 48VDC Power Supply will be required whenever DID trunks/tie lines are equipped.

2.5.2 **Amplifiers, speakers, etc, will be** required on those systems equipped with the public **address system feature**

2.5.3 Audible signaling components (such as horns, bells, etc.) will be required on systems equipped with:

(a) Unassigned Night Answer (UNA) - Audible signaling devices for

- Audible signaling devices for UNA must be capable of being driven by a 30 Hz signal interrupted at one second on, one second off.

- (b) Code Call Audible signaling
- devices for use with code call must be capable of following a switch closure of

1 ips

NOTE: These satellite equipments may not be supplied by ITT. See Equipment Specifications for ITT supplied equipment.

2.6 EPABX 1 input and Output Circuits

 Table 2-4 shows the Inputs/outputs of various EPABX circuits that are connected to satellite equipment

Table 2-4. EPABX Input and Output Circuits

Cırcuit	Output
Trunks/Tie Lines (E-M)	E and M Leads - One set per circuit
Trunks/Tie Lines (All)	Tip and Ring - One set for each circuit
Lines	Tip and Ring - One set for each circuit
UNA	URL1, URL2, URL3, URL4
Code Call	CCM1, CCM2, CCM3, CCM4
Public Address	TR1, TR2, TR3, APC1, APC2

2.7 Environmental

2.7.1 Temperature 0° to 100°F

The TE-400 does not **require air conditioning.** However, the TE-400 must be placed in a ventilated room. The room ventilation and air move. meat must be equivalent to a system cooling fan of cubic feet per minute listed below. The recommended ambient temperature is **100°F.** maximum.

Air <u>Movement</u>
CFM
64 76
94 125

2.7.2 Humidity

Humidity 10 - 90%

2.7.3 <u>Atmosphere</u>

(a) Non-corrosive atmosphere

(b) Non-explosive atmosphere

NOTE: If it is necessary to operate the TE-400 III an environment that exceeds the figures listed in paragraph 2.7, the Director of Engineering must be consulted

3.0 PREINSTALLATION PLANNING

3.1 Equipment Location

The assigned equipment space(s) should be surveyed for:

- (a) Size
- (b) Environment
- (c) Floor loading
- (d) Cable(s) access
- (e) Location of console
- (f) AC power supply outlets and ground

3.2 Equipment Specifications (E-S)

Each system produced IS provided with an Equipment Specifications

3.2.1 Sheet 1 of the E-S will be provided for each cabinet of a system

The cabinet pictoral shows the shelf position of all equipped PCB' s

The numbering plan shows the assigned access digits for each circuit

Table A lists cabinet(s) and associated equipment

 Table B lists PC boards associated with the cabinet

General and Shop notes are of interest to the installer(s) Shop notes are performed by the shop prior to shipments.

3.2.2 theetE-S will be provided for each system.

Strapping Chart A refers to equipment strapping options.

The installer(s) will be required to complete those items marked "Installer. "

Strapping Charts B and C show special strapping. The installer(s) will be required to complete these items marked "Installer."

Table C lists all ship loose Items.

The installer(s) will complete items listed under "Installation Notes."

3.2.3 Sheet 3 IS supplied when or if any special drawings are required.

3.3 Trunks/lie tines Assignments

Table 2-5 IS provided for the installer(s) to fill in.

3.3.1 Table 2-5 Column Explanatory

A - Quantity - The first <u>equipped</u> trunk in cabinet 1 is designated as trunk 1, and the count IS made in sequence from that point using all equipped trunks and tie lines.

NOTE: Equipment Specifications details console appearance.

- **B** Group Identity- ITT circuit number. Refer to Table 2-3 and E-S.
- c Type Supervisory signaling presented/received to/from cen-
- tral office. Refer to Table 2-3. D DCR Loop The DC resistance

of the trunks tip and ring cable between the MDF and CO This information must be obtained from the CO.

Resistors TR2, MR5, MR6, and E MR7 are equipped by the installer to adjust the trunks to the correct loop resistance

Refer to Table 2-3 to determine which trunks are equipped with TR2, MR5, MR6, or MR7.

Three values of each resistor are supplied.

DCR Loop TR2 MR5 MR6 and MR7 1200 ohms 1000 ohms 390 ohms 0-800 ohms 820 ohms 620 ohms 180 ohms 401-800 ohms

801-1200 ohms 390 ohms 220 ohms 0 ohms

- NOTE: See Equipment Specifications, Ship Loose Material.
 - F 48VDC requirement may be determined from Table 2-3.
 - G E&M terminal equipment re-

quired may be determined from Table 2-3.

H - Incoming only, outgoing only, or two-way may be determined

from the Equipment Specifications. ALL TE-400 trunks are two-way trunks but may be engineered for one-way operation.

All TE-400 trunks that have Ι-

attendant access may have a console appearance. See Table 2-3. However, outgoing only trunks **may** NOT use this feature. See Equipment Specifications.

- K Refer to Equipment Specifi-cations for the appropriate Access Digit(s).
- L Consult with the CO to determine city number assignment.
- 3.3.2 Table 2-5 Usage Example: Trunk 1, Bay 1 has a group

type is ground start. DCR loop equals 605 ohms, thus TR2 equals 820 ohms. Resistors MR5, MR6, and MR7 are not installed. The 48VDC or external terminating equipment is not required. The trunk is a two-way trunk with console appearance. The access digit is 9 and the assigned city number is 686-7401.

Station/Line Assignments 3.4

Each EPABX cabinet is wired for 100 lines, X00 to X99. Four hundred IS the maximum number of wired lines.

The number of lines equipped is in multiples of five (each line PCB has five line circuits), from 5 to 400.

The number of stations equipped is determined by the equipment user's requirements. All lines wired (100 in each cabinet) are terminated on the MDF. A sta-tion refers to the actual user's location and instrument. Thus, station 200 is connected to the MDF, which is connected to line 200 in the equipment cabinef

Table 2-6 is presented to allow the Installer to determine Line and Feature assignments.

The installer must consult the Equipment Specifications (E-S) and the customer to determine what lines will be equipped for the items listed in Table 2-6. A check should be inserted to indicate assignment

3.4.1 Table 2-6 Column Explanatory

A - EPABX Line - Refer to Equip-

ment Specifications numbering scheme The first entry should be the lowest (nominally 200) line equipped and proceed sequentially upwards to the highest line number equipped.

B - Not Assigned - Line circuit

equipped but not active. The Line Assignment - Each tie line installed removes one line С

from service as a station. Consequently, any line appearance assigned for tie line usage cannot be assigned to a station.

Consult Equipment Specifications to determine the lines installed and the number assigned.

D - Attendant (Attd) Line - This line is assigned to the attendant and may be selected from any line.

Trunk Conference Line - This lines as assigned to the trunk ac-

cessing conference and can be any lines may assigned as an attendant line or tie line.

F - Station Equipment (See Table 2-1) - Establish thee station equipment **required** or presently installed. From Table 2-1 determine the **number** corresponding to station equipment and enter under Station Equipment column.

G - Group Hunt (Rotary Service) -

SIX group-hunt circuits may be provided in each hundred line group (see Equipment Specifications). The first station number in the group is the master number. One master number plus a maximum of four slave stations comprise one group-hunt circuit.

NOTE: A slave station can only appear in one hunt group.

Croup hunting IS not restricted to consecutive numbering but may start with any extension number (master) and include a maximum of any four slave numbers within the same hundred group.

Any station can be assigned as a master or slave.

Consult equipment user for desired master numbers and assigned slaves for each group hunt circuit. Do NOT mix slaves between groups.

Example: Group Hunt 1 - If 220 is the master, enter "M" under Group Hunt 1 column corresponding to 220. Identify slaves as "S1," "S2," "S3," and "S4."

H - Power Failure Cut-Through (PFCT) - Three lines may be assigned per bay for power failure cut-through These lines will be connected directly to a central office trunk in the event of a power or complete (loss of DC power for any reason) equipment failure

Central office trunking - ground start

Type of instrument required - ground button to activate CO

Type of service supplied - normal inward and outward.

Central office trunking - loop start

Type of instrument required - Normal (K-500)

Type of service supplied - normal inward and outward.

Central office trunk - DID (loop)

Type of instrument required - Norma! (K-500)

Type Of service supplied - outward only.

Central office trunk - DID (E&U)

The PFCT will not function.

Where a mixture of trunks are provided (i.e., ground start and loop start) it is desirable to place the PFCT lines to the type of trunk providing both in and out service.

Determine what systems are to be assigned to PFCT and enter under PACT column.

J - Assigned Night Answer (ANA) -

Each ANA PCB has provisions for six stations to he assigned ANA service. One PCB is normally equipped per station (see Equipment Specifics

Consult with equipment **user** to determine **stations to be assigned to** ANA.

NOTE: ANA station should not be a group hunt slave station.

K - Single Digit Access (SDA) - Each equipped SDA and PCB has provisions for five stations to be assigned as SDA's Consult Equipment Specifications to determine if SDA is equipped Consult equipment user to deter-

L - Executive Override - Consult

Equipment Specifications to determine if executive override IS equipped. If equipped, one station may be assigned as executive override. Consult equipment user to establish station

M - Intercept Number - Refer to Equipment Specifications if

intercept is equipped.

mine stations to be assigned.

Consult equipment user to determine line to be intercepted

N - Class of Service Restriction -

Each equipped station may be restricted from five features This restriction prevents the station(s) from accessing the specific feature(s) when the access code(s) feature digit is dialed A busy tone IS returned to the restricted station(s) upon dialing a restricted feature digit(s)

This restriction is accomplished by equipping or not equipping a diode on the class of service (COS) PCB

The option of either equipping or not equipping a diode to restrict a station is established prior to shipment and is detailed in the Equipment Specifications installation notes

Use the information on the E-S installation notes and consult with the equipment user to determine what stations should be restricted Then enter the type of restriction under the class of service (COS) column

TM 11-5805-652-14 ISSUE 1

An "X" under Y column shows that the station should be restricted and a diode must be equipped. An "X" under the N column shows that station will be restricted if a diode is not equipped.

Example:

	Access Level	Install Diode to Restrict	Install Diode to Unrestrict
Toll Restriction (316067)			Х
City Trunk (316075)	9	Х	
City Trunk (316075)	81		Х

This shows that the equipment has toll restriction and all stations will be toil restricted unless **a diode** IS equipped. City **trunk** (dial 9) may be **accessed by** all stations <u>unless</u> a **diode** IS equipped. City trunk (dial 81) cannot be accessed <u>unless</u> a diode IS equipped.

Image Max Max </th <th></th> <th>8</th> <th>С</th> <th>D</th> <th>E</th> <th>F</th> <th>G</th> <th>H</th> <th>L</th> <th>K</th> <th>L</th> <th><u></u></th> <th>N</th> <th>0</th> <th>P</th> <th>Q</th>		8	С	D	E	F	G	H	L	K	L	<u></u>	N	0	P	Q
	Trunk Tie/Line Qty	Group Identity	Туре	DCR Loop	• TR2	MR 5	MR6	MR7	48 VDC Required	6&M Term Equip Req	Inc Only	Out Only	Two Way	Console App.	Access Digit(s)	City Number Assigned
3	1															
	2															
· · <td></td>																
	6															
	7															
I I <td></td> <td> </td> <td> </td> <td></td>														 	 	
															╂	
P N	n	· · · · · · · · · · · · · · · · · · ·														
	12															
	13														Ļ	
	14								<u> </u>							
P P <td>- 15</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td><u> </u></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	- 15								<u> </u>							
B I	17													<u>+</u>	1	
	18															
	19															
	20								 			 				
	-21															
	- 22															9 M
b b <td>24</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1</td> <td></td> <td><u> </u></td> <td></td> <td>1</td> <td><u> </u></td> <td><u>t</u></td> <td></td>	24								1		<u> </u>		1	<u> </u>	<u>t</u>	
3	25															n an
	26															
											 			nen en en en en en	-	-
	28															
1	30				<u> </u>					+			İ		+	
R A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A A A A B A A A A A	31								1		I			an a		
B	32															
A I	33										L					
	34							·			<u> </u>	<u> </u>				
	35								╡───				ļ			
9	37								<u>†</u> -							
9	38	·· <u>·</u> ·····	<u> </u>								<u> </u>		<u> </u>			
40 A	39	· · · · · · · · · · · · · · · · · · ·														
	40														1	
	41															
	42								<u> </u>			ļ	L			<u> </u>
\overline{a}	43							ł	+			<u> </u>	<u> </u>		-	
6 8 <td>45</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>· · · ·</td> <td></td> <td></td> <td></td> <td> </td> <td> </td> <td>· · · · · · · · · · · · · · · · · · ·</td> <td>1</td> <td>·</td>	45							· · · ·				 	 	· · · · · · · · · · · · · · · · · · ·	1	·
1	46						1								1	
88	47															
49	48								L				ļ		_	
30	49					<u> </u>					<u> </u>	<u> </u>			 	
92 93 94 <	51			<u></u>		·····	<u> </u>				+					
33	52									1	1			1		n pitista ana
34	53	· · · · · · · · · · · · · · · · · · ·														
55	54										ļ					
30	55		·		ļ									-		
38							+		+		<u> </u>					
99	- 58					<u> </u>		1	1		 					
60	59															
61 -	60															
02 02 <td< td=""><td>61</td><td></td><td></td><td> </td><td></td><td></td><td></td><td><u> </u></td><td> </td><td><u> </u></td><td>ļ</td><td><u> </u></td><td>Į</td><td></td><td></td><td></td></td<>	61							<u> </u>	 	<u> </u>	ļ	<u> </u>	Į			
64	63				<u> </u>				 				 			
65	64			-					†	+	<u> </u>			1	-	
66 67 68 68 69 67 68 68 69 68 69 60 <	65							I		L			Ĺ	L		
67 68 69 68 69 68 69 68 69 69 69 60 <	66															
co co co co co co co co 69 co co co co co co co 70 co co co co co co co 71 co co co co co co co 72 co co co co co co 73 co co co co co co 74 co co co co co co 75 co co co co co co 76 co co co co co co 78 co co co co co co 80 co co co co co co 81 co co co co co co 82 co co co co co co 84 co co co co co co 86 co co co co co co 88 co co co	67				 		<u> </u>	<u> </u>	+	 	 	 				
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	60									+	┨────		<u> </u>	<u> </u>		an a
11 1 1 1 1 1 12 1 1 1 1 1 13 1 1 1 1 1 14 1 1 1 1 1 14 1 1 1 1 1 16 1 1 1 1 1 17 1 1 1 1 1 17 1 1 1 1 1 17 1 1 1 1 1 18 1 1 1 1 1 181 1 1 1 1 1 182 1 1 1 1 1 183 1 1 1 1 1 184 1 1 1 1 1 186 1 1 1 1 1 188 1 1 1 1 1 188 1 1 1 1 1 188 1 1 1 1 1 189 1 1 1 1 1	70						†	1	<u> </u>	†	t	<u> </u>	<u> </u>	<u>+</u>	+	
12	71								1		1	1		1	1	CONTRACTOR OF STREET, S
73 . </td <td>72</td> <td></td> <td>I</td> <td></td> <td></td>	72													I		
14 1 1 1 1 1 75 1 1 1 1 1 1 76 1 1 1 1 1 1 77 1 1 1 1 1 1 78 1 1 1 1 1 1 79 1 1 1 1 1 1 80 1 1 1 1 1 1 81 1 1 1 1 1 1 82 1 1 1 1 1 1 83 1 1 1 1 1 1 84 1 1 1 1 1 1 85 1 1 1 1 1 1 86 1 1 1 1 1 1 88 1 1 1 1 1 1 99 1 1 1 1 1 1	73		·			· · · ·		ļ	 	<u> </u>	 				<u> </u>	
76 77 77 78 78 78 78 79 70 <	- 14			 -	<u>+</u>	<u> </u>	<u> </u>		+		 	 		+	 	<u> </u>
17 1 1 1 1 1 1 78 1 1 1 1 1 1 79 1 1 1 1 1 1 80 1 1 1 1 1 1 81 1 1 1 1 1 1 82 1 1 1 1 1 1 83 1 1 1 1 1 1 84 1 1 1 1 1 1 85 1 1 1 1 1 1 86 1 1 1 1 1 1 87 1 1 1 1 1 1 88 1 1 1 1 1 1 89 1 1 1 1 1 1	76		 		<u> </u>			<u> </u>	<u>+</u>	+	 	<u> </u>		+	+	
78 <	77							<u> </u>	1	1	1	t		1	<u>†</u>	1
79	78														1	
80	79															
81 1 1 1 1 1 1 82 1 1 1 1 1 1 83 1 1 1 1 1 1 84 1 1 1 1 1 1 85 1 1 1 1 1 1 86 1 1 1 1 1 1 87 1 1 1 1 1 1 88 1 1 1 1 1 1 89 1 1 1 1 1 1	80		ļ				ļ									
83			+		<u> </u>		<u> </u>	 	 	<u> </u>				 	 	
84	82 83		<u> </u>	<u>+</u>	 	· · · · · · · · · · · · · · · · · · ·	<u> </u>			<u> </u>	┣	 		 	<u> </u>	<u> </u>
85	84		t	<u>+</u>			†	t	<u> </u>	<u>† – – – – – – – – – – – – – – – – – – –</u>			<u> </u>	1		
86 87 9	85		1	1			<u> </u>		†	<u>†</u>				t	1	<u> </u>
87	86														L	
88	87															
	88		 	 	 									<u> </u>		
	89		+	<u> </u>			<u> </u>		┟			i		 	┣	

T a b 1 e 2 - 5 Trunks/Tie Unit Assignment

									[
							<u>† </u>		t	+	-	+	<u>+</u>
						t	1 1		1	1	1	1	
					·				1	+	+	+	1
				<u> </u>		t			t	 	<u> </u>	+	<u> </u>
			†	i i			<u>├</u>		 	<u> </u>	+		
				t		 	┼──┤		<u> </u>	 	 	<u> </u>	
		1								ļ			
									L				
	1					L		_					
												T	1
												1	
											Contraction of the local division of the loc		
 										1974 - 1975 - 1975 - 1975 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976 - 1976	L		
 		· · · · · · · · · · · · · · · · · · ·											
				·····									
•													·
									+		i		
 								l					
 	+					ļ							
 													·
]		
 <u> </u>									T				
 									T				and analysis of the produced second
 											-		(Alaine
 								1					annen alt som nyfte omrigterer
 						\rightarrow		-+			f		
						+		-+		+	+	-+	ويستقو ومعاوية والمحمولية والمحمولية والم
								+	+-		+		
								-+		+			
							-+-						
												-+	
 	+												
 				·									
 		+											
 									1				
 	[
						-+							
 							<u>L</u>		l				

2 - 9

A	В	С	D	E	F	G	Н	L	ĸ	L	Μ	N	0	Ρ	Q
Trunk Tie/Line Qty	Group I dentity	Туре	DCR Loop	TR2	MR5	MR6	MR7	48 VDC Required	E&M Term Equip Req	inc Only	Out Only	Two Way	Console App.	Access Digitts)	City Number Assigned
101															
102															
104															
105															
107															
108															
109															
111															
112							······································	 i							
115															
115															
116			···—·					<u> </u>						<u> </u>	
118															
119								ļ					 	 	
120					······			<u> </u>							
122								<u> </u>							
123												<u> </u>	<u> </u>		
124				· · · · · · · · · · · · · · · · · · ·											
126								 	┣					┼──	
127															
129								ļ			<u> </u>				
130								<u> </u>							
131	·							1	1						
133							<u> </u>	┥──					+		
134															1
136										—		ļ			
137					<u> </u>		ļ	+	+		<u>+</u>			+	
138															
140															
141		<u> </u>		<u> </u>	<u> </u>	<u> </u>			+	†	+	+			
142								1					1_		
144					<u> </u>				+		+	+	+		+
145		<u>+</u>									1				
147															
148		<u> </u>		·		+			+	+	+	+	+-	+	
149	┢─────														
151						<u> </u>				+	+	+	+		
152															
154											+		+		a and a second
155										+	+	+	+-		and the second
150															
158													_		and the second state of th
159	+		<u>}</u>					1-							
161								1-			+	+	_		
162	+			+	-			+		-	+	<u> </u>	1		
164								1		T		T		T	
165					+					-		+			
160												—	-		
168					+			+				+	+		
169	+									1	1	1			
171										+			+	+	
172															
175								T			1		+	+	
175								+-	+-	+			+-	+-	-
176	+														
178							+							+	
179				+	+										
180				<u></u>		_		-						+-	
182				+				+-	+	+-		+	+	+	
183															1
185										+		+-			
186								+		-+	+-	_	+-		
187							_		1	1		1			
189										+	+	+		+-	
190	_									\pm		\pm	1-	1	
191	+									1			1		
					1								1		1

1113	ىرىكى بارىمىيە 4-يەر بەلەر خەركىيە . كەلىكەرىدۇ بەر	aan jaraa ka ka kaani ta ka	literes interference and histolite and	a and the state of	an in an ang manganang mangang mang mang man	All and a second se		· /postpont	·····					
115									. On \$6. 3	Config.		nike Colorada	Sheine an a'	
114												+	+	
115														
116								_						
117														
119														
110													+	
119														
120										1				
121														
122														
122														
123														
124														
125														
124														
120														
127														
128														
129														
130														
131														
132														
133														
124														
134														
135														
136								_						
137														
						L								┝┩
158					L					ļ	ļ			
139							L							
140							[]							
			<u> </u>		· · · · · · · · · · · · · · · · · · ·	·				<u> </u>				
									i					
142								L		L	L	L	L	
143	1											L		
100					· · · · · · · · · · · · · · · · · · ·		· · · · ·		<u> </u>					
			· · · · · · · · · · · · · · · · · · ·				 		 		<u> </u>	t		
145					ļ		 	ļ	 		 	 	 	
146						L					L	L		
147													1	
149							<u> </u>				t	t		
140			<u> </u>		ŀ		 	 	{	 	┢────			
149														ى يەرىپى چەدەرىي زىلىد بىرى مەدەرىي تە
150									1		1			
151								[
162				·			<u> </u>	ļ	<u> </u>		t	<u> </u>		1
132							 	 	<u> </u>		<u> </u>			
153					[<u> </u>	<u> </u>	l	l		L		
154										1	1			
155														
155							<u> </u>		<u> </u>	<u> </u>				a second seco
156							ļ		Ļ			<u> </u>		
157								[Í	1		Ĺ	L	
158											I	Ι		
150				· · · · · · · · · · · · · · · · · · ·			<u>+</u>	<u> </u>			<u>+</u>			
109				h				f			<u> </u>		Ļ	
160							<u> </u>				<u> </u>			
161					ļ			l I						
162								<u> </u>		Г <u> </u>		I	I	
163			······					t	<u> </u>	+	†	<u> </u>		
105					}		<u> </u>	↓	ļ	┟───		<u> </u>		
164							1	<u> </u>	[\bot	L	\bot	L	
165	Í				ł		1	ł		1		1		
166				[1			t	1		
167							t		t		t	<u> </u>	<u> </u>	1
			ł	┟			+	t	<u> </u>	+	+	<u> </u>	+	<u> </u>
168			ļ	ļ	ļ		 	 	 		 	┣──	 	<u> </u>
169			1 =											L
170									<u> </u>				[
171			<u> </u>	t	t	t	t	t	t	1	1	<u>† </u>	t	1
			<u> </u>	<u> </u>	 	<u> </u>	+	┼───	t	╂───	t	<u> </u>	+	<u> </u>
172			ļ	ļ	·	ļ	 	 	 	┣──	Į		 	}
173			l	L								L	L	ļ
174							1	1	1	1	1	L		
175			l	l	1	T			1	1	1	,		
			<u>+</u>	<u> </u>	<u> </u>		<u> </u>	t	<u>├</u>	+	t	<u> </u>	t	<u> </u>
1/0					 		├ ───	<u> </u>	<u> </u>		╂────	╂	╂───	+
177			L	L							1	L		L
178								1	1	L		1		
170				1	T	F	<u> </u>	<u> </u>		<u> </u>	1	T		
			ł	<u> </u>	<u> </u>	 	+			+	+	+	 	1
180			}		·····	<u>↓</u>		┣	┣───			┢────		
181			L	L	L								1	1
182			1	1										
102		<u> </u>	t	t	1	<u> </u>	1	1	<u> </u>	1	1	1	1	
163			}	f	<u> </u>	<u> </u>	 	 		<u> </u>				
184				l	L	L	<u> </u>	 						
185							1				1	L	1	L
186			1	1	1	1	<u> </u>	<u> </u>		1		<u> </u>	1	
		<u></u>	<u>}</u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>	 	<u>+</u>	 	t	<u>+</u>		
187	·		 	ļ		 	ļ	 	 					
188			L	l	L	<u> </u>			1	L.				
189			1	r i						1	1		T	
100			t	t	<u>†</u>	<u> </u>	<u> </u>	t	t	<u>+</u>	t	t	1	
190			ł	<u> </u>	 	<u>+</u>			┝	 	├ ───	┢		
191			<u> </u>	L	L		L	L			1	L	 _	[
192					1	L		1	1	1	1	1	1	
103			1	1	r	T	1	1		Γ	T	<u> </u>	T]
		<u>├</u>	t	 	+	<u> </u>	+	 	t	+	t	t	f	<u> </u>
194			 	 	l	ł	<u> </u>	 	 	 	 	 		<u> </u>
195			L		L	L		1						
196	· · · · · · · · · · · · · · · · · · ·		1	I	[1					1			1
107		·	t	t	1	<u> </u>	1	1	1	t	t	1	t	1
177		<u>├</u>	┟────────────────────────────	<u> </u>	 	†	+	{	┢╼╼╼	{	 	┣───	┫	<u> </u>
198		L	ļ	<u> </u>	 		Į			Į	ł	 	L	L
100							1	1		1	1	1	1	
			1	1	T	T	1		1	1	1		1	T
200			1			i		•					1	

TM 11-5805-652-14 ISSUE 1

2-11

A	В	С	D	E	F				G			н	J	ĸ	L	M	—				N			
<u> </u>	· · · ·	T	<u> </u>	T	Ci	0	0		0	0-	A -				-		 	Clas	ss al	Serv	rice	<u>Restr</u>	ictic	<u>, n</u>
EPABX	Not Assigned	Line	Attd Line	Conf	Equip-	Hunt	Hunt	Hunt	l Group Hunt	Group Hunt	Group Hunt	Power Failure	ANA	Single Digit	Exel Over-	I ntercept		1	2		3	4		5
		Assigned		Line	ment	1	2	3	4	5	6	Cut Thru		Access	ride	110.	Y	N	Y	NY	N	Y	N	YN
00							1													T				土
01			<u> </u>		<u> </u>	<u> </u>			├──								\vdash		-	+	\square	⊣∔	\square	+
03		<u> </u>									<u> </u>	<u> </u>					\vdash		-+	╉		┝─╂	+	╋
04																-			+	+				+
05					 															T	\Box	\Box	\square	\bot
07									<u> </u>										-+	-	+	┝╌╉	+	╇
08					<u> </u>												\square		+	+	+	\vdash	+	-+-
09																								T
																			_			\square	_	-
12																	\vdash		+	+	+	┝╌╋	-+	+-
13																	\square		+	╈	╆		+	╋
14	·																					\Box		T
15												<u> </u>					Н		+		+	┝╌┥	_	╇
17					ł												Н	\square	+	+-		┝─╋	╉	+-
18																	Π						1	+
19									ļ										\square	T		\Box		T
20																	\square			+	+-	┝┥	+	╋
22																<u> </u>	\square	Η	+	╈	+	H	+	+
23																	\Box							1
24 25									<u> </u>	ļ							\square	Ц	ļ	\downarrow	+	⊢↓	-	+
26										<u> </u>							\mathbb{H}	$\left \cdot \right $		╋	+	┝┥	+	╋
27																				╈	+	H	+	╋
28																	\Box			Ţ		♫		1
29 30																			-	+-	╋	┝╌┥	-+	╇
31																	\square		\rightarrow	╈	+-	┝╼╋	+	╋
32																								1
33																			_		4_	H	_	
35																	\vdash		+	╉	+	┝╼╋	+	╺╋╸
36																			+	+	╈	\vdash	\neg	╋
37																						\Box		土
38 39																			_		╇	\vdash	_	+
40																			+		+	┝╋	+	╋
41																	Η		-			\square		
42																						\Box		T
43																	\vdash		-	+-	┿	╞╌╉	+	+
45																	$\left \right $		+		╉╼┥	┝╌╋	-+-	╉
46															_								T	
47																					1_	\square	_	╇
49																		\neg	╉	+		┝╌╊	+	╋
50														_										
51																						\Box	\square	\bot
53																			╉		+	\vdash	+	┿
54																			-	+		\vdash	-	+
55																								土
56 57																	\square	4	_[ſ	\square	Ц	Ţ	T
58																	Η	-	+	+	+	┝╼╋	╉	┿
59																	h		+	+-			╉	╋
60]																		\square		T
61 62																		[-	+	\square	$\vdash \overline{+}$	Ţ	╀
63																	⊢┦	+	+	╈	+	┢╋	+	╋
64																					П	口	1	T
65																	\square	Ļ	-	+	H	$\mid \downarrow$	Ţ	Ŧ
67																		╉	╉	+	H	┢╋╋	╉	╋
68																			+		┢┤		+	+
69																		1	T	Ţ	П	口	T	T
<u>70</u> 71																	-	+	+	+	⊢┤	┢╋	+	╇
72					_												+	╉	+	+	┼┤	+	╉	+
73																			1	T	\square	土	1	T
74 75																		_	Ŧ	\downarrow	П	_	Ŧ	
76																	+	+	+	+	┝┤	-+	╋	╋
77																	+	╋	+	+-	\square	+	+	+
78																	1	1		T	\square	士	1	T
79]				\square	Ţ	T	F	П	1	T	F
81																	-+	+	+	╋	┝╌┥	+	+	╋
82																	\uparrow	+	+	╈	┢┦	+	+	+-
83																			T	T	\square	土	1	T
84 95																	\downarrow	-	+	\downarrow	НĪ	\bot	Ļ	\downarrow
86																	+	+	+	+	\vdash	-+-	╋	+
87																	+	+	+-	+	┝─╋	+	+	╋┯┥

Line and Feature Assignments

Table 2-6.

10																					1 (
11																		Т	Т	Τ				
12	1																	T	Т	Т			Т	
13	+																		T			Τ	Т	
14	+									+								-	+					
14	+																	-+	-+-	┿		-+		
	+																	-+	+-	╉	+	-+	+	
10																		+	+-	┿	+	-+	+	+
17																		-		+-	╉╼┥		-+-	+
18																		-+	-+-	+	+			+
19																		_	_	4-	-			$ \rightarrow $
20																								
21	T																							
22	1												-							Т				
23	+																			T				
24																		-	-		-		+	
24	<u>↓</u>																		-+-	+	+		+	
25																			+	+	+		-+-	╉┥
20	<u></u>																		-+-	╉	+	┝─╋	-+-	+
27	ļ																	_	\rightarrow	+-	_	┝╌┽		+
28																			_	4		┝╍╋	_	\rightarrow
29																			\bot					\square
30																								
31																					\Box			
32	1	†																	T	Т	Τ		T	
33	1													h					+	1			1	
34	+																	1	+	+	1		+	1-1
25	+	<u> </u>							<u> </u>				 					\vdash	+	+	+	┢─┤	+	+-+
	+	<u>├</u> ──											 				+	\vdash	\rightarrow	+	+-	┢─┤	-+	+
						ļ												\vdash	-+	╉	+-	┢╌┤	-+-	+-+
37	+	 								ļ						 	 	\vdash	┍╼╋	+		\vdash	+	+
38	<u> </u>	L			L								L	ļ		 	1	\square	┢┻╋	+		┢╌┥	+	+
39																L					4	\square		
40																						\square		
41																								
42														ſ					iΠ		1			
43														1					П	Т		Π	Т	
44																			\square	T	_			
45	+												<u> </u>	<u> </u>		<u> </u>	+		r-+-	+			-+	
4)															<u> </u>	<u> </u>	+		┢╋	-			-+	
40	+														 		╉──		┢╼╋	+	+	+	-+	
4/														 	ļ	ļ			┢─╋	-			-+	
48														 		 			┢─╋				-	-
49														ļ	ļ	ļ	+		┢╼╋					
50													ļ	ļ	ļ	L	_		┢╍╋					
51															ļ		4		\square	_				
52																								
53																								
54																								
55		1															T			Т				
56	1															1								
57	1					<u> </u>			1					1		1	1			T				
58				 		<u> </u>							 				1		\vdash t	-			-	
50	+								<u> </u>				+	ł			1-		H++	+	-1-			
27	+				<u> </u>	 	┟────	ł	ł	<u> </u>		·	+	1		<u> </u>	+-	+	┢╍╋	-+-	+-	╉┈┥	┝─╋	+-
00	<u> </u>		i	 		 	Į	<u> </u>	<u> </u>		<u> </u>			<u> </u>	+	<u> </u>	+	+	┝╍╋	+	-+-	╋┥	┝─╋	
61	+	 	1	ļ	 	 	 	[}	╂───	╂────	<u> </u>	<u> </u>	+		┢╍┥		+-	+	\vdash	
62	_	 		 	ļ	 	ļ			 	 	 	╂───	+	┨────		+	╂	┝╍┥	+	-+	+	┝─┥	
63	+	!		 	 	 			 	L	 	 	↓ −−	 	 	┥────	+	╂──	⊢┥	+	+	+	\vdash	
64		L		L	ļ	ļ	ļ	ļ	ļ					↓	 	ł	+	_	⊢∔	+		+	┝╼╋	
65				I		L	L	l	I		ļ	ļ	 	 	 	 	+		H	\rightarrow		+	\square	
66							L	L				L		<u> </u>	L	L	+	1	⊢∔			+	\square	
67		<u> </u>	1									1					1		\square				Ц	
68	1	1	1	1	<u> </u>		[1											LT					
60	+	t	1	<u> </u>	1	1			1		1	1					Γ		Π		Ţ			
70	+	<u> </u>	<u> </u>	<u> </u>		1		 	1	<u> </u>	<u> </u>		1	1	1	Γ	1		\square	1	T	Г	Π	Т
71-	+	t	+	<u> </u>	<u> </u>	t	<u> </u>	1	t	t	<u> </u>	t	1	1	1	1		\mathbf{T}	Π	-†	╈			
	+	 	<u>├</u>	 	<u> </u>	ł	ł	 	<u> </u>	<u> </u>	<u> </u>	ł — — –	 	<u>+</u>	<u>† </u>	1	+	1	+	-+	+	+	\vdash	+
72	+	ļ	<u> </u>	 	 		 	┠		┣────	+	 		+	 	+	+	┨──	┢─┫	-+	-+-	+	⊢┤	
73		 	 	I	ļ	 	ļ	 	┣	 	 	 	+	+			+	+	┝╍╉	┥		+	⊢┦	
74	1	ļ	ļ	L	ļ	 	 	 	 	ļ	ļ	<u> </u>		+		 	+	–	┝─┥	_	-+-	+	┝─┥	
75						L	ļ			ļ	 	 	_	 	 	 	╋		╄╼┩	-		+	⊢┥	
76									L		L	L	1	<u> </u>	L	ļ	4_	Į	┯┪	_		+	\square	
77	-	1	[[\Box					
70		t		1	T	Τ	<u> </u>												\Box		Γ			
1 10	+		-	-		4	t	<u> </u>	T	1	1	I		1	1	1	1	T			-			T
70				1							1		1		1				1 1	- 1		1		
79						ļ							1		+	+	┿	+	┝┥	+	+	+	\vdash	+-
79 80													<u> </u>			<u> </u>	+-		╞┥	+	+	╞		
79 80 81																			┝╼┫ ┝╼┫ ├╴┨					
79 80 81 82																								
79 80 81 82 83																								
79 80 81 82 83 84																								
79 80 81 82 83 84 85																								



ŝ

Α	В	С	D	E	F			(;			Н	L	ĸ	L	M					N	-		
	filet	Tie	A 44 -1	Trk	Stations	Group	Group	Group	Group	Group	Group	Power		Sinale	Exel	Intercent	-	Clas	<u>is o</u>	Ser	2 2	Rest	<u>ictio</u>	<u>n</u>
EPABX	Assigned	Line	Line	Conf	Equip-	Hunt	Hunt	Hunt	Hunt	Hunt	Hunt	Failure	ANA	Digit	Over-	No.			ر ار		, , , , , ,		+	Ţ
		Assigned		Line	ment			,	- 4					ALLESS			Ľ	N	4			$\left \right $		1
01																	\square		-	+	+		+	\pm
02																					T		\neg	Ŧ
03																	\vdash			+	╋	H	+	+
05		·																			T	\square	t	\pm
06																		\Box					\neg	\mp
07																	\vdash			+	╋	+	-+	╋
09																	-	$\left \right $		+	╉	+	- +	┿
10																					T		\square	1
11									:											\rightarrow	+	+	┢╍╋	┿
13																	\uparrow			+				\pm
14																							\square	\mp
15																<u> </u>	┢				+-	+	┢╼╋	╋
17											÷						T				+			1
18																					_		П	\downarrow
19 20								ļ				ļ		 							+	┼╌	┢┼┥	╉
21																	\Box	\square					日	1
22								ļ			<u> </u>		<u> </u>		<u> </u>	<u> </u>	1		Ц		+	+	\square	-
23																+	╀╴	H	H	\vdash	╉	+	\vdash	+
25																					1	T	口	土
26							ļ						ļ			ļ	\downarrow			\square	+		μ	+
21 28											<u> </u>		┼──				+			\vdash	+	+	┝┥	+
29												<u> </u>					1			口		T	ば	エ
30														<u> </u>			+	+		μŀ	+	+	H	4
32								<u> </u>				<u> </u>					┢	┢		┝┼	+	╈	┝╉	┽
33												:					T				1	T	□	1
34												·				 		╞	╞	\square	+	+	\square	╉
35												┣	<u> </u>	<u> </u>			+	┢─	┝	++	+	┿	┝┥	╉
37												È										T	Ħ	土
38											ļ	ļ							L		4		\square	
39				<u> </u>							<u> </u>	┨	╂──	┨			╋	+	┼─	+	┉╋┉	┿	┝┥	-+
41								<u> </u>				┝┤────									1	1	\square	
42												Ĩ					-				+		П	\neg
43				Į								 	<u> </u>			+	┼╴	┝	╂─	┠┼	+	╋	┞┤	\dashv
45		<u> </u>								<u> </u>					<u> </u>				T			\uparrow		士
46																	+			$\left[+ \right]$	+	-	Н	\vdash
47													-				┼╴	┢	┼─	++	╉	┿	┼┦	\dashv
49													1				T					T	\square	
50											[<u></u>			┢		\vdash		_	4	\square	⊢┥
51 52	<u> </u>												+	<u> </u>			╋	╋╌	┢	┞┼	-+-	╉	⊢┤	┝╌╉
53	<u> </u>					1											t	t					\square	
54											ļ				_		+	+	\vdash	\vdash	+	+	\square	┝┥
 	<u> </u>	<u> </u>				<u> </u>					<u> </u>		-		+	<u>+</u>	╋	┢	┢─	┝┼	╉	+	╉	\vdash
57										1				1			T			口		T	\Box	
58				ļ		ļ							<u> </u>	<u> </u>		<u> </u>	╀	\downarrow		⊢↓	+	+	\mathbf{H}	┝╼┫
59 60	<u> </u>			<u> </u>	<u> </u>		 		+			<u> </u>	+	+	+	+	+	+	╋	┝┼	╉	╋	Η	⊢┥
61							1	1		1	1		1				1			\square	1	T	\square	口
62			<u> </u>		<u> </u>					 	╂───	╂───	┼		 	+	+		╂	$\left \cdot \right $	-	╋	╄┦	Н
64	+																	t			1	1	\square	┢
65	1			1		<u> </u>											-	F	F	ļП	Ŧ	F	\square	ļЩ
66	<u> </u>	 			<u> </u>	 	+	<u> </u>	╂	+	╂	╂	+	+	+		+	+	┢	┝╌┥	╋	╋	┢┥	┢╼┥
07 58	<u> </u>	+	<u> </u>		<u> </u>	<u></u>	1										1	\mathbf{T}	t			1	\square	┌╴
69						[-		<u> </u>		\bot	F	F	H	-	+	\square	H
70	<u> </u>		 	+	 		+	<u> </u>	+	+	+		1	+	+	1	+	+	┢	┢╌╂	╉	+-	┢┥	┢╌┥
72	<u> </u>		<u> </u>								1			1	1	1	T	T	T	\square	1	1	\square	口
73							+	+			\vdash	+	+		+		+	+	+	┞┨	-	+	H	┝╼┥
74	+	<u> </u>	<u> </u>		<u> </u>		+	+	+			+	+	-	+	+	╋	+	+	++	+	╋	Η	┝╌╉
76	+										1			1		1		T	1	\square	1	T	口	口
77	1		_									↓			<u> </u>		+	\downarrow	\downarrow	\square	Ŧ	+	ЦĪ	⊣
78			+				+	+	+	+		+	+	+	+		╀	+	+	┟╌┨	+	╉	H	\square
80	+	+		+	<u>t</u>			<u> </u>					1				\pm	T		Ħ	1		\square	
81	1		<u> </u>	1			Ţ						+				-	\downarrow	Ĺ	┟┦	Ŧ	+	П	ļ
82 97		╂	<u>+</u>	<u> </u>	+	+	+		+	+	+	+	+	+	+		+-	+-	+	┟┼	╉	╋	H	$ \neg \uparrow$
84 84	+		+	+	+	+	+	<u>†</u>	1								1	\mathbf{T}	T		1	1	Ħ	
85						1	1		1				Ţ	1			T	F		П	1	T	\Box	口
86	1	ļ		+	l	 		<u> </u>	- <u> </u>	+				+	+		╋		+	┝╌╿	+	+	H	-+
87				1		<u> </u>		+		<u> </u>	+	+			+		╋	+-	╀	┝╌╋	-+-	+-	⊢┥	-+

Table 2-6 (Cont'd) Line and Features Assignment

	a bardated and a state	a na siyo nafarataka nafash	ine coliseitere disc	in series and the	io i Gibara ila mala	ىلا - بەمدارلىنىلەمبە مەركە		1	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		and the second states of the second se	an and the first second se		n o n oc mang	والالالم والمعالم		and a	will for a	() 	Sec.	ومغربا فينس		198 (1996)
12							Į		 	ļ	ļ	ļ	ļ	 				_	<u> </u>	\square	_	11	
12		+	<u> </u>					1	ļ	ļ	ļ		ļ							\square		\square	
13	L																						
14																			T	Т		ПТ	
15					1	1		<u> </u>	<u> </u>	 	<u>†</u>		· · · · ·					-+-	+	┢─╋	+	╋╋	+
16		+	+	+	<u>+</u>	+	ł	 	<u> </u>								\vdash		+	╉╍╋		╉┈╋	-+
	+	+				<u> </u>		<u> </u>					L					_			_	+	_
11	 	+	J	_	 	1	L		L					L								\square	
18																						ΠT	
19						1				1				1								$^{++}$	
20					+	+			t		ł			<u> </u>				-+-	+	+	-+-	┿╍╋	
21					ł	ł	<u> </u>		 	ł								_	+	┝─┟		++	_
								Ļ															
22									1													T	
23		1												<u> </u>					1			TT	
24				1		<u> </u>	1			1	t			1			\vdash	-+-	+	┼╌┼		╉╌╋	+-
25	<u>+</u>									<u> </u>			ł							┢━┠		╉┯╋	_
26	<u> </u>		+	+	···		ļ		 				I	Į					╇	++		╄╌╋	
- 20																							
									1											II		TT	
28																			1			\mathbf{T}	1
29				1														-+-	+	╋╼╋		╋╌╋	
30		+	+	+						ł	<u> </u>	·	 				\vdash	-+-		┢╍╋		╋╋	
21	 	+	+	+		+			ļ					ļ			\rightarrow	\rightarrow		$ \downarrow \downarrow$	_	+	
21	<u> </u>	+	+	∔	ļ	L	L		L	L													
32		L								L							T	Т	T		Т	T	T
33			1			1	1		1		1			Г					1	$\uparrow \uparrow$	-1-	\dagger	
34		1		1	· · · · ·		l	l	1	1	t		t~	t	 	 	\vdash	+	+	┼╌┼	-1-	╈	+
35	· · · ·	†	+	<u>+</u>	<u> </u>	+	ł		<u>+</u>	t	<u>+</u>		┟──~	<u> </u>			\vdash		+	╋╌╋		┿╋	-+
24			+	├ ───	<u> </u>	<u> </u>	ł		 	 	<u> </u>	ļ	 	 			\square		+	++		∔∔	_
	L	 	_	ļ	L	ļ	L																
37						L			1					[T				
38				[Γ		· · · · ·	· · · · · · · ·		t			\vdash	-	+	+	-	╈	-1-
39		1	1	1	<u> </u>	<u>†</u>			t	1		·	 ~	<u> </u>			\vdash		+	┢╼╋		╈	-+-
<u></u> 20		1	+	+	t	<u> </u>	┝		├ ───		 		 	 		 	\vdash	-+-	+-	┢┼┥	—	╉╌╋	
		+	<u>+</u>	+		<u>+</u>		<u> </u>	 	<u> </u>	 		—	ļ			\vdash		+	┢┈┢		┿╌╄	
41		ł	<u> </u>	╋───				ļ	 	ļ	ļ	L	ļ	L						\square		\square	
42																			1			ТТ	
43																							
44													1						1-			† †	
45			1	1									<u> </u>						+-		-+	┿┽	
46		<u> </u>	<u> </u>	t									<u> </u>							+		╇╌╋	
47		<u> </u>	<u>+</u>										L							\downarrow		+	_
41		ł								L													
48											l											TT	
49																	ľ				Т	TT	
50																							
51																			-	┝╌╂		╈	
52																				┝─┼		╉╌╋	
53		<u> </u>																		┝─┥	_	┿	_
50		[<u> </u>										ļ										
55																			T			ТТ	
56			{																				
57			1														-+		+	t t		╋╋	
58			1																+	┝─┢		╉╌╋	-+
50								·····									\rightarrow		_			∔	_
		ł	<u>├</u>										L						1	\square			
ou		ļ								L									1			ΤT	_ _
61																		T	Τ				
62						1												-	1		-1-	$\uparrow \uparrow$	1
63													<u> </u>				+		1-			++	
64						├── -						L					-	+-	+	╞╌╋	-+-	╈	
45			<u>├</u>			┝╌─┤							—				+		+	┝╌┠		╉╋	
02		<u> </u>	┢──┤	ļ													\rightarrow		+	+	+	++	_
00		L		i					L	ļ			L						1		_	\vdash	
67								1															
68								_									Т	T	Γ	Γ	T		T
69			1														\neg		1		1		1
70		1	tl			<u> </u>			I	··· · · ·							+	-+-	+		+	++	
71			<u> </u>			└──── 											+	-+-	+	┢┼┼	-+-	╆╌╋	-+
		ļ	<u> </u>						<u> </u>								-		+	┝━╋	+	╂╌╊	
12		L	 			 				ļ							\downarrow		\vdash	\square		+	
73																							
74																	_T		Γ	LT		ΙT	
75																	\uparrow						
76	· · · · ·		<u>†</u>														+	+	1-1		+	++	+
			<u>├</u>														+		╉╌┤	┢╌┽		╞╌╇	
			<u> </u>														_		+	┢┈╇╴	—	\vdash	
18		ļ																		\square		\square	
79																				LT		T	
80																	1	T					1
81			<u> </u>					i									+			\vdash		┝╌╋╴	+
			<u>├</u>														+	+	+-	┞╼╋		┝─┣─	-
<u> </u>			├ ────┥											 			_			┝╌┿		┢╼╟	-
55			ļ							ļ			l	I			_		\vdash			\vdash	
84		L																					
85																	T	T		T		T	T
86																	T	Τ					T
87		İ — — — — — — — — — — — — — — — — — — —	<u> </u>		b												-+	-+	┿╌┥	\vdash	+-	+-+	+-
<u>.</u> .			-				-										-	-					-



2-15

TM 11-5805-62-14 ISSUE 1

N Class of Service Restriction Μ Α В С D ε F G н J K L Group Group Hunt Hunt Exel 2 3 4 Single 5 Group Group Group Power 1 Tie Trk Stations Group Intercept Not Attd Digit Access Equip-ment EPABX Line Conf Hunt Hunt Hunt Hunt Failure ANA Over-No. Assigned Line YNYNYN ΥN Y N Cut Thru ride Assigned Line 1 2 3 4 5 6 00 01 02 03 04 05 06 07 08 09 10 11 12 13 14 15 16 17 18 19 20 ٠ 21 22 23 24 25 26 27 Table 2-6 (Contd) Line and Feature Assignments 28 29 30 31 32 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 ł 54 K 55 56 57 58 59 60

61												 	 	 			┝╍╍┠╸		┢┝╽	
62													 						\square	
<u>ó</u> 3												 					┝╼╇╴	\perp	\square	$ \rightarrow $
64												 	 			-	\rightarrow		┢─┥	
65												 	 	 			┝━╋╴		\square	$- \square$
66												 	 				┝┻╋		\vdash	
67												 	 	 $ \downarrow \downarrow$	-		┝──╋	——	┢─┥	
68												 	 	 \vdash	_	+	⊢ -		╄	-+-
69							L					 	 	 ┝┥		+			∔∔	\dashv
70												 	 	 $ \rightarrow $	_	+			┟─┼	-+-
71												 	 	 \vdash		┥━┥	⊢		┢╌╁	-+-
72												 	 	 \vdash	_	+	┝─╋		\vdash	
73				L								 	 	 \vdash			-+	<u> </u>	\vdash	
74												 	 	 \vdash		+	-+-	- 	┢─┟	++
75												 	 	 \vdash			-+-	'	┢─┤	
76					 		L					 	 	 \vdash	_	+	┝──╂╸	- -	╄╌╋	-+
77						L						 	 	 		+	┝╾╌┫╌	- 	┢─┥	\rightarrow
78						<u> </u>						 	 	 $ \rightarrow $	-	+		'	<u>∔</u> -i	
79				L								 	 			+	-+-	_	┢─╁	-+
80					L			ļ	ļ				 	 \vdash	_	+		<u> </u>	\vdash	
81				L	L			ļ	L				 	 $ \downarrow $	_	+		- 	┢╌┧	-+
82						<u> </u>	ļ					 	 	 $ \rightarrow $		+	-+		┢	
83			1		L	Ļ	ļ	ļ	ļ	L		 	 	 \vdash	-+-	+	┝━╋╸		┢╾┥	
84						L		L					 	 \vdash	-	+		<u> </u>	┢──┟	
05	L	L						L							_		┝━╋		\vdash	$ \rightarrow $

					_	-	-						1170 (0.000 v.Ders	Contention of the local division of the	- 4 - 90 - 10 - 10 - 10 - 10 - 10 - 10 - 10	Contract of the local division of the	and the second second			A				
		-		-			Research and the second second second second second second second second second second second second second se				The second second second						Γ							
<u>++</u>	and the second statement of the second second second second second second second second second second second s		TO BE WEIGH ADALINE										T	Ţ	I		Τ	T. Same	Π					
12						and deriver ordered and		T	T	and the state of the state of the state of the state of the state of the state of the state of the state of the		Contraction - Million - Annal - Annal		1910 - Gill Indone	Concession of the local division of the loca	Contractionage with		ALCONO.			-127		r i	
13	and to 1 per obliga antisolation of 1004	- And Andrewski, All. 4	- araana	an sugge of the second second	and the second se	TT BRIDGENER, T-BRIDGE		al constraints a second second second second second second second second second second second second second se	Contraction and	and a second second second second second second second second second second second second second second second			-	Manazaonowen veze				-						a and the local da
14	and the state of t		and the state of the state of the	na caracterization in			ar a great and a second and a second and a second and a second and a second and a second and a second and a se	1									-					-		
EG			and a state of the second second second second second second second second second second second second second s	an an an an an an an an an an an an an a	The second state			-			and an an an an an an an an an an an an an					and the second second second			600 100 http://	1	-			
and designed and	and the second second second second second second second second second second second second second second second	and according to the	-		and the state of the second	an call - courses - quint																		
in the second		the second second second	-	and the second															Π	T	Ī			
11			L				1		I		ALCONG TARBUTCHE	and the second second			a and a second second second second second second second second second second second second second second second		and the second		1		States and States and	Transfer 1		itinati nativate
Į.	L		I		Contraction of the local division of the loc	and the state of t					The second second second second second second second second second second second second second second second s			and the second second second second second second second second second second second second second second second		and a second second second second second second second second second second second second second second second	-				2017-20 2017-20			
10	en les la source de la comparte de l	COMP & ALLANDONNES 1	Calendar - Calendar	- Carlo Carlos and Carlos of		an an an an an an an an an an an an an a	6 6 00 - 1000- 100	an ann an Anna an Anna	-		-	-					-	-				-		Annual Second
26	an an air an an an an an an an an an an an an an	WAY YOUR DOLL AR .	content instants ?	an an an an an an an an an an an an an a	er en norskriver storet	an an an an an an an an an an an an an a	-		-				-					-			-			
- 000 - 10 2 72	and the state of t	A ROAD AND AND A MADE			internet water and	contractory of the		-					<u> </u>											
FF.	and the second sec	10 - Sector - An and the	A DESCRIPTION OF THE PARTY OF	-	1											· · · · · · · · · · · · · · · · · · ·	Τ				2521-74 1263 2			•
- 22													an laine an anna an an an an an an an an an an a		The second second second second second second second second second second second second second second second s		-							-
29		A REAL PROPERTY AND A	Designation of the state of the state		and the second second second second second second second second second second second second second second second			The second second					-			and a survey of the survey of the survey of the survey of the survey of the survey of the survey of the survey	+						┝╌┿	-
24	Contraction of the local distance of the loc	Contribution and the second	10 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10 IN. 10				-		-					and all sectors and and						_	_			-
36		A CONTRACTOR OF THE OWNER OF THE OWNER			-		-	-		-								-						
34		and a state of the	-	State and the second states of the	a second of the second designed	-		-								l								
60	and the second second	Second Second Second Second	Division of the second section	-													1^{-}				Т		Т	
<u>[</u>									I	T							and a strength to		T					
28								Ī					-		and to be seen as the		-				-	-	-	-
20		Calendramy, way prod	a all the following and a subscription		and the second second second second second second second second second second second second second second second			İ							1	<u> </u>	-			-	-		-	_
Ď				-		-	1	1	<u> </u>	-	+		<u>+</u>		<u>į </u>	ļ		L	\square					
11	And the second second second	S MANUTA OF COLUMN STREET	-				<u> </u>						 					L						
1				-		<u> </u>	1			1								[Π	Т	Т		T	
<u>×</u>													[I	l			ГŤ	-	1			
33				L					T		T	1							┢╍╋	╈	+-	1-1	+	
34			1	T			1		1	1	t	t	 		<u> </u>		+		┝╌╋		+-	╉╼╴	1	+
5	1000-100-0-00-00-00-00-00-00-00-00-00-00		l'			1	1	1	<u> </u>	+	+	+							⊢∔	-		+		
14			-	-		-		<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>							Ш					
		STARL WALLARD	-		-		-		<u> </u>		<u> </u>								LT	_[LT	
<u> </u>	Section Change Section		ļ														Γ		T	T	Т	П		
38	· ····································					L					1		1		I					-†				
30						I	I	İ	T			-	-				\mathbf{H}		┝╼╋	┿	+-	┿╌┥	\vdash	
46		T	and and and		T		Construction of the second		1	1	1		t				+		┝─╋	-	+-	╋╾╼	┝╼╋	
41				Danie w Gebiednich w V	and the second second					† –	1						┢┥		┝╌╬	- -	_	╋┈┥	┝╼╋	
42	1	Contraction of the second second			International Debutto														┝─╁	4		-		_
43	and the second second second second second second second second second second second second second second secon	and the second second second second second			Contraction of the operation of the oper											Second pointing of								
W.S.	Mantenand - Monopopor	a sector and the sector sector		Contraction and the second																				
6(6) The second second second second second second second second second second second second second second second s	Constant over " Antonic Statements	STATES ANT COUNTRY AND		Concession of the local division of the loca																Т	T			
43	Station womanan w									Τ										1			Ť	
46																	-			-		╋╼╍╢		
47		11 THERE COMPANY AND AND ADDRESS	a an ann an an an an an an an an an an a	and the second second second second second second second second second second second second second second second						-	-			a version of the local party page							-	+	-	
48	and a state of the state of the state	Constant Descention, 1997		-	neonensisten av (ma) (sy	Distriction and the state of th			and a construction				÷					-	-	_	_		-	┯┥
AND DECK LODGE DURING	Streampion of Concerning	Tabletic of The State Street Str	0					3																
446 1				10.00 - 0.000 - 0.000	THE REPORT OF THE PARTY OF THE PARTY OF THE PARTY OF THE PARTY OF THE PARTY OF THE PARTY OF THE PARTY OF THE P	Contractor property in the					Contraction of the	17 designation of a special	<u></u>			******		_			-			
44 60	and the second second second second second second second second second second second second second second secon	New Color	and the second second second second second second second second second second second second second second second	composition of the later								A - Charles and a species										\square		
<u>44</u> 50		andra andra da andra da angla da angla da angla da angla da angla da angla da angla da angla da angla da angla Angla da angla			angenerative of the statement																		7	
44 50 51																								
44 50 51 52																								
40 51 52 53		Northanist, name and a state																						
44 50 51 52 53 54																								
44 50 51 52 53 53 54 55																								
44 50 51 52 53 53 54 55																								
44 50 51 58 53 54 55 55 56																								
44 50 51 52 53 54 55 55 57																								
44 50 51 52 51 52 51 52 55 55 57 58																								
44 50 51 52 53 53 54 55 57 58 58 59																								
44 50 51 52 53 53 54 55 54 57 54 54 50 60																								
44 50 51 52 53 54 55 55 57 58 57 58 59 60 60 61																								
44 50 51 52 53 54 55 55 57 58 57 58 59 60 61 43																								
44 50 51 52 53 53 54 55 57 57 58 59 60 61 62																								
44 50 51 52 53 53 54 55 57 58 57 58 59 60 61 62 63																								
44 50 51 52 53 53 54 55 57 58 57 58 59 60 61 62 63 64																								
44 50 51 52 53 54 55 55 57 58 57 58 59 60 61 62 63 64 65																								
44 50 51 52 53 53 54 55 57 58 57 58 59 60 61 62 63 64 65 66																								
44 50 51 52 53 54 55 57 58 57 58 57 58 57 58 59 60 61 62 63 64 65 66 67																								
44 50 51 52 53 54 55 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 58 57 57 58 59 60 61 62 63 64 65 64 65 64 65 64 64 65 64 64 65 64 64 65 64 64 64 65 64 64 64 64 64 64 64 64 64 64 64 64 64																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 65 66 67 68																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 65 66 67 68 69																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 65 66 67 68 69 70																								
44 50 51 52 53 54 55 54 57 58 50 61 62 63 64 65 66 67 68 69 70 71																								
44 50 51 52 53 54 55 56 57 58 50 61 62 63 64 65 66 67 68 69 70 71 72																								
44 50 51 52 53 54 55 56 57 58 50 61 62 63 64 65 66 67 68 69 70 71 72 73																								
44 50 51 52 53 54 55 56 57 58 54 57 58 54 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74																								
44 50 51 52 53 54 55 56 57 58 54 57 58 54 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74																								
44 50 51 52 53 54 55 36 57 58 59 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75																								
44 50 51 52 53 54 55 36 57 58 50 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 63 66 67 68 69 70 71 72 73 74 75 76 77																								
44 50 51 52 53 54 55 36 57 58 59 60 61 62 63 64 63 66 67 68 69 70 71 72 73 74 75 76 77 78																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79																								
44 50 51 52 53 54 55 56 57 58 59 60 61 62 63 64 65 66 67 68 69 70 71 73 74 75 76 77 78 79 80																								
44 50 51 52 53 54 55 56 57 58 59 51 54 57 58 60 61 62 63 64 65 66 67 68 69 70 71 73 74 75 76 77 78 79 80 81																								
44 50 51 52 53 54 55 56 57 58 50 57 58 50 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81																								
44 50 51 52 53 54 55 56 57 58 50 57 58 50 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82																								
44 50 51 52 53 54 55 56 57 58 59 51 54 57 58 59 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83																								
44 50 51 52 53 54 55 56 57 58 59 51 54 57 58 59 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 .54																								
44 50 51 52 53 54 57 54 50 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 -744																								
44 50 51 52 53 54 57 54 50 57 54 50 60 61 62 63 64 65 66 67 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 .44 85 86																								





A	6	C	0	f	F	G					н	н ј	К	L	M	N Class of Service Restriction								
	Net	Tie	Aitd	Trk	Stations	Group	Group	Group	Group	Group	Group	Power		Single	Exel	Intercept			<u>ss α</u> 2	T	vice 3	Kestr 4		5
EPABX	Assigned	Une	Line	Conf	Equip-	Hunt	Hunt	Hunt	Hunt	Hunt	Hunt	Failure Cut Thru	ANA	Digit	Over-	No.	-		<u>,</u>					
64		Assigned		Fille	ment		6	,	4		•	Cut Inru		VCCE22			–	N	4			╫╹		┦╝┨
	and the second second second second second second second second second second second second second second second	**************************************			antan karangan karang					na supointe a faith a f		-					┢─	\square		╈	╈	┼┼	+	H
64			10000000000000000000000000000000000000								a an an an an an an an an an an an an an									\square		П	\square	\Box
03			ana ang ang ang ang ang ang ang ang ang	a constant														μ		-+-	╋	+		+
05																	┼─			╉	╈	╉╌┨	+	
Ŕó																							T	П
07																	┞		\vdash	_	+	+		╇┈┥
<u>89</u> 00																	┢	┼─	\vdash	+	+	+	-+-	+-1
10																				1			T	口
11																<u> </u>	\vdash	-	\vdash	-+		+	+	++
13					and a subscription of the								<u> </u>			<u> </u>	╀╴	┢	H	-+	+			
14										andre des de la competencia de									П		_			П
15						ند. مراجع ومعرار						ļ		 			+-		$\left \cdot \right $	-		+	┝╌╋╼	+
10												<u></u>				<u> </u>	+	┢	$\left \cdot \right $	-	-+			+ -
18																		Γ				T	\square	\square
10					tere erede diverse					lenger, and picture de						 	+-		\vdash	-			┝─┿╸	╇
20		a a maint an fairt a state an															\uparrow	+				+		+
22																							\square	\square
23							<u> </u>					<u> </u>		<u> </u>		 	+	+	\vdash		+	+	┝╋	┿┥
24 25														1			+	╈	H	H				╉┥
26																	T	T	\Box			T	F	
27							<u> </u>							-			-	╋	\vdash	$\left \cdot \right $	+	+-	┝╌┠╴	+
<u>68</u> 26				_	a an an an an an an an an an an an an an			+		+								\mathbf{T}		H				
30					a an an an an an an an an an an an an an													Ţ					П	-
11					-	-				<u> </u>							-	_	+		+	+	┝─╋╸	+-'
11															+							+		
34																							\square	\square
19																-	+	_	-		+		┢╍╋	+
									+						+		╈							
38																			-				П	-
30																	+	_			_		┝╼┿	+-
		-	-												1		+	+						
42																					_		П	—
43					_			-		<u> </u>							_	+-	+	_	_	_	┝┼	+
44	-						+			1			-				╈		+				++	
46					-													Ţ					П	—
47	_			the the substance.									-	+			-	┢		-	$ \rightarrow $		┢╍╊	+
48							+				1						╈		┢				\square	
50															_		1	_	Ļ		_		\square	\bot
51			*******									_	-				┿	┿	┝	\vdash	-+	+	╆╋	+-
53		+					+										T						Ħ	上
54																	_	_	-				╆╋	+-
<u>55</u> 64		_															╉	╈	+-	┢		+	++	+-
70 97	-				-				-								T							土
58			-				_										Ļ	-	+	ļ		+	╇	
50			-		-		+	+	+		+				(m) (1		╉	-+-	╉─	-	┝─┤	+	┿╋	+-
61																	1	1	1	F			廿	十
62			<u> </u>					+	+		-				+	+	╉	╋	╋	+-	┝╌╢		╉╌╋	+
63 64	+		+	+													1	1	T		\square	1	\square	土
65									1			_					+	+	+-	┢	\square		╄╋	+
66		_								+			-	-	+	+	╉	+	+	+	⊢	_	╆╉	_
61 61			-	+													1	1	T		\square		口	\mp
69									I	1	<u> </u>		-			_	╇	+	+	┢	⊢┨	+	╆╋	+
70		-			-		-	-		-		-					╈	╈	\pm	1	H		$\pm \pm$	1
12																	1	1	T	L	П	T	\square	\mp
n							_	-			_		-			+	╉	╋	+	╋	┝┤		╋	+
74			-	-	-		+	-	-		-						1		1		\square		廿	工
10																	1	1	Ţ	F	П		$\downarrow \downarrow$	+
11							+		-	_				-	-	_	+	_	╋	╋	H		┨─┨	
18			+	_		-	-	_	-	_	_						\pm	1	\mathbf{T}	\mathbf{T}	Ħ		$\pm \pm$	
19	-																1	1	T		П		H	-
81			-					_			_					-		+	╋	╋	┝┤		╉╋	
82		2010 C 10 C 10 C 10 C 10 C 10 C 10 C 10		_							_						+	1	1	\mathbf{T}			\square	
83 M																		Ţ	T		\square		П	T
85													_	_			-	+	╋	-	⊢┤		╉╼┫	
80				_	-	_	+			+							\uparrow		╈	1	\square			1
			-																			T.	\square	T
						in the second					_					1								1

Table 2-6 (Cont'd) Line and Features Assignments

Sector Contractor	R F	a sea, garangiji ji	r 1		and an anna i sa	I		1		1	- 1		1	I	1	1	1	1	1	1	1 1	1.1	1.1	
(0)				oolidaa ahaanaa kaada				A COMPANY AND A COMPANY AND A COMPANY	antani tangananananya					i i i i i i i i i i i i i i i i i i i			T		Î	STREET	Π	П		
10			angeng pag anterneting anterneting		and a second second second second second second second second second second second second second second second			nieli anti-ang-ana a	an a condition of the second second second second second second second second second second second second second		Personal Contraction												_	
<u> </u>												and the second second second second second second second second second second second second second second second					_						antori Antarian	anaren a
12			o kolono in antina antina				ta dina mining di seconda da seconda da seconda da seconda da seconda da seconda da seconda da seconda da secon	and a start of the start of the start of the start of the start of the start of the start of the start of the st	and an and the second second			and the survey survey					_				-			-
13		1997-19-19-19-19-19-19-19-19-19-19-19-19-19-				12-12-10-10-10-10-10-10-10-10-10-10-10-10-10-		an an an an an an an an an an an an an a	gengele - individu	and the second second second second second second second second second second second second second second second		Constant of the Association					_			-	-	┿		
14		aline conservation of			Princer Carlos Constant (1998)							and a subscription of the last							-		-	╈┉╋	17/221 (19/2016)	
47 14			annessennes	- TORNESSICOL FORMUS		10 h. 2 million and 10 million and 10 million and 10 million and 10 million and 10 million and 10 million and 1	and the second second second second second second second second second second second second second second secon	ananan ana manganan	ander verseuwarder								-		-	-	-	++		ogooren an
- 10			No. of the Party of Contracting of the	r <u>u annara</u> raith	an a ai n chaileadh		ar. na sub-sub-subara	Reveals Country Mandag	ann - 297 - 3	and the Constant of Street	and and a second							12-035	-	-				a na sa sa sa sa sa sa sa sa sa sa sa sa sa
		an an an an an an an an an an an an an a	and a constant	ionaa a dhiingee	ana ana ang ang ang ang ang ang ang ang			er felgene følge Grand	angener versterenste		a ta ang mangangang ang mangang pang pang pang pang pang pang pan	and the second second second second second second second second second second second second second second second	Nucleasure and	A CONTRACTOR			t			-	1	\dagger	anata annaid	
10		Property and a second standard	erer ministeren son Same	regis-leader to association and a		and the state of the state of the state of the state of the state of the state of the state of the state of the					an agas na san sa sa sa sa sa sa sa sa sa sa sa sa sa	and a substant of the substant							-		1	1		
20		annan an Annais, a' a' Talain		CONTRACTOR OF CONTRACTOR	tida porencesistillita	and a state of the second state of the second	**************************************										-				Ť.	tt		a for a starter
21	ar Bangarian an an an an an an an an an an an an a	a. Texture distance of the second	an an an Anna Anna Anna Anna Anna Anna	1 antoningan (7 managana (7	parato II desenante actanta inte	Constant and Constant															T			
22			an an an an an an an an an an an an an a	and and a set of the s																		Ш		
23				and production of the second sec	and a second second second second second second second second second second second second second second second																		_	
24																				-		\square		
25			No. Contractor Concerns			170-00-00-00-00-00-00-00-00-00-00-00-00-0				-		1999	ur Braisseraute in				_	-		_		_		
26	-	tin , standardstade									and Contract Contractor							_			_	╇		
27	an Parti Crissioner un	-			and any straight of the second second		a les esterates a dipitat										_	_	_	_	_	┿		
20					articularia disrivus es							وروز بنار مارد و و و						-		_	_	┥─┥		
<u> </u>	-		-															-	-	_	-	┿┿		
											1							-	÷		+	+ +		
12				ali daga ya mananana - m																-		+	-	
33					String Dependences Stationer													Ĩ	T	1		11		
M						an an an an an an an an an an an an an a													1	1				
35	and a state of the							Andrew State (State											T	T		\Box		
36																			I	Ι	Ι	\Box		
11				1. (1. (1. (1. (1. (1. (1. (1. (1. (1. (Ţ	Ļ				Ц
38			enternetinenen, * ca		Fortici versi versi visco									-					ļ	4	╞	ᆔ	_	Н
39			- 100 - 100			an men	-	an and a star of the star of the star									Щ		4	╞	_	ᅱ	+	┢─┥
	-	and a subscription of the	व्यान्त्रीयात्र ज्ञ, ज्ञावन		and the second second second second		name income				erares ditented	الاستان المارين المارين الاستان المارين					H	-	-	_	_	+	-	┿┥
41	-			librate for an example		inervers strainslife	and the second second second second second second second second second second second second second second second		entration and the second								\vdash		+	┿		╞┥	-+-	+
AR -	#{ == =======	angene anne en anne anne		and a second second second second second second second second second second second second second second second	Barkman din Japan Kepilanangan		anima and						anain interne	and the second second						+	+	+	-	╉┫
44	and the second statement of the second second	arangin anananini dari da	Septembring and an other states of the	Contract + Contract Contract			Briefflanz, volgbeinnen i												t			+		
45	al and the second second second second second second second second second second second second second second s	an an an an an an an an an an an an an a	Coloradottorio antico y sense	Later and the second				hereitetti tottettisettisetti											-	Ŧ	╈	+ +		
46	All the second second	ana ana amin'ny soran'ny soran'	and a second second second second second second second second second second second second second second second		en a conservation and a second second second second second second second second second second second second se			a paging a constrained and a second		Conference on Conference	ter dama terte finisien													
41	and the same of the second second second second second second second second second second second second second	Sintantia, esperantiplanari enni	ang in a second space. I constant	energies and an an an an an an an an an an an an an	antilities a frank an a constant and a star and a	au - Aran maan ta room		Course - concrete singly												Ι	Τ			
44			Taklahin Glin, Bank																					
44	In the second seco																		-	_				
\$\$				_	entres-residues and a state of the little little									_			_		-		_	∔⊣		
<u>61</u>		a tag a transformation of the test															_	_	_	╞				
<u> %</u>					i natara ang panaki na pa																-			
32							and the second second second second second second second second second second second second second second second	(laranana atinto hongi											+	-	-			+
66																			-	+	+			Ħ
<u>- 22</u> 54																İ	H					1		† †
\$7					C																			
58		8																		Ι	Ι		\Box	
50			for conservation and provided	Contraction of the second second second second second second second second second second second second second s	The second second like in second																			
60							and the second se	6									Television	Contraction of the local division of the loc	the second second second second second second second second second second second second second second second s			_		
61	_													and the second second										
62													<u> </u>							1	-			
63	-																							\mathbf{H}
65	-																							
1 1 1 2 1 2																								
66																								
<u>66</u> 61																								
66 67 68																								
66 67 68 60																								
66 67 68 60 70																								
66 67 68 60 70 71																								
66 67 68 69 70 71 72																								
66 67 68 60 70 71 71 72 73 73																								
66 67 68 60 70 71 71 72 73 74 74																								
66 67 68 60 70 71 71 73 73 74 75 76																								
66 67 68 69 70 71 71 72 73 73 74 75 76 77																								
66 61 68 69 70 71 72 73 73 74 75 76 77 78																								
66 61 68 60 70 71 72 73 74 75 76 77 78 79																								
66 61 68 69 70 71 72 73 74 75 76 77 78 79 80																								
06 01 64 68 60 70 71 72 73 74 75 76 77 78 79 80 81																								
06 01 08 09 10 11 12 13 14 15 16 17 18 19 80 81 82																								
06 01 08 09 70 71 72 73 74 75 76 77 78 74 80 81 82																								
66 61 68 60 71 72 73 74 75 76 77 78 79 81 82 83 84																								
06 61 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 84																								
06 01 08 60 70 71 72 73 74 75 76 77 78 79 80 81 82 83 84 85 86 87																								
06 61 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 84 85 86 87 88																								
06 01 08 09 11 12 13 14 15 16 17 18 19 19 11 12 13 14 15 16 17 18 19 80 81 82 83 84 85 96 97 98 97 98 97 98 97 98 97 98 98 98																								
06 01 08 09 71 72 73 74 75 76 77 78 74 80 81 82 84 85 86 87 88 89 80																								
66 61 68 69 70 71 72 73 74 75 76 77 78 81 82 83 84 85 86 87 88 89 90 91																								
66 61 68 69 70 71 72 73 74 75 76 77 78 79 80 81 82 83 84 85 86 87 80 90 91 92																								
06 01 08 01 12 13 14 15 16 17 18 19 13 14 15 16 17 18 19 80 81 82 83 84 85 86 87 98 90 91 92 93																								
06 01 08 09 11 12 13 14 15 16 17 18 19 11 12 13 14 15 16 17 18 19 80 81 82 83 84 85 86 87 88 89 91 92 93 94																								
06 01 08 09 11 12 13 14 15 16 17 18 19 19 11 12 13 14 15 16 17 18 80 81 82 83 84 85 86 87 88 89 90 91 02 03 04 95																								
66 61 68 69 70 71 72 73 74 75 76 77 78 79 81 82 83 84 85 86 87 88 89 80 81 82 83 84 85 86 87 88 89 80 81 82 83 84 95 96 91 92 93 94 95																								
66 61 68 69 70 71 72 73 74 75 76 77 78 81 82 83 84 85 86 87 88 89 90 91 92 93 94 95 96 97 98																								

TM 11-5805-652-14 ISSUE 1

Section III

CONTENTS

Page

1.0	Preinsta	Illation Procedure	3-1
2.0	System	Cabinet(s) Description	3-1
3.0	Principa	al items Locations	3-1
	3.1	Junctor Test Panel	3-1
	3.2	System Teat Panel	3-1
	3.3	Fuse Panel Assembly	3-1
	3.4	PCB Layout	3-1
	3.5	AC Power Source, DC Power Supplies, and Blower Locations	3-1
	3.6	Miscellaneous Panel Assembly	3-2
	3.7	Horizontal and Vertical	3-2
	3.8	PCB Locations	3-2
	3.9	Strapping Field	3-2
	3.10	Punching Block	3-2
4.0	Sub-Rac	k D . awing Numbers	3-2

FIGURES

3-1	Typical Multicabinet System Layout	3-3
3-2	Basic Cabinet	3-4
3-3	First Cabinet	3-4
3-4	Second Cabinet	3-5
3 - 5	Third and Fourth Cabinets	3-6
3-6	Optional Equipment Sub-Racks	3-7
3-7	Junctor Test Panel	3-8
3-8	System Test Panel	3-8
3-9	Puse Panel Assembly	3 - 8
3-10	Miscellaneous Panel Assembly I"	3-9
3-11	Horisonial Bus Bu	3-10
3-12	Vertical Bus Bar	3-11
Section I I I

CABINET FAMILIARIZATION

1.0 PRELIMINARY INSTALLATION PROCEDURE

Place equipment in the assigned place 1.1 Refer to Fig. 3-1.

Unpack the equipment and Inventory. The Equipment Specifications contain. complete shipping list.

Fig. 3-1 shows a typical four-cabinet system Two, three, or four single cabinets are connected in a like manner.

Place the supplied rubber feet in place and insure cabinet(s) are level. Observe caution while tripping or raising cabinet(s) to install feet. If a multicabinet is being Installed, level and align all cabinets.

Route interbay cabling along top in space provided. Do not connect but place cabling in approximate position.

On a muticabinet system, insure that no cabling is caught between cabinets while aligning and leveling system.

2.0 SYSTEM CABINET(S) DESCRIPTION

2.1 Remove cabinet(s) front door(s). Face front of equipment and refer to the appropriate figure(s) of this section as required.

211 **Basic Cabinet Description (Fig. 3-2)**

Each cabinet provides seven levels CA-G) of mounting space.

Each level accommodates one sub-rack of printed circuit boards (PCB's). The basic cabinet will always contain levels A and B.

- Level A contains 20 line posi-(a) tions.
- Level B contains the Control (b)) Circustry.
- 2.1.2First Cabinet Description (Fig. 3-3)

The first cabinet of any system will contain levels A. B. and C. Level C is designated Attendant Control Sub-Rack. The DC power supplies will be equipped on levels C., D., E., F., and G The PC boards and sub-racks on levels D through G are optional and will be equipped as required.

Second Cabinet Description (Fig 3-4) 2.1.3

The second cabinet of any system will contain levels A. B. and C. Level C is designated Multiple Attendant Control Sub-Rack. The DC power supplies will be equipped on levels D through G Levels D-G will be equipped with PCB sub-racks on an as needed basis.

The third and fourth cabinets of any system will con-tain levels A, B, and C Level C sub-rack is designated Toll Restrictor Sub-Rack. Levels D-G will contain the DC power supplies. Levels D through G will be equipped with PCB sub-racks on an as needed basis

2.1.5 Optional Sub-Rack Description (Fig.

The four optional sub-racks are shown on Fig. 3-6 Any cabinet can contain:

Sub-Rack Level

- D City Trunk or DID Trunk
- Е
- F
- City Trunk or DID Trunk City Trunks, DID Trunks. or Tie Lines City Trunks, DID **Trunks, or Hotel/Motel** G
 - Features (or additional features)

3.0 PRINCIPAL ITEMS LOCATIONS

Junctor Test Panel (Fig. 3-7) 3.1

Facing the front of the cabinet, the junctor test panel is located on the right side of cabinet(s).

3.2 System Test Panel (Fig. 3-8)

Fig. 3-8 shows the alarm reset builton (labeled Reset), voltage check button (labeled VC)., regular (Reg), and test phone jack.

3.3 Fuse Panel Assy. (Fig. 3-9)

A fuse panel is located on each shelf of those systems equipped and/or wired with trunks/ tie lines requiring external 48VDC. Fig. 3-9 shows the fuse panel.

- 3.4 Refer to Equipment Specifications for specific PCB layout and equipped positions.
- 3.5 AC Power Source, DC Power Supplies and Blower Locations

Remove rear door and face rear. Refer to Dwg 130097 (sheet 2), Section E-E.

3.5.1 The blower input and filter input are located at the bottom left..

NOTE: Blockage of blower input will remove power from the system.

3.5.2 The AC power transformers and thear associated capacitors are positioned at the bottom right.

ISSUE 1

3.5.3 The AC power cord access entrance is located between the blower input and AC power transformer.

Miscellaneous Panel Assv. (Fig. 3-10) 3.6

Refer to Fig. 3-10. Facing the rear of the cabinet, the miscellaneous panel assembly is on the inside right.

Letters A, B, C. etc., have been assigned to individual items of the panel assembly. The items are identified below:

- A AC Circuit Breaker(s). All system AC circuit breakers will be located in cabinet 1.
- B VC Transformer fuse, 5VDC supply, and blower fuses.
- C Spare fuses.
- D Voltage Check Transformer.
- E Power Relay.
- F Ring Generator.
- G Ring Generator Suppression Assembly.
- H PFCT1, PFCT2, Auxiliary Relay, and RL1.
- J MC1 through MC5 (5VDC supply).
- K MB1 (5VDC supply rectifier).
- L Q1. MRI. and MCB (5VDC supply). M - TE resistor assembly and RV1,
- RV2, SCR1, RV21, RV22.
- TE resistor assembly PCB. - Miscellaneous Panel Assembly.
- Connector block.

NOTE: Items A, D, J, M, and N will note be equipped on cabinets 2, 3, or 4 of multicabinet systems.

3.7 Horizontal Bus (Fig. 3-11) and Vertical Bus Bar (Fig. 3-12)

A horizontal DC voltage/ground bus bar is located on each level. The horizontal bar is connected on the right to a vertical bar, which is connected to the DC power relay. Refer to Figs. 3-11 and 3-12.

3.8 Printed Circuit Board Locations

Refer to Dwg. 130097 (sheet 4) The PCB's are assigned positions on each shelf. Each PCB is inserted into the approvriate female connector. Each PCB has a distinct plug arrangement that prevents inadvertent insertion of the PCB into an incorrect position.

Facing thee rear of the cabinet, the PCB positions are counted 1, 2, 3, etc., from right to left.

3.9 Strapping Field

The strapping field (SF) is located on the left and is inside the cabinet. The connectors of each 12x35 S.F. point directly to the right.

Each level is assigned one 12 x 35 S.F.

3.10 Punching Block

The punching block is on the left rear and its connectors point directly outward. Each punching block is equipped with 12 x 7 connectors on an as needed basis. Each level is assigned its own punching block.

Dwg. 130071 shows the level A sub-rack punching block and strapping field. The pin count starts at the upper left as 00.0 and counts downward to 034.0 (lowest left pin). The assigned horizontal count starts from the extreme left vertical count and progresses in numerical sequence to the right ((00.0 to 00.11). Thus, GHA 25 is found to be on 025.7. It is 25 from the top and 7 from the left. Likewise, 255 is found on 16.4.

The method of pin court is the same for the punching block and strapping field.

4.0 SUB-RACK DRAWING NUMBERS

Dwg. No.	Sub-Rack	Level	Bay((5))
130071 130072 130073 130074 130075 130076 130077 130078 130079 130050	Lines Control Attd. Control Attd. Multiple Toll Restrictor City Trunk DID Tie Line Hotel/M: "~1 Festure	ABCCC****	Aill Aill 2* 2 of 3* * *

* Refer to appropriate paragraph on Section II for Level and Bay (casinet) information.



Fig. 3-1 Typical Multicabinet System Layout

Typical Four Calimet System Layout



3 - 4



BND. OVE

TOLL REST.

36-06		5 95-99		LINE CIRCUIT SUB-DACK
		LINE		130071
	▼ (3)	PRI-SEC. (4)	TERT. QUAT.	CONTROL SUB-RACK 130072
8		8	(3)	ATTENDANT CONTROL SUB-RACK
GRP. HUNT		GRY. HUNI	GRP. HUNT	130073

B		~	8	a		A YUS
POWER	POWER	POWER	POWER		ORIG, CONT. & LP GEN, (U)	Lines 00-04
Supply -36, -3	Supply +36, +3	\$UPPLY *12.5	-1 an 2094/a		ORIG, CONT. & LP CEN. (2)	f 05-69
-	an an an an an an an an an an an an an a				<u>→ • • • • • • • • • • • • • • • • • • •</u>	19-14
					2-	15-19
						20-24
		-3				25-29
		SWN		ATT. MULT. (Q)	SVS. ALLOT. (2)	70.74
		S OR		ATT. MULT. (2)	AUX. IUNC.	
				CONF TRK. ACC. (3)		35-39
				COMP IRK, ACC. (2) Invit sect mean i	<u> </u>	40-44
	Ħ			1 - 2 J1 -	(4)	6-47
	<u>↓ </u>			1 2 -		10-54
	i vi			2 3		62.43
				TOLL REST. MON. 2 3	(8)	25.54
				ATTR CONT. (0)	(10)	60-64
-				4 (2)	(11)	65-69
DE E					ATTEND OVP (1)	70-74
TION				ATTR. C.Q.B. (1)	ATTEND, OVR. (2)	75-79
				ATTD. C. O. B. (2)	LINELO.	20-24
20				ATTO C.O.B. (3)	C.O.S. (2)	
RES				ATTOL TRK. (1)	PRI-SEC. (1)	87-89
					(2)	90-94
				GRP. HUNT (2)	PRI-SEC (A)	T LINES 95-99
		<u> </u>			TERT. QUAT.	
		• · #_		MULTIPLE ATTENDANT CONTROL SUB-RACK 130074	CONTROL SUB-RACK 130072	LINE CIRCUIT SUB-BACK 130071

Fig. 3-4 Sound Control

3 - 5

LEVELS n ----m Ø C > • ORIG. CONT. & LP GEN. (1) POWER SUPPLY -36, -3 POWER SUPPLY +36, +3 POWER SUPPLY +18.5 POWER SUPPLY -18V LINES 00-04 ORIG. CONT. & LP GEN. (2) 05-09 REG, 10-14 CITY TRUNKS 2 -15-19 20-24 <u>}</u>, -REG. SYS. ALLOT. SYS. ALLOT 25-29 **(1)** ີ ຊີ (2) t bid TRUNKS 30-34 AUX. JUNC. JUNC. 35-39 a) (2) ()) TOLL. REST. MON. 1) 40-44 (4) (5) II 45-49 <u>______</u>55 90-54 <u>60</u> (7) Ī IDAL. REST MON. 2) **53-59** (8) 60-64 防刑 AINC. ATTEND, OVR, ATTEND, OVR, U^{ME}LO, C.O.S. C.O.S. PRI-54C. HON 70-70 75-79 OTEL OR 596595 *1*89 94 16-19 10-14 GAP, INDAT GAP, INDAT GAP, INDAT unes D PRI-SIC D TINT. OUAT. 45-49 TOLL ALSINGCOM SUB-SUCK Look to the second seco

T M 11-5805-652-14

Fig. 3-5 Third and Fourth Cabinet

3 - 6

BD. 1 8D. 2 BD. 1 BD, 2 BD. 1 8D. 2 BD. 1 BD. 2 BD. 1 BD. 2 80. 1 [']

BD. 2

(1)

(2)

()

(2)

13076

POWER SUPPLY

			ו	5			-						
	POWER SI	IPPLY			POWER SUPPI	LY		POWER	SUPPLY]		POWER	SUPPLY
	MES. WAIT.	00- 09		TIEL	NE		1	TRUNK 1	BD A	4	CITY		
₹	•	10-10	_		4		1 _	1	BO R	1		A	80, 1
ĨEU				<u> </u>		(2)	Ē		BD. C	1 9	 	╂	8D. 2
MOT		20-29	Ā			(3)		TRUNK 2	BD. A		┝	╋──	BD. 1
S 1		30-39	SUB			(4)			80. B	Ĩ	<u> </u>	┣—	BD. 2
Se-		40-40	-RA		ł		s v		8D. C	e se	 	┣	BD. 1
Ñ		40-47	×		ļ	(5)		TRUNK 3	BD. A	a z	<u> </u>	┣	BD. 2
		50-59				(6)	ÄCK		80, 8	ACK	<u> </u>		BD. 1
		60-69				(7)		+	BD, C				BD. 2
					 			TRUNK 4	BD. A				BD. 1
		N-14				(8)			8D, 8				BD. 2
		80-89				(9)		*	8D. C				8D. 1
	MES. WAIT.	90-00	1	_				TRUNK 5	BD. A		CITY TR	IKS.	BD. 2
			ł			(10)			80, B		TRK. AL	lot.	a
	WK LINE APPL	00-19				an		1	BD. C		TRK. AL	10T.	(2)
ŀ		20-39	ſ	TIE LIN	£	(12)		INUNK 6	80. A		TRANS.	JUNC.	BD. 1
		40-59	F						80.8	[TRANS.	JUNC.	BD. 1
	<u> </u>	60-79	Ļ	TRK. A	LOT.	(D	}	TOP ALLOT	BQC	ſ	T.R. AC	CESS.	(1)
ŀ	WAR LINE APPL.	80-99	l	1	l	(2)		TRAC ALUUI,		t	T.R. AC	CESS.	(2)
ŀ	A TRA. APPL.	1-7	Г			(1)	ł	TRANS HINC	BD 2	T T	UNA		
		8-14	F	Y	ور و موجود می مقد ما از ما از ا		ł	T.R. ACCESS		L L	ANA	-	
F	SING MG ACC	12.19	L L	TRK. AL	LOT.	(4)	ľ	T.R. ACCESS	(2)	r		-	
F							r	INTERCEPT	<u>a</u>				
L			L				Ī	والمتراب المعرفي المتركب المتركب المتركبة والتكامية					
	NOX1				130018		-	130077	and and an and a second second second second second second second second second second second second second se	L		13007/	

Fig. 3-6. Optional Equipment Sub-Racks



Fig. 3-8. System Test Panel.



Fig. 3-9. Fuse Panel Assembly





3 - 9



Fig. 3-11. Horizontal Bus Bar.



3-11

Section IV

CONTENTS

			Page
1.0	Visual	Inspection,	4-1
2.0	Power/	Ground Connections	4-1
	2.1	AC Power Input	4-1
	2.2	Ground Cable	4-2
	2.3	Power Multiple Cables	4-2
3.0	Consol	e(s) Cable(s) Connections	4-2
	3.1	Single Position Console	4-2
	3.2	Multi-Position Consoles	4-2
	3.3	Bell Console	4-6
4.0	Interca	abinet Cables	4-6
5.0	MDF I	Layout and Assignment (Typical)	4-10
6.0	Prewi	red Lines Tip-Ring Connections	4-10
	6.1	Prewired Cables	4-10
	6.2	Cable Run	4-10
	6.3	Cable Connections	4-10
7.0	Prewi	red Trunk/Tie Lines Tip-Ring Connections	4 - 3 5
8.0	Prewi	red PFCT and Miscellaneous Connecting Cable and Trunk E-M and	
	A-B I	eads Connections	4-35
	8.1	Prewized PFCT and Miscellaneous Connecting Cable	4 - 3 5
	8.2	Trusk E-M and A-B Loads	4-35
9.0	Instal	lor Wired Cabling	4-35
	9.1	Cable Layout, Longth, and Identification	4 - 3 5
	9.2	Cable Termination	4 - 3 5
	9.3	Lines	4 - 3 5
	9.4	Trunks	4-35
	9.5	The Lines	4-35
	9.6	Miseellaneous Culputs	4-35
	9.7	Hotel/Hotel Connections	4-40
10.0	PPCT	Connections with Prewiged Cabling	4 - 40
11.0	Unade	ligned Night Answer Prewiged Cabling	4 - 41
12.0	Public	s Address Connection - Provided Cable - recreate contract - creaters	4 - 4 1
12.0	Other	Peatures Reapping	4 - 4 1

CONTENTS (Cont'd.)

		Page
14.0	Class of Service Assignment/Equipping	4-41
15.0	Trunk/Tie Line Loop Adjustment Resistor	4-41
16.0	Satellite Equipment Cabling	4-41
17.0	Power-Off Measurements	4-41

TABLES

4-1	TE-400 Connecting Cable	4-3 through 4-15
4-2	Cable Connection to MD F - (EPABX Circuits)	4-11 through 4-19
4-3	Cable Connections to MDF - (Satellite Equipment)	4-21 through 4-29
4-4	Line Tip and Ring Connecting Cable	4-31
4-5	Line Tip and Ring Connecting Cable	4-32
4-6	Line Tip and Ring Connecting Cable	4-33
4-7	Line Tip and Ring Connecting Cable	4-34
4-8	Trunk Tip and Ring Connecting Cable	4-36
4-9	PFCT and Miscellaneous Connecting Cable	4-37
4-10	EGM - A4B Connecting Cable #1	4-38
4-11	彩色M - A-6-8 Connecting Cable #2	4-39

FIGURES

		Page
4-1	Single Console Positions	4-1
4-2	Cable Connector	4-2
4-3	Typical Multi-Position Console Installation	4-3
4-4	Diode Box Assemily	4-7
4-5	Typical Bell Connole Installation	4 - 8
4-6	Typical Single Cabinet TE-400 System and MDF Layout	4-9
4-7	Typical Instaliation Showing Connections Between TE -480 Cabinets and	
	NEF Wardsolp ,	4-9

Section IV

POWER-OFF CONNECTION

AND TESTS

2.0 POWER/GROUND CONNECTIONS

2.1 **Power Input**

Inspect the equipment for:

- (a) Broken wires
- Shorted pins, connectors, and/or (b)
- wires.(c) Miscellaneous damage.

- Insert AC power cord(s) as shown in Fig. 3-1, Section III Connect AC power cords as shown in Fig. 3-1, Section III (a)
- **(b)**

CAUTION: Insure that these connections are correct for the voltage and frequency supplied before applying power to the system





ITT Single Console/Single Cabinet

Notes:

- 1. Reference Table 4-1 for cable(s) number.
- 2. BLF equipped.
- 3. 24 trunk appearance (maximum).





ITT Console

1.0 VISUAL

1.1

2.2 Ground Cable

Connect ground multiple cable and external each ground cable as shown in Fig. 3-1, Section III. Insure that earth ground cable IS connected to a <u>known</u> earth ground of no more than 1 ohm.

- 2.3 Power Multiple Cables (Multicabinet Installations)
- 2.3.1 Connect DC power multiple cables as shown in Fig. 3-1, Section III.
- 2.3.2 AC Power Multiple
 - (a) Remove three (3) bolts from me frame miscellaneous assembly on cabinet 1.
 - (b) Swing hinged panel out to allow access.
 (c) Connect AC neuron multiple
 - (c) Connect AC power multiple cables as shown in Fig. 3-1, Section III
 - (d) Swing panel to original position, then replace bolts

3.0 CONSOLE(S) CABLES(S) CONNECTIONS

3.1 Single Position Console Wig. 4-1)

3.1.1 Refer to Equipment Specifications, Table A, for equipped cabling. A description of each cable is given in Table 4-1

3.1.2 Each cable will be 25 pair (50 wires), 24 gauge, end terminated in a <u>female</u> amphenol connector

3.1.3 Each connector will be stamped as to identity.

3.1.4 The cables will be pre-routed to the top cable trough. The cables may be routed out of the cabinet to either left or right

3.1.5 On multicabinet systems, equipped with multihundred busy lamp field(s), the busy lamp field (BLF) cables will be marked for cable identification. The Equipment Specification will furnish *necessary* information since this will be special cabling.

3.1.6 Determine the number of cables to be connected to the console. Position the console in the designated operating space Route the required cabling between the equipment and the console.

3.1.7 The prewired console cabling is end connected ted In a female amphenol connector. The cable is 24 gauge, 25 pair. See Fig. 4-2.

3.1.8 The connector of the cabling to be used should be similar to the connector to the equipment male and console female onnecotrs. 3.1.9 Mark both ends of each cable as they are run from the equipment to the

console. After cabling is run, connect each end. Insure that connecttons are snug and holding *screws* are tight

3.1.10 Information pertaining to *iron* works and/or *other* supporting devices is not supplied in this manual since installation space will dictate how the external cabling will be supported.

3.2 Muti-Position Console (Fig. 4-3)

3.2.1 Cabling and connections for multiattendant consoles may be identified as in paragraph 3.1.1 The Equipment Specifications will contain identification information to enable the installer(s) to distinguished between position 1 and position 2 consoles



Fig. 4-2. Cable Connector.

	Use Connecting	•	1
Catle No.	Chart No.	Description	Purpose
130103-1	130120	Line Tip & Ring Conn. Cable -	Used on Bay 1 to connect lines
			00-25 to distributing frame.
130101	130121	Line Tip & Ring Conn. Cable	Used on Bay 1 to connect lines
			26-49 to distributing frame.
130103-1	130122	Line Tip & Ring Conn. Cable	Used on Bay 1 to connect lines
			50-74 to distributing frame.
130103-1	130123	Line Tip & Ring Conn. Cable	Used on Bay 1 to connect lines
			75-99 to distributing frame.
130103-7	130120	Line Tip & Ring Conn. Cable	Used on Bay 2 to connect lines
			00-24 to distributing frame.
130103-7	130121	Line Tip & Ring Conn. Cable	Used on Bay 2 to connect lines
			25-49 to distributing frame.
130103-7	130122	Line Tip & Ring Conn. Cable	Used on Bay 2 to connect lines
			50-74 to distributing frame.
130103-7	130123	Line Tip & Ring Conn. Cable	Used on Bay 2 to connect lines
	1		75-99 to distributing frame.
130103-2	130124	Trunk Tip & Ring Conn. Cable	Used on Bay 1 to connect trunks
			1-24 to distributing frame.
130103-8	130124	Trunk Tip & Ring Conn. Cable	Used on Bay 2 to connect trunks
			1-24 to distributing frame.
130103-3	130125	Power Fail Cut-Thru & Misc.	Used on Bay 1 to connect power
		Conn. Cable	fail lines, public address
			amplifier, UNA bell & code call
			bell to distributing frame.
130103-9	130125	Power Fail Cut-Thru & Misc.	Used on Bay 2 to connect power
		Conn. Cable	fail lines to distributing frame.
130103-4	130126	Bell Telephone Console Conn.	Used on Bay 1 to connect Bell
	1	Cable #1	console 757A.
130103-4	130127	Bell Telephone Console Conn.	Used on Bay 1 to connect Bell
		Cable #2	console 757A.
130103-4	130128	Bell Telephone Console Conn.	Used on Bay 1 to connect Bell
		Cable #3	console 757A.
130103-4	130129	Bell Telephone Console Conn.	Used on Bay 1 or Bay 2 to connect
		Cable #4	Bell console 757A to trunks 13
			to 24.
130103-4	130130	Bell Telephone Console Conn.	Used on Bay 2 to connect Bell
		Cable #5	console 757A to trunks 25 to 30.

Table 4-1TE-400ConnectingCable

ſ	Use Connecting		
Cable No.	Chart No.	Description	Purpose
130103-4	130131	Bell Selector Console	Used on Bay 1 to connect Bell
		Conn. Cable #1	selector console 1Al or 1A2.
130103-4	130132	Bell Selector Console	Used on Bay 1 to connect Bell
100100		Conn. Cable #2	selector console 1A1 or 1A2.
130103-4	130133	Bell Selector Console	Used on Bay 1 to connect Bell
100100 -		Conn. Cable #3	selector console 1A1 or 1A2.
130103-3	130132	Bell Selector Console	Used on Bay 2 to connect Bell
100100 0		Conn. Cable #5	selector console 1A2.
130103-3	130133	Bell Selector Console	Used on Bay 2 to connect Bell
100100 0		Conn. Cable #6	selector console 1A2.
130103-9	130134	Bell Selector Console	Used on Bay 2 to connect Bell
100100 0		Conn. Cable #4	selector console 1A2.
130103-3	130135	Attendant's Console Conn.	Used on Bay 1 for 1st attendant
200400 0		Cable #10	or Bay 2 for 2nd attendant
			consoles.
130103-1	130136	Attendant's Console Conn.	Used on Bay 1 for 1st attendant
	_	Cable #13	or Bay 2 for 2nd attendant to
			connect trunks 26 to 42 to the
			consoles.
130104-3	130149	Attendant's Multiple Cable	Used on Bay 1 to cable K & TL
			leads to Bay 2 when two
			attendants positions equipped.
130104-4	130149	Attendant's Multiple Cable	Used on Bay 2 to cable K & TL
			leads to Bay 1 when two
			attendants positions equipped.
130103-6	130153	RS Connector Cable (Male)	Used on Bays 1, 2 & 3 to connect
			RS terminals when two digit
			in-dial registers equipped.
130103-5	130153	RS Connector Cable (Female)	Used on Bays 2, 3 & 4 to connect
			RS terminals when two digit
			in-dial registers equipped.
130103-3	130154	Attendant's Console Connect-	Used on Bay 1 for 1st attendant
		ing Cable #11	or Bay 2 for 2nd attendant con-
			soles.
130103-3	130155	Attendant's Console Connect-	Used on Bay 1 for 1st attendant
		ing Cable #12	or Bay 2 for 2nd attendant con-
	1		soles.

Table 4-1 (Cont'd)

TE-400. Connecting Cable

Table 4-1 (Cont'd)TE-400 Connecting Cable

Cable No.	Use Connecting Chart No.	Description	Purpose
130103-6	130160	Bay 1 to Bay 2 Common Multiple Connecting Cable #3	Used on Bay 1 to connect to Bay 2.
130103-5	130160	Bay 2 to Bay 1 Common Multiple Connecting Cable #3	Used on Bay 2 to connect to Bay 1.
130163-1	130164	Wire Braid (Ground Strap)	Used on Bay 1 to Bay 2, Bay 2 to Bay 3 & Bay 3 to Bay 4.
130103-2	130189	E&M-A&B Connecting Cable #1	Used on Bay 1 to connect trunks 1-12 to distributing frame.
130103-2	130190	E&M-A&B Connecting Cable #2	Used on Bay 1 to connect trunks 13-24 to distributing frame.
130103-8	130189	E&M-A&B Connecting Cable #1	Used on Bay 2 to connect trunks 1-12 to distributing frame.
130103-8	130190	E&M-A&B Connecting Cable #2	Used on Bay 2 to connect trunks 13-24 to distributing frame.

3.2.2 Two diode box assemblies (Fig. 4-4) are provided. These boxes are connected in series (one for each console position) with cables 10 and 13.

3.2.3 Although a busy lamp field is available, a BFL will not normally be provided since a system can be equipped with 42 trunks. Fig. 4-3 shows a typical installation.

- 3.2.4 Connect cables with care and insure they match.
- **3.3** Bell Console (Fig. 4-5)
- 3.3.1 Cabling identification may be determined as in paragraph 3.1.1

3.3.2 Fig. 4-5 shows a typical Bell console installation.

- 3.3.3 The following cables will not be required on a single cabinet system:
 - (a) CON 6 (b) SEL 4 (c) SEK 5
 - (d) SEL 6

4.0 INTERCABINET CABLES

4.1 All equipped intercabinet cabling will be marked for identification.

4.2 Refer to Equipment Specifications, Table A, for equipped cabling.



Fig. 4-3. Typical Multi-Position Console Installation.







Fig. 4-4. Diode Box Assembly.



Fig. 4-6. Typical Single Cabinet TE-400 System and MDF Layout.



Fig. 4-7. Typical installation Showing Connections Between TE-400 Cabinets and MDF Verticals.

4.3	Connect	each	cable,	like	identification
	to like i	identif	ication.		

- Insure firm fit and that holding screws 4.4 are tight.
- 5.0 MAIN DISTRIBUTION FRAME(MDF) LAYOUT AND ASSIGNMENT(TYPICAL)

The purpose of any MDF is to provide 5.1 cross connection points between the EPABX equipment and the distribution plant

- Figs. 4-6 and 4-7 are provided to 5.2 show a typical installation.
- 5.3 Refer to Fig. 4-6. A top view of a single cabinet TE-400 and MDF layout

is shown

5.3.1 Each vertical is equipped with a number of feed-through terminals. A connection can be made on each terminal side with continuity.

5.3.2 Five verticals are shown Each vertical is assigned to:

- EPABX lines(s) T-R outputs
- (b) Subscriber stations T-R inputs/ outputs.
- EPABX trunk/tie lines T-R out-(c) put.
- Central Office or tandem equip-ment line T-R input/outputs (d)

(e) EPABX miscellaneous circuits outputs and installation miscel-laneous devices (i.e., UNA Beils, Public Address Systems).

5.3.3 EPABX outputs may be connected to other equipment(s) by connecting Jumpers between vertical terminals.

Fig 4-7 shows connections between the cabinet(s) and MDF Verticals 5.4

Although connections are only shown 5.4.1 for lines T-R, these are typical for trunk/tie lines T-R and PFCT miscellaneous connecting cable

5.5 Vertical Terminal Assignment

To prevent confusion and miswiring, 5.5.1 the terminal assignments must be completed and designated before terminations are attempted.

- 5.6 Tables 4-2 and 4-3 are to be completed by the installer
- These tables supply information con-5.6.1 cernig connections to the MDF
- 5.7 Table 4-2 Column Explanatory

A - EPABX circuit such as line numbers, trunk/tie line number. or miscellaneous circuits (i.e., UNA, Code Call, E&M Leads, etc.)

- **B** Cable Number Identification of cable between TE-400 cabinet and **MDF** vertical. - Vertical - Vertical to which
- C cable terminates.
- Output Output of EPABX cir-cuit such as T-R, URLI (UNA), D etc.
- E Terminal numbers of vertical.
- F _ Wire color of cable pair terminated to vertical terminal.
- Connect to Vertical and termi-nal number to connect to for G -

operation. This part of the table should be filled in after Table 4-3 IS completed.

5.8 Table 4-3 Column Explanatory

A - Satellite Equipment - Subscriber station numbers, trunk/tie line numbers, miscellaneous equipment outputs

- Cable number, vertical, outputs, B terminals, wire color - Same as Table 4-2
- C Location Physical location of equipment.
- CO Line Numbers The number assigned by the CO to an incom-D ing trunk

NOTE: Columns B, C, and D should be filled in when performing "Cut-Over Procedures" in Section VII of thus manual

6.0 PREWIRED LINE(S) TIP-RING CONNECTIONS

6.1 Prewired Cables	
---------------------	--

- 6.1.1 Each cabinet(s) will be equipped with four line T-R cables
- 6.1.2 Identify these cables as described in paragraph 3.1.1
- Use Tables 4-4 through 4-7 and Figs. 4-2, 4-6, and 4-7 for reference 6.1.3
- Each pair is 25 pair 24 gauge and 6.1.4 female ended amphenol connector
- 6.2 Cable Run
- 6.2.1 Determine cable length

6.2.2 Insure cables are terminated with male connector(s) that will mate with the associated female amphenol connector(s).

- 6.2.3 Identify each cable at both ends
- 6.2.4 Refer to Table 4-2, then butt, strip, and terminate cables to MDF vertical(s)
- **Cable Connections** 6.3

	A EPARX	B Cable	La		E	F	G Connect To			
	Circuit	Number	Venic	ore.	Numbers	Color	Terminal	Vertical		
		+	1				(TURDET	- Monider		
	 	<u> </u>		+	+					
	ļ	<u></u>	<u> </u>					1		
					+	+		-		
		1						1		
		ł			<u> </u>	ł	<u> </u>			
		ļ	ļ		ļ	1		1		
				-	┝	 	╉╼───	4		
		<u> </u>			<u></u>		<u> </u>			
]		
			L		<u> </u>	1				
	1	ł			ł	╀		4		
		<u> </u>			1	<u> </u>				
					<u> </u>	<u> </u>				
	ļ							1		
				<u> </u>	 	<u> </u>		-		
								1		
					<u> </u>	┼────	╆			
								1		
					<u> </u>					
					ļ					
						<u> </u>	<u> </u>	<u> </u>		
				\square				 		
						<u> </u>				
-										
						 	}			
<u>,</u>										
Τa				-						
tion of										
e e		_				<u> </u>				
4 ₽										
H PI										
X A X										
<u>;</u>										
ruite										
ت										
				_						
	┝]			
			ł							
			1							

•

٠

A	8	C L		E	F	G Conne	ct To		
Circuit	Number	/ertic	B	Numbers	Color	Terminal	Vertical		
	-	Ť							
	+	╋	╉─	_					
		+	\square				1		
		<u> </u>	F	1	<u></u>		1		
	1	† -		†			1		
	+	+		<u> </u>		1			
	+	+	╂				<u> </u>		
			\square			1			
		<u> </u>			İ				
				<u> </u>					
			┣—	+	<u> </u>				
		1	<u> </u>	ļ	ļ		_		
	†	+		<u> </u>	 	<u> </u>			
		+ -	E	<u> </u>					
	+	╂	<u> </u>			·			
	<u> </u>	_		L					
	1	Γ							
		\uparrow							
	<u> </u>	+							
	+	+							
		–							
	ļ	 							
						<u> </u>			
	1								
	<u> </u>								
		-							
		\vdash				F			
									
						 			
		┝─┤							
	<u> </u>	┝─┤							
		┝Ĭ							
						<u> </u>			
							{		





	A	B	C	D	E	F	1 0	3
	EPABX Circuit	Cable Number	Vertical	Output	Terminal Numbers	Wire Color	Conn Terminal Number	ect To Vertica Number
				<u> </u>				
		+	+				<u> </u>	+
							1	
								-
				-				-
			+	<u> </u>				
		<u> </u>		+	<u> </u>			
		ļ						1
								-
								_
		1	┼──			1		
								1
								-
								-
			\square					
			<u> </u>	+	+			
	· ··=							1
					<u> </u>	<u> </u>	<u> </u>	-
		1						
	·····		<u> </u>					+
								1
		1						-
		 						
			ļ			<u> </u>		1
								1
								<u> </u>
								<u>}</u>
								1
								4
la								<u> </u>
ble								
4-								
2								
on								
t d)								
					·····			
ſ								
ł								
ŀ								
					•			

Cable Connection to MDF (EPBAX Circuit)

•	8	C	0	E	F	G	ct To	
EPABX Circuit	Cable Number		hput	Terminal	Wire Color	Terminal	Vertical	
		╞	ð			Number	Number	
	1		<u>, ;</u>	+				
·	1	+		1				
		+			<u> </u>		-	
	<u> </u>	-					-	
			-	-	+	Hall Constitution and the section	•	
	+	+			+		n in the second second second second second second second second second second second second second second seco	
			F	1				
				+	<u> </u>			
	1			ļ			1	
	<u> </u>		+	+	 	+		
	ļ			1				
						<u>+</u>	-	
	1	+		1				
	 	+	+	+				
<u> </u>	 		1				1	
	1			+				
	1							
		+			<u> </u>	and a second distance of the second second second second second second second second second second second second		
	Ĺ							
	╆────	-		1				
	ļ	_				an international statements and	course another succ	
			<u> </u>					
		Γ	<u> </u>	1				
	<u> </u>							
		╂—	-					
		ļ	\vdash	<u> </u>				
								
		+	╋──					
		 	F					
	L							
		1	┣—					
		1						
			-					
			F					
			<u> </u>			a data tanan panakana ang		
_			├					
	and the second second second	L			in an an an an an an an an an an an an an			
							7	
	and a state of the							
	میں نوبید دعلیٰ (کو تع				ilia di kata ta ta sa sa sa sa sa sa sa sa sa sa sa sa sa		- annotation	
		<u> </u>	-					
				0			5.575.000 (F10)22	
	the second second second second second second second second second second second second second second second s				in successive and			
				a di seconda di seconda di seconda di seconda di seconda di seconda di seconda di seconda di seconda di second	international and an and an and an and			





Cable	
Connections	
to	
MDF	
(EPABX	
Circuit)	

Table 4-2 (Cont'd)

EPABX	Cable	1		Terminal	Wire	Conn	ect To
Circuit	Number	E T	1	Numbers	Color	Terminal	Vertica
		<u> ></u>	<u> </u>			Number	Numbe
				+			-
				1			
		<u> </u>					
				+			_
		+		+			
		1		+			-
							+
							1
				<u> </u>	+	_	-
+				+			+
						-{	-
		1		1	1		
				┢────			-
				+	+		
					+		1
					+		
							1
I							
		┝──┤			+		
							-
				<u>† </u>	1	+	1
							1
					+		
		}		<u> </u>	+		1
				t	1		t
							1
		' H					4
					f		<u>+</u>
							1
						+	
		ŀ					
					1		
					ł		
		H			+		
		-+			<u> </u>		
		F					
					├ ────	+	
						+	
		[
		F					
						+	<u> </u>
		H		<u> </u>	·	+	
						++	
		Γ					
		F					
						┼────┤	
		\vdash				╃━━━━┫	
			-+			┽	
	-1						
						↓]	
	1					+	
		-+-	-+			┼───┼	
			-+			t{	
						<u> </u>	
		-	-				
						<u>↓</u>	
1		\vdash	-+			+	
						<u>+</u> +	
1			-+			t	
	1					1 '	

A	8	C)	E		f		Cenn	i ect lo
Circuit	Number	Vertic	Otton		Numbers		Color		Terminal Number	Vertica
			F	_				_		-
						_		_		
		_	F	_						1
								-		
			\vdash			-		-		-
_		1	F							
		+	+	_		+		+		
		+	F	_		-		_		
				_				1		1
			-	-		┥		┥		-
		Τ	-			7		4		
		+	T					1		<u> </u>
		+	+-	-		+		+		+
		<u> </u>	\square	-		-		4		1
			F					+		
			-	_		-		+		
						1		1		1
		+-		-		+		+		
		+	-			4		4]
						1				1
			-	+		+	and the support of the support	┽	enzerina enzeri de la com	
						1				-
		+				1	in ma tria della companya		nan mana ang kang sa sa sa sa sa sa sa sa sa sa sa sa sa	and the second second second second second second second second second second second second second second second
		+	+	_		+	a code di minocodi ci angla	-	inge sandlærde stat	
						1				
				+	ويترجه بورعتهم	+		+	antikanak producerka panat	{
				_		Ţ		_		
		+		1						
		<u>+</u>	+-	+		+		ŀ	والإليون أنحك أنشر	
	<u> </u>	┝	F	-		Ţ		t		
			F	İ		t		t		
				+		╀		╉		
		1	—	1		Ŧ		1		
				t		t		$^{+}$		
		┣		ł		F		Ļ		
				t		t		t		
				t		t		+		
				ł		F		F		
				t		t		t		
+				ł		+		+	and a management of a single part of the ball	
	<u> </u>		—	F		F		L	an de la company de la company de la company de la company de la company de la company de la company de la comp	a a la companya a companya a companya a companya a companya a companya a companya a companya a companya a comp
				t		t			in the second second second second second second second second second second second second second second second	
				t		┝	ويعتقر ومحمد المترجم بدراني	-		
				Ļ		Ľ			an de la completado	ana ana ang ang ang ang ang ang ang ang
				t		_		Ĺ	dinan (Ching) ang j	
				╞		-		F	and the state of the state of the state of the state of the state of the state of the state of the state of the	
				F		_				
				t		-		E	anan anan ang si ang	
				F		_			nanda ana amin'ny tanàna mandritry tanàna mandritry tanàna dia mandritry tanàna dia mandritry tanàna dia mandri	
				L		_	inidenteta sigari da	_	an in the second second second second second second second second second second second second second second se	alisense alie
				E			na anna anna anna		and the second second second second second second second second second second second second second second secon	the party of the party of the
				L			territe at gange	-	in Ideanse Enclose Arristo	
					Contraction of the local division of the loc	-		-	No amoustantina (USA)	



			 	 ·	
~	 		 	 ·	
7			 	 	
<u>,</u>	 		 		
5/0					
4		L			
<u> </u>					
6				 	
<u>-</u>				 	
<u> </u>					
n f			 	 	
<u>र</u>			 	 	

Table 4-2 (Cont'd) Cable Connections to MDF (EPABX Circuit)
--

EPABX	Cable	F		<u> </u>	Termin	لچ	W		Cor	nect T	0
Circuit	Number	Vertic			Numbe	a1 rs	Color		Termina	at V	ertic
		+	t						Numbe		umb
			\square								
				+			<u> </u>				
											-
			_	_				_			
		1	┢	+				_			
		ł	<u> </u>	-+							
				-+							
				-+-							
				+		-		-+	·	-	
				_							
				+		-		_			
				1				-		-	
				-							
				+		-		+			_
				1						1	
				-		Ţ		1			
				+		+		+			
				1				+		-	
				+		1		1		-	
				+		+		+		+	
								T			
		ŀ		+-		+		+		-	
				1				+		+	_
						_		Т.		1	
		ł		╋		-+-		-+-		-	
								+		+	
				+_		\downarrow		Ţ		1	
		F		+		+		+		-	
						1					
				+		+		+		7	
		-				+		+-		1	
		-				Ŧ		T			
				-		╉		+		+	
										1	
						1		-			
		-+-		-		+		╋		+	
						T					
		-				╀		+		-	
						\dagger	·	+	· <u> </u>	+	
						T		L		1	
		-		\vdash		+		╋		-	
						+		+			
—— 				ļ		F		F		 	
		-				┢		+-		1	
		L				L				<u> </u>	
		-+-				╞		-			_
		-				┢		+			
				_		┝					
						t		<u>† </u>			
		_				Γ					
						-					
			-			Ē					_
			+								
		-	\neg			L					
			-+					_			_
						-					
	T		1								Τ
			+								
			1								
			-+								
					•						

A	8		<u> </u>	D	3		F		G		t To	
EPABX	Cable Numbe	-	erica	hopy	Termina	6 6	Wire Color		Termina	1	Vertical	
		+	4	_		_		-	Number	-	Number	
			[
			_ }			-		-		-		
		Τ	T									
			-			_		-				
						_		1			6-80-10-10-10-10-10-10-10-10-10-10-10-10-10	
				_								
			-							_		
	-	-	-									
	+		-+-							_		
				_						-		
			┝			_		-		_		
				-								
			-			_		-		7		
	·											
			⊢			-		+		_		
				_	+	+		+		+		
	<u> </u>	_	_					1				
				_		+		+		\neg		
		T	-			4		‡		1		
		+				+		+		╉		
						_		1				
	L					-		╉		-		
						1	Silvey - Convert	1		_		
	<u> </u>						ي ويوسين في ويوني ويوني	+		-		
•			-	_		7		-	an an an an an an an an an an an an an a		a personal debage	
			F						tal avenation) exercise			
			\vdash			-		Ļ	and an appropriate the party of		and a state of the	
						1				_	in and the second second	
				-	i Marina di Angela di Kasarangan	╉		-	an an an an an an an an an an an an an a	-	anga ang ang ang ang ang ang ang ang ang	
			F			t		t				
			\vdash			+		╀		-		
		1				1		t	بمحدد مناهدة			
		┿╌	+-			╉		╀╴		╉		
		4	F	_		T		L		1		
						╋		╀		┨		
			F	_		Ţ.		L		t		
		+				╉		+		╀		
		╂		-		Ļ		L		1		
						┢		┝				
				-		F				1-		
		-	L			t		F		┢		
		┼──	+	-		ŀ				1_		
				1		t						
			\vdash	+		╞		_		ŗ		
						E				┢		
			+	+		-+		_		-		
		L		1								
		ĺ	\vdash	╉		┝		_				
			F	7						-		
			t	+		-		-		_ .		
<u> </u>		h	F	Ŧ				_			-	
_			\vdash	+					and the second second second second second second second second second second second second second second secon		7	
				_					and a second state of the	-		
				+		-	n ministratives and		li maning biomer	and the second		
			F	Ŧ								
			-	╋		-						
T			\vdash	Ţ		_				-	manalatina di sunt	
				╋						d tooline or	- Colores	







A	8	C	P	E	1	Cons	i ect Jo
EPABX Circuit	Cable Number	e the	1 de	Terminal Numbers	Wire Color	Terminal	Vertica
		+-	┝		+	Number	Number
			F				
			-	+			-
							1
		+	+			+	+
			F	1			1
				+		+	-
			<u> </u>		<u>+</u>	+	+
		-					1
				<u>+</u>			
			<u> </u>				
		+					<u> </u>
			├	 	ļ]
		<u> </u>					1
		1					4
						<u> </u>	
				<u> </u>		+	-
Ī							1
		\vdash				<u> </u>	<u>†</u>
				ļ			
			_				
			_				
					_		
		T					
			-+				
		-					
		F	_				
			_			_	
			\rightarrow				
T		F					
			-+				
		F	-				
		⊢		+			
			-+-				
		F	·				
	T	F					
			_			+-	
		_F	-				
			-+-				
							i

.

•

.

.

4-19/(4-20 blank)

TM 11-5805-652-14

Satellite Equipment	Cable Number	Vertical	Output	Terminal Numbers	Wire	Location	
							Ι
						-	Τ
							T
						1	t
					ļ		╈
			_		<u> </u>	<u></u>	+
						<u> </u>	╉
							+
							╀
							╀
						-	\downarrow
					L	ļ	Ļ
		F					Γ
		-					T
							t
+							t
	l 1		 				1
							┢
							┝
							ļ
		-	-				
		_¦					
		-					
		-					
		-					
		-					
	+	-					
							· · · ·
			-+				
		F	+				L.
	<u> </u>						
			-				
		F	T				
			+				
	+						
			+			······	
			\pm				
		-+					
			-				
		·+-	_				_
		<u> </u>	_				
 		-	-+				_
			-				_
1			-				
			-+-				
1							

Satellite	Cable	3	5	25		1	C.O.
Equipment	Number	Vertic	ş	Termi.	Cele E	Location	Lines Numbe
			_				
		<u> </u>					
		+		+			
		ļ				!	<u> </u>
						-	
_		\uparrow	<u> </u>			1	1
		+	┝				
			-			1	
						•	
						-	
		1	—				†
		+	+				<u> </u>
		-				ļ	
					1	1	
				ļ		4	
		1		1			1
	<u> </u>	+	+	<u> </u>	<u> </u>	<u> </u>	
		1		<u> </u>		1	
				<u> </u>		1	
				+	+	4	
		1			1		1
			+	+			
		<u> </u>	-				
			-				
		1			ļ		1
		<u> </u>					
							<u> </u>
		┼──	╞				
		L					
			┝	<u> </u>			
		†		<u> </u>			
		┝	—				
			<u> </u>				
			┣				
			-				
		†		L			
	· · · · ·						
		L					
			<u> </u>				
		[—			······	
		┝					
			<u> </u>				
			L				
		├ —					
						1	

4-21/(4-22 blank)

Cable Connection to MDF -(Satellite Equipment)

			r		R			r r	
		Satellite Equipment	Cable Number	Vertical	Output	lerminal	Wire Color	Location	C. O Lines Number
								<u></u>	
					-				
		ļ							
								<u> </u>	
									ļ
									L
				[
				ŀ					
Ca									
ble Co									
nnectior	Tab								
to N	le 4-3								
IDF- ((Coi			İ					
Satellite	ıt'd)		1	ł					
Equi		1							
pment				+					
		+							
				F					
				‡					
	ĺ			-	-Ŧ				
		†		1 	+				
		+							

A			B			- <u>-</u> C	D
Satellite Equipment	Cable Number	Vertical	Output	Terminal Numbers	Kire Color	Location	C.O. Lines Number
		 					
		ļ					
		[
	[
						1	
							1
							<u> </u>
						1	
Ļ							
						·····	
						· · · · · · · · · · · · · · · · · · ·	







4-25/(4-26

Cable Connections to MDF-(Satellite Equipment)

A		Γ=				ι.	
Satellite Equipment	Cable Number	/ertica	Jutput	umber	W.r.e Color	Location	Lines
				<u><u> </u></u>			
),				
			 				
		L .	<u>↓</u>				
	/ L						ļ
			 				
• •• ••••••		1	•		• · · ·		
		† ·		 	⊷ _ ≁ ⊨		+ 1
			ļ				
							<u> </u>
	 						ļ
	! 				<u>↓</u> ↓		ļ
i			·	}			
					I		
•	•	+ 			⊷		
-	'	<u> </u>			↓ ↓↓		1
		<u> </u>	1				<u> </u>
	·	 					
	·	 }					+
		L			·		İ
	L	i_		· !			
	N 1	1	H	<u>↓</u>	·		
		1					1
	i	<u>†</u>			+		t
		<u> </u>			+		<u> </u>
		t	<u></u>		·		
	i	; ;		L			+
		ļ	ļ		+		
		ļ	<u> </u>		└────┤ ↓ +		ļ
•••••••	+	1					1
	i }	 1	ļ		<u>├</u> +		
					i		
	†						
		i			<u> </u>		
	·	1					ļ
L _	1				<u>↓ </u> ↓		
-		1	L -		↓ →		
	+ ,	1			·		
'	+	+	<u> </u>	L	+ + +		1
	+	+		<u> </u>	+		+
	I		+	+	•		+
1	,	1		1	+		1

					r		0
A Satellite Equipment	Cable Number	Vertical	Output	¹ erminal Numbers	Wire Color	Location	C. O. Lines Number
••••							
<u> </u>		+	+	 			┝
		ł					
				<u> </u>			
						4	
							1
		ļ					
	ŀ			<u> </u>		1	
	T	• 1		+	+		
	+	<u> </u>	.	<u> </u>			
	L				↓	1	
					·		
	+			<u>+</u>		<u> </u>	
	ļ		<u> </u>	Į			ļ
	1	1	<u></u>		<u> </u>	4	1
	1	+ 	+	+	+	4	
		ļ				1	+
						1	
							1
	+	+					+
							ļ
						4	
	+	1			1	<u>}</u>	1
	·					<u></u>	÷
			-	+		-	•
	1	1	1			1	1
	+	+				+	+
		i 				1	
			-			-	
<u> </u>	+	+				<u></u>	
		∔				<u> </u>	1
				+		4	
	1	+		1	1	1	1
	+	+ —			+	<u>+</u>	+
			↓		<u></u>	-	
		T	-			-	
	+	+	- <u> </u>	1	+	<u>+</u>	+
	+	+					
				+	· +	4	
		• + • •				+	1
	+	÷				+	+
						+	
		1					1
<u> </u>	-+	+-			+	+	+
		1				-	
		+ -	+	+	+		-+





4-27/(4-28 blank)

			r		P			1	T
		Satellite Equipment	Cable Number	Vertical	Output	Terminal Numbers	Wire Color	Location	C. O. Lines Number
						ļ	ļ	4	1
				<u> </u>					<u> </u>
				<u> </u>				<u> </u>	+
					<u> </u>]	
							ļ		
						}	╆		
					ļ	+ +	+ +		<u> </u>
									<u> </u>
							ļ		
						<u> </u>	ļ		
							+		
					+ ⊢	·			
					L.,				
					┣				
		+					·		
							<u> </u>		
				l	 				
			1				••		
	l							······	
Cab									
le C									
onne									
ctio	Tal				+				
ns to	ble		i +						
M	4-3			ļ	+				
)F-((C)								
Sate	nt'd)								
llite	ŀ								
Equ	ł								
ipme									
ent)	ļ				+	+ +			
				H			i		
	Γ			+- -					
	ľ		·	 		• • ·			
	ŀ			+	- 		+		
	ŀ						+		
	-		+						
				_		İ			
		+	4			+			
	F	+-	++	- +		+			
	-		+		+				

			-				T
Salellite Loupment	Cable Number	Vertical	Output	Terminal Numbers	Wirre Color	Location	C. O. Lines Number
							
					ļ	ł	
		ļ	┣	┢			
				┼───	<u> </u>	4	
			t	<u> </u>			<u> </u>
		Į .			<u> </u>	1	
				ļ	 		
			┣	{	<u> </u>	4	
		<u>+</u>	<u> </u>	<u> </u>			
		L				1	
	_		<u> </u>	L			
		 	 	 			
			<u> </u>			ł	
		1		<u> </u>		<u> </u>	†
						L	
					ļ		
		<u> </u>	<u> </u>	<u> </u>			
						4	1
		 				and a state of the second second second second second second second second second second second second second s	
						1	
					_		<u> </u>
		1				4	
					-		1
		l 				1	
						1	
•	and and a second second second		-				
1		1			the conservation of		
لد معد مده مه ا						and a state of the	
		┝──╺					
							ł
							•
Ţ							
				Of June 10 Film		وندمر برين ورون ورون	
ļ		i					· ·
						and the second second second	
			-				
				and an and a state			
1				مند کارد معاد			
						ور الساري بالسي بيمايين الماري و	
T							
 	 				ander de andersonel		
	1						
		2-122 -					
			-				
				m ne.m m			
			-		-		
		A-04 -14					
		_ [18. manual a sur	





4-29/(4-30	
blank)	

	r						·····	····
Satellite Equipment	e nt	Cable Number	Vertical	Output	Terminal Numbers	Wire Color	Location	C.O Lines Number
i k							-	
 							-	
							4	
	T			-				
1							ł	
							+	
;								
·	1		1					<u> </u>
, 	-†	·						
i	+							
	1							
·	+		+					
· · · · ·	-+							
1	+							
	-		<u> </u>					
	╉		+					
	+		+					
	+							
	╉						}	
	+							
	+	·						
;	-							
	1		<u> </u>					
	-							
	-+-							
 	_							
	_	<u> </u>						
2 0								
	Ţ							
	T							
	Ì							
	-							
	+							
	+							
	-+-							
	t		╞──┤					
	-+-		┝──┤					

A 1							0		
Satellite Equipment	Cable Number	Vertical	Output	Terminal Numbers	Wire Color	Location	C. O. Lines Number		
									
		ŧ	•						
		Ļ					ļ		
		1	Ĺ						
			<u> </u>						
							L		
							1		
		1	<u> </u>	<u>↓</u>					
							l		
			 	ļ					
					-				
		<u> </u>			Contrast of the second s		a alfanan ya anani dagan na di s		
		İ							
		1		and the second second	1				
		<u> </u>				a for a fight of the fight of the fight of the fight of the fight of the fight of the fight of the fight of the			
						en andere andere andere andere andere andere andere andere andere andere andere andere andere andere andere an			
		1							
		<u> </u>	 		-	Andrika Andria and Antonia	and a state of the second		
					arekani e nagyoya pikar	t a may mail bit	The second second second second second second second second second second second second second second second s		
		<u> </u>							
			-						
		·							
		1							
							1		
			<u> </u>				 		
		Í	├──						
		 					<u> </u>		
		Ĺ				<u> </u>			
			<u> </u>						
		<u> </u>							
			 	<u> </u>					
		 -							
					<u> </u>				
		i							
		L	<u> </u>						
						inensiyilin territorin energia	[
				<u> </u>	hite come ape	101811			
					a an daaqayaa garaa ahaa				
						a an an an an an an an an an an an an an			





Table 4-4

Line Tip and Ring Connecting Cable

This Chart Used on Cables: 130103-1 130103-7

St amp

(Connector) Line T & R 00-24"									
Pin	Serie	Color	Desig.	"A" Sub-Rack Pchg.	Pin	Serve	Color	Desig.	"A" Sub-Rack Pchg.
26		₩-B	T-00	00.0	1		B-W	R-00	00.1
27		₩-0	T-01	00.3	2		0W	R-01	00.4
28		W-G	T-02	00.6	3		G-W	R-02	00.7
29		W-BN	T-03	00.9	4		BN-W	R-03	00.10
30		W-S	T-04	01.0	5		S-W	R-04	01.1
31		R-B.	T- 05	01.3	6		B-R	R- 05	01.4
32		R-0	T- 0	01.6	7		O-R	R-06	01.7
33		R-G	T-07	01.9	8		G-R	R-07	01.10
34		R-BN	T-08	02.0	9		BN-R	R-08	02.1
35		R-S	T-09	02.3	10		S-R	R-09	02.4
36		вк-в	T-10	02.6	11		В-ВК	R-10	02.7
37		вк-о	T-11	02.9	12		О-ВК	R-11	02.10
38		BK-G	T-12	03.0	13		G-BK	R-12	03.1
39		BK-BN	T-13	03.3	14		BN-BK	R-13	03,4
40		BK-S	T-14	03.6	2.5		S-BK	R-14	03.7
41		Y-B	T-15	03.9	16		B-Y	R-15	03.10
42		¥-0	T-16	04.0	17		0-Y	R-16	04.1
43		Y-G	T-17	04.3	18		<i>с</i> ~у	R-17	04.4
44		Y-BN	T-18	04.6	19		BN-Y	R-18	04.7
45		Y-S	T-19	04.9	20		S-Y	R-19	04.10
46		V-B	T–20	04.0	21		B-V	R-20	05.1
47	[V-0	T-21	05.3	22		0-V	R-21	05.4
48		V-G	T-22	05.6	23		G-V	R-22	05.7
49		V-BN	T-23	05.9	24		BN-V	R-23	05,10
50		V-S	T-24	06.0	25		S-V	R-24	06.1

4-31

Table 4-5 line Tip and Connecting Cable

This Chart Used on Cables: 130103-1 130103-7

Stamp

(Connector) "Line T & R 25-49"

(CON	Dector)	Line T&	K 25-49						······································
Pin	Serve	Color	Desig.	A Sub-Rack Pchg.	Pin	Serve	Color	Desig.	Sub-Rack Pchg.
26		₩-B	T-25	06.3	1		B-W	R-25	06.4
27		W-O	T-26	06.6	2		0-W	R-26	06.7
28		₩-G	T-27	06.9	3		G-W	R-27	06.10
29		W-BN	T-28	07.0	4		BN-W	R-28	07.1
30		W-S	T-29	07.3	5		S-W	R-29	07.4
31		R-B	T-3 0	07.6	6		B-R	R-3 0	07.7
32		R-O	T-31	07.9	7		O-R	R-31	07.10
33		R-G	T-32	08.0	8		G-R	R-32	08.1
34		R-BN	T-33	08.3	9		BN-R	R-33	08.4
35		R-S	T-34	08.6	10		S-R	R-34	08.7
36		BK-B	T-35	08,9	11		B-BK	R-35	08,10
37		BK-O	T-36	09.0	12		O-BK	R-36	09.1
38		BK-G	T-37	09.3	13		G-BK	R-37	09.4
39		BK-BN	T-38	09.6	14		BN-BK	R-38	09.7
40		BK-S	T-39	09.9	15		S-BK	R-39	09.10
41		Y-B	T-40	10.0	16		B-Y	R-40	10.1
42		¥-0	T-41	10.3	17		0-У	R-41	10,4
43		Y-G	T-42	10.6	18		GY	R-42	10.7
44		Y-BN	T-43	10.9	19		BN-Y	R-43	10.10
45		Y-S	T-44	11.0	20		S-Y	R-44	11.1
46		V-B	T-45	11.3	21		B-V	R-45	11.4
47		~v-0	T-46	11.6	22		0-V	R-46	11.7
48		V-G	T-47	11.9	23		G V	R-47	11.10
49		V-BN	T-48	12.0	24		BN-V	R-48	12.1
50		V-S	T-49	12.3	25		S-V	R-49	12.4

Table 4-6

Line Tip and Ring Connecting Cable

This Chart Used on Cables: 130103-1 130103-7

Stamp

"Line T & R 50-74" (Connector)

	Aector)	mme r e	A 00-11			_			10 - 10
Pin	3erve	Color	Desig.	"A" Sub-Rack Pchg.	Pin	Serve	Color	Desig.	"A" Sub-Rack Pchg.
26		₩-В	T-50	12.6	1		B-W	R-50	12.7
27		₩-0	T-51	12.9	2		0-W	R-51	12,10
28		₩-G	T-52	13.0	3		G-W	R-52	13,1
29		W-BN	T-53	13.3	4		BN-W	R-53	13.4
30		W-S	T-54	13.6	5		S-W	R-54	13.7
31		R-B	T-55	13.9	6		B-R	R-55	13.10
32		R-O	T-56	14.0	7		O-R	R-56	14.1
33		R-G	T-57	14.3	8		G-R	R-57	14.4
34		R-BN	T-58	14.6	9		BN-R	R-58	14.7
35		R-S	T-59	14.9	10		S-R	R-59	14.10
36		BK-B	T-60	15.0	11		B-BK	R-60	15.1
37		BK-O	T-61	15.3	12		O-BK	R-61	15.4
38		BK-G	T-62	15.6	13		G-BK	R-62	15.7
39		BK-BN	T-63	15.9	14		BN-BK	R-63	15.10
40		BK-S	T-64	16.0	15		S-BK	R-64	16.1
41		Y-B	T-65	16.3	16		B-Y	R-65	16.4
42		¥-0	T-66	16.6	17		0-Y	R-66	16.7
43		Y-G	T-67	16.9	18		G-Y	R-67	16.10
44		Y-BN	T-68	17.0	19		BN-Y	R-68	17.1
45		Y-S	T-69	17.3	20		S-Y	R-69	17.4
46		V-B	T-70	17.6	21		B-V	R-7 0	17.7
47		v-0	T-71	17.9	22		0-V	R-71	17.10
48		V-G	T-72	18.0	23		G-V	R-72	18.1
49		V-BN	T-73	13.3	24		BN-V	R-73	18.4
50		V-S	T-74	18.6	25	Γ	S-V	R-74	18.7
Table 4-7 line lip and Ring ConnectingCable

This Chart Used on Cables: 130103-1 130103-7

Stamp

(Connector)	"Line	т	&	R	75-99"	
		_	_	_		

Pin	Serve	Color	Desig.	"A" Sub-Rack Pchg.	Pin	Serve	Color	Desig.	"A" Sub-Rack Pchg.
26		W-B	T-75	18.9	1		B-W	R-75	18.10
27		₩-0	T-76	19.0	2		0-W	R-76	19.1
28		₩-G	T-77	19.3	3		G-W	R-77	19.4
29		W-BN	T-78	19.6	4		BN-W	R-78	19.7
30		W-S	T-79	19.9	5		S-W	R-79	19.10
31		R-B	T-80	20.0	6		B-R	R-80	20.1
32		R-O	T-8 1	20.3	7		O-R	R-81	20.4
33		R-G	T-82	20.6	8		G-R	R-82	20.7
34		R-BN	T-83	20.9	9		BN-R	R-83	20.10
35		R-S	T-84	21.0	10		S-R	R-84	21.1
36		BK-B	T-85	21.3	11		B-BK	R-85	21.4
37		BK-O	T-86	21.6	12		0-вк	R-86	21.7
38		BK-G	T-87	21.9	13		G-BK	R-87	21.10
39		BK-BN	T-88	22.0	14		BN-BK	R-88	22.1
40		BK-S	T-89	22.3	15		S-BK	R-89	22.4
41		Y-B	T-90	22.6	16		B-Y	R-90	22.7
42		¥-0	T-91	22.9	17		0- Y	R-91	22.10
43		Y-G	T-92	23.0	18		G-Y	R-92	23.1
44		'Y-BN	T-93	23.3	19		BN-Y	R-93	23.4
45		Y-S	T-94	23.6	20		S-Y	R-94	23.7
46		V-B	T-95	23.9	21		B-V	R-95	23.10
47		V-0	T-96	24.0	22		0-V	R-96	24.1
48		V-G	T-97	24.3	23		G-V	R-97	24.4
49		V-BN	T-98	24.6	24		BN-V	R-98	24.7
50		V-S	T-99	24.9	25		S-V	R-99	24,10

7.0 PREWIRED TRUNKS/TIE LINES TIP-RING CONNECTIONS

- 7.1 Each cabinet will be **equipped with trunk tip-ring connecting cable(s).**
- **7.2** The Table 4-6, Figs. 4-2 and 4-7, for reference.
- 7.3 Cable Run

(a) Determine cable lengths.

- (b) Insure they are terminated with male connector(s) that will mate
- with the female amphenol connector(s) that will

(c) Identify each cable at both ends.

(d) Using Table 4-2, butt, strip, and terminate cables to MDF vertical(s)

7.4 Connect cables via connectors - male to female. Check all connectors for snug fit and that all holding screws are tight.

8.0 PREWIRED RED PFCT AND MISCELLANEOUS CONNECTING CABLE AND TRUNK E-M AND A-B LEADS CONNECTIONS

8.1	Prewired	PFCT	and	Miscellaneous
	Connectin	ng Cab	le	

- 8.1.1 Each cabinet will be equipped with one cable.
- 8.1.2 Identify this cable as in paragraph 3.1.1
- 8.1.3 Use Table 4-9, Figs 4-2 4-6, and 4-7 for reference.
- 8.1.4 The cable is 25 pair, 24 gauge and has a <u>female</u> ended amphenol connector.
- 8.1.5 Connect cable as outlined on Steps (a) through (d) of paragraphs 7.3 and 7.4

8.2 Trunk E-M and A-B Leads

8.2.1 Each cabinet will be equipped with one or two cables leach cable provides leads for 12 trunks)

- 8.2.2 Identify this cable as in paragraph 3.1.1
- 8.2.3 Use Tables 4-10, 4-11, *Figs* 4-2, 4-6, and 4-7 for reference
- 8.2.4 Connect cable(s) as outlined in Steps (a) through (d) of paragraphs 7.3 and 7.4.

9.0 INSTALLER WIRED CABLING

9.1 Cable layout, identification. and length should be arrived at by following instructions outlined on paragraphs 6.0, 7.0, and 8.0.

survisions outlined on paragraphs 6.0, 7.0, and 8.0.

9.2 The Installer will terminate cable(s) at both ends.

9.3 Lines

9.3.1 Refer to Dwg. 130071 and Fig. 4-7.

block for line T-R is located parallel with level A in each cabinet.

- 9.3.3 Refer to Table 4-2 and terminate cable(s) at both ends.
- 9.4 Trunks
- 9.4.1 **Refer to Dwg :** 13076, 130077, and **Fig 4-7.**

9.4.2 Dwgs 130076 and 130077 show that both DID and city trunks utilize the same terminal block for T-R.

9.4.3 Refer to Section III of this manual and Equipment Specifications and establish which levels are equipped with trunks. The punching block is parallel with each equipped level.

9.4.4 Using the information supplied by Table 4-2, terminate cable(s) at both ends.

9.5 Tie Lines

9.5.1 mute, identify, and determine cable(s) length as described in paragraphs 6.0, 7.0, and 8.0, then terminate cable at both ends.

9.5.2 Refer to Dwg. 130078 and Fig. 4-7.

9.5.3 Tie lines may be terminated the same as in the trunk' lines except that 12 T-R terminals are provided.

9.6 Miscellaneous Outputs

Refer to Equipment Specifications to

determine:

- (a) If the following outputs and inputs are required
- (b) Location of level equipped with circuit
- 9.6.1 Unassigned Night Answer (UNA) Outputs
 - (a) Four outputs are available -URL1, URL2, URL3, and URL4
 - (b) Provide cabling as required to URL1, URL2, URL3, and URL4.

Table 4-8 Line Tip and Ring Connecting Cable

This Chart Used for Cables: 130103-2 130103-8

Stamp Cunector "TRK T & R"

Pin	Serve	Color	Desig.	Sub-Rack Pchg.	Pin	Serve	Color	Desig.	Sub-Rack Pchg.
26		₩B	TI	D-00.0	1		B-W	Rl	D-00.1
27		₩-0	T2	D-00,2	2		0-W	R2	D-00.3
28		₩-G	T 3	D-00.4	3		G-W	R3	D-00.5
29		W-BN	T4	D-00.6	4		BN-W	R4	D-00.7
30		W-S	T 5	D-00.8	5		S-W	R5	D-00.9
31		R-B	T 6	D-00.10	6		B-R	R6	D-00.11
32		R- 0	T 7	E-00.0	7		O-R	R7	E-00.1
33		R-G	T8	E-00.2	8		G-R	R8	E-00.3
34		R-BN	T9	E-00.4	9		BN-R	R9	E-00.5
35		R-S	T10	E-00.6	10		S-R	R10	E-00.7
36		BK-B	T 11	E-00.8	11		B-BK	R11	E-00.9
37		BK-O	T12	E-00.10	12		O-BK	R12	E-00.11
38		BK-G	T13	F-00.0	13		G-BK	R13	F-00.1
39		BK-BN	T14	F-00.2	14		BN-BK	R14	F-00.3
40		BK-S	T15	F-00.4	15		S-BK	R15	F-00.5
41		Y-B	T16	F-00.6	16		B-Y	R16	R-00,7
42		¥-0	T17	F-00.8	17		0-Y	R17	F-00,9
43		Y-G	T18	F-00.10	18		GY	R18	F-00,11
44		Y-BN	T19	G-00.0	19		BN-Y	R19	G-00.1
45		Y-S	T20	G-00.2	20		S-Y	R2 0	G-00.3
46		V-B	T21	G-00.4	21		B-V	R2 1	G-00.5
47		V-0	T22	G-00.6	22		0-V	R22	G-00,7
48		V-G	T23	G-00.8	23		G-V	R23	G-00.9
49		V-BN	T24	G-00.10	24		BN-V	R24	G-00.11
50		V-S		SPARE	25		S-V		SPARE

Table 4-9 PFCT and Miscellaneous Connecting Cable

This Chart Used for Cables: 130103-3 130103-9

Stamp

(Connector) "P F & Misc"

Pin	Serve	Color	Desig.	Sub-Rack Pchg.	Pin	Serve	Color	Desig.	Sub-Rack Pchg.
26		W-B	LT1	B-19.0	1		B-W	LR1	B-19.1
27		₩-0	LT2	B-19.2	2		0-W	LR2	B-19.3
28		₩−G	LT3	B-19.4	3		G - ₩	LR3	B-19.5
29		W-BN	ST1	B-19.6	4		BN-W	SR1	B-19.7
30		W-S	ST2	B-19.8	5		S-W	SR2	B-19.9
31		R-B	ST3	B-19.10	6		D-R	SR3	B-19.11
32		R-O	TTI	B-20.0	7		O-R	TR1	B-20.1
33		R-G	T12	B-20.2	8		G-R	TR2	B-20.3
34		R-BN	TT3	B-20.4	9		BN-R	TR3	B-20.5
35		R-S	CT1	B-20.6	10		S-R	CR1	B-20.7
36		ВК-В	CT2	B-20.8	11		В-ВК	CR1	B-20.9
37		BK-O	СТЗ	B-20.10	12		O-BK	CR3	B-20.11
38		BK-G			13		G-BK		
39		BK-BN			14		BN-BK		
40		BK-S			15		S-BK		
41		Y-B	TR1	C-00.0	16		B- Y	APC1	C-00.3
42		Y-0	TR2	C-00.1	17		0-Y	APC2	C-00.4
43		Y-G	TR3	C-00.2	18		G- Y		
44		Y-BN	URL1	D-06.6	19		BN-Y		
45		Y-S	URL2	D-06.7	20		S-Y		
46		V-B	URL3	D-06.8	21		B-V		
47		V-0	URLA	D-06.9	22		0-V	CCM1	
48		V−G			23		G-V		
49		V-BN			24		BN-V		
50		v-s			25		S-V		

4-37

Table 4-10E&M -A&B Connecting Cable #1

This Chart Used on Cables: 130103-2 130103-8

Stamp (Connector) "TRK. E&M-A&B1"

Pin	Color	Desig.	P	chg.	Pin	Color	Desig.	Р	chg.
26	W-B	E-1	••1	D" 02.0	1	B-W	M-1	"	D" 02.1
27	W-0	E-2		02.2	2	0-W	M-2		02.3
28	₩-G	E-3		02.4	3	G-W	M-3		02.5
29	W-BN	E-4		02.6	4	BN-W	M-4		02.7
30	W-S	E-5		02.8	5	S-W	M~5		02.9
31	R-B	E-6		02.10	6	B-R	M-6		02.11
32	R-O	A-1		03.0	7	O-R	B-1		03.1
33	RG	A-2		03.2	8	G-R	B-2		03.3
34	R-BN	A-3		03.4	9	BN-R	B-3		03.5
35	R-S	A-4		03.6	10	S-R	B-4		03.7
36	BK-B	A-5		03.8	11	В-ВК	B-5		03.9
37	ВК-О	A-6	1	03.10	12	о-вк	B-6		03.11
38	BK-G	E-1	"1	5" 02.0	13	G-BK	M-1	"I	8" 02.1
39	BK-BN	E-2		02.2	14	BN-BK	M-2		02.3
40	BK-S	E-3		02.4	15	S-BK	M-3		02.5
41	Ү-В	E-4		02.6	16	B-Y	M-4		02.7
42	¥-0	E-5		02.8	17	0 Y	M-5		02.9
43	Y-G	E-6		02.10	18	G Y	M-6		02.11
44	Y-BN	A-1		03.0	19	BN-Y	B-1		03.1
45	Y-S	A-2		03.2	20	 S-Y	B-2		03.3
46	V-B	A-3		03.4	21	B-V	B-3		03.5
47	V-0	A-4		03.6	22	 0-V	B-4		03.7
48	V-G	A-5		03.8	23	G-V	B-5		03.9
49	V-BN	A-6	1	03.10	24	BN-V	B-6		03.11
50	V-S				25	s-v			

Table 4-11E&M- A&B Connecting Cable #2

This Chart Used on Cables: 130103-2 130103-8

Stamp (Connector) "TRK. E&M- A&B2"

Pin	Color	Desig.	P	chg.	Pin	Color	Desig.	P	chg.
26	W-B	E-1	"1	F" 02.0	1	B-W	M-1	"1	F" 02.1
27	W-O	E-2		∩2.2	2	0-W	M-2		02.3
28	₩-G	E-3		02.4	3	G-W	M-3		02.5
29	W-BN	E-4		02.6	4	BN-W	M-4		02.7
30	WS	E-5		02.8	5	S-W	M-5		02.9
31	R-B	E-6		02.10	6	B-R	M-6		02.11
32	R-O	A-1		03.0	7	O-R	B-1		03.1
33	R-G	A-2		03.2	8	G-R	B-2		03.3
34	R-BN	A-3		03.4	9	BN-R	B-3		03.5
35	R-S	A-4		03.6	10	S-R	B-4		03.7
36	BK-B	A5		03.8	11	B-BK	B-5		03.9
37	вк-о	A-6		03.10	12	0-вк	B-6		03.11
38	BK-G	E-1	"(G" 02.0	13	G-BK	M-1	"0	G" 02.1
39	BK-BN	E-2		02.2	14	BN-BK	M-2		02.3
40	BK-S	E-3		02.4	15	S-BK	M-3		02.5
41	Y-B	E-4		02.6	16	B-Y	M-4		02.7
42	¥-0	E-5		02.8	17	0-Y	M-5		02.9
43	Y-G	E-6		02.10	18	G Y	M-6		02.11
44	Y-BN	A-1		03.0	19	BN-Y	B-1		03.1
45	Y-S	A-2		03.2	20	 S-Y	B-2		03.3
46	V-B	A-3		03.4	21	B-V	B-3		03.5
47	V-0	A-4		03.6	22	0-V	B-4		03.7
48	V-G	A-5		03.8	23	G-V	B-5		03.9
49	V-BN	A-6		03.10	24	BN-V	B-6		03.11
50	V-S				25	s-v			

- Refer to Dwg. 130076 for punch-(c) ing block connection.
- Furnish one ground wire for punching block 130071 for each (d) output.
- 9.6.2 Public Address Outputs
 - Three outputs are available-(a) TR1-TR3 for 600 ohms impedance TR1-TR2 for 300 ohms Imped
 - ance
 - APC1-APC2 for switch closure **Refer to Dwg.** 130073 **punching** (b) block for terminal location.
 - Provide cabling as required. (c)

9.6.3 External 48VDC supply input

- Only systems equipped with DID trunk tie lines will require (a) -48VDC

(b) Cable -48VDC and ground as re-quired to each level so equipped Refer to Dwgs. 130077 and 130078 for punching block terminal locations

On those systems equipped with EM 9.6.4 Trunk/Tie Lines the E-M loads will have to be connected to the terminating satellite equipment A maximum DCR loop of 100 ohms is allowed

Refer to Dwgs 130077 and 130078 for punching block terminal locations Cable and connect as required

9.7 Hotel/Motel Connections

9.7.1 Line Cabling - Line cabling will be the same as described in paragraph 6.0 and Steps 9.3.1 through 9.3.3 of paragraph 9.3 except:

(a) All lines utilized for message waiting are connected to the R1 terminal instead of the R terminal on punching block 130071

Example.

X00-TO and R1-0 X00-T1 and R1-1

- 9.7.2 Trunk Cabling- Trunk cabling will be the same as described in paragraph
- 9.7.3 Trunk Loop Adjustment Resistors -Refer to paragraph 15 0 and Dwg 130079
- 9.7.3.1 Trunks 316061-316075
 - **RBR1 same value** as **TR2** (b) Equip with RBR1 Instead of MR5
- 9.7.4 Hotel/Motel Attendant Position
 - Refer to Dwg. 130079 (a)

MW terminals are provided for (b)

100 lines per cabinet MW-0 corresponds to line X00.

PC (Peg Count) terminals are provided for 100 lines per cabi-(c)

net. Terminal PC-O corresponds to line X00 Cable and connect as required (**d**)

10.0 PFCT CONNECTIONS WITH PREWIRED RED CABLING

Assignment - Three PFCT stations 10.1 may be assigned per cabinet Refer to Section II, paragraph 2.10

Establish the station(s) assigned from Table 2-6 of Section II. Refer to 10.1.1 Table 4-2 of this section and determine vertical assignments

10.1.2 Insure that central office (CO) trunks are not E&M trunks E&M trunks cannot be assigned to PFCT

If a mixture of DID loop, loop start, 10.1.3 and/or ground start trunks is provided, utilize (if possible) in the following order:

> (a) Loop start (b) Ground start (c) **DID** loop

10.1.4 Ground star: trunks will require PFCT station instruments be equipped with a groundbutton

10.1.5 See Table 2-5 of Section II for trunking information and Equipment Specifications for cabinets equipped in

NOTE: The following connection" information IS for initial installation Prewire <u>but do not connect</u> Jumpers if installation has an<u>other system in opera-</u> tion

10.2 Refer to Table 4-9 and perform the following steps:

From EPABX Line Vertical, connect the first PFCT Line No T to miscel-10.2.1 laneous connect cable vertical terminal LT1 Repeat for R to LR1 If jumper has been connected from line vertical to station vertical, remove T-R Jumpers

10.2.2 From station vertical connect the first PFCT station T to miscellaneous connect cable vertical terminal ST1 Repeat for R to SR1

From the EPABX trunk vertical, con-10.2.3 nect the first trunk assigned as PFCT T to TT1 of the miscellaneous connect cable vertical Repeat for R to TR1

From the CO trunk vertical connect the first CO trunks T to the miscel-10.2.4 laneous connect cable vertical Repeat for R to CR1

Repeat above until all PFCT circuits/ 10.2.5 stations are assigned

4-40

11.0 JNASSIGNED NIGHT ANSWER PREWIRED CABLING

11.1	Assignment - Four outputs are avail-
	able to connect to bells for audible
sionaling of	a truck call during UNA operation.

11.1.1 The four available cutputs are:

(2)	URL1
(Ь)	URL2
(c)	URL3
(d)	URLA

11.1.2 Each output is activated until the call(s) is answered via the UNA circuit.

11.1.3 Connect cable(s) as required to PFCT and miscellaneous connect vertical terminals URL1, URL2, URL3, and URL4 as required.

11.2 Refer to Table 4-9.

A ground common to the system should be provided from the external bell(s) to system ground

12.0 PUBLIC ADDRESS CONNECTION- PRE-WIREDCABLE

12.1 The public address (PA) PCB allows an unrestricted EPABX station to access an external PA system.

- 12.1.1 The following three outputs are available:
 - (a) TR2 to TR3 300 ohms imped-
 - (b) TR1 to TR3 600 ohms impeda n c e
 - (c) APC1 to ACP2 Closed switch when PA is accessed.

12.2 Connections

12.2.1 Determine the input impedance of the PA system. Connect cable pair to PFCT and miscellaneous connector vertical terminals TR2-TR3 or TR1-TR3. Connect cable pair to terminals APC1-APC2 if a switch closure is desired.

12.2.2 Refer to Tables 4-2 and 4-9

13.0 OTHER FEATURES STRAPPING

- **13.1** Refer to Equipment Specifications, Table A, **and** Installation Motes.
- **13.2** Use Dwg. 130117 to determine straps required. Use appropriate figure and page number, then strap as required.

14.0 CLASS OF SERVICE ASSIGNMENT/ EOUIPPING

14.1 Refer to Table 2-6 in Section II of this manual.

from each cabinet. The lines and inputs to the COS circuit

Remove class of service (COS) PCB's

14.3The lines and inputs to the COS circuit
are .dentified as 1, 2, 3, 4, 5.

14.3.1 The Equipment Specifications shall detail the circuits associated with

each COS point.

14.4

14.2

Example:	Access Code (Dial 9) Trunk	+	1
-	Access Code (Dial 5) PA	-	2
	Toll Restrictor	•	3

- Table 2-6 of Section II lists the class of service for each station assigned.
- 14.5 Install diodes as required to restrict system stations. Solder diodes to PCB connection terminals.
 - 14.5.1 Perform this step an all cabinets of the system.
- **14.6** Replace COS PCB's but do not reinaert into female connectors.

15.0 TRUNK/TIE LINES LOOP ADJUSTMENT RESISTOR

- **15.1** Refer to Equipment Specifications, Table 2-5, and paragraphs 3.3.1-E and 3.3.2 of Section II.
- **15.2** Table 2-5 provides location and **value** of resistors to be installed.
- **15.3** Refer to Dwg. 130077 for DID trunk information.
- **15.4** Refer to Dwg. 130076 for city trunk information
- **15.5** Refer to Dwg. 130078 for tie lines information.
- 15.6 Install apperciate resistors.

16.0 SATELLITE EQUIPMENT CABLING

16.1 Initial Installations

(a) Refer to Table 4-3.

(b) Install, designate, and **cable any** terminal appearances required for Satellite Equipment on the MDF verticals

16.2 Replacement Installations

16.2.1 See Section VU of this manual.

17.0 POWER-OFF MEASUREMENTS

17.1 Remove all PCB's from their female connectors by pulling the boards outward. Leave **disconnected** PCB's in their respective mounting slots.

tion Motes. 16.2

17.2 With VOM on Rx1 scale, measure the resistance of the following DC power supply outputs to ground. Refer to Fig. 3-1 of Section III.

-18VDC	5 ohms minumum
+18.5VDC	5 ohms minimum
+36VDC	15 ohms minimum
+3VDC	5 ohms minimum
-36VDC	15 ohms minimum
-3VDC	5 ohms minimum

17.3 Keasure DC resistance on each horizontal bus bar: Ground to Ground - zero ohms maximum Measure DC resistance on one horizontal bus bar:

> Each voltage pin to ground - infinity (open) Each voltage pin to all other voltage pins - infinity (open)

CAUTION: Do not perform any "Power-On Adjustments" of Section V until performing all s**teps and** meeting all requirements of paragraphs 17.1, 17.2, 17.3, and 17.4.

Section V

CONTENTS

			Page
1.0	Genera	al	5-1
2.0	Initial	Adjustments	5-1
	2.1	AD Power On	5-1
	2.2	Blower Operation	5-1
	2.3	DC Power Supplies Access	5-1
	2.4	DC Power Supplies Adjustment	5-1
	2.5	+3VDC and -3VDC Adjustments (Multicabinet)	5-1
	2.8	5VDC Power Supply Adjustment	5-1
	2.7	Power Removal	5-1
	2.8	Reapplying AC Power	5-1
	2.9	Voltage Check and +5VDC Adjustments	5-2
	2.10	48VDC Alarm	5-5
	2.11	3VDC Alarm Adjustments	5-5
	2.12	AC Power On-Off Test	5-5
3.0	Commo	on Control Circuits Tests	5-5
	3.1	Printed Circuit Board Insertion	5-5
	3.2	Tones Supply Circuit (318037) Adjustment Test	5-5
	3.3	Routine Test	5-5
	3.4	Register(s) (318302) Test	5-8
4.0	Junctor	· (318024) Test	5-8
	4.1	Function	5-8
	4.2	Junctor Selection	5-8
	4.3	Junctor Test	5-8
5.0	Line Ci	ircuit (318040) and Feature(s) Circuit Test	5-7
	5.1	Preliminary	5-7
	5.2	Equipped Features Test	5-7
	5.3	Line-to-Line and Line-to-Features	5-7
6.0	Feature	Circuits Test	5-7
7.0	Trunk	Allotter (318034) Test	5-9
8.0	Console	Operation	5-10
	8.1	ITT Console	5-10

CONTENTS (Cont'd)

	8.2 Bell Console	Page 5-12
	8.3 Multi-Attendant Operation	5-12
9.0	Tru nk/ Tie Line Test	5-12
10.0	Toll Restrictor (316067) Test	5-12
11.0	Attendant Digit Read-Out Test	5-13
12.0	Camp-On-Busy (316031) Test	5-13
13.0	Attendant Line Test	5-13
14.0	Trunk Conference Accessing (318077) Test	5-13
15.0	Restricted Station . Class of Service Test	5-13
16.0	Power Failure Cut-Through Test	5-13

FIGURES

5-1	Lamps, VO and VT Adjustment Locations	5-3
5-2	DC Power Supply Adjustment Locations	5-4
5-3	DC Power Supplies High-Low Voltage Adjustments	5-4

1.0 GENERAL

The location of all busy/seize, allott, 1.1 and alarm lamps IS shown, in Fig. 5-1. The lamp color codes are:

> Alarm - red Allott - yellow Busy/Seize - green

NOTE: Lamps may or may not be **color coded**, depending on **issue of** PCB.

The location of all voice originate (VO) 1.2 and voice terminate (VT) potentiometers requiring adjustment is also shown in Fig 5-1.

The adjustment of all VO and VT po-1.2.1 tentiometers IS accomplished in the following manner:

Place a 20,000 ohms per volt VOM on the positive DC scale (a) (3VDC scale is recommended, if available) Connect positive meter lead to appropriate VO or VT poten-tiometer Test Point (TP) Connect negative meter lead to ground (any convenient chassis point may be

used). (b) Adjust associated VO or VT po-tentiometer for +1VDC

NOTE: Circuit under Adjustment must be seized before adjustment is attempted. Walt 60 seconds before adjustment is attempted.

CAUTION: Use insulated clips to connect VOM to VT/VO Test Point. Do not short VT/VO test point to other PCB components, printed circuit tracks, or ground <u>Damage will result if a short is placed on</u> <u>PCB</u>. Use an insulated screw driver to adjust potentiometers.

1.3 The following abbreviations are used in this Section:

TJP - Test phone equipped with Jack for insertion into test circuit.

TVP - Test phone equipped to connect to MDF vertical terminals

An Instrument must be connected as a 1.4 conventional city line so that it may be used for the performance of Section V tests

- 2.0 INITIAL ADJUSTMENTS
- 2.1

Apply AC power by:

AC Power On

- Connect AC power cords to AC (a) source
- Position AC circuit breakers (cabinet 1) to ON (h)

NOTE: All AC circuit breakers for complete system are located in cabinet 1.

2.2 **Blower** Operation

Insure that blowers are operation by checking each air output (lower front).

DC Power Supplies Access 2.3

Access to the DC power supplies (for adjustment) is achieved by lifting the ventilation assembly up and swinging assembly to left.

2.4 DC Power Supplies Adjustments

Refer to Fig. 5-2 for test point (TP) and adjustment locations

Using a calibrated 20,000 ohm per volt VOM, adjust the -18VDC, +18 5VDC, +36VDC, +3VDC, -36VDC, and -3VDC supplies in first cabinet (-18V, +18.5V, +38V, and -38V are not adjusted in cabinets 2, 3, or 4)

Any convenient chassis point may be used for VOM ground connection. Place VOM on appropriate DC scale for accurate reading.

+3VDC and -3VDC Adjustments 2.5 (Multicabinet)

Adjust the +3VDC and -3VDC supplies in cabinets 2, 3, and 4 Replace DC power supplies ventilation assembly to closed position

5VDC Power Supply Adjustment 2.6

Refer to Fig. 3-10 of Section III In-sert +5VDC supply and alarm PCB into its connector Connect VOM positive lead to bottom terminal of power resistor 1R1. Connect VOM negative lead to chassis ground Adjust 5VDC power supply for +5VDC by adjusting the potentiometer located on the bottom front of the +5VDC supply and alarm PCB.

Power Removal 2.7

to OFF

2.8

2.7.2

2.7.1 Remove AC power before inserting PCB's by placing AC circuit breakers

Insert the following PCB's:

- Voltage check reference (a) 316070()122 (h)
- Voltage check main alarm 316069()122
- System main alarm (c) 316068(2)122

Reapplying AC Power

Return AC power by placing AC circuit breakers to ON

5-1

2.9 Voltage Check and +5VDC Adjustments

The power relay (PR) may or may not operate when the AC power is reapplied. The alarm lamps on the 316069 PCB may or may not be "on."

PR relay operates - Lamps on 316069 2.9.1and Hi-Lo lamps on 316301 are extinguished.

NOTE: If PR does not operate, proceed to paragraph 2.9.2

Using a calibrated VOM, measure the 2.9.1.1 outputs of the Voltage Check PCB (316070). Each **output** may be measured from the test point at the front of the PCB. Each TP is Identified. All voltages have a $\pm 10\%$ tolerance.

2.9.1.2 Main Alarm Voltage Check - Refer to Figs. 5-2 and 5-3

Hi-Lo Voltage Detection Adjustment

Connect a calibrated DC volt-(a)

- meter (use-ate scale) to the test Jack of the +36 volt power supply.
 - Adjust the +36 volt power supply for an output of 39.6 volts (b)
 - Press the voltage check reset (c)

push button while rotating the +36 volt high "potentiometer adjust" (second from bottom of main alarm voltage check PCB) **clockwise** until the top lamp is extinguished.

Slowly rotate the "potentiometer adjust" counterclockwise until (**đ**)

the associated lamp comes on '

- Return the +36 volt power supply (e) output to +36 volts
- Press the voltage check reset (f) push button to extinguish the lamp
- Adjust the +36 volt power supply for an output of 32 4 volts (g)
- Press the voltage check reset (h) push button while rotating the

+36 volt low "potentiometer adjust" (top of PCB) counterclockwise until the associated lamp IS extinguished

- Slowly rotate the "potentiometer adjust" clockwise until the lamp comes "on"
- Return the +36 volt power supply (j) output to +36 volts
- Press the voltage check reset (k) push button to extinguish the
- lamp Repeat Steps (a) through (k) using the -36 volt power supply, **(l)**

the -36 volt low potentiometer and lamp (third from top of board), and the -36 volt high potentiometer and lamp (fourth from bottom of board)

- Connect the DC voltmeter to the test jack of the +18 5 volt power (m)
- supply Adjust the +18.5 volt power sup-ply for an output of 20.4 volts (n)
- Press the voltage check reset (0)push button while rotating the

+18.5 volt high "potentiometer adjust" (first from bottom of board) clockwise until the associated lamp

(p) Slowly rotate the **'potentiometer** adjust' counterclockwise until the lamp comes "on."

- Return the +18.5 volt power sup-ply **output to** +18.5 volts. (q)
- Press the voltage check reset (r) push button to extinguish the lamp.
- Adjust the +18.5 volt power sup-ply for an output of 16.6 volts. (s)
- Press the voltage check reset (t)

+18.5 volt low "potentiometer **adjust**" (fourth **from** top of board) counterclockwise until the associated lamp is extinguished.

- Slowly rotate the "potentiometer (u) adjust" clockwise until the lamp comes "on."
- Return the +18 5 volt power sup-(v) ply output to +18.5 volts
- Press the voltage check reset (w) push button to extinguish the
- lamp Repeat Steps (m) through (w) using the -18 volt power supply, (**x**)

the -18 volt low potentiometer and lamp (second from **top o** of board), and the -18 volt high potentiometer and lamp (third from **bottom** of board) All lamps should now be extinguished and the power relay should remain operated

- 2.9.1.3 +5VDC Hi-Lo Adjustments - Refer to paragraph 2 6
 - Low Voltage Adjustment
 - Adjust +5VDC supply to +4 2 volts The PR Relay "drops (a)

out" and "sets up" an alarm condition causing the +5 volt alarm lamp to turn "on"

- Readjust voltage to +5 volts (h)
- Depress VC reset button. The (c)
- PR relay operates and the 5-volt alarm lamp goes "out"

High Voltage Adjustment

(a)

Adjust +5VDC supply to +5 8 volts The PR relay "drops out" and "sets up" an alarm condition causing the S-volt

- alarm lamp to turn "on'
 - Readjust voltage to +5 volts (h)
 - Depress VC reset button The (c) PR relay operates and the 5-

volt alarm lamp goes "out"

292 Power relay does not operate - lamps on 316069 and/or 3160301 remain "on"

2.9.2.1 Perform the following steps if the alarm lamp(s) on the 316069 PCB

remain "on."

Refer to Fig 5-2 and repeat all steps of paragraphs 2 4, 2.5, and (a)







Fig. 5-2. DC Power Supply Adjustment Locations.

Fig. 5-3. DC Power Supply High-Low Voltage Adjustments.

	(b) Refer to Fig. 5-3 and repeat all	
	steps of paragraphs 2.9.1.1,	
	2.9.1.2, and 2.9.1.3.	
	(c) If alarm lamp(s) remain "on,"	
	refer to Section XII, paragraph	
	7.0, and Section XIII, paragraph	
	2.1.	
2.9.2.2	Perform the Pollowing steps if alarm lamps(s) of 316301 PCB remain **on.'*	
	(a) Perform Step 2.9.1.1.	

3.2 (b) Follow procedure outlined in paragraph 2.6. **Repeat all steps of paragraph** 2.9.1.3 (c) 3.2.1

2.10 **48VDC** Alarm

Verify and apply 48VDC input to sys-tems requiring external 48VDC. The 48VDC alarm lamp should go "out" when the external 48VDC source is applied.

- 2.11 **3VDC** Alarm Adjustments (See Fig. 5-2)
- 2.11.1 +3VDC Adjustment
 - Adjust +3VDC output to 4.0VDC. (a) Alarm lamp(s) should turn 'on." Adjust +3VDC output to exactly +3.0VDC. Alarm lamps should (b) go -out.*
 - (c) Adjust +3VDC output to +2.0VDC
 - Alarm lamp should turn "on." Readjust +3VDC output to exact-ly +3 0VDC. (d)

2.11.2 -3VDC Adjustments

- Adjust -3VDC output to 4.0VDC. (a) Alarm tamp(s) should turn "on."
- Adjust -3VDC output to exactly -3.0VDC. Alarm lamp(s) should go "out." (b)
- Adjust -3VDC output to -2 0VDC. Alarm lamps should turn "on." Readjust -3VDC output to exactly -3.0VDC. (c)
- (d)
- Repeat all steps of paragraphs 2.11.1 2.11.3 and 2.11.2 for all equipped cabinets.

2.12 AC Power On-Off Test

Place AC circuit breakers to OFF position. Wait one minute then place circuit break-ers to ON position. The PR relay should operate within three to five seconds and all alarm lamps on 316069 and 316301 PCB's should go 'out." Check PR relays in each equipped cabinet to insure they have operated.

3.0 COMMON CONTROL CIRCUITS TESTS

- 3.1 **Printed Circuit Board Insertion**
- 3.1.1 Insert the following PCB's:
 - Tone supply 316037()122 (a)

TM 11-5805-652-14

- **(b)** Originate control and lamp generator 316029()122
- System allotter 31603()122 No. 1 register 316302 (Board 1 (c) (d)
- and Board 2) No. 1 junctor 316024()122 (all (e)
- cabinets
- Line PCB 316040- correspond-(**f**) ing to assigned attendant's line number.

Tones Supply Circuit (363037) Adjustment Test

- Adjustment Using Oscilloscope
 - Connect probe to TP on regular (a) tones supply front.
 - Adjust top potentiometer for 600 (b) Hz (1.66 millisecond). Adjust bottom potentiometer for
 - (c) 15V peak-to -peak signal.
 - Repeat the above procedure for standby PCB. (d)
- Adjustment Using Test Phones
 - Insert TJP into test circuit test (a) iack.
 - Place bay switch to bay 1, junc-**(b)** tor switch to junctor 1.
 - (c) Go off-hook - dial tone should be heard. Junctor 1, bay 1 will be seized.
 - Connect another test phone to the (**d**)
- output of an audio signal generator such as Northeast Electric Company Model 15B

Test Set.

- Set signal generator to 600 Hx (e) and set output to a desirable level.
- Listen to both phones and com-(f)
- pare signal levels. Adjust tones

supply top potentiometer to obtain Identical signal levels.-

- Remove regular tones supply (g)
- then repeat Steps (a) through (f) above for standby tone supply. (h) Connect 20,000 ohm per volt
- VOM to tones supply TP (front and middle of PCB). Adjust bottom potentiometer for 5VAC rms.
 - Reinsert regular tones supply (i) PCB. Repeat Step (h) for regu-

lar tones supply.

3.2.2

- **Routine Test** 3.3
- 3.3.1 Depress reset button and hold for five seconds to extinguish all alarm lamps.

Depress regular routine key and hold key depressed for 20 to 30 seconds. 3.3.2 The routine key is located on the test panel right

front.

As key is depressed, bottom allotter alarm lamps (will come "on." All four alarm lamps (red) will come "on" within 20 to 30 seconds. Some alarm allotter alarm lamps (red)

lamps on the standby originate control may come	
sole alarm lamp will turn 'on' briefly and buzzer	
will sound.	

The regular tone supply alarm lamp (red) will come on within 20 to 30 seconds. The standby allotter alarm lamp should go "out" at this time.

D&press the reset button as in Step 3.3.1. All alarms should go 'out.'

3.3.3	Depress and hold the standby routine	
	key for 20 to 30 seconds. The alarm	
lamps (red)	on the standby allotter, originate control,	
and tone supp	ply PCB"s should turn "on."	

Depress the reset button as in Step 3.3.1 All alarm lamps should go "out."

- 3.4 Register(s) (3160302) Test
- 3.4.1 Insert all PCB's except:
 - COS PCB's 316043()122
 - Transfer juctor 316076()122 Camp-on-busy 316031()122 (b)
 - (c) (d)
 - Register 316302()122 UNA 316036()122 (e)
 - (f) ANA 316042()122

NOTE: Register 1 is already inserted Leave in place

3.4.2 Place junctor test switch to OFF Insert TJP into test jack

Connect a TVP to the following EPABX 3.4.3 lines T-R on the MDF and dial the line number it is attached to from the TJP.

X00, X01, X11, X12, X22, X33, X44, X55, X66, X77, X88, and X99

X corresponds to the hundreds digit of the line attached to

If a line PCB is not equipped, do not dial that number

Observe the TVP rings but do not establish transmission

- 3.4.4 Remove register 1 and insert register 2 Repeat all steps of paragraph 3 4 3.
- 3.4.5 Repeat Steps 3 4 3 and 3 4 4 for all equipped registers
- 3.4.6 Insert all equipped registers Into the system.
- 4.0 JUNCTOR (316024) Test
- Function 4.1

The junctor teat circuit enables the selection of any equipped junctor for test purposes The bay lamp will turn "on" when a busy junctor is selected.

Bay switch- position 1 Junctor select switch- as desired 4.2.2 To select junctors in bay 2: Bay switch -position 2 Bay 1 junctor select switch- positron 1 Bay 2 junctor select switch -as desired 4.2.3 To select junctors in bay 3: Bay switch -position 3 Bays 1 and 2 junctor select switchesposition 1 Bay 3 junctor select switch- as desired 4.2.4 To select junctors in bay 4: Bay switch-position 4 Bays 1, 2, and 3 junctor select switches- positron 1 Bay 4 junctor select switch- as desired 4.3 Junctor Test

Junctor Selection

To select junctor in bay 1:

4.2

4.2.1

- 4.3.1 Insert TJP into test Jack. Connect TVP to an equipped MDF vertical line T-R then perform the following steps:
 - Select junctor 1, bay 1 and place TJP "off hook" Dial tone should (a) be received
 - Dial any TVP number that is (b)

connected to the MDF at a line appearance. Junctor seize lamp will go "off" when dialing is completed

- Observe ring back tone, audible ring of MDF test phone. Answer (c)
- MDF test phone and establish conversation (d) Place VOM on positive DC volt-
- accurate reading of +1VDC (3VDC scale, if equipped). (e) Ground may be derived from any
 - convenient chassis point. Connect VOM to VO test point of (f)

ijunctor in use <u>Insure</u> that the test point is not shorted to ground or adjacent PCB

- tracks or components.
 - Adjust VO potentiometer (top (g) potentiometer on bottom) for +1VDC
 - (h) Connect VOM to VT test point of junctor in use
 - Insure that test point is not (i)
- shorted to ground or adjacent PCB tracks or components.
- - Adjust VT potentiometer (bottom (J) potentiometer) for +1VDC. Re-
- move VOM from test point.
 - Re-establish conversation and (k) insure transmission
 - Place TVP 'on hook." Remain (1)
 - "on hook" for approximately 30
- seconds Observe that TVP does NOT re-ring.

(m) The TJP should remain "dead." Go "off hook" with TVP and in-

sure transmission. (n) From attendant console, depress **attendant** override (busy station verification) button. Attendant override button lamp will come "on." Key punch number (depress DSS button on Bell consoles) of TVP. **Insure both** TVP and TJP receive burst of **dial** tone and that attendant can enter three-way transmission. Release attendant

- by depressing release bar (**button**). (0) Place both pho
 - Place both phones 'on hook." Place TJP "off hook." Dial 1
- and observe busy tone. Place TJP (on hook."
 - (p) Repeat Steps (a) through (o) above for all equipped junctors.
 - (q) Return bay switch to position 1.
 - Return all junctor switches to
- OFF. Remove TJP and TVP.

5.0 LINE CIRCUIT (316040) AND FEATURE(S) CIRCUIT TESTS

5.1 Preliminary

Determine equipped line appearance Using two TVP's, connect to-the lowest and highest MDF line appearances.

- 5.1.1 Lowest line appearance is station number from X00 upward.
- 5.1.2 Highest line appearance IS station number from X99 downward.

5.2 A check of the equippped features is accomplished by dialing the access code of the features from the lowest station number TVP Refer to Equipment Specifications for equipped features and access code(s).

- 5.2.1 Public Address 316047 Test (if equipped)
 - (a) **Dial assess code**
 - (b) Insure burst of ring tone (RT).
 - (c) Adjust VT Level for +1VDC.

NOTE: The PA feature will be rechecked during "Cut-Over Procedure" (Section VII)

5.2.2 Code Call 316048 Test (if equipped)

- (a) **Dial access code.**
- (b) Dial three digit code
 (c) Insure code is repeated via dial
- tone bursts.
- (d) Dial termination code from TVP connected to highest line appearance.
- (e) Establish transmission.
- (f) Adjust VT1 and VT2 for +1VDC.
- (g) Place both TVP's 'on hook''

NOTE: Code Call will be rechecked during "Cut-Over Procedure" (Section VII) .3 Attendant Trunks (U equipped)

Single Console System

- (a) The attendant has two trunk circuits:
 - 316073 (Trunk 1) 316099 (Trunk 2)
- (b) Dial access code for attendant. A **trunk will be seized. The**

attendant's trunk will be seized and the attendant's trunk lamp (located on **console**) will flash at 120 ipm **bright.** Answer call at the console' by depressing the lamp **button that is flashing.** The lamp will turn "on."

- (c) Establish transmission. Adjust the VT of the trunk seized for +1VDC
- (d) Depress release bar on console. Place TVP 'on hook."
- (e) Repeat Steps (b), (c), and (d) above and adjust level of other attendant's trunks.

Multrconsole Systems

- (a) Each console has two trunks.
- (b) Dial access code (as in Step (b) under Single Console System),

answer, and adjust each trunk's VT level for +1VDC.

- 5.3 Line-to-Line and Line-to-Features
- 5.3.1 Call fro-m TVP to TVP, both directions Insure ringing and transmission
 - (a) Dial an equipped feature (PA,

code call, or attendant trunk) Establish connection and transmission.

5.3.2 Work inwards (toward middle) and repeat Step 5 3 1 for all equipped lines.

Example (1):	200 to 299 299 to 200 200 to PA 299 to PA
Example (2).	201 to 208

Example (2):	201 to 298
• • • •	298 to 201
	201 to PA
	298 to PA

6.0 FEATURE CIRCUITS TEST

- 6.1 Test the following features if equipped.
 - 6.1.1 Executive Override 316052

(a) **Connect a TVP to the MDF line** vertical terminal assigned to

executive override. (b) Connect TVP's to two equipped

transmission between them.

TM 11 5004	5 650	14	
1111-3803	(c)	From the executive override TVP, dial one of the two TVP's.	
After receivin	g busy	y tone, dial the override access	
Cout.	(d)	Both TVP's should receive a burst of dial tone, and three-way	
transmission s	should (e)	be possible. Place all three TVP's "on hook." Disconnect from MDF.	NOTI may
6.1.2	Grouj	p Hunt 316045	6.1
	(a)	Connect TVP's to MDF line ap-	
hunt circuit N	o. 1.	pearances assigned to group	
	(b)	Connect a TVP to a non-group hunt MDF line appearance.	an ala
	(c)	Place TVP assigned to group hunt master "off book"	
	(d)	From non-group hunt TVP, dial group hunt master TVP number	will s
TVP Slave 1 s	hould	ring.	
	(e) A	Answer TVP slave 1 Establish transmission. Place both	will g
phones "on ho	ok." 1 (f)	Place TVP slave 1 "off hook." Repeat Step (d). Slave 2 should ring.	
	(g)	Repeat Steps (d), (e), and (f) for all slaves	alarm
	(h)	Repeat Steps (a) through (g) for all group hunt circuits equipped.	
6.1.3	Line	Lock-Out	
	(a)	Connect SIX TVP's to MDF line	will g
	(b)	Place all SIX TVP's "off hook" All six corresponding busy	6.1
lamps will co	me "o	n" Check all junctors with junctor	
and not huge	(t)	test switch to establish that they	
are not busy.	(d)	Busy tone should be received	and r
	(e)	Repeat Steps (a) through (d) for each line lock-out PCB equipped	pondi
6.1.4	Auxi	llary Junctor (316079)	
	(a)	Remove all junctors except No.	"out. Place
	(b)	Insert TJP into test Jack	1 lace
	(U) (J)	MDF line appearance.	remov
	(a)	TJP A busy tone should be re-	
ceived	(e)]	Dial any equipped feature pre-	

lish transmission. Remove TJP and TVP's. (e) E: SDA is normally a hotel/motel feature but be equipped in other systems. 1.6 System Main Alarm 316068 Major Alarm Depress regular routine key, (a) placing both system allotters in arm condition Console alarm light will come (**b**) "on" bright and console buzzer sound alarm Depress attendant reset key for (c) five seconds. All alarm lamps go "out" and buzzer will go "off." **Minor Alarm** Depress standby routine key, (a) placing standby allotter into an

Dial the SDA access code assigned to each line appear - - ance. Answer each SDA line. Estab-

(c)

(d)

- condition Console alarm lamp will come (b) "on" dim and buzzer does not sound. Depress attendant reset key for (c) five seconds All alarm lamps go "out" and buzzer will go "off." 1.7 Hotel/Motel Feature Tests Message Waiting (316057) Test Connect a telephone equipped with neon lamp to MDF line tip (a) ring designated as a message waiting station. At attendant position, place **(b)** (b) At attendant position, place message waiting switch corres-ing to station under test to the operate position. (c) The lamp on phone will "wink" at approximately 60 ipm. Place phone "off hook." Wink signal is removed and lamp goes (**d**) "Dial the attendant Establish transmission. phone "on hook" and "wink" will return
- (e) Place message waiting switch to OFF position. Wink signal is ved and lamp will go "out." Repeat Steps (a) through (e) for (\mathbf{f}) all message waiting stations Message Register (316080) Test
 - Connect a TVP to lowest MDF (a) line appearance equipped with message registration. Place TVP "off hook." Dial (b)
- trunk access code correspondequipped with message registraing to trunk group tion
 - (c) Observe peg count meter of station under test. Dial city

be accessible.

viously checked Feature should **Re:** move TJP and TVP. Insert all junctors

- (f) (g)
- 6.1.5 Single Digit Access (316086)
 - Connect TVP's to MDF line ap-(a) pearances assigned as SD.?, lines.
 - **(b)** Connect TJP to test jack.
- 5-8

number and let city phone ring but do not answer. Peg count meter should not count.

- Answer city phone. Peg count (d) meter should count. Place both
- phones "on hook." (e) Repeat Steps (a) through (d) for

all lines equipped with message registration, and all trunks equipped with message

registration.

NOTE: One call through each trunk is sufficient.

6.1.8 Traffic Recording Adapter Circuit (316046) Test

Line Circuits

Identify traffic recorging equip-(a) ment assigned to lines. Note

that equipment is showing idle.

(b) Connect a TVP to an \mathbf{MDF} line appearance corresponding to the first Line connected to the traffic recording adapter

circuit.

- Place phone 'off hook" for 30
- seconds. Note that traffic re-
- cording equipment shows busy for 30 seconds. Place phone on hook." Equipment shows idle.
- (d) Repeat Steps (b) and (c) for all lines connected to the traffic

recording equipment.

Junctor

(c)

- Refer to paragraph 4.0 of this (a) section.
- (b) Identify traffic recording equip-Note that equipment display indicates idle.

Repeat Steps (a) through (c) of paragraph 4.3.1 in this Section. (c)

- (**d**) Observe peg count meter counts
- as junctor is seized. Remain "off hook" for 30 sec-onds. Note that traffic record-(e)
- ing equipment shows busy for 30 seconds. Place phone 'on hook." Equipment shows Idle.
 - (**f**)
 - Repeat Steps (a) through (e) above for all junctors connected
- to traffic recording adapter. Remove all junctors from female
 - (g) connectors but leave in mounting

slots. Note that traffic recording equipment shows all circuits busy.

(h) Insert junctor No. 1 Note "all circuits busy" clears. Insert all junctors.

Registers.

Identify traffic recording equip-(a) ment assigned to registers. Note that equipment display indicates idle.

- Remove all register PCB's ex-cept register No 1 (two PCB's) Connect a TVP to an equipped (h)

 - (c) MDF line appearance and go

"off hook." Note register seizure (green lamp)

(d) Peg count meter counts and register is seized. Register remains seized for ten seconds. Equipment should indicate "all circuits busy" and duration of seizure.

- (e) Repeat Steps (a) through (d) for
- all equipped registers.
- Remove TJP. (Ú

NOTE: It is recommended that the following test (Attendant Control Test) be deferred until after the completion of the "Trunk/Tie Line Test," paragraph 9.0 of this Section.

Attendant Control

- Identify traffic recording equip-(a) ment associated with the attendant control.
- Connect a TVP to an equipped (b)

MDF has appearance. Place TVP "off hook" and dial the attendant's access code. Answer call at console then depress release bar. Peg count meter counts upon release. Place TVP's "on hook" and **remove** TJP.

Trunks/Tie Lines and Transfer Junctor

Identify traffic recording equip-(a) ment associated with trunk/tie

> Using appropriate trunk test procedure, place incoming or

lines and transfer junctor. Note that all displays are

(b) outgoing calls.

idle.

As each trunk is seized, note (c) that peg meter counts upon

trunk seizure and that the duration of seizure is displayed.

During the test, at appropriate (d) point, note that peg **meter** counts

upon seizure and that duration of seizure is displayed.

Meconnect all trunk/tie lines (e) and transfer junctor PCB's.

Note that equipment displays all equipment busy. The duration indication will remain idle.

Insert all trunk/tie lines and (f) transfer junctor PCB's.

7.0 TRUNK ALLOTTER (316034) TEST

7.1 Refer to Equipment Specifications for trunk/tie lines equipped, assigned access codes, and trunk allotter location.

- 7.2 A trunk allotter is associated only with two-way and one-way (out)
- Remove all trunk/tie lines, except the 7.3 first equipped trunk/tie line, associated with the trunk allotter under test
- 7.4 Note that the trunk tie/line allott lamp (yellow) is "on'

7.5 Observe that one of the four trunk group allott lamps (yellow) on the trunk allotter is "on."

trunk/tie lines

7.6 Dial the trunk/tie line access code.

- The trunk/tie line busy (green) (a) tamp turns "on."
- Truck allotter truck group (b) (yellow) lamp turns "off." pro-

ceeds to sent lamp, and turns "on."

7.7 Continue dialing the access code. Note that the trunk allotter yellow

lamps proceed to come "on" in sequence and that the alarm (red) lamp remains "off."

7.8 **Repeat Trunk Allotter Test (Steps 7.1** through 7.7) for all equipped trunk allotters.

8.0 CONSOLE OPERATION

The following is presented to aid the installer(s) in trunk and related circuit tests.

ITT Console 8.1

8.1.1 Incoming Trunk call (City Trunks)

8.1.1.1 The incoming call flashes the associated trunk button (120 ipm bright) sad sounds the console buzzer.

The Attendant (Attd) answers by de-8.1.1.2 pressing the flashing trunk button. The trunk button changes to steady bright, the buzzer turns "off," and the talk city (TC) lamp comes "on' steady bright.

8.1.1.3 The Attendant determines the station number desired and key punches the number on the console station number pad.

8.1.1.4 If the station is not busy, it will ring.

Upon answer, the TC lamp will go out and the talk entension (TX) lamp will turn "on" steady bright. The Attendant announces the call and, if call is accepted, depresses the release bar. The city trunk party and the station will be connected and the TE lamp will turn "off." The trunk button (TB) will turn 'on" dim steady.

The Attendant may depress the release 8.1.1.5 bar if the station is not busy and the

Attendant does not wish to announce the call. Ring back tone (RBT) will be transmitted to the city party. The TB lamp will flash at a dim 60 ipm. The TC lamp will go "Out." Upon answer by the station, the TB lamp will turn "on" dim steady.

To release from a mis-key, busy, or 8.1.1.6 ringing station (prior to depressing the release bar), depress the release extension key and re-key.

If the station is busy the Attendant 8.1.1.7 may

> Place the trunk in a hold condi-(a) tion by depressing the hold bar.

The trunk busy (TB) lamp will flash a wink signal (.9 seconds on, .1 second off). (b) Place the

Place the call on camp-on-busy (COB)

8.1.1.8 Camp-on-Busy Operation

After key punching the desired (a) number and receiving busy

depress the COB button. NOTE: Group hunt master numbers may not be camped on unless all associated slave out bers are

- Depress the release bar. The **(b)** trunk button lamp will flash 120 iom dim.
- (c) Two station numbers per circuit may be placed in the COB mode

at the same time. When the second number is camped on, the COB button will turn "on" bright steady and no more station numbers will be accepted.

- The COB circuit will attempt to (d) ring the called number every five seconds.
- The trunk button lamp will turn (e)

dim steady when the camped-on station has answered.

busy.

8.1.1.9 Attendant Recall

(a) The trunk_button_lamp will flack 120 ipm bright when the EPABX

station connected to the city trunk recalls the Attendant.

- Depress the trunk button. The (b)
- TB lamp will turn bright steady.
- The talk city lamp will not turn (c) "on.
 - The Attendant may spilt the (d)

trunk manually by depressing the talk extension or talk city button, or the Attend-

ant may participate in a three-way conversation. (e) If it is desirable to transfer the call, instruct the EPABX station to "hang up." &press the talk city button

and proceed as though it were an incoming call. (f) If the call had been lost and

there is a recall, depress the

TC button and proceed as though it were an incoming call.

Incoming Trunk Call (DID with 8.1.2 Attendant Access)

The associated trunk button lamp will 8.1.2.1 turn "on" bright steady when the trunk

is seized and will flash a dim 60 ipm indicating a call has been terminated. The trunk button lamp will turn dim steady indicating an answer.

If the Attendant is recalled, the call 8.1.2.2 may be treated as an incoming City

trunk call except:

camp-on-busy is not accessible.

8.1.3 outgoing Call

Any EPABX station may reach an out-8.1.3.1 (digit 0) and requesting a city trunk. This action defeats the class-of-service (COS) restriction feature when city number is dialed by the Attendant

The Attendant's trunk button (a) lamp (Op 1 or 2) will flash 120 inm bright and the console buzzer will sound. The Attendant will answer by depressing the trunk button. The trunk button lamp will turn bright steady.

The Attendant may, if desirable, (b)

depress an appropriate city

trunk button and release from the call by depressing the release bar.

The EPABX station will receive (c) city dial tone and may dial the

(d) The Attendant, if time and charges (T and C) are desired,

depresses the hold bar instead of the release bar. The trunk button lamp will flash wink upon completion of the call, indicating that the call has been completed and that T and C can now be obtained.

number directly

(e)

ing the number directly, using an appropriate trunk button. Upon answer, the Attendant may release from the call OF proceed as in Step (d) above.

(f)

If requested, the Attendant may book the call and place it at \mathbf{a} later time. The Attendant must release the station

connected to the Attendant's trunk.

8.1.3.2 Placing Outgoing Call from Console

The Attendant may seize an out-

going trunk by depressing an appropriate trunk button and dialing the desired number

(b) If a mid-dial occurs, the city may be released and reseized by &pressing the release city (RC) button.

After the called number has (c)

answered, the Attendant may

transfer it to an EPABX station in the same manner as an incoming call.

NOTE: The Attendant must not extend a booked call to an EPABX station until the distant party has answered.

8.1.3.3 An outgoing call from a station will turn the associated trunk button "on" dim steady.

NOTE: The Attendant must not place outgoing calls on hold until the called party has answered.

8.1.4 Attendant Line

(a)

8.1.4.1 The Attendant may originate calls to all EPABX stations and features by:

Depressing the attd. line button (a) (ALB). The ALB lamp will turn 'on" and dial tune will received.

- Using the rotary dial and dialing (b) the desired number(s).
- (c) The Attendant must depress release bar to release from call.

8.1.5 Attendant Override (Busy Verification)

8.1.5.1 The Attendant override feature may be employed by:

Depressing attendant override (a) button (AOB). The AOB iamp

will come "on" bright steady (h)

- Key punching the desired station number. The busy station will
- receive a burst of dial tone to announce the orertide. Depressing release bar to re-(c) lease.
 - 8.1.6 Unassigned Night Answer (UNA)

Upon departure from the console, the Attendant **turns** the UNA key ON. All incoming calls will be answered through the UNA circuit.

- 8.1.7 Assigned Night Answer (ANA)
- 8.1.7.1 Same as UNA

8.1.8 Attendant Controlled Conference

Any extension user can call the 8.1.8.1

Attendant via an attendant trunk to request a conference with a maximum of five other extensions or tie lines (can be mixed) and one trunk party.

> After acknowledging the confer-(a)ence request and obtaining the

list of conferees, operate the CONF key. The attendant trunk lamp will go dark and the conference lamp will light bright steady, indicating the conference circuit has been seized and the attendant and the extension have been transferred into the conference circuit.

- Disconnect from the conference (h) circuit by operating the release button
- To call an extension conferee, (c) operate attendant trunk #2

(will light steady) and key punch the desired station number

(d)

When the extension answers, advise him of the conference.

To add the extension, operate the conference key. Repeat the procedure in Steps (e)

(b), (c), and (d) for the other extension conferees.

To add an outgoing trunk* to the (f) conference, operate an idle out-going trunk key and dial the desired number.

When the outside party answers, (g) advise him of the conference; to

add the outside party, key punch number assigned as trunk conference accessing number (attendant transferred into conference to announce trunk party) and then the release button to disconnect from the conference (trunk party added to conference when attendant releases)

If an incoming trunk party re-(h)

quests a conference, obtain the list of the conferees and place the trunk on hold.

Repeat the procedure in Steps (i) (b), (c), and (d) for the other

extension conferees*

* At least one station must be in the conference before the trunk party is added.

If desirable, the Attendant may

place the station's call by dial-

(j) Re-enter the incoming trunk and add the incoming party by key punching the assigned number (see Step (g) above). To release the trunk from the (k) conference, station user in the

conference must dial the digit 2 or greater. If a conferee wants to contact the attendant he must go "on hook" and then dial "0." The attendant may then return him to the conference by operating the conference key.

8.1.9 Trunk Lamp Signals on ITT Console

TRUNK LAMP SIGNALS

CALL CONDITION	TYPE TRUNK	TRUNK LAMP
Incoming CO Trunk	2-Way Central Office	Flashes bright 120 ipm. Buzzer operates.
Trunk Answered (by attend- - 1	All Trunks	Bright steady.
Station Ringing	2-Way Central Office DID	Flashes dim 60 ipm.
Request for Transfer	%-Way Central Office DID DID	Rashes bright 120 ipm.
Camp-On Established	2-Way Central Office DID	Flashes dim 120 ipm.
Station Answers	2-Way Central Office DID	Dim steady.
Trunk on Hold	All Trunks	Dim Wink.
30 Second Delayed Answer	DID	Flashes bright 120 ipm. Buzzer operates.
Extension Dials "0"	Attendant Trunk	Flashes bright 120 ipm. Buzzer operates.

Bell Console 8.2

The operation of the Bell console is similar to the operation of the ITT console.

8.2.1 Substitute the following terms in the text of paragraph 8.0 (Console Operation) when referring to a Bell console

ITT Console	Bell Console
 (a) Key punch station	 (a) Depress station
number (b) Release bar (c) Hold bar (d) Attendant override	DSS number (b) Release button (c) Hold button (d) Busy verification

8.2.2 Since the Bell console contains a built-in busy lamp field (BLF), a busy station may be determined by referring to the DSS console.

8.3 Multi-Attendant Operation

- All incoming calls will be split (odd and even) between the two consoles 8.3.1
- All calls may be transferred from one 8.3.2

console to the other by turning the transfer key on the console from which the call is to be transferred.

9.0 TRUNK/TIE LINE TEST

9.1	Refer to Equipment Specifications for trunks/tie lines equipped and location.	
9.2	Test ail trunks/tie lines by using ap-	
Section VI	propriate test procedure(e) listed in	
9.3	Multi-Console Systems	
9.3.1	All trunks having a console appearance	
even basis.	will appeal at the console of all odd-	
9.3.2	Test each trunk with the consoles split Transfer all trunks to position	
1 and retest each trunk. Transfer all trunks to posi tion 3 and retest each trunk.		

10.0 TOLL RESTRICTOR (316067) TEST

Use the "Toll Restrictor Test" of Section VI as necessary with assoica-10.1 ted trunk test

11.0 ATTENDANCE DIGIT READ-OUT TEST

Complete an incoming call from a city 11.1 number to the Attendant.

11.2meet a TVP to MDF line T-B terminals representing X00 (lowest

equipped number is usually 200).

- Key punch X00. Insure ring but do not Text answer. Depress release extension key. Ringing should stop.
- 11.4 Repeat above step for X01, X11, X22, X33, X44, X55, X66, X77, X88, and X99. Repeat for each hundred group equipped.

NOTE: Do not test line appearances not equipped With PCB'S.

- 12.0 CAMP-ON-BUSY (316031) TEST
- Refer to "Camp-On-Busy Test" in Section Vi. Perform test if COB is 12.1 equipped.
- 13.0 ATTENDANT LINE TEST

NOTE: Attendant line will be inoperative whenever TJP is inserted.

Depress attendant line button on con-sole. Attendant line button turns "on" 13.1 bright steady. System dial tone is returned.

13.2 Using console rotary dial, dial the number of a TVP connected to the MDF line terminals.

- Insure ring back, answer, and estab-13.3lish conversation.
- Release by depressing release bar. Place TVP "on hook." 13.4
- 14.0 TRUNK CONFERENCE ACCESSING (316077) TEST
- Systems Arranged for Station Access 14.1
- Connect five TVP's to MDF line (a) terminals that are not restricted from the conference access code.
 - Dial access code from each TVP and establish five-way trans-(b)
- mission insure conference busy lamp on console is
- 'on" bright steady. Adjust each (total of five) VT (c)
- potentiometer for +1VDC. Use calibrated VOM for measurement
 - Complete call from city phone (d) via city trunk to attendant.
- Attendant key punches trunk conference accessing number and depresses release key
- Six-way transmission should (e)

 - now be possible Dial digit 2 from any one of the TVP'S. The trunk should re-(f)
- lease from the conference

TM 11-5805-652-14

Place one TVP "on hook." Go (g) "off hook" and dial the attendant.

Attendant answers, depresses conference button and release key. The TVP should now be connected back in conference.

Place all TVP's "on hook." Place one TVP 'off hook" and (h)

dial the attendant. Attendant places the TVP into conference as in Step (g).

(i)

(c)

Attendant depresses OP2 and key punches number of and idle TVP. The TVP rings and answers. Attendant places TVP into conference as in Step (g). Repeat for other three TVP's.

Place all TVP's "on hook."

- Systems Arranged for Attendant 14.2 Controlled Conference
 - Connect five TVP's to MDF line (a) terminals that are not restricted
- from the conference access code. (b) Place all TVP's 'on hook." Go "off hook" with one TVP and

dial the attendant. Attendant places the TVP in conference by depressing the conference button and release key.

Attendant &presses OP2 and

key punches number of an idle TVP. The TVP rings and answers. The attendant places TVP into conference by depressing conference button and release key

- (d) Repeat Step (c) for the remain-ing TVP's. Adjust each (total Of five) VT potentiometer for +1VDC. Use calibrated VOM for measurement.
 - Complete call from city phone (e)
 - via city trunk to attendant. At-

tendant key punches trunk conference accessing code and depresses release key.

- Six-way transmission should now be possible. (f)
- Dial the digit 2 from a TVP. (g) The trunk should release from

the conference.

16.1

15.0 RESTRICTEDSTATION -CLASS OFSERVICE TEST

15.1 Connect a TVP to MDF line appearance of a restricted station. Dial each restricted code of that station. Busy tone should be returned for each restricted digit(s) dialed.

> Repeat above procedure for all (a) restricted stations of the systern.

16.0 POWER FAILURE CUT-TROUGH TEST

ment.

- Connect an appropriate TVP to MDF line vertical terminals of PFCT.
 - (a) For city trunks: Ground start - TVP must have ground button.

Loop start - Normal instru-

	(b) For DID trunks:	16.4	DID Loop Truaks:		
	Loop - Normal instrument.		(a) Place outgoing call to city num-		
16.2	Place AC circuit breakers to OFF.		1464 ·		
16.3	For PFCT with city trunks:	16.5	Repeat Steps 16.3 and 16.4 for all PFCT numbers assigned.		
TVP rings initiate cal	 (a) Place in and out calls between TVP and city place. Insure that (Ground button must be depressed to Il on ground start trunks.) 	16.6	Restore AC power.		

5-14

Section VI

CONTENTS

				Page
1.0	City Tr	unk Test	(316061 and 316075)	6-1
	1.1	Outgoin	g -With Console Appearance	6-1
	1.2	Outgoin	g - Without Console Appearance	6-1
	1.3	Incomin	g (316075 only)	6-2
	1.4	Incomin	g (316061 only)	6-2
2.0	Tie Lir	ne Test (3	16051 and 316060)	6-2
3.0	Direct Inward Dial and Tie Trunk Test Procedure			
	3.1	Proced	ure	6-3
		3.1.1	Outgoing - With Console Appearance	6-3
		3.1.2	Outgoing - Without Console Appearance	6-4
		3.1.3	Incoming - With Console Appearance, Without Second	
			Dial Tone	6-4
		3.1.4	Incoming - With Second Dial Tone and With Console	
			Appearance	6-4
		3.1.5	Incoming - Without Console Appearance, Without Second	
			Dial Tone	6-5
		3.1.6	Incoming - With Second Dial Tone and Without Console	
			Appearance	6-5
4.0	Toll F	Restrictor	Test	6-5
5.0	Camp-	On-Busy	(316031) Test Procedure	6-6

FIGURES

6-1 Partial View of Toll Restrictor PCB	6-5
---	-----

TEST PROCEDURES

1.0 CITY TRUNK TEST (316061 and 316075)

In the following procedures, TVP refers to a test phone connected to the MDF vertical terminals of an equipped EPABX line. Insert the trunk/tie line under test only. Remove when test is completed.

Outgoing - With Console Appearance 1.1

Connect a ground start or loop

start (as appropriate) city line to the MDF tip and ring of the trunk to be tested. (b)

station line appearance that is unrestricted. Dial "0" and have Attendant extend the call to the desired trunk. After the Attendant releases, dial the city number. Answer, then adjust the "VT" potentiometer to +1VDC, then release. (c) Dial "0" and have the Attendant

extend the call to the desired trunk. The Attendant now dials the city number. Answer the city phone. The Attendant now depresses the hold bar on the console. A two-way conversation should be possible between the TVP and the city phone. Place both phones "on hook." The Attendant should receive a wink lamp indication on the assoiated trunk button. The Attendant should now enter the trunk and then push the release bar The trunk should clear in approximately six seconds. (đ)

The trunk allott lamp of the

desired trunk should light. Dial the trunk access code from the TVP. The allott lamp will go "out" and the busy lamp will come "on." The city dial tone will be heard. Dial the city number and answer Immediately dial zero; the digit will be ineffective. Watt at least ten seconds after dialing. Now dial "0" again and the Attendant's trunk lamp should flash a bright 120 ipm The Attendant now is able to talk to the city and the TVP The At-tendant can depress the talk city key and converse with the city and the TVP will be cut off. The Attend-ont can depress the talk vartancian key and converse ant can depress the talk extension key and converse with the TVP and the city will be cut off. The At-tendant now releases, and the TVP remains connected to the city

(e) If equipped, insert transfer junctor No 1 PCB Depress reset button to extinguish all alarms.

From the TVP, continue to talk with the city and dial digit "2" (f)

Nothing should happen to Interrupt the conversation

Dial digit "1" from the TVP.

The city party is cut off and the TVP receives the EPABX dial tone. Depress the TVP hook switch for a period not to exceed ten seconds, and release The city party should now be heard again

Dial digit "1" from the TVP. A (h)

dial tone will be heard Dial

another "1," and a busy tone will be heard. Dial another "1," and the TVP should now be able to con-

verse with the city again.

(i) Dial digit "1" from the TVP. A dial tone will be heard. Connect a second TVP to the MDF station line. Dial the numbers of the second TVP. When it rings, dial "1." The second TVP should drop out and conversation with the city should again be possible. (i)

Dial "1" from original TVP, then dial number of second

TVP. Answer the second TVP. The city party should not be able to hear. Only the TVP's should be able to hear one another. Dial "2' from original TVP. The city and both TVP's should be able to con-verse. Check and adjust the transfer junctor VT to +1VDC. Place the original TVP "on hook." The sec-ord TVP more requested by the city.

+1 VDC. Place the original TVP "on hook." The sec ond TVP may now converse with the city. (k) With the second TVP, dial "1." Then dial the number of the original TVP. Answer the original TVP. Dial "2" from second TVP. This brings the city party back tn. Now dial "1" from the second TVP and the original TVP should be cut off. Release the trunk. (l) Remove transfer junctor No. 1

Remove transfer junctor No. 1 (1) and insert transfer junctor No.

Repeat Steps (f) through (k). (m)

Repeat Steps (f) through (k) for ail transfer junctors accessible

by trunk under test (n)

Connect a TVP to the MDF line appearance that is restricted from accessing the trunk Dial trunk access code.

A busy tons should be received.

If the system is equipped with toll restrictors and if the trunk under test accesses the toll restrictor, perform the "Toll Restrictor Test" (paragraph 4.0 of this Sectton).

- Establish a call as described in Step (d) Do NOT dial "0" (0)
- Depress attendant override (p)

button on console. The associa-ted lamp will turn "on." Key punch the TVP number. Both the TVP and-city will hear a burst of dial tone. A three-way conversation should be possible. De-press the release bar. The attendant override lamp button will turn "off" Place TVP and city phone "on hook'

Outgoing - Without Console Appear-1.2 ance

Connect a TVP to MDF station

line appearance that is unre-stricted Connect a city line to the MDF tip and ring terminals of the trunk under test (b)

(a)

The trunk allott lamp (yellow) of the desired trunk will light.

Dial the trunk access digit. The trunk allott lamp should go "out," and the busy lamp (green) should turn "on." Dial the city number. Check for ring back tone, answer, and begin transmission.

(a)

Connect a TVP to an MDF

- Dial "0" from TVP. The At-(c) tendant's buzzer should not sound.
- Place TVP and city phones "on hook." The trunk should (d)
- clear in approximately six seconds. Řepeat Steps (e) through (p) of paragraph 1.1. (e)

Incoming (316075 Only) 1.3

All incoming city trunks have console appearance.

> Place city phone "off hook" and dial number assigned to the (a)

EPABX trunk to be seized. The associated trunk lamp button should flash 120 ipm, and the buzzer in the console will be heard. (b)

Depress attendant trunk button.

The lamp indication will be bright steady. The Attendant can now converse with city. Depress Attendant release bar. The trunk lamp will be dim steady.

(c) After approximately six seconds, the trunk should indicate recall. The lamp will flash bright at 120 ipm, and the buzzer will sound.

- (d) Attendant answers and key punches the TVP number. The TVP rings.
- Attendant depresses release ex-tension key and TVP drops out. Key punch TVP number again. (e)
- (f)

When TVP answers, talk city lamp goes "out" and talk extension lamp lights. The City should not hear the TVP or attendant.

- The Attendant then releases and (g)
- city and TVP can converse. Place TVP "on hook." The At-(h)
 - tendant is recalled in approxi-

mately six seconds as in Step (c).

Attendant answers and key punches the TVP station num-(i)

ber. Attendant releases by depressing the release bar. Lamp indication is a dim 60 ipm. (j)

If transfer junctor equipped, the

TVP station answers and im-mediately dials "1." A dial tone will be received Place TVP "on hook." Trunk should clear in approximately 16 seconds. (k)

Repeat Steps (a) and (d) above.

Place city phone "on hook." The TVP stops ringing, and trunk should clear in approximately six seconds.

(1) Repeat Step (a) above. Place city phone "on hook." The trunk button will go "out" and buzzer turns "off."

Depress trunk button of trunk (m)

under test and then the hold bar Place city phone "off hook," and dial the city number associated with the EPABX trunk under test. The city busy tone should be returned.

NOTE: If number dialed is city master rotary number, the bay tone may not be returned.

If camp-on-busy is equipped, place TVP "off hook" Insert camp-on-busy circuit No. 1.

(0) Answer trunk recall and key punch to busy TVP. The busy

- lamp on console will light and busy tone is returned. (p) Depress COB button on console.
 - Depress release bar. The trunk
- lamp indication is dim 120 ipm. (q) Place TVP "on hook." In seven seconds or less the TVP will

ring and trunk indication change to dim 60 ipm. Answer TVP and then place phone "on hook." (r) Remove COB circuit No. 1 and

insert COB circuit No. 2. De-

press reset button to extinguish all alarms.

- Repeat Steps o) through (r) (s)
- above. Place city phone "on book."
- (t) Turn key on console to UNA

position, if equipped.

- Initiate a call from the city phone. The relay in UNA will
 - pulse. Place TVP "off hook" and dial (v) the assigned UNA digit. The

trunk lamp will change from a bright 130 ipm to dim steady. Transmission to the city should be possible.

- If the system is equipped for MA rather than UNA, insert

the ANA printed circuit board.

NOTE: Connect TVP to EPABX station(s) assigned as ANA as required.

> Initiate a city call to the EPABX. (X)

The extension assigned to this trunk will ring. Answer and establish transmission. Place both phones "on hook."

- 1.4 Incoming (316061 Only)
 - Repeat Step (a), paragraph 1.3. Depress Attendant trunk button. The trunk lamp will be bright (h)

steady, and the Attendant can converse with city.

- Key punch the TVP number. The TVP will ring. Repeat Steps (e) through (g), (c)
- (d)
- paragraph 1.3. Place TVP and city phone "on hook." (e)
- (f)
- Repeat Steps (a) and (b) above, Key punch the TVP number. The TVP will ring. (g)
- Repeat Steps (e) through (g), (l), (h) and (k), paragraph 1.3 Answer the TVP. Place both city phone and TVP "on hook."
- (1)
- Repeat Steps (n) through (x), (j) paragraph 1.3.

2.0 TIE LINES TEST (316051 and 316060)

2.1 Connect tip and ring from distant equipment (DE) to tie line tip and ring terminals of MDF. Connect a TVP to equipped MDF line appearance.

2.1.1 From distant equipment, dial access code to seize EPABX tie line.

6-2

(a) The tie line busy lamp (green) will come "on." The phone con-

nected to the DE receives a dial tone from the EPABX.

- Dial the TVP number. Check (b) for ringing and ring back tones.
- Answer TVP and establish transmission. (c)
 - Adjust VT potentiometer for +1VDC.
 - (d) **Depress Attendant override** button on console. Key punch

the TVP number. The TVP and DE phones should receive a burst of dial tone, and three-way conversation should be established. Depress release bar on console. Attendant is released.

- Place both TVP and DE phones (e) "on hook." The tie line becomes idle
- 2.1.2 Check to make sure that the tie line (that is inserted and under test) is allotted. Allott lamp (yellow) will come "on."

(a) Dial the access code for the tie line under test. Note that the

tie line busy lamp turns "on," and dial tone is received on DE

- (b) Dial DE phone number. Note ring back tone. Answer and establish transmission.
- Place DE phone and TVP "on (c) hook " The tie line becomes idle.

3.0 DIRECT INWARD DIAL AND TIE TRUNK TEST PROCEDURE

3.1 Determine which trunk PCB number the EPABX is equipped with Refer to the following information for the correct test procedure:

Trunk	Perform Steps
316089	(a) through (n) of paragraph 3.1.1 and (a) through (o) of paragraph 3.1.3.
316056	(a) through (e) of paragraph 3 1 2, (a) through (c) of paragraph 3 1 5, (1) through (o) of paragraph 3 1 3
316054	(a) through (n) of paragraph 3 1 1 and (a) through (o) of paragraph 3 1 3
316090	(a) through (e) of paragraph 3 1 2, (a) and (b) of paragraph 3 1 6, (n) and (o) of paragraph 3 1 3
316091	(a) through (n) of paragraph 3 1 1, (a) through (d) of paragraph 3 1 4, (n) and (o) of paragraph 3.1 3
316092	(a) through (n) of paragraph 3 1.1, (a) through (d) of paragraph 3 1.4, (n) and (o) of paragraph 3.1.3.
316093	(a) through (e) of paragraph 3.1.2, (a) and (b) of paragraph 3.1.6, (n) and (o)

of paragraph 3 1.3

316094 (a) through (n) of paragraph 3.1.1 and (a) through (o) of paragraph 3.1.3.

- 316095 (a) through (e) of paragraph 3.1.2, (a) through (c) of paragraph 3.1.5, (1) through (o) of paragraph 3.1.3.
- 316096 (a) through (e) of paragraph 3.1.2, (a) and (b) of paragraph 3.1.6, (n) and (o) of paragraph 3.1.3.
 - 3.1.1 **Outgoing - With Console Appearance** (Attendant Access)
- Connect a city line to the MDF (a) tip and ring terminals of the trunk under test. Connect a TVP to an equipped

MDF line tip and ring that IS not restricted from the access code of trunk under test (b) Dial "0" with the station TVP

and have Attendant extend the

call to the desired trunk After Attendant releases, dial the city number Answer the city phone and adjust the "VT" potentiometer for +1VDC (c)

Dial "0" with the station TVP and have Attendant extend the

call to the desired trunk The Attendant now dials the city number Answer the city phone The Attendant now depresses the hold bar on the console. Conversation with the city should be possible Place both phones 'on hook." The Attendant should receive a wink lamp indication on the associated trunk button The Attendant should now enter the trunk and then depress the release bar The trunk should clear in approximately six seconds

The trunk allott lamp of the de-sired trunk should light Dial (d)

the trunk access digit from the TVP The allott lamp will go "out" and the busy light will light City dial tone will be heard Dial the city number and answer Immediately dial "0" The digit will be ineffective Walt at least ten seconds after dialing Now dial "0" and the Attendant's trunk lamp should Row dial '0' and the Attendant's trunk ramp should flash bright 120 ipm The Attendant now enters and is able to talk to the TVP and the city Depressing the Attendant's talk city key will cut off the TVP Depressing the Attendant's talk extension key cuts the city off and the Attendant converses with the TVP The Attendant releases.

If equipped, insert transfer junctor No 1 printed circuit (e) Depress reset button to extinguish all boards alarms

The TVP now talking to the city (\mathbf{f}) party dials a "2" Nothing

should happen to interrupt the conversation (g) From the TVP dial "1" The (g)

city party is cut off and the TVP receives a dial tone from the EPABX Depress the TVP hook switch for a period not to exceed ten seconds and release The city party should now be heard again

> Dial "1" from the TVP A dial (h) tone should be received Dial

another "1" and a busy tone will be heard Dial another "1" and transmission with the city should be possible

Connect VOM to VT test point.

Dial "l" from the TVP. A dial (1) tone should be received. Con-

nect a second TVP as in Step (a) above. **Dial second** TVP number. When second TVP rings, dial "1." Thee second TVP drops out and the city should be heard again.

Dial "1." Dial the second TVP (j) number. Answer the second

TVP. The city party should not be able to hear. Only the TVP's should be able to hear one another. Dial "2." The city and both TVP's should be able to converse. Check and **adapt** the transfer junctor VT potentiometer for +1VDC.

Place the original TVP "on hook." The second TVP should now be able to talk to the city.

> **Dial** "1" from the second TVP. (**k**)

and then the number of the original TVP. Answer the original TVP. Dial "2" from the second TVP and the city is again connected. Now dial "1" from the second TVP and the original TVP should be cut off. Release the trunk.

(1)Remove transfer junctor No. 1 and insert transfer junctor

No. 2. Repeat Steps (e) through (j) above.

Repeat Steps (e) through (h) for all transfer junctors accessible by trunk under test.

> (m) Connect a TVP to an MDF line

appearance that is restricted from accessing the trunk. Dial the trunk access code. A busy tone should be received.

If the system is equipped with toll restriction, and if the (n)

trunk under test accesses the toll restrictor, per-form the "Toll Restrictor Test," (paragraph 4 0).

3.1.2 Outgoing - Without Console Appear ance (Attendant Access)

> Connect a TVP to an MDF line (a)

appearance that is unrestricted. Connect a city line to the MDF tip and ring terminal of the trunk under test.

(b) The trunk allott lamp (yellow) of the desired trunk will light. Dial the trunk access digit from the TVP. The allott lamp should go "out" and the busy lamp (green) should turn "on." Dial the city number Check for ring back tone. Answer, and establish transmission. (c) Dial "0" from TVP. The Attend-

- ant's buzzer should not sound.
- (d) Place city phone and TVP "on hook." The trunk should clear in approximately six seconds
 - Repeat Steps (e) through (1) of paragraph 3.1.1 if transfer (e)

junctor is equipped.

3.1.3 **Incoming - With Console Appearance** (Attendant Access), Without Second **Dial Tone**

Connect a TVP to the MDF line appearance corresponding to the city number assigned to:

Access the trunk, and

King that line.

(b) Prom the city phone, dial the number assigned. The trunk

busy and register access lamps (green) will turn 'on." The register access lamp will go "out" upon completion of dialing. The console trunk lamp completion of diamag. The sec-button will flash dim 60 ipm. (c) Check for **ringing** and ring back

tones. Place city phone "on

- hook." Trunk should go idle. (d) Place TVP "off hook." Repeat
 - Steps (a) and (b) above. A busy tone will be sent to the (e) city phone. Place city phone
- **Trunk** will go idle. Remove all registers from and TVP 'on hook." (f)
 - system. Repeat Step (a) above. The trunk busy lamp will turn "on." (g)
- The register access will remain off. The EPABX

busy tone will be returned to the city phone. (h) Reinsert all registers. Depress

reset button to extinguish alarms. Place city phone 'on hook." Repeat Steps (a) and (b) above. Allow TVP to ring for 30 sec-(i)

onds. At the end of 30 seconds (approximately), the TVP will stop ringing. The Attendant console trunk

button will flash 120 ipm.

Answer the TVP. The trunk (i)

lamp will turn "on." Key punch the TVP number and release. The trunk button will

flash dim 60 ipm

(k) Answer the TVP. Establish transmission between TVP and

city phone Place both phones "on hook."

- If equipped, dial a number assigned to Intercept from the (1)city phone.
 - Check for ring back tone. (m)

Operator Intercept - The trunk button will flash 120 ipm The Attendant answers and both parties converse. The call may then be completed to a station in the normal manner.

is started information is transmitted to calling party

> Repeat Steps (a) and (b). Answer TVP and establish (n) transmission.

Depressing Attendant override (0)

button on console will turn lamp Key punch the TVP number. The "on" bright steady TVP and city party should hear a burst of dial tone. Establish a three-way conversation with Attendant. Depress release bar on console. The Attendant override lamp will go "out." Place TVP and city phone 'on hool

- Incoming With Second Dial Tone and 3.1.4 With Console Appearance (Attendant Access)
 - (a) Repeat Step (a) of paragraph 3.1.3
 - From the city phone, dial the (b) number assigned to access the

trunk. The trunk busy and register access lamps (green) will turn "on." A dial tone will be returned to the city phone. Dial the TVP number. The register access lamp will turn "off" upon completion of dialing, and the trunk button lamp will flash dim 60 ipm **during** ringing.

- (c) Repeat Steps (c) and (d) of paragraph 3.1.3 and Steps (a) and (b) of paragraph 3.1.4.
- (d) Repeat **Steps** (e) through (h) of paragraph 3.1.3.
- (e) Repeat Step (a) of paragraph 3.1.3 and Step (d) of paragraph 3.1.4
- 3.1.5 Incoming Without Console Appearance (Attendant Access), Without **Second Dial** Tone
 - (a) Repeat Step (a) of paragraph 3.1.3.
 - (b) From the city phone, dial the number assigned to th nk.

The trunk busy and register access lamp n) will turn **"on."** The register access lamp turn "off" **upon** completion of dialing.

- (c) Repeat Steps (c) through (m) of paragraph 3.1.3
- 3.1.6 Incoming With Second Dial Tone and Without Console Appearance (Attendant Access)
 - (a) Repeat Step (a) of paragraph 3.1.3 and Step (b) of paragraph

3 1 4. The trunk button lamp will not flash, as it is not equipped.

(b) Repeat Steps (c), (d), and (e) of paragraph 3.1.4

4.0 TOLL RESTRICTOR TEST

4.1 Toll restriction samples the digits dialed **to the** city, and disconnects those Calls from stations that are restricted from these digits If a station dials a restricted number, a system **busy** tone will be returned to that station

4.2	The tol	1 restrictor	contains	the	fol-
	lowing	circuits:			

1-316067(11)112	Monitor Board One
1-316067(12)122	Monitor Board Two with class of
	service, programable for ten
	restricted codes.
1-316067(16)	Monitor Board Two equipped to
· · ·	add an additional ten program-
	able codes.
1-316067(31)	Access Board, provides access
()	for three trunks .
1-316067(15)	Monitor Board Two without
. ,	class of service.

4.3 For systems equipped with one circuit, the following PCB's would be equipped:

1-316067(11)122 1-316067(12)122

1-316067(16)122	(if more than ten codes were to
1-316067(31)122	be restricted) For each group of three trunks to be accessed.

For additional circuits, the PCB's would remain the same except 316067(12) would be 316067(15).

A maximum of five circuits may be equipped in each system.

- 4.4 From the Equipment Specifications, determine the toll restrictor PCB locations.
- 4.5 **Insert** clip into busy **out** terminals on all 316067(11) toll restrictor PCB's. See Fig. 6-1.

CAUTION: Equipment may be damaged **due to inad**vertent **shorts.**



Fig. 6-1. Partial View of Toll Restrictor PCB.

4.6 Connect a TVP **to an MDF** line appearance that is not restricted from accessing the trunk(s) **under test but** is toll restricted.

- **4.7** Dial **trunk** access **code** from TVP. A system busy tone should be returned.
- **4.8** Remove **busy** clip from one toll restrictor circuit.

(a) Dial access **code** and seize trunk. The toll restrictor ac-

cess lamp (green) on 316067(31) associated with the trunk under test will turn "on" The monitor lamp (green) on 316067(12) or 316067(15) of circuit under test will turn "on."

Refer to Equipment Specifications for restricted code(s). Dial the first restricted code. The trunk should release from the city and the EPABX system **busy tone should be returned**.

- (b) **Repeat Step** (a) for each restricted code
- (c) Repeat Steps (a) and (b) for each trunk.

Reconnect clips to **busy out** terminals of circuit tested In paragraph 4.8.

4.9

Remove clip from busy out terminals of next circuit to **be tested** Repeat Steps (a), (**b**), and (c) of paragraph 4.8.

4.10 Repeat Step 4.9 for all equipped toll restrictor circuits.

5.0 CAMP-ON-BUSY TEST PROCEDURE (316031)

5.1 Insert three camp-on-busy PCB's in system. Depress the reset button to extinguish all alarm lamps.

- Using two city lines, establish (a)
- osing two city lines, establish an incoming call on two EPABX city trunks. Connect a TVP to line 200. Place TVP 'off hook," setting **up** a busy line. (b)
- Answer one of the incoming (c)

trunk calls and key punch line 200 The busy lamp will come "on" and a busy tone will be heard by the Attendant and city phone. (d) Depress the COB button. The trunk lamp will change from

bright steady to dim 120 ipm. Depress Attendant release bar

(e)

Answer the other Incoming trunk call. Key punch line 200 The busy lamp will light and a busy tone will be returned.

Depress the COB button. The (f)

120 ipm and COB lamp will light dim steady. This

(g) Place the TVP "on hook." Line 200 will ring in seven seconds

or less. The COB lamp will be extinguished, and the associated trunk will indicate dim 60 ipm.

Answer TVP on line 200. The trunk lamp will turn "on" dim (h) steady.

Place TVP 'on hook." In seven seconds or less, the remaining trunk lamp will change from dim 120 ipm to dim 60 ipm, and TVP on 200 will ring. Answer and go 'on hook." In six seconds both trunks will be indicating recall.

> Move the TVP to line 211 and (i) repeat Steps (a) through (h) above.

This procedure should be continued until all of the following numbers have been camped on: 200,211, 212,233,244,255,266,277,288,299.

If the system is over 100 lines, perform further tests to insure that camp-on is capable of accessing lines beginning with the other hundred digits used.

Section VII

CONTENTS

Page

1.0	Gener	al	7-1	
2.0	Pre-Cu	utover- Initial Installation	7-1	
	2.1	Customer Training	7-1	
	2.2	Procedure	7-1	
	2.3	Station Tests	7-1	
	2.4	Trunks/Tie Lines Test	7-1	
	2.5	Code Call -UNA- PA	7-1	
	2.6	Directory Check	7-1	
	2.7	Placing Equipment in Service	7-1	
3.0	Pre-Cu	Pre-Cutover (Telecommunication System in Current Operation)		
		3.1 Procedure	7-1	
4.0	Post-C	Cutover	7-1	
	4.1	Procedure	7-1	

Section VII

CUT-OVER PROCEDURES

1.0 General

This Section provides certain cut-over procedures. Although each cutover is different, these procedures should be followed as a guide.

2.0 PRE-CUTOVER-INITIAL IN-**SRALLATION**

2.1**Customer Training**

2.1.1All station users and Attendants should be properly trained before using the EPABX system.

2.1.2The service assistant should provide the training. If a service assistant is not available, schedule training at a time convenient to the equipment user.

- Refer to the Customer Training 2.1.3Manual.
- 2.2 Procedure

2.2.1Refer to Table 4-2, Section IV. Table 4-2 provides the MDF vertical number and the terminal numbers of each EPABX circuit with an output to station and satellite equipment.

- 2.2.2 Refer to Table 4-3, Section IV.
 - (a) Enter all information required on Table 4-3.
- 2.2.3 Using Table 4-3 as a guide, cable and connect station and satellite equipment
- 2.2.4Connect jumpers between EPABX circent outputs (Table 4-2), MDF vertical

terminals, and satellite equipment (Table 4-3). Enter these connections on Table 4-2 under "Connect to" column.

- 2.3 Station Tests
 - 2.3.1 Make a station-to-station call and check that:
 - Number assigned to station is (2) correct.
 - ((b)) Station is receiving a dial tone. (c) Station is receiving a ring back
 - 10mæ. (d)) Transmission is possible.
 - Audible ring and/or visual sig-(e) nal is present.

2.3.3Check each station to insure it cannot dial via restricted access codes without receiving a busy tone.

2.3.3

Check each station to insure that it can access those feature circuits not restricted

Trunks/Tie Lines Test 2.4

2.4.1Utilizing the appropriate test proce-dure (Section VI), check each trunk/ tie line equipped.

- Code Call UNA PA 2.5
- Make sure satellite equipment is functioning properly. Refer to Section 2.5.1V for test procedures.

- 2.6 Insure that Directory is correct and distributed
- 2.7 Place TE-400 equipment into service.

3.0 PRE-CUTOVER (TELECOMMUNICA-TION SYSTEM IN CURRENT OPERA-TION)

- 3.1 Procedure
 - (a)
 - (b)
 - Repeat Steps 2.1.1, 2.1.2, 2.1.3, 2.2.1, end 2.2.2. Remove all EPABX PCB's listed **under** "EPABX Equipment" in
- Table 4-2, Section IV, from female connectors.
- Leave PCB's in mounting slots. Repeat Steps 2.2.3 and 2.2.4. (c) (d) Remove power from present
 - system in service. Remove cabling from previous (e)
 - system to MDF. (f) Insert all PCB's into female
 - connectors on EPABX. Repeat tests in paragraphs 2.3, (g)
 - 2.4, and 2.5.

4.0 POST-CUTOVER

4.1 Procedure

4.1.1 Observe operation of TE-400 system for a minimum of one day. Make sure that Atlendant understands console operation, and that station users are able to utilize equipment.

4.1.2 Refer to Customer Service Manual for operational questions and/or comolaints.

4.1.3 Refer to Maintenance portion of manual (Sections IX through XIV) for troubleshooting procedures, (if necessary).

Section VIII

CONTENTS

Page

1.0	Preliminary	8-1
2.0	Power Removal from In-Use Cabinets	8-1
3.0	Section V Tests Reference	8-1
4.0	Section VII Test Reference	8-1
5.0	System Observation	8-1

Section VIII

SYSTEM EXPANSION PROCEDURES-

130097 TE-400 CABINET(S)

1.0 PRELIMINARY

1.1 Read Sections I through VII and refamiliarize yourself with this installation. Particular attention should be paid to Section III (Cabinet Familiarization)

1.2 Refer to Tables 2-3, 2-5, and 2-6 of Section II These tables were filled in upon initial installation, and should be amended to reflect this expansion.

1.3 Follow Section III, 1 1 in general. Do not disturb cabinet inn operation Align and level new cabinet(s) to level of m-use cabinet.

1.4 Perform on the <u>new cabinet only</u>: Section IV, paragraphs 1 0 and 2 0

1.5 Using paragraphs 2 2 and 2.3 of Section IV as a guide, connect new cabinet. (Tag, for identity, cables to m-use cabinet) Do not connect to m-use cabinet

1.6 Consult Equipment Specifications and customer Refer to Section IV, paragraph 5.0, Tables 4-2 and 4-3 Fill in as required for expansion.

 1.7
 For the new cabinet, do as required, Section IV, paragraphs 6 0, 7 0, 8.0,

 9.0, 10.0, 11.0, 12.0, 13 0, 14 0, 15 0, 16 0, and 17 0

NOTE: If lines/trunks, tie lines are to be distributed from m-use cabinet(s) to expansion cabinet(s), reflect these changes on Tables 2-5 and 2-6 of Section II and Table 4-3 of Section IV.

1.8 Familiarize yourself thoroughly with the old and new Equipment Specifications. ITT Dwg. 130117 will be referred to extensively

1.9 At this time the expansion cabinet should be wired for AC input, the

multicabinet power cables should be wired, EPABX circuit outputs/inputs (T-R-satellite) to the MDF wired

1.10 Remove all PCB's from new cabinet Leave in position

2.0 REMOVE POWER FROM IN-USE CABINET

- 2.1 Connect cables as in Section IV, paragraphs 2 2, 2 3, 3 0 (if required), and 4.0
- 2.2 Refer to and perform all Installation Notes listed on Equipment Specifications

CAUTION; Exercise <u>extreme</u> care to insure correct connections on punching block and strapping field Refer to Section III, paragraphs 3 9, 3.10, and 4.0, for familiarization

- 2.3 After wiring is complete, repeat all steps of paragraph 17 0, Section IV, on all cabinets
- 3.0 REPEAT ALL TEST AND ADJUSTMENTS OF SECTION V COMPLETELY. USE SECTION VI AS REQUIRED.
- 4.0 REPEAT ALL TESTS AND PROCEDURES OUTLINED IN PARAGRAPHS 1.0, 2.0, AND 4.0 OF SECTION VII.

NOTE; It may be <u>assumed</u> that the satellite equipment (instruments, etc) are in place and connected to the MDF to facilitate cutover and minimize down tume

5.0 IT IS RECOMMENDEDTHE INSTALLER REMAIN IN CLOSE PROXIMITY WITH THE SYSTEM FOR 72 HOURS AFTER IT IS REPLACED IN SERVICE.
Section IX

CONTENTS

		Page
1.0	General	9-1
2.0	Cabinet Assemblies	9-1

FIGURES

9-1	Typical TE-400 320136 Cabinet Assembly	9-2
9-2	Typical TE-400 320136 Cabinet Assembly	9-3

Section IX

DESCRIPTION OF CABINET

ASSEMBLIES 320136

2.2

1.0 GENERAL

1.1 The second generation cabinets were listed as 320136 assemblies. These assemblies are readily identifiable from the third generation 130097 assemblies by the swing-out door on the 320136.

1.2 Numerous cabling variations exist on the 320136 cabinets. It is imperative that the Equipment Specifications be studied carefully due to the numerous cabling variations

2.0 CABINET ASSEMBLIES 320136

2.1 Front views of two typical cabinets are shown in Figs. 9-1 and 9-2

ITT Dwg 329972 details all strapping patterns for TE-400 320136 assemblies.

2.2.1 EPABX circuit outputs/inputs are located on the punching field. Job strapping is done on the strapping field. See ITT Dwg. 329972

2.3 Refer to Equipment Specifications for further familiarization with 320136 assemblies.

NOTE: Refer to Section XI for interchangeability of PCB's between different cabinet assemblies.



Fig. 9-1. Typical TE-400 320136 Cabinet Assembly



Fig. 9-2. Typical TE-400 320136 Cabinet Assembly

Section X

CONTENTS

			Page
1.0	Genera	al	10-1
2.0	Index	of Assemblies	10-1
3.0	Assem	blies General Description	10-2
	3.1	Junctor	10-2
	3.2	Registers	10-2
	3.3	Consoles	10-2
	3.4	Originate Control and Lamp Generator	10-3
	3.5	Camp-On-Busy	10-3
	3.6	Trunk Allotters	10-3
	3.7	Attendant's Control	10-4
	3.8	Unassigned Night Answer	10-4
	3.9	Tone Supply	10-4
	3.10	System Allotter	10-5
	3.11	Line Circuit	10-5
	3.12	Tertiary- Quaternary	10-5
	3.13	Assigned Night Answer	10-6
	3.14	Class of Service	10-6
	3.15	Group Hunt	10-6
	3.16	Traffic Recorder	10-6
	3.17	Public Address	10-6
	3.18	Code Call	10-6
	3.19	Five Party "Meet-Me" Conference	10-7
	3.20	E&M Tie Trunk	10-7
	3.21	Executive Override	10-7
	3.22	Power Supply	10-7
	3.23	Loop Tie Trunk	10-7
	3.24	Loop DID Trunk	10-7
	3.25	Message Waiting	10-9
	3.26	Loop Tie Line	10-9
	3.27	Loop Start City Trunk	10-9
	3.28	Attendant Override	10-9

CONTENTS (Cont'd.)

3.29	Intercept	10-9
3.30	Primary-Secondary Matrix	10-10
3.31	Toll Restriction	10-10
3.32	System Main Alarm	10-11
3.33	Main Alarm Voltage Check	10-11
3.34	Voltage Check Reference	10-11
3.35	Attendant's Trunk and Allotter	10-11
3.36	Attendant's Trunk No. 1	10-11
3.37	Ground Start City Trunk	10-11
3.38	Transfer Junctor	10-12
3.39	Conference Trunk	10-12
3.40	Line Lockout	10-12
3.41	Auxiliary Junctor	10-12
3.42	Message Register Line Applique	10-12
3.43	Message Register Trunk Applique	10-12
3.44	Attendant Multiple Position	10-13
3.45	Single Digit Access	10-13
3.46	Loop DID Trunk	10-13
3.47	DID Loop Tie Trunk (316090)	10-13
3.48	DID Loop Tie Trunk (316091)	10-13
3.49	DID Loop Trunk (316092)	10-13
3.50	DID Loop Tie Trunk (316093)	10-13
3.51	Two-Way DID E&M Trunk	10-14
3.52	Two-Way E&M Loop Tie Trunk (316095)	10-14
3.53	Two-Way E&M Loop Tie Trunk (316096)	10-14
3.54	Attendant Trunk No. 2	10-14
3.55	The 24VDC Supply	10-14
3.56	Console Applique	10-14
3.57	Five-Volt Supply and Alarms	10-14

TABLES

Section X

PRINTED CIRCUIT BOARD

FUNCTIONAL DESCRIPTION

1.0 GENERAL 1.1 The following number scheme is used on ITT drawings for identification purposes 31xxxx()68 31xxxx()77 - Schematic (a) (b) - Narrative (c) 31xxxx()122 - Assembly 1.2 An assembly *may* consist of one or more than one PCB. Example; 316075(4)122 Ground Start City Trunk Assembly Board 1-316075(11)122 Board 2-316075(15)122 1.3 In addition to the above, the PCB's that an assembly is comprised of may be numbered in such a manner that no direct relationship is recognizable. 316061(4)122 Loop Start City Trunk Board 1-85860-1 Board 2-85861-4 Example: 1.4 Any PCB of any assembly is composed of two parts - the drill board and the component parts. 85862 - Drill board for PCB 85860-1 which is part of assembly Example: 316061(4)122 1.5 A higher () number is assigned to an assembly when the assembly is changed or modified. The PCB dash number (85861-4) is advanced also, and the drill board issue number may change. 1.6 The drill board number is etched in the full. Example: **885862** - Drill hoard The issue number will be stamped after the drill toard number. 85862-2 Example: The 85862-2 is the second drill board issued and replaces issue number 1 1.7 The PCB assembly issue number of any assembly of more than one PCF can change (advance) separately or at the same time The same also applies to drill boards • • • e

1.8	Each PCB is stamped Number.	with	an	Issue
Example:	P (B 858 60-1 ISS. 1R5			

1.9 A serial number is also stamped on the PCB to allow activity tracing of the PCB

1.10 The Issue Number bears very little relation to the actual number of different assemblies An Issue Number can change for reasons that have no bearing on circuit operation or function

2.0 INDEX OF ASSEMBLIES

Junctor	316024
Register	316025
Console	316028
Originate Control and Lamp Generator	316029
Camp-On-Busy	316031
Trunk Allotter	316034
Attendant's Control Circuits	316035
Unassigned Night Answer	316036
Tones Supply	316037
System Allotter	316039
Line Circuit	316040
Tertiary Ouaternary Matrix	316041
Assigned Night Answer	316042
Class of Service	316043
Group Hunt	316045
Traffic Recording Adapter	316046
Public Address	316047
Code Call	316048
Five Party (Meet Me) Conference	316049
E-M Tie Trunk	316051
Executive Override	316052
Power Supplies	316053
Loop Tie Trunk	316054
Loop DID Trunk	316956
Message Waiting	316057
Loop Tie Line	316060
Loop Start City Trunk	616061
Attendant Override	316062
Intercept	316063
Primary-Secondary Matrix	316065
Toll Restrictor	316067
System Main Alarm	316068
Main Alarm Voltage Check	316069
Voltage Check Reference Supply	316070
Attendant's Trunk and Alloter	316073
Attendant's Trunk No. 2	316074
Ground Start City Trunk	316075
Transfer Junctor	31676
Conference Trunk Accessing	316077
Line Lockout	316078
Auxiliary Junctor	316079
Message Register Line Annlique	316080
Message Register Trunk Annlique	316081
Attendant's Multiple Position	316062
Single Digit Line Access	316086
Loop DID Trunk	316089
DID Loon Tie Trunk	316090
DID Loop Tie Trunk	316091
DID Loop The Trunk	316092

DID Loop Tie Trunk	316093
E-M Trunk	318094
E-M Tie Trunk	31609
L-M Tie Trunk	316096
Attendant's Trunk No. 2	316099
24VDC Supply	316141
Console Applique	316184
Power Supply	316300
+5VDC Supply and Alarms	316301
Register	316302
• •	

3.0 ASSEMBLIES GENERAL DESCRIPTION

3.1 Junctor 316024()122, ()68 Dwg., Fig. 1

3.1.1 One PCB is required per circuit. A maximum of twelve may be equipped per 100 line cabinet

3.1.2 The junctor provides supervision and connection of all originating calls. On local calls it provides termination and voice paths. The junctor is released after all successful origination and terminations to feature circuits, and may be reseized immediately. On a failure to terminate, it is held to the originating line to provide busy tone.

3.1.3 One junctor (nominally No. 12) is reserved for the executive line on systems equipped with executive override.

3.1.4 Five outputs (DIW, DIX, DIY, D12, OVJ) are provided for the override circuits.

3.1.5 The seize lamp (green) will turn "on" when a junctor has been seized and is

in the process of originating or terminating a call. The lamp will go "out" after the call is terminated or atondoned.

3.2 Registers Assemblies (316025 and 316302)

3.2.1 Two PCB's are required per register circuit. A list of register assemblies (and their associated PCB's) employed by the TE-400 are listed below;

- (a) 316025(1)122 Assy ,()68 Dwg., Fig. 1, 5 316025(11) - Board 1 316025(12) - Board 2
- (b) 316025(2)122 .Assy. ()68 Dwg., Fig. 1, 3, 5
 316025(13) Board 1
 316025(12) Board 2
- (c) 316025(3)122 Assy,()68 Dwg., Fig. 1, 3, 5 316025(14) - Board 1 316025(12) - BoaFd 2
- (d) 316302(1)122 Assy,()68 Dwg., Fig. 1 130108-1 - Board 1 130109-1 - Board 2
- (e) 316302(2)122 Assy. ()68 Dwg., Fig. 1, 2
 130108-1 Board 1
 130109-1 Board 2

3.2.2	Digital	Store

	Local	Feature	Maximu Lines
316025(1)	3 digit	1 digit	100
316025(2)	2 digit	1 digit	200
316025(3)	3 or 4 digit	1 and/or 2	400
316302(1)	3 or 4 digit	1 and/or 2	+1000
316302(2)**	3 or 4 digit	1 and/or 2	**400

TE-400) has a maximum of 400 lines.

- * * 316302(2) has a 2-digit in-dial completion for DID trunks.
 - 3.2.3 Provision is made for insertion of three register circuits (six PCB's) per cabinet.

3.2.3.1 Access - Access to registers is provided via the local junctor circuit for local originating calls, and directly from the dial trunk or tie line on incoming calls. Access is possible to a maximum of five registers from any one junctor, tie line, trunk, or transfer junctor.

3.2.3.2 Dial Tone - Dial tone is extended from the tone supply to the junctor, trunk, or tie line via the register upon seizure. The local junctors are arranged to always extend the dial tone to the originating subscriber. The DID tie trunks are arranged to extend or withhold the dial tone on an optional basis (Section II, Table 2-3). The DID trunks are arranged to always withhold the dial tone.

3.2.4 Typical Numbering Plan, 316302(1)122

Cabinet	Hundreds	Feature Digits	
1	2xx		
2	3xx		
3	4xx		
All		0 - Operator	
All		6 - Public Address	
All		7 - Tie Line	
All		81 - Features	
All		9 - City Trunk	

3.2.5 A seize lamp (green) will turn "on" when the register is accessed.

3.2.6 Failure of the register will cause a

minor alarm to appear on the Attendant's console. The alarm lamp (red) on the register board 2 will also turn "on."

3.3 Consoles(316028)

- 3.3.1 Two ITT Attendant Consoles are available.
 - (a) Attendant Console 316028()122
 - Assy., () 68 Dwg., Figs. 1 and 2
 (b) Attendant Console 316028(2)122
 Assy., () 68 Dwg., Figs. 1 and 3

- 3.3.2 The operation and functions of the consoles are identical with the following exceptions:
 - (a) 316028(1) employs 24 trunk keys and is equipped with 100-line busy lamp field
 - (b) 316028(2) employs 42 trunk keys and a busy lamp field is not equipped

3.3.3 The console IS arranged to serve a

PABX having up to 42 trunk circuits All functions are performed by push-button operations No circuit operation requires the use of more than a single push button. Lamps and push buttons for each circuit are combined, having the lamp mounted inside a transluscent push-button knob Attendant consoles of 24-trunk capacity are arranged with an integral busy lamp field consisting of 100 line busy lamps However when more than 24 trunks are equipped, additional trunk push buttons are added in place of the busy lamp field and the busy lamp field may be relocated to an adjacent busy lamp field console. This busy lamp console is optional

In addition to one combined push-button lamp for each trunk circuit, the Attendant's turret contains a single 'release" bar to be used for releasing the Attendant from any circuit, a single 'hold' bar used for putting any circuit in a hold condition, a buzzer cut-off button, a ten-key keyset for completing incoming trunk calls, attendant override, or placing a call into conference by key sending, an error button, and a turn key used for night answering services.

In addition to the above meentioned equipment, the console is equipped with a standard instrument dial of the K-500 type, with lucite finger wheel, for completing outgoing city trunk calls, and calls within the EPABX utilizing the Attendant line The console is equipped with jacks at both ends for the optional use of a headset or a handset.

The dimensions of the Attendant console are approximately 24" wide, 12" deep and 5" high The top forms a gently sloping surface on which all buttons, keys, dials, etc., are mounted.

The console is equipped with a flexible cable terminated at one end inside the Attendar.t console and at the other end with a connector. This cable is arranged to be connected by the type of cable normally used in key systems for the interconnection to the EPABX cabinet.

3.4 Originate Control and Lamp Generator Assemblies(316029)

- 3.4.1 **316029(1)122** Assy., () 68 Dwg , Figs 1 and 2
- 3.4.1.1 One PCB is required per circuit Two PCB's (regular and standby) are

squipped in the first cabinet of all systems

3.4.1.2 The lamp generator (Fig 1) provides lamp supervisory signals of 120 ipm, 10 ipm, and "wink" (0.9 second off, 0.1 second on) The maximum load is:

42 trunks (with console appearance)

4 attendant trunks 2 attendant control circuits

2 attendant control circuits

3.4.1.3 The Originate Control (Fig. 2) provides an originate control pulse (OH) to the lines for origination control.

3.4.2 Originate Control 316029(2) Assy., ()68 Dwg, Fig. 2

3.4.2.1 One PCB is required per circuit. The 316029(2) IS identical to the 316029(1) except that the lamp generator (Fig. 1) is not equipped

3.4.2.2 The 316029(2) will be equipped, two per bay, in all bays except Bay 1, unless the system requirement exceeds the capacity of the 316029(1)

3.4.2.3 Both the 316029(1) and (2) are equipped as a regular and a standby, and if the regular should fall, transfer to the standby is automatic If either the regular or standby fall, a minor alarm will occur A failure of both will generate a major alarm

- 3.5 Camp-ON-Busy 316031(2)122 Assy.,() 68 Dwg., Figs. 1 and 2
 - 3.5.1 Three PCB's are required per two circuits
 - (a) 316031(11)122- Boards 1 and 3 (b) 316031(12)122 - Board 2
- 3.5.2 One set may be equipped per each attendant control circuit.

3.5.3 The COB furnishes the Attendant with the ability to place an incoming trunk call in ramp-on" to a busy extension The extension will ring within 5 (±2) seconds, once the extension IS idle

3.5.4 Two circuits are provided per assembly The COB lamp button will light (at Attendant's console) when both circuits are busy.

- 3.5.5 Refer to Section II. Table 2-3. for trunks capable of accessing this feature
- 3.5.6 Failure of either circuit will result an a minor alarm
- 3.6 Trunk Allotter Assemblies (316034)
- 3.6.1 Trunk Allotter 316034(1)122 Assy., ()68 Dwg., Fig 2
 - 3.6.1.1 One PCB is required per circuit

3.6.1.2 One circuit is required for each trunk group that is to be accessed arrectly by the subscriber

P

3.6.1.3 Capacity - Each trunk allotter has a capacity of four allott groups with a total of seven trunks in each group for a sum total allotter capacity of 28 trunks. A trunk allotter is required for each access code that represents multiple matrix appearances of individual circuits, i.e., trunks, tie lines, dictation trunks, etc

3.6.1.4 Operation - The trunk allotter assures the sequential selection and seizure of trunks within a trunk group by progressively advancing the trunk selection from one allott group to the next group each time a trunk is seized.

3.6.1.5 Class of Service - The trunk allotter recognizes the presence or absence of a class of service indication from the line circuit as having or not having the privilege of accessing the trunk.

3.6.1.6 Allott Group Busy - In the event that all trunks within the selected allott group are busy, the trunk allotter will allott a trunk at random from all allott groups

- 3.6.2 Trunk Allotter 316034(2)122 Assy., () 68 Dwg., Fig 2
- **3.6.2.1** The 316034(2) is identical to the 316034(1) except that:

(a) The 316034(2) has outputs for shop wiring to enable the class of service PCB to require or not require a class of service **diode to be** Installed for station restriction.

- 3.7 Attendant's Control 316035(1)122 Assy., ()68 Dwg., Figs. 3,4,5,6,7, and 8.
- **3.7.1 One** circuit consists of four PCB's:
 - (a) **316035(11)122 Board 1** (b) **316035(12)122 - Board 2**
 - (b) 316035(12)122 Board 2(c) 316035(13)122 - Board 3
 - (d) 316035(14)122 Board 4
- 3.7.2 One circuit is required **per** attendant console.
- **3.7.3** Failure of the attendant's control circuit will provide a minor alarm
- 3.7.4 Functions
 - (a) Enables the attendant console position to access any trunk,
- local extension, or attendant's trunk. (b) Enables the attendant to transfer an extension from attendant's trunk to an outgoing trunk or conference.

(c) Provides control of the lamps on the attendant's console to indi-

cale the condition of the trunks and attendant's trunks.

(d) Provides for sounding the buz-

tion when needed

zer to gain the attendant's atten-

- (e) Enables the attendant to extend call from a trunk to an extension
- 3.8 Unassigned Night Answer Assemblie (316036)
- 3.8.1 Unassigned Night Answer 316036(1)122 Assy, () 68 Dwg, Figs. 1 and 2
- 3.8.1.1 One PCB is required per circuit. Each can provide UNA service for a maximum of seven city trunks
- 3.8.2 Unassigned Night Answer 316036(2)122 Assy, () 68¹Dwg, Figs 1 and 2.

3.8.2.1 One PCB is required per circuit. Each can provide UNA service for a maximum of 14 city trunks.

- 3.8.3 Unassigned Night Answer 316036(4) and (5), 122 Assy., () 68 Dwg., Fig. 2
- 3.8.3.1 The 316036(4) extends the 316036(1) by seven trunks and shares common control equipment with the 316036(1).

3.8.3.2 The 316036(5) extends the 316036(2) by 14 trunks and shares common control with the 316036(2).

- 3.8.4 Unassigned Night Answer 316036(6)122 Assy., () 68 Dwg., Figs. 1, 2, and 3
 - 3.8.4.1 One PCB is required per circuit.
- Each circuit can furnish UNA service for six city trunks.
 - 3.8.5 Unassigned Night Answer 316036(7)122 Assy. () 68 Dwg., Fig. 1

3.8.5.1 The 316036(7) is identical to the 316036(6) except it services an additional six trunks and uses common control equipment located on the 316036(7).

- 3.8.6 Operation
- 3.8.6.1 The UNA circuits provide output(s) enabling audible/visual signaling (from a satellite device) of the presence of an in-

coming trunk call when the turret is unattended and the console UNA key is operated in the UNA position.

The call may be answered from any extension by dialing the UNA access digit(s) and, if equipped, the call can be transferred to another extension by dialing the digit 1. The system must be equipped with a transfer junctor 316076 assembly to make it possible to transfer a call from an extension.

- 3.9 Tone Supply 31c037(1) 122 Assy., () 68 Dwg., Figs. 1 and 2
 - 3.9.1 One PCB is required per circuit.

3.9.2 The Tone Supply circuit provides:

- (a) 600 Hz dial tone (DT)
- (b) 600 Hz 0 5 second on/O 6 second off busy tone (BT)
- (c) 600 Hz modulated at 12 ips

interrupted at a rate of **I.0 sec**ond on/3 0 seconds off ring tone (**RT**) (d) **RG1**, 2, 3, and 4 to the ring

generator for start/stop of ring

3.9.3 The amplitude and frequency of the dial tone are controlled by potentiometers. See Fig. 5-1 of Section V

3.9.4 Dial tone (DT), ring tone (RT), and busy tone (BT) circuits are fused by 0.5 amp, 8AG fuses located on the front of the PCB

3.9.5 One regular and one standby circuit

are equipped per system. Transfer from regular to standby is automatic in case of failure of the regular. If either regular or standby fails, a minor alarm will result. If both fail, a major alarm will be generated. The alarm lamp (red) located on the PCB will turn "on."

3.10 System Allotter Assemblies (316039)

- 3.10.1 System Allotter 316039()122 AD ()68 Dwg., Figs. 1 and 2
- 3.10.1.1 One PCB is rewired per circuit.
- 3.10.2 System Allotter 316039(2)122 Assy , ()68 Dwg.. Fig- 2.
- **3.10.2.1** One PCB is required per circuit

3.10.2.2 The 316039(5) is equipped with a 12stage ring counter. It will normally be equipped in Bay 2 of systems requiring a total regular interval time (RIT) count of 32 or less.

NOTE: The 316039(2) is available on the 130097 assembly a special engineering only.

- 3.10.3 System Allotter 316039(3)122 Assy., () 68 Dwg., Fig. 2
- 3.10.3.1 Cne PCB is required per circuit.

3.10.3.2 The 316039(3) is equipped with a 20stage ring counter. It will normally be equipped in Bays 2, 3, or 4 in systems requiring a total RIT count of 40, 60, 80, or less.

3.10.4 System Allotter 316039(1) Operation

3.10.4.1 The allotter supplies an RIT output fe each cycle of DT input. The RIT

pulses are used for system timing. Each RIT is 1.6 millisecond duration. A regular enable (REN) putput enables all circuits requiring an RIT input (SIT-SEN for standby allotter). The 316039(1) provides 20 RIT time slots.

3.10.4.2 In addition, the assigned register time slot will be extended to 3.2 millisec-

onds when terminating to a feature This extended time slot permits completion of four operations - drop-junctor (**DJ**), trap, termination, and drop-back-to-junctor (**DBJ**) in the event of a failure to terminate

3.10.4.3 One regular and one standby are equipped in the first cabinet of all systems Transfer from regular to standby, in the case of failure, will be automatic A minor alarm will be generated if either fails A major alarm will be generated if both fall

- 3.10.4.4 The alarm lamp (red) located on the front of the PCB will turn "on" in the event of a failure
- 3.10.4.5 The trap lamp will turn 'on" if the trap circuit becomes permanently marked This is a failure.
 - 3.10.5 System Allotters 316039(2) and 316039(3) Operation
 - 3.10.5.1 Both the **316039(2)** and (3) are equipped with regular and **standby**
- 3.10.5.2 Failure of the 316039(2) or (3) allotters will generate alarms/ transfers as the 316036(1). See paragraph 3.10.4.
- 3.11 Line Circuit 316040(1)122 and 316042(2)122 Assemblies, ()68 Dwg., Figs. 1, 2, and 33.
- 3.11.1 One PCB IS required for five line carcuits
- 3.11.2 The 316040(1) mounts inn odd line positions The 31604012) mounts in even positions

3.11.3 The lane circuit provides a trasition from the subscribers balanced loop to the unbalanced (single path) EPABX switching system

3.11.4 The line circuit recognizes supervisory signals (ringing. on-hook, offhook), and provides a transmission path. beusy lamp field (BLF) output. and lamp busy output

- 3.12 Tertiary Quaternary**TQ) Matrix** 316041(6)122 Assy., (**):68 Dwg.**, Figs. 1,2, and 3.
 - 3.12.1 Tertiary Quaternary (**TQ**) **Matrix** 316041(6)122 Assy _ () 68 Dwg., Figs, 1,2 and 3

3.12.1.1 The TQ matrix has 41 quaternary horizontal (QH) outlets Each outlets as connected to the VT (voice terminate) circuit. Each QH outlet is connected to the cathode of five quaternary diodus Each QH diode anode is connected to the cathode of five tertiary horizontal (TH) diodes Twenty-five TH inlets are provided to the system strapping field for system gracing to the 25 secondary horizontal (SH; outlets from the P-S matrix

6.12.1.2 Any terminated call will be connected via the VT circuit, QH diode, TH diode, SH diode, and PH diodes.

- **3.12.1.3** Grading connection is covered in the ITT Dwg. 329972.
- 3.12.2 Tertiary Quaternary (TQ) Matrix 316041(7) Assy., ()88 Dwg., Fig. 2
- 3.12.2.1 One PCB is required per circuit One circuit is equipped per cabinet.

3.12.2.2 Provision is made for the Insertion of one teritary/quaternary PC board.

The TQ matrix has 40 quaternary horizontal (QH) outlets. Each QH outlet is connected to a VT (voice terminate) circuit on a one to one basis. Each outlet (QH) is connected to the cathode of five quaternary diodes. Each anode of a QH diode is connected to the cathode of five tertiary horizontal (TH) diodes. Twenty-five TH inlets are provided to the system grading terminal for grading connectings to the 25 SH outlets from the P-S matrix.

3.12.2.3 The TQ matrix IS controlled, in that, upon termination attempt, the associated TH and QH horizontal/vertical cross-connects are applied bias voltage. This decreases fan-out and insures firing of the devices.

3.12.2.4 Any terminated call will be connected via the VT circuit, QH diode, TH diode, SH diode, and PH diode.

- 3.12.2.5 Failure of the 316041(7) will generate a minor alarm
- 3.12.2.6 Grading connections are covered in ITT Dwg 130117
- 3.13 Assigned Night Answer 31 **6042**(1) Assy., ()68 Dwg., Figs. 1 and 2.
- 3.13.1 **One** PCB is required per circuit

3.13.2 One ANA PCB is equipped to allow direct station ringing from a trunk for an incoming call when the console: is unattended.

3.13.3 Each PCB can service six trunks.

3.13.4 The maximum number of ANA circuits that can be assigned is equal to the number of junctors equipped.

3.14 Class of Service 316043(2)122 and (3) 122 Assemblies, (168 Dwg.

3.14.1 - Both assemblies re uire one PCB for each circuit. Each circuit can accommodate fifty lines.

- 3.14.2 Restriction of five (maximum) features is accomplished by equipping or not equipping a diode.
 - 3.14.3 On TE-400 320136 assemblies the class of service panel is located on

the left frontRestriction is accomplished as described in paragraph 3.14.2

- 3.15 Group Hunt 316054(1)122,()68 Dwg. Figs. 1 and 2.
- 3.15.1 One PCB is required for two circuits
- 3.15.2 Each circuit has one master line (directory number) assignment and

four slave line assignments. Hunting is sequential. The fourth slave may be assigned to 25 lines. Hunting will then be sequential over the first three slave lines and random hunting for the 25 lines assigned to the fourth slave.

3.15.3 Three PCB's may be equipped in each bay (130097 assemblies only).

3.15.4 On TE-400 320136 assemblies, the number of equipped 316045(1) PCB's may be determined via the Equipment Specifications

- 3.16 Traffic Recorder Adapter Assemblies (3160461
- 3.16.1 Traffic Recorder Adapter 316046(1)122 Assy., ()68 Dwg., Figs. 2A, B, C, D and E.

3.16.1.1 One PCB is required for each circuit Each circuit can furnish traffic information from three transfer **junctors**, ten lines, three, registers, and two attendant control circuits.

3.16.2 Traffic **Recorder Adapter 316046(2)122, ()68** Dwg., Figs. C, D, and F.

3.16.2.1 One PCB is required for each circuit Each circuit can provide traffic information from 12 junctors.

3.16.3 Traffic Recorder Adapter 316046(3)122 Assy., Fi**igs. 1, A, C,** and E

3.16.3.1 One PCB is required for each circuit. Each circuit can furnish traffic information from 12 trunks.

- 3.17 Public Address 316047(1)122 Assy., () 68 Dwg., Fig. 1
 - 3.17.1 One PCB is required for each circuit.

3.17.2 One circuit provides access for nonrestricted lines to an external public address system.

3.17.3 The output impedance is 300 or 600 ohms. An additional output (dry contact closure) is furnished for control of external equipment.

- 3.18 Code Call 316048(1) and 316048(2)122 Assy., () 68 Dwg., Figs. 1 and 2.
- 3.18.1 Two PCB's are required for one cir-

3.18.2 Non-restricted lines may access the code call by dialing d feature digit(s). By dualing one of 125 possible three-digit codes, uduble signaling will be provided. The party sig-nated may answer the call by dialing a feature

ligit(s)

- 3.19 Five Party "Meet Me" Conference 316049 ()122 Assy.,()168 Dwg., Figs. 1 and 2.
- 3.19.1 Two PCB's are required per circuit.
- 3.19.2 A maximum of five parties may dial the access code(s) and meet for a conference
- 3.19.3 The Attendant does not have access to the conference.
- 3.20 E&M Tie Trunk 316051()122
- 3.20.1 Refer to Table 2-3, Section II.
- 3.20.2 See paragraph 17.0 of Section XII.
- 3.21 **Executive Override Assemblies** (316052)
- **Executive Override** 316052(1)122 **Assy , ()**68 Dwg., **Figs.** 1 and 2 3.21.1
 - 3.21.1.1 One PCB is required per circuit.

3.21.1.2 The executive override permits the executive line to override busy extensions by dialing a feature digit(s) after receipt of busy tone. The override is announced by a short burst of dial tone. Three-way conversation is then

- 3.21.1.3 Use of the 316052 requires one junctor (nominally No. 12) to be dedicated to · executive override.
- 3.21.1.4 The 316052(1) may override:

mossible.

- 12 junctors, 12 trunks 'the lines 6.21) 10 june tors, 14 trunks/the lines (b)
- 3.21.2 Executive Override 316052(2)122 Assy . () 68 Dwg., Fig 2
- Executive override 316052(2) employs 3.21.2.1

the control circuit on the 316052(1) and is used to expand the number of junctors and trunks/we lones that may be overrudden by 12-12 or

Power Supply Assemblies 1316053 and 3.22 3163300

3.22.1 The EPABX power supply is dreaded men have modeline; the -18 volt supply, - 18.5 valit supply, -36 valit and -3 valit supply, and the witnesser siles 2. Bong village AR.

3.22.2 The operation of each suggety is identicall. They diller in run respects : the

amount of current that each IS rated to supply, and the output voltage and **polarity**.

3.22.3 The AC line voltage is stepped down to the prescribed full-wave bridge input voltage and is full-wave rectified. This rectified voltage is then filtered. After filtering, the voltage is fed to a regulator circuit which maintains a constant voltage at the output of the supply to within a $\pm 5\%$ tolerance from 0 to 100% rated load over the prescribed temperature range.

3.22.4 The supply is resistive against a short circuit by means of current limiting diodes.

3.22.5 The AC line voltage is stepped down to 23-26 volts (rms) for the 18-volt potentials and 42-45 volts (rms) for the 36-volt potentials.

This secondary voltage is full-wave rectified through a bridge rectifier circuit, and the negative output of the bridge is connected to the collectors of the power transistors.

The -3 volt and +3 volt potentials are derived from the -36 volt and +36 volt modules.

- 3.22.6 Table 10-1 lists the Power Supply Assemblies, Outputs, and Locations.
- 3.22.7 In multibay operation, only those supplies in Bay 1 have the control cir-

cuitry equipped. This control circuitry (see Section XII, paragraph 7.0) is connected to the supplies in surreeding bays to provide regulation control.

3.22.8 In multibay operation for TE~600 assemblies 320136 and 130097, a strap must be removed from the bay 1 supplies. See Section XI

- Loop Tie Trunk 316054(1)122 and 316024(4)122 Assemblies - Refer to Table 2-3. Section 11.
- 3.23.1 Two PCB's are required for one cir-Cuit
- The 316054(1) has resistors MRS, 6, 3.23.2 and 7 mounted on board 2 of the assembly
- 3.23.3 The 316054(4) has resistors mounted on TE-400 assembly (130097) punching field (PF) assembly

NOTE: See Section XII. paragraph 16:0

3.23

- Loop DID Trunk \$16056(1)122 and 3.24 316056(4)122 Assemblies - Refer to Table 2-3 of Section 11.
- 3.24.1Three PCB's are required for one CIFCUI
- 3.24.2 The \$16056(1) has remators MR5, 6, and 7 mounted on board 2 of the secondity

 Table 10-1

 Power Supply Assemblies- Outputs and Locations

Power Supply Assembly	Output Voltages	Cabinet Assembly	Bay(s)	
316053(1)122	-18VDC	320136 (internal)	1	
316053(2)122	-36VDC	320136 (internal)	1	
316053(3)122	+18.5VDC	320136 (internal)	1	
316053(4)122	+36VDC	320136 (internal)	1	
316053(5)122	-18VDC	320136 (internal)	2, 3, or 4	
316053(6)122	-36VDC	320136 (internal)	2, 3, or 4	
316053(7)122	+18.5VDC	320136 (internal)	2, 3, or 4	
316053(8)122	+36VDC	320136 (internal)	2, 3, or 4	
316053(9)122	-18VDC	85924 (auxiliary cabinet)	1	
316053(10)122	-36VDC	85924 (auxiliary cabinet)	1	
316053(11)122	+18.5VDC	85924 (auxiliary cabinet)	1	
316053(12)122	+36VDC	85924 (auxiliary cabinet)	1	
316053(13)122	-18VDC	85924 (auxiliary cabinet)	2, 3, or 4	
316053(14)122	- 36VDC	85924 (auxiliary cabinet)	2, 3, cr 4	
316053(15)122	+18.5VDC	85924 (auxiliary cabinet)	2, 3, 67 4	
316053(16)122	+36VDC	85924 (auxiliary cabinet)	2, 3, 67 4	
316300(1)122	-18VDC	130097	1	
316300(2)122	+18.5VDC	130097	L	
316309(3)122	-36, -3VDC	130097	L	
316300(4)122	+36, +3VDC	130097	L	
316300(5)122	- 18VDC	130097	2, 3, er 4	
316300(6)122	+18.5VDC	130097	2, 3, er 4	
316300(7)122	-36VDC, -3WDC	130097	2, 3, av 4	
316300(8)122	+367000, +37000	130097	2. J. er 4	

3.24.3 The 316056(4) has resistors MR5. 6, and 7 mounted on the TE-400 assembly 130097 PF assembly.

NOTE: See Section XII, paragraph 16.0

- 3.25 Message Waiting 316057(1)122 and 316057(2)122 Assemblies, () 68 Dwg., Figs 1 and 2.
- 3.25.1 One PCB is required for ten circuits

3.25.2 The 316057 supplies a "message waiting" signal to a station. The "message waiting" visual signal is a flashing lamp on the

station instrument.

3.25.3 The 316057(2) expands the message waiting capacity by ten lines per-

equipped PCB

- Loop Tie Line 316060(1)122 and 3.26 316060(4)122 Assemblies - Refer to Table 2-3 of Section 11.
- 3.26.1 One PCB is required per circuit
- 3.26.2 The 316060(1) has resistors MR5, 6, and 7 mounted on the PCB.
- 3.26.3 The \$16060(4) has the resistors mounted externally

NOTE. See Section XII, paragraph 17.0

- 3.27 Loop Start City Trunk 31606:1/122 and 316061(4)[22 Assemblies - Refer to Table 2-3 of Section 11
- 3.27.1 Two PCB's are required for each of influences
- 3.27.2 The 316061(1) is composed of:

ALC: NO. States 1 **6346**1 Bours 2 (Bernstein 1902 merenenting are mainted 24

3.27.3 The \$16061(4) as composed of:

> des de la com Westernet It CEREN & Browner 2

3.27-3.1 Предлания ТМС на таки нача стана на в inen utrafficien fri innenen friefige fårefige

15 C (1997) . Soor Section VIII genright 15 C

3.28 attentant Overrite Assemblies (Too 10.2)

- Automation Ocertifiate \$156062/UH22 3.28.1 Augurs (1) 68 Share Flans 1 2 and 9
 - 3.28.1.1 Cover 101 100 and investigation and games a core-and Cover wanter a company is month with

3.28.1.2 The 316062 is accessed from the Attendant's console via the attendant control and trunk 2 circuits The Attendant must key punch the desired number. A short burst of dual tone will announce the override.

3.28.1.3 Also equipped is the attendant's alarm reset and automatic reset. The automatic alarm reset exunguishes all alarms if AC power or DC power is cycled off/on by supplying a

- 3.28.1.4 The power failure cut-through relay is equipped on this PCB. See Section II,
- 3.28.2 Attendant Override 316062(2)122 Assy , () 68 Dwg , Fig. 1

reset pulse to all circuits with an alarm output.

3.28.2.1 The 316062(2) is used to extend the number of circuits that can be over-

radden by:

- 12 junctors and 12 trunks, or (a)10 junctors and 14 trunks (b)
- 3.28.2.2 The \$16062(2) is partially equipped utilizing common circulary on the 316062(1)
- 3.28.3 The 316062(1) and (2) assemblies may be equipped in cabinets 1 and 2. The

316062(1) will be equipped in cabinet 2 only when the multiple attendant's position is equipped

- Attendant Override \$16062(\$) and (4) 3.28.4 Assemblies
- The \$16062(\$) is identical to the (1) 3.28.4.1 Assombly and the \$10062(4) is identical to the (\$) Assendity, enough;

- tal Power failure out through relay IS ANT CONSIDER Matrix control test alarm is 100
- ednithhed

NOTE: See Section W

3.29 Intercent Assemblies (\$16063)

- Intercept 016060(1)102 Accy . 1 1 66 3 29 1 Dwg Page 1 and 2
 - 3.29.1.1 On PCB is required per circuit
 - 3.29.1.2 One circulat is required for first five (1) transie to be intercepted
- 3.29.2 Antercom 916069(21102 Amer 1) 96 Prop Pro 2
- The DIBOORD is more to compare term 3.29.2.1 with the StillOStill for each unbitconal tion transition to be intercepted
 - hoterverp: \$16063/41122 Anny (i 40 3.29.3 Prog. Pype 1 2 and 3

10 - 9

one PCB is required per circuit aid one circuit is required for first six 3.29.3.1 DID trunks to be intercepted.

Intercept 316063(5)122 Assy.,() 3.29.4 Dwg., Fig. 2

The 316063(5) is used with the 3.29.4.1 316035(4) for each additional six trunks to be intercepted.

The 316063 Intercept circuit permits the routing of DID calls directed to 3.29.5

vacant feature codes and unassigned or unequipped telephone numbers, to either a recorded announcement or to the console attendant.

3.30 Primary-Secondary Matrix Assemblies (316065)

- The primary-secondary (P-8) matrix assemblies are listed below. 3.30.1
 - 316065(3)122 Asey., () 68 Dwg., (2) Pig. 1
 - 316065(5)122 Assy., () 68 Dwg., **(b)**
 - Fig. 1 316065(7)122 Assy., () 68 Dwg., (c)
 - Figs. 1 and Y 316065(8)122 Assy., () 68 Dwg., (d) Figs. 1 and Y
- The operation of all P-S matrix 3.30.2 assemblies used is identical.
- 3.30.3 Operation

Each board provides a primary matrix consisting of 25 primary horizontal PH inlets. Each PH inlet (line-up appearances) is connected to the anode of five primary diodee. Each primary diode's cathode in connected to the anode of five secondary horizontal (SH) diodes, for a total of 25 SH outlets.

In a single cabinet with four PCB's, a total of 100 PH inlets and SH outlots is available. These SH outlets are pregraded to provide 15 outlets for local junctor originating points, and 25 for connecting to the tertiary quaternary matrix for terminating connections and system grading.

System grading provides proper cross connections. These PNPN diodee provide transmission paths.

3.31 Toll Restriction Assemblies (316067)

- 3.31.1 Tall Restriction is made up of the lubtowing PCB'st
 - (a) Monitor Board 1 316067(11)122 Masy., Fig. 1
 - Monitor Board 2 316067(12)122 (b) Assy., Figs. 1, 2, and 4
 - (ie.); Monitor Beard 2 316087(13)122
 - Assy., Figs. 1 and 4 Monitor Board 2 316067(10)122 (48) Assy., Fig. 4
 - Access Board 1 310007(13)122 (e) Agere

(f) Access Board 2 316067(14)122 Assy.

Access Board 2 316067(31)122 (g) Assy.

8.31.2 The toll restriction circuit interfaces with the trunk to detect dialed digits. It may be strapped to reject or pass digits.

- 3.31.3 AN individual station may be restricted from the toll restriction cult
- 3.31.4 The (11) and (12) assemblies check class of service and store digits for detection. The (12) assembly may be programmed to pass or reject a maximum of ten 1, 2, or 3 digit codes. Codes may not be mixed.
 - 3.31.5 The (16) assembly extends the maximum codes by ten each.
- 3.31.6 The (15) assembly is the (11) assembly without the class of service. A (15) assembly would be used in conjunction with an

(11) assembly, (i.e., second, third, fourth, or fifth circuits).

- 3.31.7 The (13) assembly is used to access four trunks.
- The (14) assembly is used to access 3.31.8 three trunks.
- The (31) assembly to used to access 3.31.9 three trunks.
- 3.31.10 The first circuit of a system with time trunks could be:

316067(11)132 316067(13)122 314047(31)133 314047(31)122 310007(31)122

The second circuit:

316067(11)132 316067(19)122

The third circuit:

310007(11)132 110007(12)132

3.31.11 **Onevation**

As a station accesses the trunk, the (31) secondly will gase the digits to the (11) and (11) assemblies of the circuit accessed. This access is random, in that a four-layer divide on the beard with the and enters one of the three title circuits.

If all circuits are busy, the station will be rejected and busy tone gamed to the aubacetions

If the circuits are free and the called number is out centricted, the call will proceed an annual.

If the called number is restricted, busy tone will be passed to the subscriber and the called dropped.

- 3.32 System Main Alarm Assemblies (3)6068)
- 3.32.1 System Mun Alarm 316068(1)122 Assy., () 68 Dwg., Figs. 1 and 2
 - 3.32.1.1 One PCB is required per circuit,

3.32.1.2 The system main alarm provides a major or minor alarm indication to

the attendant's console and equipment cabinet alarm lamps on receipt of logic inputs from the system.

3.32.1.3 A ground is supplied to the PPCT resistor (located on TE bus, 320138 assemblies without external power cabinet), when a major alarm is encountered.

- 3.32.2 System Main Alarm 316060(2)122 Assy., () 68 Dwg., Figs. 1, 2, and 4
- 3.32.2.1 Refer to paragraphs 3.32,1,1 and 3.32,1,2,
- 3.32.2.2 The \$16068(2) does not provide an output to PPCT.
- 3.32.2.3 Fig. 4 of () 68 Dug. applies if S16066(1) is required.
- 3.32.2.4 The \$14008(2) provides four outputs to external curvality:
 - (a) two to connele (b) two to the calumet alarm lamps
- 3.32.2.5 The 318006820 provides a uniner alarm in the overs of a failure of the SWEC or 40VEC supplies

NOTE: An opened as associable to convert the sTVDC allored from a monter to major allored.

3.33 Nilyin Alarm Vellage Check Assembilies (318004)

- 3.33.1 Blain Alberton Voltage Check 300000(1) U22 Austry ... (1) Gb Datg... Fag. 1
 - 3.33.1.1 One PCB is required for each scratt

3.33.1.2 The SUBDE compares the DC outputs of the power supplies and the SU070 willing clinck card and,

> (a) growthe a ground output on the Alfa do not the task 200 and

and the algorithm a state of the state of th

itst: promerous the ground from the approximation over

condie = 1899. This results in reminist of DC press Brow the system by drapping the press relay 1996;

3.33.2 Main Milarm Weilinge Cherik 310050(2) 1000 Marcy ., H 100 Dag ., Fig. 1

- 3.33.2.1 Refer to paragraph 3.33.1.8.
- 3.33.2.2 The 316069(2) is used on the 130097 cabinet assemblies only.
- 3.34 Voltage Check Reference Supply Assemblies (3)6070)
- 3.34.1 Voltage Check Reference Supply 316070(1)123 Assy., () 68 Dwg., Pige. 1, 2, and 3
- 3.34.1.1 One PCB is required per circuit. One circuit is required per system.
- 3.34.1.2 The 316070 provides -18, +18.5, -36, +36, -56, +56 VDC to the 316069 main

+36, +56, +56 VDC to the 316069 main alarm voltage check circuit for comparison with the system power supplies outputs. A reset output is also applied to the 316069 when the comparison of these voltages are within ±10%.

3.34.2 Voltage Chock Reference Supply \$10070(2)122 Assy., () 98 Dwg., Figs. 1 and 2

4

- 3.34.2.1 Operation of the \$16070(2) is the same as the operation of the \$16070(1) ex-DC output (Fig. 5) is deleted.
- 3.35 Attendent's Trunk and Allotter 316079(1)122 Assy., (1)68 Dwg., Figs. 1 and 2.
- 3.35.1 One circuit per PCD. One PCB per attentiant control
- 3.35.2 The trunk allotter allows rotation of attendant trunk calls between trunks 1 and 2.

3.35.3 The attendant's trunk allows an extension to dial the attendam. The attendar ant may service the trunk call by:

	 Supplying unformation. Transferring the extension to an idle and transferring
	 to only communication to hang to and dial an outside number
for the exten	sion. (d) Transfer the extension into a conference til equipped).
3.35.4	Pailure of the trunk will cause a minur alaria
3.36	Attendant's Trunk No.2, 316074(1122 Assy., + 168 Owg., Fig. 1.
3.34.1	The \$16074 will function the same or the \$16000 assembly except 41 <u>same</u> s
graph 9.54	is to the conference ctroutt. See pare -
3.37	Ground Staft City Trunk Assemblies (S16075)
3.37.1	Ground Stori City Trunk \$16075(1312) Anny

3.37.1.1 The 316075(1)132 assembly consists Of:

(a)	316075(11)123	-	Board	1
(b)	316075(12)122	-	Board	2

3.37.2 Ground Start City Trunk 316075(4)122 Assy,

3.37.2.1 The 316075(4) consists Of:

(a) 316075(11)122 - Board 1 (b) 316075(14)122 - Board 2

3.37.3 The 316075(1) has TR2 mounted internally on the 316075(12) assembly.

3.37.4 The 316075(4) has TR2 removed from 316075(14) assembly and TR2 mounts on the punching field.

NOTE: See Section XII, paragraphs 15.0 and 16.0

3.38 Transfer Junctor 316076(1)122 Assy., ()68 Dwg., Fig. 1.

- 3.38.1 **Two PCB's are required for each cir**curt.
 - (a) 316076(11)122 Board 1
 - (b) 316076(12)122 Board 2

3.38-2 One circuit may be equipped for each trunk sub-rack. See Table 2-3 of Section II to determine what trunks may access the transfer junctor.

NOTE: See Section XII, paragraphs 15.0 and 16.0

3.38.3 Operation

Any city or **DID** trunk call, whether incoming or outgoing, may be transferred from one extension or may be held while the PABX extension calls **another** extension for consultation. For this operation, the party on the **PABX** extension dials digit one. This causes the trunk circuit **to** be placed **in a hold condition and connects the extension to** a **register circuit** The extension party receives dial tone and places a call to another **extension in a** normal manner. After **talking** to **the other** extension, either party may be connected to the trunk, depending upon which extension goes "on hook." **Prior to either party going 'on hook" to effect transfer, the** dialing of a digit 2 by the origi **nal PABX extension** will connect the city party, for **ming a three-party** conference.

3.39 Conference Trunk Accessing 316077(1)122 Assy., () 68 Dwg., Figs. L and 2.

3.39.1 Two PCB's are required per circuit.

(a)	85854-1-	Board 1
(b)	85855-1-	Board 2

3.39.2 Strapping option permits:

(a)

Any unrestricted station to inter

the conference by dialing the access **code** and/or the attendant to place the station **into the conference.**

(b) Only the attendant to place a station into the conference.

3.39.3	The conference circuit may be pro-
	vided by inserting the two associated

PC boards.

3.39.3.1 The basic conference circuit (316049) permits a maximum of five local

stations to dial the assigned access code on a prearranged basis, thus establishing a "meet-me" type of conference. This circuit may be modified to provide access to the conference from the primary (line) side of the matrix, thus permitting the addition of a conferee located in a distant office (316077).

3.39.3.2 Attendant access to the 316077 confer' ence circuit is accomplished via a

conference key located on the console.

- 3.40 line Lockout 316078(1)122 Assy., ()68 Dwg., Fig. 1.
- **3.40.1** One PCB is required for SIX circuits.

3.40.2 A station permanently "off hook" for any reason will be seized by a lockout circuit when the register times-out and an associated permanent signal lamp IS caused to light. Busy tone will be passed to the station. The station will release from lockout when the "on hook" condition is receive*.

3.41 Auxiliary Junctor 316079(1)122 Assy.. ()68 Dwg., Figs. 1 and 2.

3.41.1 One PCB is required per circuit.

3.41.2 The auxiliary junctor reserves one junctor for feature calls. If N of N + 1 junctors **are busy, then any attempt to call local-to-local** will result in busy tone.

- 3.42 Message Register Line Applique 316080(1)122 Assy., ()68 Dwg., Fig. 1.
- 3.42.1 One PCB is required for 20 circuits.

3.42.2 This circuit in conjunction with the trunk applique circuit provides a 'peg count" output whenever a city Gill is completed from a station. See paragraph 3.43.

3.43 Message Register Trunk Applique Assemblies(316081)

3.43.1 The 316081 in conjunction with the line applique circuit (316080) provides a "per count" output whenever a city call is completed.

10-12

DID Loop Tie Trunk Assemblies

Refer to Table 2-3 of Section II

3.43.2	The 316081(1), (2), (3), and (5) assemblies provide service for seven trunks	NOTE: See Section X10, paragraph 16		
	each.	3.47	DID Loop Tie Trunk Asso (316090)	
3.43.3	The 316081(4) assembly provides service for five trunks	3.47.1	Refer to Table 2-3 of Sect	
3.44	Attendant Multiple Position Assemblies	3.47.2	Two PCB's are required cuit.	
3.44.1	Attendant multiple position board one 316082(11)122 assy., ()68 Dwg., Fig. 1	3.47.3	DID loop tie trunk 316090 (3) are composed of:	
3.44.2	Attendant multiple position board two 316082(12)122 assy., ()68 Dwg., Fig. 1	3.47.4	(a) 85796 (b) 85797-1, -2, -3 - E DID Loop Tie Trunk 3160	
3.44.3	Diode Box 130113-1 Assembly		composed of:	
3.44.4	Two PCB's and two diode box assemblies are required for one circuit.		(a) 85796 - (b) 85797 -4 -	
3.44.5	The 316082 provides multiple console operation. Each trunk will be split	3.47.5	Resistors MR5, 6, and 7 a on the punching field	
between cons basis. One a	sole 1 and console 2 on an odd-even attendant may gain full control by oper-	3.47.6	See Section XII, paragraph	
ating the atte	endant transfer key on the other console	3.48	DID Loop Tie Trunk 3160 Assy.	
with a 316082(13)122 assembly mounted on the termi-		3.48.1	Two PCB's are required f circuit.	
3.45	Single Digit Access 316086(1)122 Assy.		(a) -] (b) -]	
	()68 Fig., Fig. 1	3 48 2	The Board 2 has M	
3.45.1	One PCB IS required for six circuits.	5.40.2	7 mounted externally on th	
3.45.2	The 316086 permits single digit ac- cess to six individual lines.	3.48.3	Refer to Table 2-3 of Secti	
3.46	loop DID Trunk Assemblies	2 40	DID Lease Trank 21(002)1	
3.46.1	Refer to Table 2-3 of Section If	3.49	DID Loop Trunk 510092(1	
3.46.2	Three PCB's are required per circuit	3.49.1	Refer to Table 2-3, Section	
3.46.3	Loop DID Trunk 316089(1), (2), and (3) assemblies are composed of:	3.49.2	Two PCB's are required a cuit.	
	(a) 85793 - Board 1		$ \begin{array}{c} (a) & -1 \\ (b) & -1 \end{array} $	
3 16 1	(b) $85794-1$ 2, -5 - Board 2 (c) 85795 - Board 3	3.49.3	The board 2 has 1 7 mounted externally on th	
5.40.4	posed of:		field.	
	(a) 85793 - Board 1 (b) 85794-4 - Board 2 (c) 85795 - Board 3	3.50	DID Loop Tie Trunk 3160 Assy.	
3.46.5	Loop DID trunk 3/16099(1). (2). and (3)	3.50.1	Refer to Table 2-3, Section	
assemblues have MR5, 6, and 7 mounted internally on the 85794-1, -2, -3, Board 2.		3.50.2	Two PCB's are required	
3.46.6	The 316089(4) has resistors MR5. 6.		cuit	
	THE TAXABLE AND A WALVALL LIANS I VI			

The 316089(4) has resistors MR5, 6,
and 7 located externally on the punch-
ing field.

3.47.2	Two PCB's are requir cuit.	ed for each cir-
3.47.3	DID loop tie trunk 3160 (3) are composed of:	090(1), (2), and
	(a) 85796 (b) 85797-1, -2, -3	- Board 1 - Board 2
3.47.4	DID Loop Tie Trunk 3 composed of:	16090(4) is
	(a) 85796 (b) 85797 -4	- Board 1 - Board 2
3.47.5	Resistors MR5, 6, and on the punching field	7 are located
3.47.6	See Section XII, paragra	aph 16.0
48	DID Loop Tie Trunk 3 Assy.	16091(1)122
3.48.1	Two PCB's are require circuit.	ed for each
	(a) (b)	- Board 1 - Board 2
3.48.2	The Board 2 has 7 mounted externally or held.	MR5, 6, and the punching
3.48.3	Refer to Table 2-3 of S	ection II.
49	DID Loop Trunk 31609	92(1)122 Assy.
3.49.1	Refer to Table 2-3, Sec	tion II
3.49.2	Two PCB's are requir cuit.	ed for each cir-
	(a) (b)	- Board 1 - Board 2
3.49.3	The board 2 h 7 mounted externally or field.	as MR5, 6, and n the punching
50	DID Loop Tie Trunk 3 Assy.	16093(1)122
3.50.1	Refer to Table 2-3, Sec	tion II.
3.50.2	Two PCB's are require cuit	ed for each cir-
	(a) (b)	- Board 1 - Board 2

10-13

3.50.3	Resistors MR5, on Boar	6, and 7 are equipped d 2.		
NOTE: See Se	ection XII, paragr	aph 16.0.		
3.51	Two-Way DID E Assy.	&M Trunk 316094(1)		
3.51.1	Three PCB's arcircuit.	e required for each		
	 (a) 85793 (b) 85935-1 (c) 85795 	- Board 1 - Board 2 - Board 3		
3.52	Two-Way E&N 316095(1)122 As	A Tie Trunk sy.		
3.52.1	See Table 2-3, S	ection II.		
3.52.2	Two PCB's are cuit.	required for each cir-		
	(a) (b)	- Board 1 - Board 2		
3.52.3	The Bo 7 mounted extern field.	ard 2 has MR5, 6, and nally on the punching		
3.53	Two-Way E&M I	Loop Tie Trunk (316096)		
3.53.1	Refer to Section	II, Table 2-3		
3.53.2	Two PCB's are required for each circuit.			
	(a) Board 1 (b) Board 2			
3.53.3	The Board 2 has mounted externa field.	8 MR5, 6, and 7 ally on the punching		
3.54	Attendant Trun	k No. 2, 316099(1)122		

Assy., ()68 Dwg., Fig. 1. 3.54.1 One PCB IS required per circuit. One circuit for each attendant control.

It further provides: Access to the attendant's line. (a) Access to the attendant's over-(b) ride. By seizing OP2, the attendant may key a station into a confer-(c) ence (if equipped), or extend the call to a trunk. Failure of trunk No. 2 will generate a 3.54.4 minor alarm. The 24 VDC Supply 316141(1)122 Assy. 3.55 ()68 Dwg., Fig. 1. One PCB is required for two circuits. 3.55.1 3.55.2 The 316141 provides two +24VDC out-puts for operation of external equip-ment. The +24 is derived from the +36VDC system supply. 3.56 Console Applique 316184(1)122 Assy., ()68 Dwg., Fig. 1. 3.56.1 One PCB is required per circuit. 3.26.2

3.54.2

3.54.3

The 316099 provides all services of the 316073 assembly (see paragraph

3.35) except allotting.

One circuit is required for systems utilized by Bell companies. The console applique provides Interface between the attend-ant control circuits and consoles similar to Bell 757A, and associated direct station selection consoles:

- 3.57 **Five-Volt Supply and Alarms** 316301(1)122 Assy., ()68 Dwg., Figs. 1, 2, and 3.
- 3.57.1 One PCB is required per circuit. One circuit is required for each system

3.57.2 The five volt supply provides +5VDC for the integrated circuitry. In addi-tion, alarm circuitry IS provided for ±3VDC, -46VDC supplies.

Section XI

CONTENTS

		Page
1.0	General	11-1
2.0	Table 11-1 Notes	11-7
	TABLES	

11-1	Interchangeability of Printed Circuits	•	···· · · · · · · · · · · · · · · · · ·	11-1

FIGURES

11-1	Power Supply 316053 PCB	11-8
11-2	Partial View of Ground Start City Trunk 316075 Assembly	11-9

Section XI

PRINTED CIRCUIT BOARD

INTERCHANGEABILITY

1.0 'GENERAL

1.3 Both Drill Board Issue numbers and Assembly Issue numbers are listed

1.1 Table 11-1 shows the Interchange-ability of printed circuit boards between the 320136 cabinet assemblies and the 130097 third generation cabinets

- An 'X" under columns A, B or C of Table 11-1 indicates the following: 1.2
 - A Used in 320136 Cabinet assemblies only
 - B Used in both 320136 and 130097
 Cabinet assemblies
 C Used in 130097 Cabinet assem-
 - blies only

The Drill Board Assembly Issue numbers art? the lowest **issue** numbers 1.4

used Unless otherwise noted, all higher issue numbers may be used

1.5 Many boards are listed that have never been used This must be taken into consideration when using this table to determine which areas parts to stack which spare parts to stock.

Paragraph 2.0 defines the notes listed **in the Notes column**

Table 11-1

1.6

Interchangeability of Printed Circuits

Description	Drilî Board Issue	Assembly Issue	A	В	с	Notes
316024	3	16	x			
316024	3	17		<u>x</u>		
316025	all	all	X			
316029 (1)	4	10 R3		X		
316029 (2)	5	13		X		
316031 (11)	2	9R1		X		
316031 (12)	2	9R1	X			1
316031 (12)	3	10	X			
316031 (12)	4	12	X			
316031 (13)	2	9R1		X		
316031 (13)	3	10		X		
316031 (13)	3	12		X		
316034 (1)	4	15	X			
316034 (1)	6	17	X			
316034 (2)	1	20			X	
316035 (11)	2	17R7		X		
316035 (11)	3	24		X	-	
316035 (11)	4	26		X		
316035 (12)	4	17R7		X		2
316035 (12)	5	19		X		
316035 (12)	6	26		X		
316035 (13)	4	17R7		X		
316035 (13)	5	24		X		
316035 (13)	6	26		X		
316035 (14)	1	17R7		X		
316035 (14)	1	24		X		
316035 (15)	4	17R12			X	
316036 (1)	3	12 R3	X			
316036 (1)	4	15	X			

Table 11-1 (Cont'd)Interchangeability of Printed Circuits

	- <u>T</u>	7		T		
1	Drill					
	Board	Assembly		1		1
Description	Issue	Issue		R	l c	Notos
J					l č	notes
316036 (2)	3	12R3	X			
316036 (2)	3	15	X		+	1
316036 (3)	3	12R3	X	1	1	†
316036 (3)	4	15	X	<u> </u>	+	
316036 (4)	3	12R3	X	1	+	
316036 (4)	4	15	T Y	+	+	+
316036 (5)	3	1283	T T	+	+	+
316036 (5)	4	15	+	 	+	
316036 (6)		16	<u>+ ^ </u>	{		<u> </u>
316036 (7)		16	+	+	<u> </u>	
316036 (8)		10	+			
316037 (1)	811	811	+	1	<u> </u>	
216027 (1)		11R3	+	X	ļ	
316037 (1)		15		<u>X</u>	L	
	8	17		X	ļ	
316039 (1)	5	<u>18R2</u>		<u>x</u>		
316039 (1)	. 6	23		X		
316039 (2)	5	18 R2		X		3
316039 (2)	6	23		X		3
316039 (3)	5	18R2		X		
316039 (3)	6	23	1	x	1	
216040 (1)	+		<u>+</u>			
316040 (1)		<u>12R1</u>		X		
316040 (1)	2	16		X		
316040 (2)		<u> </u>		X		
316040 (2)	3	16		X		
316041 (1)	2	4R1	X			4
316041 (1)	2	5	X			4
316041 (2)	2	4R1	X			4
316041 (2)	2	5	X			4
316041 (3)	2	4R1	X			à
316041 (3)	2	5	X			A
316041 (4)	2	481	X			
316041 (4)		5				
316041 (5)	+				 	
316041 (6)		e e				
316041 (6)		ل 493	<u> </u>			
316041 (6)	+	981				4
316041 (0)	+		X			
316041 (7)	- <u> </u>	<u> </u>		ي و و ا	X	
<u></u>		6	J		X	
J16042	4	5R1		X		
316042	5	6		X		
316042	6	8		X		
316043 (3)	1	6			X	
316045	4	10R1		X	1	ana 2014 ang at a subscription of the
316045	6	13	and the second second second	X	1	an manager Carrow and Carlot and Station
316047	3	14R1	X	ALTON DE LOUIS AUTOR		New Contractor Charles
316047	3	14R2	-	X	+	
316047	4	16		X	+	and the second second second second
316048 (11)	1 1	1021	X			
316048 (11)	1 2 1	14	t v		+	
316048 (12)		1001	+	and a second second second second second second second second second second second second second second second	ф	
C-26020 (10)	adamenta da manarada	Shu E	dimminant de la construcción de			land the second s

Interchangeability of Printed Circuits

	Det 11					
	Board	Assembly				def
Description	Issue	Issue	A	8	C	Notes
316048 (12)	2	14		X		
316048 (13)	1	1084			X	
316048 (13)	3	15			X	
316049 (11)	1	1581	X			
316049 (11)	2	18	X			
316049 (12)	2	15R1	X			
316049 (12)	3	18	X			
316051	1	3		X		
316051	2	4		X		
316052	2	783		X .		
316052	3	10		X		
316053 (1)			X			5, 6
316053 (2)			X			5,6
316053 (3)			X			5, 6
316053 (4)			X			5,6
316053 (5)			X			6, 7
316053 (6)			X			6, 7
316053 (7)			X			6, 7
316053 (8)			X			6, 7
316053 (9)			x			7.8
316053 (10)			x			7.8
316053 (11)			x			7.8
316053 (12)			X	1	1	7.8
316053 (13)		<u></u>	x	<u> </u>		8.9
316053 (14)			x			8.9
316053 (15)			X			8.9
316053 (16)			X			8,9
316057	2	1181	1	X		
316057	2	3	1	X		
316060 (1)	1	181	x			
316060 (2)	1	181	X			
316060 (3)	1	1 81	X			
316060 (1)	1	182	x			
316060 (2)	1	182	X			
316060 (3)	1	182	X			
316060 (1)	1	3	X			
316060 (2)	1	3	X			
316060 (3)	1	3	X			
316060 (1)	2	4	X			
316060 (2)	2	4	X			
316060 (3)	2	4	X			
316060 (4)	1	3R1			X	
316060 (4)	2	4	1		X	
316061 (85860)	2	1	X			
316061 (85860)	2	1R2	1	X		
316061 (85860)	2	3	T	X		
316061 (25861)	2	1	X			
316061 (85861-4)	2	1R3	Í		X	
316061 (85861-4)	4	4	1		X	
316062 (1)	4	11	X			

Interchangeability of Printed Circuits

	1		-	1	1	
	Dr191					
	Board	Assembly				
Description	Issue	îseue	A	8	C	Notes
316062 (1)	6	2016			1 10	a an an an an an an an an an an an an an
316062 (1)	7	17	and an an an an an an an an an an an an an	and the second second second	W N	
316062 (1)	8	18	and the second second second second second second second second second second second second second second second	-	w w	
316062 (2)	6	81	t x		480 2	Condite Tritere-constant - Galattalager
316062 (2)	6	1106		1	8	and the second second second second second second second second second second second second second second second
316062 (2)	2	19				and the second second second second second second second second second second second second second second second
316062 (2)	8	18		-	w w	
316062 (3)	all	all			Y N	
316064 (4)	all	all	-			
316063 (1)	1 1	1	W N		100	
316063 (1)	2	- i	Y X	Contractor Construction (Sec		and the second second second second second second second second second second second second second second second
316063 (3)	3	1	Y Y	Construction of the local division of the lo		ana da matana ing kana ang kana ang kana ang kana ang kana ang kana ang kana ang kana ang kana ang kana ang ka
316063 (4)	1 1	1		h	- y	
316063 (5)		7			$\left \frac{2}{y} \right $	
316063 (6)	t i	The second second second second second second second second second second second second second second second se	1	and the second second second	y w	ana di Germani di La casa di La Casa di La Casa di La Casa di La Casa di La Casa di La Casa di La Casa di La C
315365 (1)	i i	7	×		494 -	an an an an an an an an an an an an an a
3,6065 (2)		nere and the second second second second second second second second second second second second second second		Contraction of Contra		
316065 (3)		Ŷ	t w			an an an an an an an an an an an an an a
	-					
310095 (4)	1	7	X			
	<u>)</u>	7	<u>*</u>			
316065 (6)		7	X			
316065 (7)	1	9		x		
316065 (8)	1	9		X		
316067 (11)	1	4R1		X		
316067 (11)		9		X		
316067 (12)	11	<u>4R1</u>		X		
316067 (12)	1	9	State State State	4		
316067 (15)		4R1	Set of The Second Second	X		
316067 (15)		9		X		
316067 (16)	1	4R1	A CONTRACTOR OF CONTRACTOR	X		
316//67 (16)		9		X		
316067 (13)		481	X		Į	
316067 (13)	3	9	X			
316067 (14)	1	481	X	Adam - Trace - Lange - Lange		
310097 (14)	<u> </u>	9	<u>λ</u>	Constant of the log of the log		
J16067 (31)		10			X	
316088	2	12R)	X	Contract of the second state	Alling the second statements	
316368	1	13	X			
316068	+ 4	14	<u> </u>			
J19008 (Z)	2	1282			X	
316068 (Z)	4	14			X	
<u></u>	2	13	<u> × </u>			
316069 (2)	2	14			X	
316070 (1)	<u>Z</u>	<u>5R1</u>	X			
316070 (1)	4	8	X			
316070 (2)	2	<u>681</u>		CONTRACTOR DE LA CONTRACTÓRIO DE LA CONTRACTÓRIO DE LA CONTRACTÓRIO DE LA CONTRACTÓRIO DE LA CONTRACTÓRIO DE LA	X	
316070 (2)	4	10		and the same statement of	X	
316073	2	9R1]
<u></u>	1 3	10	X			
316073	2	9R5	The second secon	X		

Interchangeability of Printed Circuits

Preservition	Drill Board Issue	Assembly		B	с	Notes
Description	15346	13500	-			
316073 .1)	3	14		<u> </u>	<u> </u>	
316074	1	<u>9R1</u>	X		<u> </u>	
316074	2	10	X			
316075 (11)	3	9R1	X	L	Ļ	9
316075 (11)	4	13	X		Ļ	9
316075 (11)	3	937	J	X	ļ	9
316075 (11)	4	16		X	<u> </u>	9
316075 (12)	6	9R1	X	I	ļ	9, 10
316075 (12)	9	15	X			9, 10
316075 (15)	7	9R8		L	X	9, 10
316075 (15)	9	20		L	X	9, 10
316076 (11)	1	9R1		X		
316076 (11)	1			X	÷	
316076 (12)	2	<u>9R1</u>	<u> </u>			ļ
310076 (12)	2	9R2		X		
316076 (12)	3	11		X		
316076 (12)	4	12		<u>x</u>	<u> </u>	
316077 (86854-1)	1	1R1	<u>x</u>			
316077 (86854-1)	1	1R2		X	<u> </u>	
316077 (85854-1)	2	3		X		
316077 (85855-1)	1	1R1	X			
316077 (85855-1)	2	1R2		X		
316077 (80855-1)	3	2		X		
316078	1	1R1	X			
316078	2	2	<u>X</u>			
316078	1	1R4		X		
316078	2	5		X		
316079 (1)	1	1R2		X		
316079 (1)	1	3		X		
316079 (1)	2	4		X		
316080	1	1R1		X		
316080	1	3		X		
316081	1	1R1	X			
316081	2	2	X			
316081 (3)	1	1R2	1		X	
316081 (3)	2	5			X	
316081 (4)	1	1R2			<u>X</u>	
316081 (4)	2	5			X	
316082 (11)	1	1R1		X		
316082 (11)	2	2		X		
316082 (12)	1	1R1		X		
316082 (12)	2	2		X		
316082 (13)	2	1R1	X			
316082 (13)	3	2	X			
316086	1	1R1		X		
316086	1	2		X		
316089 (85793)	2	1R1	Х			
316089 (85793)	3	4	X		1	
316089 (85793)	2	1R4	1	X	1	
316089 (85793)	3	ż	T	X		

8

Table 11-1	(Cont'd)
------------	----------

Interchangeability of Printed Circuits

	Drill			1	T	T
Recription	Board	Assembly				
acact i priton	Issue	Issue	A	В	С	Notes
316089 (85794-1)	2	2R1	X	1		+
(85794-2)	2	2R1	X		+	l
(85794-3)	2	2R1	X	†	+	+
(85794-1)	4	7	X	1	+	
(85794-2)	4	7	X	1	+	
(85794-3)	4	7	X		1	
316089 (85794-4)	2	225			X	
316089 (85794-4)	4	9		I	X	
316089 (85795)	1	2R1		X	1	
316069 (85795)	1	5		X		
316090 (85796)	2	<u>2R1</u>	X			
316090 (85796)	2	2R3	<u> </u>	X		
316050 (85795)	3	5		X		
		2R1	<u> </u>			
	3	2R1	X			
(03/9/-3)	3	2R1	X			
316090 (8579 1)	4	4	x		<u>├</u>	
(85797-2)	4	4	X		<u> </u>	
(85797-3)	4	4	X			
316090 (85797-4)	2	2R3			X	
316090 (85797-4)	4	6			X	
316094 (85793)	2	1R1	X			
316094 (85793)	2	1 R4		X		
316094 (85793)	3	4		X		
316094 (85935-1)	3	1R1		X		
316094 (85935-1)	3	1R1		X		
316094 (85935-1)	4	2		X		
316094 (85935-1)	4	2		X		
216094 (85795)		2R1		X		
316096 (85795)	1	5		<u>X</u>		
316096 (85796)		281	<u> </u>			
316096 (85796)	2	- 21(3		<u> </u>		
316096 (85799)	3	501		<u> </u>		
316096 (85799)		- JRI R	+	- <u>x</u>		
-						
-				1		
316099	3	181	- <u>x</u> +			
316099	7	187		X		
316099	8	7		x +		
316141	1	1		x		
316184	1	181	x			
316184	3	1R4			X	
316184	4	5			x	
316300 (1)	all	a 11			x	11
316300 (2)	all	al1			X	11
316300 (3)	all	all			X	11
316300 (4)	all	all			X	11
316300 (5)	all	all			X	12
316300 (6)	al1	all			X	12
316300 (7)	all	all			X	12

1

Interchangeability of Printed Circuits

Description	Drill Board Issue	Assembly Issue	A	B	С	Notes
316300 (8)	all	all			X	12
316301	1	1			X	
316302 (1)	1	1			X	
316302 (2)	1	1			X	
316046	1	1		X		

2.0 Table 11-1 Notes

NOTES

- The 316031(12) is equipped with one COB circuit. It is not provided with the 130097 TE -400 assembly
- 2 The 316035(12) issue 17R7 and below must have track from relay AL2-2 cut to operate with 316099(1)122 attenuant trunk No. 2. This track must not be cut for operation with the 316074(1) 122 attendant trunk No. 2
- The 316039(2) allotter will be provided on TE-400 assemblies 130097 as special engineering only
- 4. These assemblies are listed for reference only They were never equipped.
- 5 These assemblies of the 316053, when utilized for multicabinet operation, are used in cabinet 1 only A strap between H-G must be removed for multicabinet operation and the strap replaced for single cabinet. Refer to Fig. 11-1.
- 6. These assemblies of the 316053 are always utilized for cabinets 2, 3, or 4 of multicabinet

systems. Strap between H-G must be removed. (Note 5.) Refer to Fig. 11-1 (PCB will be partially equipped.)

- See Note 6. These supplies will be located in the auxiliary power cabinet, and are used for cabinet 1 supplies only Refer to Fig. 11-1.
- See Notes 6 and 7. These supplies will be located in auxiliary power cabinet, utilized for cabinets 2, 3, or 4 only. Refer to Fig 11-1
- 9 No issue lower than 9R6 may be utilized in systems equipped with a 31607(14) having a "piggyback" assembly.
- Equip strap "A-A" for local connectors which put battery on the tip ("T" lead) as soon as they are ring tripped. Equip strap "B-B" at all other times Refer to Fig. 11-2.
- 11 These assemblies are utilized on TE-400 assembly 130097. Remove straps as shown in Section III, Fig. 1, for multicabinet operation. These assemblies used on cabinet 1 only.
- 12 See Note 11. These assemblies used on cabinets 2, 3, or 4 only



Fig. 11-1 Power Supply 316053 PCB



Fig. 11-2 Partial View of Ground Start City Trunk 316075 Assy.

CONTENTS

		Page
1.0	General	12-1
2.0	Time Costrol Mark Generator	12-1
3.0	Origination	12-2
4.0	Termination (Station-to-Station)	12-3
5.0	Termination (Feature Call)	12-4
6.0	PNPN Matrix	12-5
7.0	DC Voltage Generation	12-7
8.0	Three-Volt DC Supplies	12-10
9.0	Attendant Override	12-10
10.0	Group Hunt	12-10
11.0	Single Digit Access	12-11
12.0	Auxiliary Junctor Applique	12-11
13.0	Executive Override	12-11
14.0	Trunk-Tie Line Seizure	12-12
15.0	Two-Way City Trunk	12-13
16.0	Two-Way DID Loop Trunk (Loop Signaling) and Tie Trunk (E&M Signaling)	12-19
17.0	Two-Way Tie Line	12-24
18.0	Console Operation	12-25

TABLES

12-1 Typical Trunk Group Assignment		12-13
-------------------------------------	--	-------

FIGURES

12-1	Block Diagram of Time Control Mark Generator	12-2
12-2	Call Originations from One Line Group (25 Lines)	12-2
12-3	Block Diagram Showing Station-to-Station Termination	12-3
12-4	Grading Block Diagram	12-4
12-5	Block Diagram of Feature Call Termination	12-5
12-6	Oscillograph Showing Feature Call Control Marks	12-6

CONTENTS (Cont'd)

12-7	Oscillograph Showing Failure of a Feature Call	12-6
12-8	PNPN Matrix Block Diagram	12-7
12-9	Originating Permanent Grading Cable and Terminal Strapping - 100 Lines	12-8
12-10	Block Diagram of DC Voltage Generation	12-9
12-11	Group Hunt Cercuit Block Diagram	12-11
12-12	Block Diagram of Trunk Allotter	12-12

Section XII

GENERAL OPERATIONAL DESCRIPTION

2.3

1.0 GENERAL

The operation of the TE-400 EPABX can be more readily understood if ITT Dwg. 316023 is referred to while studying this Section. It is suggested that Section X be read carefully, referring to figures supplied and the 316023 EPABX Block Diagram.

2.0 TIME CONTROL MARK GENERATOR

2.1The tone supply provides a 600 Hz dial tone (DT) to the system allotter. Refer to Fig. 12-1.

2.2 The allotter generates a 1.6 millisecond pulse for each cycle of the 600 Hz input. This pulse is called **a regular** interval time (RIT) or standby interval time (SIT).

(a) A 20- or 12-stage ring counter produces the RIT pulses. A single cabinet system is normally equipped with a

20-stage ring counter

(b) **The** ring counter will provide 20 sequential counts, and then repeat itself; thus RIT1 will occur and recur every 32 milliseconds.

- Each RIT (1 through 20) occurs (c) for a discrete (1.8 millisecond) time period, and then recurs every 32 milliseconds.
- An RIT pulse is assigned (one to (d) each circuit) to the following:

	, 0
RIT 1	Register 1
RIT 2	Attendant Control
RIT 3	Camp-On-Busy No. 1
RIT 4	Register 2
RIT 5	Junctor 1
RIT 6	Camp-On-Busy No. 2
RIT 7	Junctor 2
RIT 8	Register 3
RIT 9	Junctor 3
RIT 10	Transfer Junctor No. 1
RIT 11	Junctor 4
RIT 12	Transfer Junctor No. 2
RIT 13	Junctor 5
RIT 14	Junctor 6
RIT 15	Junctor 7
RIT 18	Junctor 8
RIT 17	Junctor 9
RIT 18	Junctor 10
RIT 19	Junctor 11

- **RIT 20** Junctor 12
- With the above assignment. (e)

register 1 will mark-for 1 6 millisecond and remark every 32 milliseconds The same is true for all other circuits

Each circuit, having an assigned (f) time slot, cannot switch except

during that time slot Only one circuit may switch at any given time

Under normal operation, the (g) regular allotter will provide a

regular enable (REN) sutput to the circuits requiring an RIT pulse. If the regular allotter fails, the RE will be removed and the standby allotter will provide a standby enable (SEN) output to the circuits, and switching from RIT's to SIT's occurs. A minor alarm will be provided on the console.

> The junctor will receive its RIT and REN input from the allotter.

The junctor will also receive (a) from its five accessible registers (three per cabinet). (b) With an RF, RIT, and REM in-

voice originate (VO) lead from ground to -18VDC for 1.6 millisceond, or one RIT pulse. (c)

As the VO marks, the junctor

will provide a signal originate mark (SOM) pulse to the originate control (316029).

2.4 The originate control will compare the time period of the SOM input with the cycle of DT that produced the RIT_SOM; and, if correct, will provide an originate horizontal (OH) pulse to the lines circuits (318040).

> In an idle system, each junctor (a) will be marking in sequence

providing an SOM pulse in sequence to the originate control. The originate control then provides an OH pulse in sequence to tie lines.

2.5For mulitcabinet systems, the allotter RIT count may be expanded by 12 or 20, depending on the needed time slots. The allotter will continue to count sequentially. The total time between a specific RIT may be determined by multiplying 1.66 by the number of ring-counter stages For example:

<u>RIT Count</u>	Time (milliseconds)
20	33
40	66
60	99
80	132
	<u>RIT Count</u> 20 40 60 80

2.6 For correct origination, a QH pulse must-sent on the line The origination is random in that the junctor will be seized that is marked at the time the "off hook" condition of the station instrument is recognized

2.7If a station, in a system with traffic, attempts origination during a time slot that is assigned to a busy junctor, the line must wait until the time slot from a free junctor occurs



Fig. 12-1. Block diagram of Time Control Mark Generator.

3.0 ORIGINATION

To originate a call the subscriber goes 'off hook." The line VO will 3.1 pulse towards +18VDC.

3.2 In each 100 liner, the lines are divided into four group of 25 lines each. Each group is connected to the primary matrix on a one-line to one-input basis. gee Fig. 12-2.

Each input has access to the anode of five primary diodes. (a) Each diode is connected to one group of five second-ary diodes. The outlets of the secondary matrix are 25 secondary horizontal (SH) leads.

(b) Four primary-secondary ma-trices (316065) are equipped per 100 lines. These outlets are graded to present 25 outlets to the tertiary matrix for terminating paths, and 15 to the junctor VO's for origination.

(c) The 15 junctor outputs are graded so that in a 100-line, 12-junctor system each may access one out of ten junstors.

3.3 As the lines VO **mark** toward +18VDC (Fig. 12-1), the junctors VO mark toward +18VDC. The associated diodes **fire and** connect the line VO to the junctor VO. After the diodes

have fired, a low impedance path exists. The junctor/line combination provides holding current for the matrix. The junctor seize lamp will be "on." A junctor may access one of five registers. The register seize lamp (green) will be "on."



Fig. 12-2. Call Originations from One Line Group (25 Lines).

3.4 A dial tone is provided to the registers from the tone supply. On ser-

zure, the DT is passed via the register seize lead (RS) to the junctor, the primary-secondary (P-S) matrix, and to the line. See Fig. 12-1.

(a) Dial pulses are passed via the line VO, P-S matrix, junctor, and to the register stores.

(b) The register will store the

digits. Upon completion of the dialing, the register will send (mark) out its stored digits on the terminate bus during the register's next assigned time slot (RIT pulse). A terminate pulse will be sent via the register senze (RS) lead to the junctor. The register and junctor senze lamps will go "out." The duration of the zegister mark for a local call is 1.6 millisecond (RIT).

4.0 TERMINATION (STATION-TO-STATION)

4.1 Upon receipt of the terminate pulse, the junctor will mark its voice terminate (VT) lead to -18VDC. Refer to Fig. 12-3. The hundreds (H), tens (T), and

units (U) output of the register is graded to the lines via the terminate bus. The 1.6 millisecond mark from the register will cause the addressed lines VO to rise toward +18VDC. (b) Each junctor VT is tied to the

(a)

Each junctor VT is tied to the tertiary-quaternary (T-Q) ma-

trix on a quaternary horizontal (QH). There are 40 QH outlets available for connection to all circuits that have a VT connection. Each VT output occupies one QH outlet. Thus, 12 junctors will have 12 QH outlets assigned.

NOTE: (T-Q) matrix 316041(6) used on most 320136 TE-400 cabinet assemblies has 41 QH inlets.

(c) The 40 QH outlets are graded in the (T-Q) matrix **for 25** tertiary

horizontal (TH) inlets. These inlets are graded and connected to the 25 SH outlets from the P-S matirx.

(d) With the he VO at +18VDC and the junctor VT at -18VDC, the

primary, secondary. tertiary, and quaternary diodes will fire.



Fig. 12-3. Block Diagram Shaving Station-to-Station Termination.

4.2

The junctor will recognize this as termination. The ring tone (RT) will be

gated-out to the line terminated in the junctor VT. At the same time, a ring generator start (RGS) output is sent to the interrupter on the tone supply The ring back tone (RBT) is provided to the originating line.

- (a) The terminated line will integrate the RT and close its ring up (RU) relay.
- (b) The RGS input to the tone supply will start the interrupter The

interrupter will provide a one-second "on," threesecond "off" ground to the ring generator control. The ring generator will pulse "on-off" to provide ring generator current to the RG input of the called line.

(c) The rung generator (RG) will pass through the closed contacts of relay RU, via the ring lead, to the station istrument of the called line

4.3 **Upon** answer, the line circuit will detect answer of the call and open the RU relay. The junctor will also detect the answer. and will remove the outputs (1) RGS. (2) RBT, and (3) RT and turn cm the speech gates Transmission may take place.

(a) If the codes do not fire (refer to Step (d) of paragraph 4.1), due to any reason, the junctor will detect this and pass a busy tone (BT) to the originating line. The busy tone is supplied by the tones supply.

is supplied by the tones supply. (b) It should be noted that termination occurs during; the RIT time slot (1.6 millisecond) that marked the register.

4.4 As described previously, and shown in Fig. 12-3, a line IS normally addressed by marking the R's, T's, and U's leads from

the register via the TE bus

(a) This grading IS shown in Fig 12-4 It should be noted that if

an Attendant Control (316035) and Console (316028) are equipped, the output from the attendant control digit stores are essentially parallel with the register digit outputs.

(b) In either case, a ground pulse IS supplied to the H, T, and U input

of the lane circuit (c)

The circuit within the line IS a three-input AND gate

NOTE: An AND gate requires three simultaneous inputs to perform its assigned function(s)

- (d) The H's 2, T's 5, and U's 4 out--puts are used to mark the digits 2-5-4
- (e) To mark the digits 2-6-7, the 2 is again marked. Due to no in-

is again marked Due to no input to the T's 6 and U's 7, the line 2-6-7 will not be addressed

(f) Grading IS such that no line circuit(s) will have the same three inputs



Fig. 12-4 Grading Block Diagram.

4.5 The line circuit also i.as a group hunt access (GHA) output. This GHA lead may be opened to address the line Opening the GHA lead will cause the lines VO to pulse towards +18VDC. If a junctor (or feature) VT is being marked (paragraphs 4.1, 4.2, and 4.3) termination may occur.

Each line circuit's GHA lead is con-

4.6

nected to the punching field (refer to Dwg 130071 or 329972). If it IS not used for a feature application (SHA, GHA, or ANA), it is connected to ground

5.0 TERMINATION (FEATURE CALL)- SEE FIG. 12-5

5.1 If the dialed number is a feature call (trunk/tie lines, attendant, code call, public address, conference), the register detects this and will mark the required hundreds and/or tens digit plus a feature mark (FM). This will be time (1).

5.2 The addressed feature will provide a feature mark allott (FMA) pulse to the

allotter The allotter will then expand the RIT to 3 32 milliseconds It will also provide a drop junctor (DJ) pulse to the junctor

> (a) The junctor will drop the originating line by removing the

holding current (Ih) from the matrix This occurs at time (1)

5.3 The allotter will mark the Crap circuit, and the dropped line will re-fire to the trap and be held This IS time (2).

(a) The addressed feature will check the lines class of service If the line IS acceptable, the feature will provide a drop


Fig. 12-5. Block Diagram of Feature Call Termination.

back inhibit (DBI) pulse to the allotter. The DBI pulse will prevent **a** drop-back-to-junctor (DBJ) from being originated within the allotter. This occurs during time (2)

The trap circuit will drop the line, and 5.4 the line VO will mark positive. The feature VO will then mark to -18VDC. and the P-S and T-Q matrix diodes will fire, and termination will occur This takes place during time (3).

5.5 If the addressed feature does not pro-vide a DBI pulse, the allotter will gen-erate a DBJ pulse The DBJ pulse will be sent to the **Junctor dropping** the line from the trap circuit. The line VO will **re-fire through the P-S matrix to the re-marked junctor VO** This takes place during time (4)

> The junctor will recognize this (a) and send a busy tone to the line

6.0 PNPN MATRIX - SEE FIG. 12-8

6.1 Each primary matrix PCB has 25 outlets for connection to 25 line VO terminals

Each primary outlet is con-(a) nected to five primary diodes Each primary diode IS connected to five secondary

diodes Each secondary diode becomes a secondary horizontal (SH) outlet Each primary-second PCB has 25 SH outlets (Refer to Fig 12-8) The **SH** outlets (100) of four P-S

(b) matrices are graded to provide 15 outlets for assignment to the junctor VO's, and 25

for assignments to the tertiary matrix inlets

6.2 The SH grading for the junctors IS shown in Fig. 12-9 In a 12-junctor system, VO's 13, 14, and 15 are folded back to VO's 8, 9, and 10

A maximum of 12 junctors may (a) be equipped per cabinet Dwgs 130117 and 329972 detail the strapping

Each cabinet has one tertiary quarternary (T-Q) matrix (316041). 6.3

NOTE: See Table 11-1 of Section XI

(a) Each quaternary matrix has 40

quaternary horizontal (QH) in-lets Each QH inlet is assigned on a one-to-one **basis to the feature** VT's







Fig. 12-7. Osciliograph Showing Failure of a Feature Call.

(b) For a 100-line cabinet, the following circuits VT's require **a** QH outlet assignment:

Circuit	QH Required	To Possible	tal Equipped
Junctor Transfer Junctor Trunks Tie Lines Attendant Trunks Code Call Public Address Conference Line Lockout	1 each 1 each 1 each 1 each 2 each 2 each 5 each 6 each	12 4 24 0 2 1 1 1	12 4 18 12 2 1 1 1 1
Total		46	52

6.4 Although it is possible to specify a single cabinet with 46 or 52 circuits requiring QH outlets, this cannot be done due to the limitation of 40 QH Inlets each T-Q.

(a) The tertiary horizontal (TH) outlets are made available **for**



Fig. 12-8. PNPN Matrix Block Diagram.

grading to the secondary horizontal (SH) outlets. (b) Strapping information for single and multicabinet systems 16 provided in Dwgs. 130117 and 329972.

7.0 DC VOLTAGE GENERATION - SEE FIG. 12-10

7.1 The AC input is provided to a stepdown transformer located in each cabinet.

Each bay has its own separate rectifier and series regulating (SR) circuits. Bay 1 has the above, plus the control circuits. The control output regulates all power supplies by controlling the series regulator circuits. Thus, the **outputs** of all power supplies of a type (i.e., -18) are paralleled for multibay operation, and controlled from bay 1. Thus, all adjustments (except ± 3 VDC) are made on bay 1.

7.2 Each **power** supply (-18, +18.5, +36, -36 VDC) has its separate AC input from the transformer secondary to its separate rectifier assembly.

7.3 On TE-400 assemblies 320136, the rectifier assemblies are located on he rear of the cabinet or in the external power cabinet.

7.4 On TE-400 assemblies 130097, the rectifier assemblies are located

within the power supply assemblies.

7.5 On TE-400 assemblies 320136 with Internal power, the ± 3 VDC is derived from the +18 5, -18VDC supplies anti is fused and adjusted from She front of each cabinet.

7.6 On TE-400 assemblies 320136 with external power, the ± 3 VDC supply is derived as in paragraph 7.5 **but** fuse and **adjustment** location i in She rear of each power cabinet (85924).

7.7 On TE-400 assemblies 130097, the ± 3 VDC supplies are derived from the +36, -36VDC supply and ate adjusted from the front of each +36, -36VDC supply (Section V, Fig. 5-2).

7.8 The voltage check reference transformer (paragraph '7.1) provides AC outputs to the 316070 voltage check reference circuit that allows the 316010 to supply ±18, ±36, ±56 to the 316089 main alarm and voltage check circuit.

7.9 The outputs of the +18.5, -18, +36, -36 VDC supplies are also supplied to the 316069. The 316069 compares these outputs, and (if within $\pm 10\%$) provides a ground to the auxiliary relay via the auxiliary relay power signal (APS). With the auxiliary relay operated, a ground is supplied to each *power* relay of each bay. The PR is operated and power is supplied to the system voltage bus **bars.** Each PR has four "C" type contact sets. In the IDLE state (320136 assemblies with external power cabinet 85924 and 130097 cabinet assemblies only), an IDLE load IS provided.

Each contact set is composed of (one each) -18, +18 5, -36, +36 VDC

7.10 If a short is placed upon the DC bus bars, the power supply output will fall 10% below its rated output. The 316069 will detect this and remove power from the system by removing ground from the APS lead, ultimately removing ground from all PR relays and de-energizing same.



15 LOCAL JUNCTORS EQUIPPED - NO STRAPS ARE REQUIRED

Fig. 12-9 Originating Permanent Grading Cable and Terminal Strapping - 100 Lines.



Fig. 12-10 Block Diagram of DC Voltage Generation.

If the AC input goes beyond limits, the output of the 316070 will vary. Var-7.11 iance greater than ±10% will be detected by the

116089 and nower removed as described above. The AC input to the 316070 is fused.

The APS lead is multiplied through the 7.12 blower switch and the 5VDC supply on 130097 assemblies only. The DC power is automatically removed when:

- blower fails, or (a)
- (b) 5VDC power supply fails or is misadjusted.

7.13 The 316069 has adjustments and alarm lamps (paragraph 2.9 and Figs. 5-1

and 5-3 of Section V) corresponding to each power supply. The adjustments are made initially and it may or may not be necessary to readjust them.

7.14 A manual reset is provided.

7.15 A constant voltage transformer is used for the AC input, thus allowing

for large line fluctuations.

7.16 Applying an <u>external</u> ground to the APS lead when the PR relay is de-1 energized (as in paragraphs 7.10, 7.11, or 7.12) may result in serious damage to the system.

7.17 As noted in Section XI, the power supplies are not interchangeable. A -18VDC (for example) supply for bay I has control circuitry and may be Identified by the adjustment potentiometer. For bay 2 the potentiometer and control circuit will not he equipped.

On the 130097 assemblies, a 5VDC 7 18 source is equipped. See Section X, paragraph 3.57 for description.

8.0 THREE-VOLT DC SUPPLIES

The basic functions of the ±3VDC sup-8.1 plies is to provide clamp voltage to the lines, **junctors**, and all other circuits capable of originating or terminating a call.

- failure of the -3VDC supply usually 8.2 results in an inability to ring a line.
- 8.3 Failure of the +3VDC supply usually results in an inability to terminate a call.

NOTE: Termination to any feature circuit is (generally) as described in paragraph 5.0. Certain features do not utilize the VT or seizure as described in paragraph 5.0

9.0 ATTENDANT OVERRIDE

9.1 Each junctor, trunk, or tie line has the following outputs: DIW, DIX, DIY, DIZ. These outputs are connected via a grading pattern to the 316062 attendant override.

Each junctor, trunk, or tie line has an 9.2 override voice originate (OVO) or override voice terminate (OVT) lead connected to the 316062

When the attendant override botton is 9.3 depressed on the Attendant's console. the console's a T-R is connected to the override circuit, via the OPT lead

9.4 The attendant then key punches the desired station number on the console key pad. This causes the terminate bus to be marked and the line circuit addressed. This provides a posi-tive pulse via the line VO, P-S matrix, junctor VO (or trunk/tie-line VT) to the 316062 via the DIW, DIX, DIY, and/or DIZ leads

The 316062 will decode the input and cause the OPT lead to be connected to the console T-R.

A one-second burst of dial tone will be 9.5 provided to warn the parties of the incoming call.

If the desired station is not busy, BT 9.6 will be returned to the Attendant.

10.0 GROUP HUNT - SEE FIG. 12-11

leads

Each group hunt PCR contains two group hunt circuits. Fig. 12-11 shows 10.1 only inputs/outputs for one circuit only.

(a) A master group bunt line is
assigned. The H's, T's, and U's
leads are connected to thee group hunt circuit.
(b) In an idle state, ground is pro-
vided to each slave line via the
GHA lead. Refer to paragraphs 4.5 and 4.6.
(c) If the master line is addressed.
the group hunt circuit is ad-
dressed. The following outputs are provided:
(1) GHCR and GHCS to the sys-
tem allotter to inhibit the
trap circuit.
(2) RTA to the system allotter
to expand the RIT pulse to
3.32 milliseconds.

(d) The RIT expansion causes the H's, T's, **and U's** mark to **re**-main for 3.32 milliseconds. If the master line is not busy, it will mark positive on its VO and termination will occur. Refer to paragraphs 4.1, 4.2, and 4.3.

10.2 If termination does not **occur**, the group hunt will sequentially open each GHA lead (1-2-3-4). If the first slave line is free, it will be marked and will terminate (refer to paragraphs 4.5 and 4 6). If the first slave line is busy, the **second** slave line may respond.

10.3 If all slave lines are busy, termination will not occur. At the end of 3.32 milliseconds, the junctor or trunk/tie line will detect that termination has not occurred and will provide a **busy tone** to the originating line.

12-10



Fig. 12-11 Group Hunt Circuit Block Diagram.

12.1

13.1

10.4 The fourth slave GHA may be connected to a maximum of 25 line GHA's If it IS marked, all 25 lines will attempt to terminate. One will succeed This will be entirely at random.

10.5 The GHE (group hunt extend) lead will be tied to ground unless it is utilized by the single digit access (SDA) or assigned night answer (ANA) circuits.

Accessing the SDA or ANA circuit (connected to GH via GHE) will then result in addressing the master line as in paragraphs 10.1 through 10 4.

11.0 SINGLE DIGITACCESS (316086)

11.1 Six lines may be connected to the SDA circuit. These lines may then be addressed by dialing the assigned single digit.

11.2 The call is processed as in paragraphs 3 1, 3.2, 3 3, 3 4. 4 1, 4 2,

and 4.3 with the following exception

11.3 The register recognizes the single digit and addresses the SDA circuit with an Ii's and feature mark. This opens the SDA line's GHA lead (as in group hunt), and allows termination to occur as described in paragraphs 3 1, 3 2, 3.3, 3 4, 4 1, 4 2, and 4 3

12.0 AUXILIARY JUNCTOR APPLIQUE CIRCUIT (316079)

Each junctor's (maximum 12) JB

(junctor busy) output IS connected to the 316079. Each 316079 has three AJB (all junctor busy) outputs (one per register)

As long as N out of N+1 junctors are free, the AJB lead will remain at -18VDC

If N+1 become busy, then all JB inputs are at -18 (a free junctor yields a ground on its JB lead) and the AJB output goes to ground.

12.2 If an attempt is made to dial a local subscriber number through the N+1 junctor. busy tone will be returned

12 3 Feature calls will be processed normally

13.0 EXECUTIVEOVERRIDE (EOR)

The EOR circuit provides the capability of reaching a busy line. The sub-

scriber, who has access to this circuit, will be able to access a busy line via the junctor/trunk/tie line with which the busy line is associated.

13.2 The EOR circuit is permanently associated with the line which is capable of overriding a busy line. A junctor is also assigned to this line, such that this line can only seize this particular junctor which is made not accessible to any other line. This line dials for the desired number in the normal manner. If the desired line is busy, the subscriber with the EOR (Executive Override) feature could dial digit two to access the desired busy line via a set of relay contacts to the junctor associated with the desired busy line.

<u>13.3</u> As the subscriber with the EOR (executive override) feature goes off hook, the EOR line seizes the EOR applique circuit and EOR junctor. (The common OH input to the line circuits is removed from the executive's line circuit and EOH from the executive applique circuit is connected to the OH terminal of the executive's line circuit. The SOM output of the EOR junctor is isolated from the common SOM terminal of the junctors. With the above arrangement, the EOR line circuit can only seize the EOR junctor.) And, the EOR junctor seizes a register via the R-J matrix in the normal manner. When all the dialed information is stored in the register. termination is ready to take place. The register extends a pulse to the termination side of the junctor. In this case, this pulse is also extended to the EOR applique circuit via terminal MC.

During this time slot, the desired line circuit will also be marked by the terminating bus. If the line circuit is busy, this pulse is extended to its junctor and this pulse from the line circuit to the junctor will enable a gate in the EOR junctor adapter circuit. The output of this gate is fed back to the EOR applique circuit via four diodes to operate a set of flipflops identifying the junctor is which the desired line circuit is located. If the EOR party desires to override, he dials two. This will enable the relays, and a path via the relay contact will be established from the EOR line circuit to the junctor after a burst of warning tone is sent to both parties.

14.0 TRUNK-TIE LINE SEIZURE

14.1 Any trunk or tie line that is accessible by dialing a feature digit(s) must be equipped with a trunk allotter (316034). See Fig. 12-12.

- (a) Termination to a trunk feature
- was described in paragraph 5.0.
- (b) Each trunk allotter can allott a maximum of 28 trunks/tie lures.

These 28 trunks/tie lines are divided into four trunk groups (TG) for a maximum assignment of seven trunks to each TG.

(c) A trunk/tie line must **be allotted** before it can be seized directly

by a subscriber.



Fig. 12-12. Block Diagram of Trunk Allotter.

(di) The trunks/tie lines are assigned sequentially as shown

in Table 12-1.

Table 12-1

Typical Trunk Group Assignment

TG	Trunks
1	1, 5, 9, 13, 17
2	2, 6, 10, 14
3	3, 7, 11, 15
4	4, 8, 12, 16

(e) As the trunk allotter is addressed, it will step sequentially

through its four TG's.

14.2 The Allotting of the trunks in each group is described in the following paragraphs:

When power is applied to the (a) cabinet, one of the four leads of the TG will mark negative. This negative pulse will

be present on the TG lead of the **assigned trunks**. (b) Of the assigned **trunks**, one will

be allotted. This is due to the firing of a four-layer diode on the individual trunk bard.

When tie trunk allotter is ad-(c)

dressed via a feature digit(s) (FD) and feature **mark**, RFT and SFT will mark all trunks. The trunk previously allotted will accept this input and mark its VT. (d)

At the same time, the trunk

allotter will step to its next TG and mark it negative. The trunk allott lead (TA) of all assigned trunks will mark positive, and a trunk will be allotted.

As the trunk allotter is ad-(e)

dressed, the lines class of service is checked. If acceptable, a DBI pulse IS sent to the allotter from the trunk allotter. At the same time, an FMA input IS provided to the system allotter to double the RIT pulse.

Each TG will always have a first 14.3 allott, second allott, third allott, etc., trunk. The second will allott if the first is busy, etc.

> It IS Important to understand (a) that the first allott, second,

etc, may not be in sequence. This is due to the use of four-layer diodes within the trunk. For example: In TG1, trunk 17 could be the first allott, trunk 5 the second, etc.

If all trunks within a trunk (b)

group are busy, a pulse on the trunk allott overflow (TK) lead will cause any free trunk to allott

14.4	The trunk allotter assembly 316034(2)
	may be strapped so that a diode must
be provided to	o restrict a line or a diode is not
equipped to	provide restriction. This option is not
available on T	TE-400 320136 assemblies

14.5 A NAM input is provided from the unassigned night answer circuit to reallott the trunks for each unassigned night answer call. The NAM pulse will be accompanied by an FM pulse.

14.6 After the originating subscriber seizes the trunk, the central office (CO) trunk will be seized. The central office dial tone will be returned to the subscriber's extension.

15.0 TWO-WAY CITY TRUNKS

15.1 This circuit provides a means of establishing calls between an EPABX extension and the central office. Incoming calls may go through the attendant, or directly from an extension to the central office.

- 15.2 Functions:
- Marks the trunk busy when it is (a) seized by an extension or the city so that the trunk cannot be allotted.
- - Passes dial pulses to the city. Provides toll restriction.
 - (c)
 - Provides trunk hold on incoming calls. (d)
 - Provides trunk hold on outgoing (e) calls for time and charges information.
 - Provides a means of transfer-(f) ring the trunk to another EPABX
- extension on either
- incoming **or** outgoing calls. Provides for consultation calls, (g) or transferring either incoming
- or outgoing calls.
 - Provides unassigned night (h)
 - answer service. (I) Provides assigned night answer service.
 - Provides camp-on-busy access. $\begin{pmatrix} j \\ k \end{pmatrix}$ Provides operator recall on out-
 - going or incoming calls.
 - Provides attendant's access with visual and audible indication. (1)
 - (m) Provides trunk splitting.
 - Provides a means of releasing if (n) the city abandons the call at any time.
 - Provides a six-second delay (0)after the extension goes "on hook" before the trunk can be allotted.
 - Provides message registration. (p)
 - Provides attendant override. (q)
 - (r) Provides attendant recall on an
 - incoming call if the extension

should lose the call for any reason

- The following terms will be used in describing the operation of the twoway city trunks:
- BZG Busy Guard Relay Buzzer Lead ΒZ ÇТ Cut-Through Relay Dial Relay SZ
- Seize Relay RU Ring-Up Relay

15.3

T	-	Tip
R	-	Ring
VT	-	Voice Terminate
tss	-	Trunk Splitting Lead
rex	•	Release Extension
COA	-	Camp On Access
COE	-	Camp On Enable
th	-	Trank Hold
LU	-	Lock Up
tja	•	Transfer Junctor Access
TJR	-	Transfer Junctor Release
TUN	•	Trunk Unassigned Night Answer
ori	-	Operator Recall Inhibit
SAT	-	Saturate
TL	-	Trunk Lamp
rft	-	Feature Terminate
SFT	-	Feature Terminate
TG	-	Trunk Group
ТА	-	Trunk Allott
OVT	-	Override Voice Terminals

15.4 The following paragraphs describe the general operation of two-way city dial trunks for both loop and ground start operation.

- City originates call to EPABX. 15.5
 - The city placer ground on the tip of the EPABX trunk. **The BZG relay operate@.** (a)
 - (b)
 - -18 volts is extended to lead (c) BZ-1 to cause the attendant's buzzer to sound.
 - (d) The trunk is marked busy to prevent it being allotted end

seized by an EPABX extension. The Attendant's trunk lamp button flashes bright at 120 ipm. (e)

NOTE: 15.5 (a) and (b) for 316075 trunk only.

Attendant answers the incoming call. 15.6

- The Attendant depresses the (a) flashing trunk button.
- The trunk is held in a busy state (b)

to prevent it from being allotted and seized by an EPABX extension. The Attendant trunk lamp button (c)

changes from bright flashing at 120 ipm to bright steady. The console's talk city

- lamp cornea on (d)
 - The COB circuit is accessed (if a COB is idle).
 - The Attendant may now converse (e) with the city party via the TC

lead, XR2-2 relay contacts, and T1.

15.7 The Attendant terminates a city call to an EPABX extension.

The Attendant keys-in the de-(a)

sired EPABX extension via the Attendant key sender. These digits are stored by both the Attendant's and the COB storage circuits. When all the digits have been stored, the Attendant's control circuit during its allott period marks the desired extension and extends a negative pulse to the trunk via the PU lead.

- (b) The trunks VT lead is marked negative.
- If termination is complete, the (c) extension and the trunk will be

connected together via the four-layer diode matrix

- and ring tone will be sent to the line.
 - (d) The ring back tone is sent to the city party and the trunk is held in a busy state.
 - The Attendant may now release (e)

from the trunk if it is desirable or may remain connected to the trunk to announce the call.

> If the Attendant releases from (f)

the trunk, the trunk lamp button

will change from bright steady to dim 60 ison to indicate a ringing condition.

- EPABX extension answers.
 - The trunk senses the answer (a) condition.
 - The Attendant's trunk lamp (b) button changes from dim 60 ipm

to dim steady if the Attendant has released from the trunk; otherwise, the Attendant's trunk will remain bright steady.

The trunk is held in a busy state. The ring tone and ring

back tone are turned off.

(c)

Text

15.9

(d) If the Attendant remained connected to the trunk, a negative

signal is sent to the Attendant's control via the TSS lead to operate the automatic trunk splitting circuits. The automatic trunk splitting circuit will hold the **trunk speech gate** off **via** the SAT lead. The Attendant **may** converse with the EPABX extension via t&e TE lead without the city party overhearing the con-versation. The trunk speech gate will turn on when the Attendant releases from the trunk, allowing the EPABX extension to converse directly with the city party. If the Attendant released from the trunk while the trunk was in the ringing condition, the trunk speech gate will turn on as soon as the EPABX extension answers.

With the EPABX extension having answered the call, the trunk digit stores are enabled, making operator recall and transfer junctor features available to the extension as soon as the incoming trunk call is answered.

Release Extension

If the extension keyed by the Attendant does not answer, the Attendant depresses the REX button on the turret. This places a timed negative pulse of approximately one-half second duration on the REX lead, causing the trunks VT lock-up path to be interrupted. This allows the trunk to release the unanswered extension by removing matrix holding current.

NOTE: The only time an extension can be released using the REX button is while the extension is in a ringing or busy condition.

- 15.10 The COB terminates the city call to an EPABX extension.
 - If the extension to which the (a)Attendant wishes to terminate

the trunk is busy, the Attendant may place the trunk in a camp-on-busy (COB) condition by depressing the COB button on the Attendant turret.

> (b) When the Attendant enters a trunk to service an incoming

call, a negative pulse is sent to the COB circuit via lead COA. This causes the COB circuit to mark lead COB 1 or COB 2 positive, seizing the associated circuit.

(c) If the call is completed in the normal manner, the COB cir-

cuit will be released. (d) If the extension is busy and the

Attendant operates the COB button, the COB circuit is set via the COE-1 lead. The Attendant's trunk lamp button changes from bright steady to dim 120 inm.

(e) The L relay remains operated, holding the loop to the city

closed. Thus, the trunk, in effect, is in a hold condition under control of the COB. The COB circuit will mark the COB-1 or COB-2 leads positive, causing an attempt to mark the trunks VT lead. However, as long as the Attendant is in the trunk, the signal is inhibited.

(f) The Attendant may now release from the trunk. Every five seconds, the COB circuit will cause the trunk's VT lead

to be marked negative. At the same time, the COB attempts to mark the extension.

(g) When the extension becomes

idle and both the trunk and extension are marked, they will be fired together via the four-layer diode matrix. The trunk's sensing stages will detect this and:

- (1) Send a ring tone to the extension.
- (2) Cause the COB to be released.

The Attendant's trunk lamp button will change from dim 120 ipm to dim 60 ipm, indicating the ringing condition.

The call will now proceed as a normal call.

(h) If the Attendant should key in the wrong number or desire to

change the number, the COB circuit must be released to clear the COB storage circuits. The Attendant operates the REX button which sends a timed (one-half second) negative pulse to the trunk via the REX lead. When the REX pulses negative, the COB is reset.

(i) If the city party should abandon the call while the trunk is under control of the COB circuit:

- (1) The trunk will release because the loop has been opened.
- (2) The COB is reset, thus releasing the COB circuit.
- (3) The hold condition is removed.
- (4) When the six-second timer times out, the trunk will return to an idle condition.

(j) If the Attendant should re-enter the trunk while it is under coatrol of the COB circuit, the COB enable filp-flop will

be reset via the LU lead. The Attendant's trunk lamp button will change from dim 120 ipm to bright steady. If another extension is desired, the REX button should be operated in order to clear the COB storage circuits.

If the city party wishes to hold the extension for a longer period, the Attendant must re-operate the COB button before releasing from the trunk.

15.11 **Operator Recall**

(a) If it is desired, the EPABX extension may recall the Attendant

by dialing the digit 3 or above.

- (b) The operator recall is set.
 (c) A negative voltage is sent to the BZ lead to sound the Attendant's buzzer.
- (d) The Attendant's trunk lamp button for that trunk switches

from dim steady to bright 120 ipm. (e) As the Attendant enters the

trunk, the BZ lead returns to

ground, disabling the Attendant's buzzer. (f) The trunk is now under control

of the Attendant who can talk with both the extension and city party. (g) The Attendant's lamp button for

(g) The Attendant's lamp button for that trunk switches from bright

120 ipm to bright steady.

NOTE: Since there is an extension in the trunk, COB is not accessed when the Attendant enters the trunk.

(h) The Attendant may now transfer the call to another extension or release from the trunk, placing the trunk beck under control of the extension. If transfer is desired, the party requesting the transfer must go "on hook" after contacting the attendant.

15.12 Trunk Hold

(a) Any time after the Attendant enters a trunk, the trunk may be placed in a trunk-hold condition by depressing the

trunk-hold button at the Attendant's turret.

(b) Depressing the trunk-hold

button places a ground on the TH lead. The L relay is held operated, and the loop to

- the city is held. (c) Each time the Attendant enters a trunk and the trunk is in a hold
- condition, the trunk is reset via the LU lead. (d) The main functions of the hold condition are:
 - (1) To hold an incoming call while the Attendant is busy with something else.
 - (2) To obtain time and charges from the central office operation.

(e) The trunk lamp button will only give an indication of a trunk hold condition when the attendant has released from the trunk and there is no EPABX extension in the trunk. The lamp indication is a dim wink.

15.13 Transfer Junctor Access

When desirable, the EPABX ex-(a) tension connected to the trunk

may initiate a consultation call after answering an incoming trunk call, or ten seconds after the last digit was dialed on an outgoing trunk call by dialing the digit one. This causes the transfer to be accessed.

(b) The digit one is detected, causing one of the four-layer diodes connected to TJA 1 or 2 to fire (if a transfer junctor is available). When one of the diodes has fired, the circuit is held marked

- The speech gate circuits are (c)
 - turned OR.

If a transfer junctor has been (d)seized, the dial tone will be VT lead via one of the marked

passed to the trunk four-layer diodes

The transfer junctor and system register. The trunks VT lead is now held marked by two circuits:

- (1) The normal hold path which is under control of the extension connected to the trunk.
- (2) The transfer junctor.
- The city trunk speech gate IS (f) turned 'off 1'

15.14 Extension dials into transfer junctor.

> The extension connected to the (a)

trunk may now dial the number of the desired EPABX extension. Dial pulses are passed to the transfer junctor via the TJA lead.

If the extension should dial the (b)

cancel the call by momentarily going "on hook," re-leasing the transfer junctor During the time the ex-tension is 'on kook," the trunk is held marked by the When the extension goes "off hook," again the trans-fer junctor will be released, making it necessary to dial digit one again to reseize the transfer junctor

NOTE: Do not use this feature after the last digit has been dialed.

> When dialing is completed, the transfer junctor will send a (c)

pulse via the TJA lead to the trunk which sets the transfer terminate flip-flop This removes the hold condition placed on the trunk by the transfer junctor when the transfer junctor was accessed The trunk's VT lead is now under control of the extension connected to the trunk

> The digit stores in the trunk are (**d**) now enabled.

If the called extension is busy, (e) the transfer junctor will send a

busy tone to the trunk.

- To release the transfer junctor (f) it is necessary to dial the digit one.
- If the called extension is idle, (g)

the transfer junctor will send a ring back tone to the trunk.

> (b) If the called extension is not answered, it is necessary to

dial the digit one to release the transfer junctor.

15.15 Extension in transfer junctor answers.

(a) When the extension in the transfer junctor answers, the called and calling parties may converse without the city party hearing the conversation.

(b) The city party may be included in the conversation by the exten-

sion in the trunk dialing the digit two. The digits are stored in the (c)

digit stores of the trunk. The

transfer junctor conference circuit will be set, and turns the city speech gate on. The city party is now included in the conversation.

Transfer Junctor Release 15.16

The extension connected to the (a)trunk may release the transfer junctor at any time after termination by dialing the digit one.

15.17 Transfer

(a) **During** a consultation call, it may be desirable to transfer the extension connected to the transfer junctor to the trunk (placing this extension in control of the trunk).

The extension connected to the trunk goes 'on hool This signal is sent to the transfer junctor via the TJA lead. The extension IS released from the trunk.

(b) After approximately a one-

second delay and during the transfer **junctor's allott** period, the transfer junctor releases the extension connected to the transfer junctor and extends a pulse to the trunk to mark the trunk's VT lead.

If the attempt to transfer is not successful, the extension will be (c) reconnected to the transfer junctor. During the next allott period, the transfer junctor will be released as described previously Since there is no extension connected to the trunk's VT lead, the trunk will clear to an idle condition. If the original call was an incoming call, the trunk will be reseized by the city. If the attempt to transfer is suc-(d)

cessful, the extension will be

connected to the trunk, placing the trunk under con-trol of the transferred extension During the next transfer junctor allott period, the transfer junctor will be released as described previously.

NOTE: Do not attempt to transfer until after the extension connected to the transfer junctor answers.

15.21

tension.

Extension originates call to city.

TM 11-5805-652-14

15.18 **Unassigned Night Answer (UNA)**

(a) If for any reason the Attendant's turret is left unstiended, the

unassigned night answer (UNA) may be turned on (at the Attendant's turret), giving any EPABX extension the capability of servicing an incoming trunk call.

(b) When the trunk is seized by an

incoming call, the BZG relay is

operated, and a negative signal is sent to the UNA circuit via the RU lead which actuates an audible or visual alarm, indicating an incoming trunk call.

(c) Any extension may answer the trunk call by dialing the UNA

feature digit. The UNA circuit will mark the trunk's VT lead via the TUN lead. The extension and the trunk will be connected together through the fourlayer diode matrix. The trunk is now under control of the extension.

(d) Operating the UNA key places a ground on the ORI lead. Thus,

with the trunk is the UNA condition, the operator recall feature is inhibited.

Assigned Night Answer (ANA) 15.19

If for any reason the attendant (a) turret is left unattended, the assigned night answer (ANA) may be turned on (at the

attendant's turret). A preassigned extension now has (b) When the trunk is seized by an

incoming call, a negative signal is sent to the ANA circuit via the RU lead. The ANA circuit will mark the trunk's VT lead during its allotter period via the PUA lead. At the same time.

the preassigned extension is marked, causing the extension and trunk to fire together via the matrix. This is detected by allowing a ring signal to be sent to the extension and a ring back signal to the city party. The VT lead IS now held marked. When the **ANA** circuit was turned on, a negative signal was extended to the trunk via the ANF lead.

When the extension answers, the (c) VT lead moves to +l volt. The

trunk is now under control of the extension. (d)

If the city abandons the call before the ANA extension

answers and the extension is in a ringing condition, the extension will be released

(e)

Operating the ANA key places a ground on the ORI lead. Thus,

with the trunk in the ANA condition, the operator recall feature is inhibited.

Trunk Allotted (See Paragraph 14.0) 15.20

Only a trunk that has been (a) allotted may be seized by an

extension attempting to make a call to the city Refer to Fig. 12-12

For a trunk to be allotted, either lead TC or TK must be **(b)**

negative and lead TA positive, This will allow the trunk to be allotted. The trunk is now under control of the trunk allotter for outgoing calls

When an extension with the (a)

proper class of service dials the trunk feature digit, the trunk allotter places ground on leads RFT-SFT. This marks the VT lead of the allotted trunk. The Extension which dialed the feature digit will now be connected to the trunk via the matrix, placing the trunk under control of that ex

(b) The trunk is now marked and the

- trunk busy lamp is turned "on." The trunk lamp button turns "on" dim steady.
 - The ten-second timer is primed. (c)
 - The L relay is operated. (d)
 - (e) When the L relay is operated, ground is placed on the trunk's R lead.
- (f) The city equipment responds by placing ground on the trunk's T lead operating the BZG relay.

(g) The CT relay now operates,

operates the SZ relay and C relay (h) When the C relay is operated,

the speech gate to the city is

turned on and the ten-second timer is now enabled and starts to time out. With the loop to the city closed and the speech gate on, the EPABK extension will hear a dial tone from the city.

- The EPABX extension may now (I)
- dial the city number.
- (i) The C relay releases at the
- start of each digit and re-

operates 300 milliseconds after the last dial pulse of each digit. This shorts out the T1 transformer during dialing and resets the ten-second timer; therefore, the inter-digit time is always ten seconds. When the ten-second timer times out, the L relay is held (II)

operated, preventing any more dial pulses to be passed to the city. <u>Also, the trunks digit stores are</u> enabled, making operator recall and transfer junctor features available the extension.

15.22 Attendant originates call to city

The Attendant may originate a (a) call to the city by depressing the trunk button of any idle trunk

Ground IS extended to the trunk (b)

via the K lead to operate the KR relays which are then held operated by a ground on the LU lead

- (c) The L relay operates via ground on the S lead.
- (**d**) Ground is extended to the city on the R lead, and the city returns

ground on the T lead, operating the BZG relay With the BZG relay operated, the CT relay will operate, closing the loop to the city which operates the SZ relay and the C relay.

Dial pulses are passed to the (e) city via the S lead and L1 relay contacts

NOTE 1: The Attendant cannot access the trunk's digit stores; therefore, the transfer junctor is not available to the Attendant

NOTE 2: The Attendant does not enable the tensecond timer.

15.23 Release City (RC)

(a) If the Attendant should dial the

wrong digit when dialing a city number, the city may be released and reseized by depressing the RC button.

(b) Ground is removed from the S

lead, allowing the L relay to release. Thus, the loop to the city is open, allowing the city equipment to release.

(c) When the RC button is released, the city will be reseized as

described above

15.24 Attendant Transfer of an Extension to a Trunk

(a) The Attendant may transfer an EPABX extension from the

atte**ndan**t's feature circuit to a trunk by depressing a trunk lamp button of an idle trunk. (b) The operation of the trunk is the

(b) The operation of the funct

(c) The Attendant may dial the city number for the extension and release the extension from the trunk, or the Attendant may release from the trunk and allow the extension to dial the city number. However, as long as the Attendant is connected to the trunk, the L relay is <u>under control of the Attendant, and the extension</u> cannot signal the city. The extension is inhibited from operating the L relay.

(d) When the extension entered the trunk, the ten-second timer was enabled. Thus, the first digit must be dialed within ten seconds.

NOTE: The Attendant should not place the trunk in a hold condition until the city answers.

15.25 Toll Restriction

(**d**)

(a) When the class of service for an extension is such that the extension can make only limited city calls, the toll restriction circuits will monitor the information dialed

striction circuits will monitor the information dialed to the city. (b) When a trunk is allotted, ground

is extended to the toll restrictor

via the TRA lead. If a toll restiction circuit IS not available, the trunk is inhibited from being marked via the TRA lead when a restricted extension dials the trunk feature digit.

(c) When the trunk IS seized by a restricted extension, the toll

restriction circuit IS seized via the TRF lead.

As the extension dials into the

city, the first three digits are

passed to the toll restriction circuit via the TRH lead If the first three digits are an unrestricted code, the call will proceed normally. If, however, the code is restricted, the toll restriction circuit will inhibit the L relay from operating via the TRG lead and extend a busy tone to the trunk"s VT lead via the TRJ lead.

15.26 Attendant Override

(a) If the Attendant wishes to converse with a busy EPABX ex-

tension, the Attendant may initiate an attendant override. If the busy extension is connected to a trunk, the level of the VT lead will move from +1V to approximately +4V, causing the DIW, DIX, DIZ, and DIY leads to be marked.

(b) The override circuit recognizes these signals and connects the Attendant to the OVT lead of the proper trunk, thus allowing the Attendant to converse with the EPABX extension.

15.27 Reverse Battery Supervision Detection (City Answer Detection)

(a) Some types of CDO, during a heavy traffic period, wil be

seized and connected through to the EPAB; however, there may be a period of waiting for a CDO register to become available. This watting period may be longer than the time-out period of the tensecond timer. In this case, the extension would hear a dial tone from the city when the register became available but would not be able to dial the city number due-to the timer having timed out, locking the L relay operated. This option provides a resistor, diode, relay combination to detect the answer supervision sent from the central office.

(b) The ten-second timer circuit is inhibited until the CDO party

answers. This operates the RB detection relay and the timer starts to time out. The inhibit network on the timer also changes its time-out period to approximately 500 milliseconds. After the timer times out, the circuit operation IS as described.

NOTE: All the components necessary for this option are **mounted externally to the** standard trunk printed **circuit boards.**

15.28 Optional CT Relay Control

(a) When the local connector in the central office is of the type (the same as or similar to the Bell System's No. SD 31656 local connector) that places battery on the tip ("T" lead) of the city trunk and when the local connector is i e g tripped by the city trunk, strapping option "A" must be equipped as described in Note 8 on Dwg. 316075()66.

(b) This option allows the CT relay control flip-flop to be set approximately 16 milliseconds after the L relay is

operated as opposed to the normal operation. (c) This change in timing is re-

(c) This change in timing is required to insure the CT relay control flip-flop is s before the BZG relay is released by the battery applied to the tip ("T" lead) by the local connector.

12-18

Loop Trunk Comparison 15.29

(a) **Operation of the loop start trunk** is similar to operation of the

ground start city trunk.

(b) A ring up (RU) relay is provided on the loop start trunk instead of

a busy guard relay. (c)

If a city party abandons a call

while the loop start trunk is in the ringing condition, and the Attendant has released, the trunk will release if the city trunk has first party release. If the city trunk has last party release, the Attendant must re-enter the trunk to effect release.

This is not true on the ground start trunk. It will release in either case.

> If an incoming call, with a (d)

ground start trunk, should be lost (for any reason) after it has been terminated to an EPABX station, the operator will be recalled in approximately 15 seconds.

This does not apply to the loop star: trunk. The call will be lost in this case.

16.0 TWO-WAY DID LOOP TIE TRUNK (LOOP SIGNALING) AND E&M TIE TRUNK (E&M SIGNALING) OPERATION

General 16.1

> Refer to paragraph 2.0, Section X, for trunk assembly numbers

16.1.1 The following descriptions of DID

trunk circuit operation were written basically concerning the loop signaling type **trunks**; however, there IS no difference in the operation of the trunk circuit when using E&M signaling types.

Purpose of Circuit - The DID trunk 16.1.2 circuit provides a means of extablish-ing a call between an EPABX extension and the central office (CO) Outgoing calls may go through the attendant or directly from an EPABX extension to the CO. Incoming calls can go directly to an EPABX extension if the outside party dials the extension num-ber after dialing the city number assigned to the **EPABX**

City Originates to EPABX 16.2

16.2.1 The city places a resistive loop on the tip and ring (T and R) of the trunk which operates the INC relay.

16.2.2 The trunk IS marked busy to prevent it from being allotted and seized by an EPABX extension

16.2.3 The attendant's trunk lamp button turns "on" dim steady via the TL lead, indicating the trunk is busy.

The trunk is now held busy to prevent 16.2.4 the trunk from being allotted

Accessing a Register for an Incoming Call

- If a register is idle it is ac-(a) cessed via the RS leads.
- The fact that a register has been (Ъ) accessed by the trunk is detected.
- Dial pulses sent from the city (c) are detected and passed to the
- register via the RS lead to the register (d) If a register had not been ac-

16.2.5

cessed by the time the first dial

pulse is received from the city, a register may not be seized should one become idle.

(e) If the register was not accessed the busy tone gates are enabled, thus allowing busy tone to be sent to the city party.

Second Dial Tone (optional) (f)

(See Table 2-3, Section II). When register access is detected, the second dial tone flip-flop IS set. The C relay is operated, remowing the resistive shorts from across T1; idle period termination IS removed, and the dial tone gates are enabled sending dial tone to the city.

Register Termination of the Call to an EPABX **Extension** 16.2.6

When the register has received (a)

all **the digits from the** city, the

register sends a terminate pulse approximately 1 6 millisecond duration to the trunk via the RS lead. This IS detected and causes the trunk's VT lead to be marked negative. At the same time, the register will cause the desired line circuit to mark positive on its VO lead. If the extension is idle, the extension and the trunk will be fired together via the fourlayer diode matrix. This causes the trunk's VT lead to ascend to approximately +1V then descend to approximately -5

- **(b)** The inhibit on the ring signal is removed
- The VT level detectors detect (c) that an extension has been fired

to the trunk and causes ring signal to be sent to the EPABX extension via the trunk's VT lead, T-O, and P-S matrix.

- (**d**) The busy tone gates are inhibited
- Ring back tone is sent to the city (e) The trunk's VT lead is held
- (**f**) marked.
- The attendant's trunk lamp (g) button will flash dim 60 ipm.

indicating an extension is being rung

The reset condition is removed (h) from the reverse battery (RB)

relay control flip-flop

If the desired extension is busy, (i) to the trunk, causing the trunk's VT lead to be marked. However, the extension will not be fired to the trunk; thus, the VT level will go to approximately

-18V for the duration of the terminate pulse. At the end of the terminate pulse, the register terminate flip-flop is set This removes idle period termination, operates the C relay, and enables the busy tone gates, allowing busy tone to be sent to the city.

16.2.7 EPABX Extension Answers

- (a) When the extension answers, the
- VT lead potential rises from approximately -5V to +1V. The +1V level is set by

approximately 30 to 10.

- (b) The VT level detectors sense the answer condition.
- (c) Ring signal and ring back tones are turned off.
- (d) The attendant's trunk lamp button changes from dim 50 ipm
- to dim steady via the TL lead (e) The trunk's VT lead is held
 - (f) The trunk is held in a busy
 - (I) The tronk is held in a busy state
 - (g) The reverse battery (RB) relay control flip-flop is set (opera-

ting the reverse battery supervision relay).

NOTE: The L relay will operate at this time; however. it will have no effect on the circuit as the OTG relay is not operated.

- (h) When the RB relay operates:
 - (1) The C relay IS held operated and under control of the RB relay
 - (2) The trunk's digit stores are enabled, making operator

recall and transfer junctor features available to the EPABX extension if the trunk circuit is so equipped

- (i) The speech gate to the city IS turned on"
 - (j) Idle period termination IS held off
- 16.3 Transfer Junctor Access (Optional) -See Table 2-3, and Paragraph 15 13
- 16.4 Extension Dials Into Transfer Junctor - See Paragraph 15 14.
- 16.5 Extension in the Transfer Junctor Answers - See Paragraph 15.15.
- 16.6 Transfer Junctor Release See Paragraph 15.16
- 16.7 Transfer

(a)

During a consultation call. it

may be **desieable to** transfer the extension connected to the transfer Junctor to the trunk (placing this extension in control of the trunk) The extension connected to the trunk goes "on hook" This signal IS sent to the transfer junctor via the marked **four-layer** diode and the extension is **re**leased from the trunk

(b) After approximately a onesecond delay and during the transfer junctor's allott period, the transfer junctor:

- Releases the extension connected to the transfer unctor.
- (2) Extends a pulse to the trunk to mark the trunk's VT lead via the marked four-layer diode
- (c) If the attempt to transfer is successful, the extension will be

connected to the trunk, placing the trunk under control of the transferred extension Euring the next transfer junctor allott period, the transfer junctor will be released as has been described

(d) If the attempt to transfer is not successful, the extension will

be reconnected to the transfer junctor and during the next allotter period the transfer junctor will be released as has been described.

The trunk will then take one of the following actions:

(1) If an EPABX extension

will clear down to an idle condition and the call will be lost

(2) If the city initiated the call, the trunk will provide a

-18VDC voltage to the BZ lead to sound the attendant's audible alarm. The attendant's trunk lamp button switches from dim steady to bright 120 ipm via the TL lead. indicating a request for service by that trunk. When the attendant enters the trunk, the trunk IS reset. the operator recall is reset. the trunk is now under control of the attendant and may be serviced as described later.

(3) If the city party hangs up before the timer times out, the trunk will be reset, the timer will not time out, and the trunk will clear down to an idle condition.

NOTE 1: Do not attempt to transfer until after the extension connected to the transfer junctor answers

NOTE 2: If the trunk circuit is not equipped with an attendant access circuit and an attempt to transfer IS not successful, the trunk will clear down to an idle condition when the transfer junctor IS released and the call IS lost

- 16.8 Operator Access (Optional) See Table 2-3, Section I I
 - 16.8.1 Dial Zero Operator Recall

(a) If it is desired, the EPABX ex-

tension may recall the attendant by dialing **the** digit 3 **or greater**, **after answering an** incoming call or ten **seconds after the last digit was** dialed on an outgoing call

- (b) The dial pulses are detected and passed to the digit stores
- (c) At the start of the first dial

pulse, the reset condition IS removed from the digit stores, allowing the digits to be passed to the digit stores As the third pulse comes in to the stores, it IS detected and inhibited so that no more pulses are sent to the stores

- (d)) A negative voltage is sent to the BZ lezd to sound the attendant's audible alarm
- The attendant's trunk amp (e) button switches from dim steady

to bright 120 ipm via the TL lead, indicating a request for service by that trunk.

16.8.2 Extended Ringing

16.8.2.1 When an incoming call has been terminated to an extension and the extension

is ringing, the ring signal gates were enabled, removing the negative voltage necessary to hold the timer circuit recovered; thus, as soon as the EPABX extension starts ringing, the timer starts timing for 30 seconds If the timer times out before the extension answers, the operator will be recalled.

This disables the ring signal gates after a short delay, which allows the timer to recover, cuts off ring signal to the extension, ring back to the city party, and drops the matrix connection to the extension. With the operator recalled:

- -18V is extended to the BZ lead (a) to cause the attendant's buzzer to sound
- The attendant's trunk lamp (b)button changes from dim 60 ipm

to bright 120 ipm via the TL lead, indicating a re-quest for service by that trunk. The attendant may now service the call as described later

NOTE: After the operator IS recalled and the ringing gates are disabled. the city party will not hear ringback tone The city party's phone is quiet until the attendant enters the trunk.

- Attendant answers a request for 16.8.3 service
- **Operator Recall** 16.9
- 16.9.1 There are three operator recall conditions

Dial zero operator recall This (a)

IS when an extension connected to the trunk dials the digit three or greater as has been explained in paragraph 16.8.1

- (b)
 - Recall for extended ringing as explained in paragraph 16 8 2
 - Recall for a lost transfer call as (c) explained in Step (d) of paragraph 167

Recall Circuit Operation 16.10

The attendant depresses the (2)flashing trunk button, causing

the trunk's KR relay to operate by placing a ground on the K lead The KR relays are then held operated

by a ground on the LU lead

- The L relay IS held operated via (b) the S lead
- The operator recall IS reset by (c) a ground through KR1-2 relay contacts

- TM 11-5805-652-14
- The negative voltage is removed (a) from the BZ lead.
- The attendant's truck lamp (e)

button switches from bright 120 ipm to bright steady via the KRI-2 relay contacts and the TL lead.

16.11 If the request for service is a dial zero operator recall, the attendant

may now talk with both the extension and the city party, or either separately, by depressing the talk city (TC) button or the talk extension (TE) button The trunk is under control of the attendant; the attendant may now transfer the call to another extension (when the first extension hangs up) or release from the trunk, placing the trunk back under control of the extension. If the attendant is to extend the call to another extension, the extension connected to thee trunk must hang up The "on hook" condition is' sensed by the VT lead level detectors; the extension is released from the trunk by removing the holding current from the four-layer diode matrix. This puts the attendant incomplete control of the trunk

> The attendant may now extend the call to an extension by

punching the extension's number into the key sender 'These digits are stored by the attendant's storage circuits When all the digits have been stored, the attendant's control circuit during its allott period, will mark, the described extension and extend a negative pulse to the trunk via the PU lead

(a)

(d)

- The trunk's VT lead is marked **negative -18V via the** PU lead **(b)**
 - If the desired extension is idle.
- (c) the extension and the trunk will

be connected together via the four-layer diode matrix This causes the trunk's VT lead potential to increase to approximately +1V, then decrease to approximately -5V

The VT level detectors cause ring tone to be sent to the

EPABX extension via the trunk's VT lead, T-Q, P-S matrix

- Ring back tone is sent to the (e) city party
- The trunk IS held in a busy (f) state
- The trunk's VT lead is held (g) marked
- The attendant may **now** release (h) from the trunk if it is desirable

since the trunk is now under control of the ringing gates If the attendant does release from the trunk, the trunk lamp button will change from bright steady to dim 60 ipm via the TL lead

When the extension answers, (i)

the VT lead moves from ap-

proximately -5V to +1V The +1V level is set by potentiometer 3R4

- The VT level detector senses (j) the answer condition
- The attendant's trunk lamp (k)

button changes from dim 60 ipm to dim steady if the attendant has released from the trunk, otherwise, the attendant's trunk lamp will remain bright steady

> The trunk's VT lead is held (1)married

- Ring signal and ring back are (m) turned "off."
- A negative signal is sent to the (m) attendant's control via the TSS

lead to operate the automatic trunk splitting circuits if the attendant is still in the trunk. The automatic trunk splitting circuit will hold the trunk speech gate "off" via the SAT lead. The attendant may converse with the EPABX extension via the TE lead without the city party overhearing the conversation. The trunk speech gate will turn "on" when the a.tendant releases from the trunk, allowing the EPABX extension to converse directly with the city party. If the attendant released from the trunk while the trunk was in the ringing condition, the trunk speech gate will turn "on" as soon as the EPABX extension answers.

16.12 If the request for service is a recall for extended ringing, the circuit operation is the same as described in paragraph 16.10. The attendant may now converse with the city party via the TC lead and T-R leads.

The attendant may extend the call to an extension in the same manner as in the previous paragraph (16.11).

If the request for service is recall 16.13 for lost transfer, the circuit operation is similar to Steps (a) through (e) of paragraph 16.11. The only difference is that when the attendant enters the trunk, there is no EPABX extension connected to the trunk and the attendant will be in a "talk city" mode.

16.14 Attendant originates call to city.

16.14.1 The attendant may originate a call to the city by depressing any of the idle trunk's trunk lamp buttons.

> Ground is extended to the trunk (a)

via the K lead to operate the KR relays which are then held operated by a ground on the LU lead.

- (b) The L-relay operates via ground on the S lead.
- (c) Idle period termination IS removed via the SAT lead.
- The OTG relay is operated (d) This extends a resistive loop to

the city to seize the city equipment. Dial tone will now be sent from the city to the attendant via T1 and the TC lead

> Dial pulses may now be passed (e)

to the city via the S lead, KR1-3 relay contact, and L-l relay contacts, which opens and closes the loop to the city.

- The trunk IS marked busy (f)
 - The attendant's trunk lamp (g)
 - button turns on bright steady via the KR1-2 relay contacts.
- 16.15 Release City (RC) - See Paragraph 15.23

16.16 Attendant Transfer and Extension to a Trunk

16.16.1 The attendant may transfer an EPABX extension from the attendant's trunk

circuit to a trunk by depressing an idle trunk's lamp button.

16.16.2 The operation of the trunk is the same as in paragraphs 16 10 through 16.13,

with the exception that during the attendant's feature allott period the trunk's PU lead is marked negative. causing the VT lead of the trunk to be marked via the PU lead, KR1-6. At the same time, the extension is released from the trunk circuit; thus, the extension is transferred to the trunk. In this manner, a restricted extension may be connected to a trunk.

16.16.3 The attendant may dial the city number for the extension and release from the trunk. or release from the trunk and allow the extension to dial the city number.

NOTE 1: When the extension entered the trunk the ten-second timer was enabled; thus, the first digit must be dialed within ten seconds. Otherwise, the L-relay will be neld operated, preventing any signaling to the city.

NOTE 2: The attendant should not place the trunk in a trunk hold condition until the city party mswers.

As long as the attendant is in the trunk, 16.16.4 the trunk's lamp button will be bright steady: when the attendant releases from the trunk with an extension connected to the trunk VT lead, the trunk lamp button will switch to dim steady.

16.17 Trunk Hold

16.17.1 Any time after the attendant enters a trunk, the trunk may be placed in a trunk hold condition by depressing the trunk hold button

16 17 2 The trunk hold will:

> Hold an incoming call while the (a) attendant IS busy with something else

Hold the connection to the city (h) so the attendant may get time and charges information firm the central office

operator on outgoing calls.

The trunk hold button places ground on 16.17.3 the TH lead which sets the trunk hold flip-flop via the KR2 3 relay contacts.

- (a) The L-relay IS held operated for an outgoing call
- Each time the attendant re-(b)

enters the trunk it IS necessary to depress the TH button again if the trunk is to be put back into trunk hold

The trunk lamp button will only (c) give an indication of a trunk hold condition when the attendant has released from the

12-22

trunk and when there is no EPABX extension connected to the trunk. The lamp indication is a wink

16.18 **Trunk Allotted**

16.18.1 Only a trunk that has been allotted may be seized by an extension at tempting to make a city call.

- 16.18.2 See paragraphs 14.0 and 15.20.
- 16.19 Extension originates call to city.
- 16.19.1 See paragraph 15.21.
- 16.20 Toll Restriction (Optional) - See Table 2-3, and Paragraph 15, 25,
- 16.21 Attendant Override - see Paragraph 15.26
- 16.22 Intercept (Optional)

16.22.1 **Recorded Message Option**

(2) If the city party originates a call to the EPABX trunk and dials a

feature number or an unequipped extension number, the intercept circuit will recognize that the cail cannut be completed and send a recorded message to the city party.

(b) During the register terminate period, a pulse is sent to the

intercept circuit via the ITP lead. (c)

If the call cannot be completed, the intercept circuit sends sug-

nals back on the BTI and ITC leads to hold off busy tone.

A recorded message is sent to (d) the city party from the intercept circuit via the OVT lead and T1.

When the city party hangs up, a (e)

release signal is sent to the intercept circuit via lead ITR.

16.22.2 **Attendant Access Option**

If the city **party** originates a call to the EPABX trunk and (a)

dials a feature number or an unequipped extension number, the intercept circuit will recognize that the call cannot be completed.

During the register terminate **(b)** Period, a pulse is sent to the Intercept circuit via the ITP lead. (c) If the call cannot be completed,

the Intercept circuit sends a ground back to the trunk on the ITC lead to set the

operator recall

Ring back tone is sent to the city (**d** from the intercept circuit via

the OVT lead and T1 (e)

trunk to answer the request for service, a release signal is sent to the intercept circult via the ICR lead

If the call is abandoned before the attendant answers, a release signal is Bent to the Intercept circuit via the ITC lead

16.23 **Busy Out**

The trunk circuit may be busied out by 16.23.1 placing a strap across the busy out

- terminals. This places a negative voltage:
 - To turn on the attendant's trunk (a) lamp button dim sleady indi-
- cating the trunk is busy.
 - (b) To operate the L-relay, the Crelay, and the OTG relay, which

closes the loop to the city, thus preventing an incoming call.

(c) To prevent the VT lead from being marked by the trunk allotter,

thus preventing an outgoing call.

16.24 **Reverse Battery Supervision (City** Answer Detection) - (Optional) -(Loop Trunks Only) - See Paragraph 15, 27,

16.25 Miscellaneous

16.25.1 Operator Recall Inhibits

> (a) When DID trunks are used in a

system equipped with other types of trunks that have the night answer features, it is necessary to inhibit all the operator recall features in the trunk circuits. When the ABA or UNA key is operated, a ground is placed on the ORI lead. This inhibits dial zero operator recall, recall for extended ringing, and recall for lost transfer.

Reset When the System Power is Turned "On" 16.25.2

> When the system power is (a) turned on and the system has

been reset, the trunk should be in an idle condition.

If the trunk hold should be set when the power is turned on, the trunk could be held busy and I he attendant would have to enter the trunk to clear the condition. it is also possible to fire the trunk to an "off hook" extension when the power is turned "on." This condition looks normal to the trunk circuit and cannot be reset except by placing the phone "on hook.

If the central office were extending a loop at the time the power was turned "on," the condition looks normal and the trunk cannot be reset except by removing the loop.

When the power is turned "on" and there is no loop from the city and the trunk has not fired to an "off hook" extension, a pulse is extended to the transfer junctor access circuit via RK1, to insure the trunk is not connected to the transfer junctor, which could prevent the reset of the trunk

E&M Option (See Table 2-3) 16.26

The operation of the trunk circuit adapted for E&M signaling is the same as that for loop signaling with similar features except that instead of being seized by a loop the E&M trunk is

TM 11-5805-652-14

When the attendant enters the

seized by a ground on the E lead. The city is seized by placing battery on the M lead via the L relay contact. For E&M operation the trunk must be used with an E&M signaling circuit. This circuit will be mounted external to the equipment cabinet.

16.27Two Digit - In Dial Completion

In situations that require multiple assignment of the CO hundreds digit, the 316302(2) register is utilized within the EPABX.

This allows hundreds identification of the umber be ing dialed by trunk seizure, requiring only the tens and units digits to be pulsed from the city. Thus, a trunk in cabinet 1. upon seizure, automatically assumes that the station being addressed is located within that cabinet (hundreds group).

In practice this IS accomplished via recognition Of the RS lead seizure, and by assignment of the RS-RF leads, the called station is determined.

As this is a accomplished by shop wiring, changing Of after careful study of Dwg 130117.

17.0 TWO-WAY TIE II NE 316060()68 AND 316051(68) - LOOP SIGNALING AND E&M SIGNALING

17.1 **Purpose of Circuit**

17.1.1 This circuit provides a means of establishing calls between extensions in two different PABX telephone systems. Incoming calls may, by waiting for dial tone and then dialing the desired extension number, be connected to that extension or (if the line had the proper class of service) dial a feature digit ma access any of the features provided in the second EPABX system. Outgoing calls may be initiated by an extension (with the proper class of service) by dialing the tie line access digit. waiting for the second dial tone, and then dialing the desired extension or feature number in the far end EPABX.

17.1.2 **Circuit** Functions

- Provides loop or E&M signaling. Provides attendant or executive (b)
- override
- (c) Marks the circuit **busy when** seized by an incoming or out-
- going call so that the circuit cannot be allotted Provides direct, access to fea
 - ture circuits (when equipped

with the proper class of service) or extensions in the terminating EPABX

17.2 **Functional Description**

(d)

17.2.1 The originating PABX places a resistive loop on the tip and ring of the tie line.

17.2.2 This seizes the tie line, allows it to mark its VO lead towards +18VDC seizing a free junctor and register (see paragraph 30 - Origination)

17.2.3 Dial tone will be returned to the originating PABX party, who may now dial the required access digits to accomplish operation.

The line will require one PNPN P-S 17.2.4matrix PH inlet, thus requiring one line per equipped tie line, reducing the total system

by one line per equipped tie line.

- 17.2.5Seizure allotting of the tie line was described in general in paragraph 14.0.
- Extension originates to an outgoing tie 17.2.6 line call.

When an extension with the (a) proper class of service cials the tie line feature digit, the trunk allotter places ground on the RFT, SFT, and RCS leads which allow the trunk's VT lead to be marked negative for approxmately 1.6 millisecond. The extension which dialed the feature digit now be connected to the trunk via the matrix and the VT lead, placing the trunk under control of that extension.

- **(b)** When the matrix is fired and held, the level of the VT lead increases from -18V to approximately +1V.
 - The trunk's VT lead is now held (c)
 - marked. (**d**) The trunk is marked busy.
 - Ground is extended to the MET lead.
 - Approximately two milliseconds (e) after the matrix firing is de-

tected, the OTG relay IS operated.

- When the OTG relay operates: (**f**)
 - The L-relay is operated.
 - A ground is extended to lead OTG-M to operate the out-(2)

going peg-count circuit. A resistive loop is extended (3)

- to seize the far end PABX.
- (4) The secondary of T1 is con-

nected to the speech path so that tones may be passed to the EPABX extension.

(g) Approximately 180 milliseconds after the matrix IS fired and

held, the C-relay IS operated. This:

- (1) Removes the inhibit, allowing the speech gate to be turned "on."
- (2) Removes the resistive

shorts from Its primary of T1, allowing dial tone to be heard by the EPABX subscriber via T1 and the VT lead.

The EPABX extension may now (h) dial the desired far end PABX extension number The dial pulses are detected and sent out via the L-1 relay contacts

The C-relay releases at the (i) start of each digit and reoperates 180 milliseconds after the last dial pulse of each digit This shorts out TI during dialing to improve the signaling to the far end **PABX** The speech gate IS turned "off" The speech gate is then

held "off" by C-1 relay contacts during each series of dial pulses.

- 17.2.7 Attendant or Executive Override see paragraph 15.26
- 17.2.8 **Busy cut**

The tie line may be busied out by placing a strap across the busy out terminals.

17.2.9 The maximum input signal level the tie line will accept these distortion occurs is zero DBM.

17.2.10 Optional Tie Line Circuits

(a) Loop Trunk Signaling - The loop tie line described in paragraphs 17.1.1 through 17.2.9 is the basic tie line circuit.

(b) E&M Signaling (Optional) - The operation of the tie line circuit

adapted for E&M signaling is the same as that for loop signaling except that instead of being seized by a loop, the E&M tie line is seized by a ground on the E lead. Another PABX is seized by placing battery on the M lead via the L-relay. For E&M operation, the tie line must be used with an E&M signaling circuit. This circuit will be mounted externally to the equipment cabinet. Circuit description for Two-Way Tie Line 316051 () 68 (E&M Signaling) is shown in Dwg. 316060 () 77.

18.0 CONSOLE OPERATION

18.1 Refer to paragraph 3.3 of Section X for basic console fun&tons. Refer to Section V, paragraph 8, for console operation.

Section XIII

CONTENTS

1.0	Routine	e Maintenance	13-1
	1.1	General	13-1
	1.2	Periodic Inspection and Maintenance	13-1
	1.3	Visual Inspection	13-1
	1.4	Electrical Maintenance	13-1
	1.5	Troubleshooting Procedures	13-1
	1.6	Warranty	13-1
	1.7	Customer Service	13-1
2.0	Total i	System Failure	13-1
3.0	Major	Alarms	13-2
4.0	Minor	Alarms	13-2
5.0	station	n Trouble	13-3
6.0	Conso	le Troubles	13-4
7.0	Misce	ellaneous Trouble	13-6

1.0 ROUTINE MAINTENANCE

1.1 General

Routine maintenance of a solid state switching system is greatly simplified in comparison to a mechanical system.

Periodic Inspection and Maintenance 1.2

Visual inspection and routine mainten ance checks should be made quarterly--four times per year.

1.3 Visual Inspection

- Cleanliness of cabinet assembly (a) location area.
- check cables from MDF to sys-(b) tem equipment for damage and
- snug fit on cable connectors. Check equipment cabinets for (c)
 - damage. (d) Indication of water seepage or leakage
 - (e) Equipment cabinets for disturbance of PCB's.

Electrical Maintenance 1.4

- Refer to Section V Perform all (a)steps of paragraphs 2.4, 2.5, and 2.6.
- Refer to Section V Perform all (b) steps of paragraphs 3.2, 3.3, 3.4, and 4.0. From a TJP, dial trunk access
- (c) digits and observe trunk allotter

"stepping" through all four allott groups.

NOTE: IF, WREN TROUBLESHOOTING, A PCB IS **REPLACED** AND DOES NOT CORRECT THE **TROUBLE**, RE-INSERT THE ORIGINAL PCB BEFORE PEPCEDURING FURTHER

1.5 The troubleshooting procedures in this Section are performed in conjunction with the associated tables of Section XIV.

The TE-400 is under warranty for one 1.6 year. An attempt to replace components upon a PCB may void the warranty upon that **PCB** if it is damaged beyond repair during the attempt

1.7 The troubleshooting procedures in this Section are offered as an aid to the installer/repairman. Careful study of this manual, the equipment, and the Equipment Specifications should provide the necessary knowledge. If problems are encountered beyond your ability, contact:

Customer Service ITT Telecommunication 2107 Swift Drive Oak Brook, Illinois 60521

Telephone: (312) 654-3600

Telex:

2.0 TOTAL SYSTEM FAILURE

No alarms - complete system failure -refer to Section III, Fig. 3-1, for sys-2.1 tems using 130097 cabinet assemblies.

2.1.1Check the following for defects:

- AC circuit breakers
- (b) AC input
- Voltage check transformer fuse refer to Section III, Fig. (c)
- 3-10, for 130097 cabinet assemblies. Blower(s) (130097 cabinet assemblies only) (d)

Main alarm and voltage check (316069) PCB is alarmed.

(a) Check DC power supplies -18, +18.5, -36, and +36VDC. Refer to Section V, paragraph 2.4. If power **supply** output voltages are correct, proceed to Step (b). If power supply voltages are out of tolerance and will not adjust, proceed to Step (e) or (f).

(b) Check outputs of 316070 voltage

(b) Check reference **supply**. See Section V, paragraph 2.9.1.1, and Fig 6-1. (c) If 316070 output voltages are incorrect, replace 316070 PCB and reachade. Main alorge and states a check 216060

and recheck. Main alarm and voltage check 316069 may need adjustment. See paragraphs 2.9.1.2 and 2.9.1.3 of Section V.

(d). If 316070 output voltages are

correct, perform all steps of paragraphs 2.9.1.2 and 2.9.1.3 of Section V. If alarm lamps remain 'on' and DC power is not restored (PR relay Operated), perform all steps of paragraph 17.0 (power-off measurements) in Section IV. Replace PCB 316069 and repeat all steps of paragraphs 2.9.1.2 and 2.9.1.3 in Section V.

(e) On single cabinet systems, re-move AC power and replace defective supply. See Step (g) below.

NOTE: Refer to paragraph 3.22 of Section X and paragraph 7.0 of Section XII.

On multicabinet systems, use the following procedure to (f) locate the defective supply.

2.2

(1) Remove AC input power.

Remove 316089 main alarm and voltage check. Disconnect DC power output cable and control lead at last equipped bay. Insure that cables are not shorted.

(2) Restore AC power. Check

for faulty supply as in Step (a) above. If DC supply voltages (paragraph 2.4, Section V) are now within tolerance, the disconnected bay is at fault. Perform the following step if DC supply voltages are still incorrect.

(3) Repeat Steps (1) and (2) above until supplies in

cabinets 4, 3, and 2 have been eliminated. If power supply at still checks bad, trouble is in cabinet I supply.

(4) Replace defective supply

and do step (g).

Make all adjustments on 316069 (g) PCB as in paragraphs 2.9.1.2

and 2.9.1.3 of Section V.

- PCB 316301 (5VDC Supply and Alarms) Alarm Lamps "On" (130097 Cabinet 2.3 Assemblies Duly)
 - Replace PCB 316301 and follow instructions outlined in para-(a)

graphs 2.6 and 2.9.1.3 of Section V.

- 2.4 Stations do not ring. RBT normal. No alarms.
 - (a) Check fuse at ring generator.
 - Replace, if necessary.
 - On 320138 assemblies, dial TVP (b) from another TVP. Insure RBT

received Observe RL3 on 316037 (tones) pulsing Replace, if required.

Observe ring generator extension relay pulsing. If 316037 RL3 pulses, and extension relay does not, check input to replay. Repair as required. Replace as required.

2.5 No dial tone at all stations - major alarm

> Check both (regular and standby) allotters 316039 () for an alarm (a)

condition. Replace as required. Check both (regular and standby) (b)

tone supplies for an alarm condition. Replace as required.

Inspect all registers (316025 or 316302) for an alarm condition. (c)

Replace as required.

2.6 No dial tone at all stations - minor alarms

Check 316037 tone supply If in (a) an alarm condition, remove and check for dial tone at TVP If dial tone returns, replace PCB

Check 316039 allotter If in an alarm condition, remove and check for dial tone at TJP. Replace as necessary.

(c) Check all registers (316025 and 316302) for an alarm condition. Replace as required.

- 2.7No busy tone (BT) and/or ring back tone (RBT) at all stations - major alarm
 - Repeat Step (b) of paragraph 2.4. (a)
 - Wrong numbers all stations (stationto-station or station-to-feature)
 - (a) Perform all steps of paragraphs 3.4.1 through 3.4.6 of Section V.
 - Wrong numbers all stations (incoming DID trunk calls)
 - Perform all steps of paragraphs (a) 3.4.1 through 3.4.6 of Section V.
 - Insure correct digit outpulsing (h) from central office equipment.

3.0 MAJOR ALARMS

2.8

2.9

3.1 A major alarm will occur if both of the following PCB's fail. The console alarm lamp will glow a bright red and the console buzzer will sound continuously when this happens.

Common Control Circuits

Regular Allotter	316039()122
Standby Allotter	316639()122
Regular Tone Supply	316037()122
Standby Tone Supply	316637()122
Regular Originate Control	316029()122
Standby Originate Control	316029()122

3.2 As noted in Section XII, failure of any regular common control circuit will result in automatic transfer from the defective regu-lar PCB to the standby. The possibility of this not occurring was covered in paragraph 2.5.

On TE-400 assemblies 320136 with the PFCT relay mounted upon the 316062, 3.3 a major alarm will place those stations assigned to PFCT into the PFCT mode and they will be connected directly to the central office equipment.

3.4 Replace all card(s) causing a major alarm.

3.5 On some 130097 TE-400 cabinet assemblies a failure of the ±3VDC supply may cause a major alarm. Consult Equipment Specifications The alarm lamp on the 316301 PCB will be "on" Test/adjust 3VDC as in Section V, paragraphs 2.4 and 2.5

4.0 MINOR ALARMS

4.1 General

A minor alarm will be generated at times when there are no system defects. Depress the reset key when a minor alarm is indicated If

the alarm lamp goes "out" and does not come back "on," it may be assumed that the system is operating normally.

4.2 The locations of the alarm lamps are shown in Fig. 5-1, Section V. The alarm lamps are color coded on some 130097 cabinet assemblies. Refer to paragraph 1.1, Section V.

4.3 If a PCB indicates a minor alarm, replace same with PCB. If the minor

alarm condition remains, further troubleshooting will be necessary.

1 /

4.4 P	enerate minor alarms.
316025 () 122	Register
316029 () 122	Originate Control and Lamp
	Generator
316031 () 122	Camp-On-Busy
316034 () 122	Trunk Allotter
316035 () 122	Attendant Control
316037 () 122	Tones Supply
316039 () 122	Allotter
316041(7)122	T-O Matrix
316067 () 122	Toll Restrictor
316073 () 122	Attendant Trunk 1
316074 () 122	Attendant Trunk 2
316099(1)122	Attendant Trunk 2
316302(1)122	Register

4.5 Failure of the fuse that protects the external 48VDC will cause a fuse alarm on some systems.

On systems with auxiliary power 4.6cabinets, failure of the blower will cause a minor alarm.

On 130097 cabinet assemblies. failure 4.7 of the ±3VDC or external 48VDC will generate a minor alarm.

Originate Control and Lamp Generator 4.8(316029) Trouble

> Indication- Originate control **(**a) and lamp generator 'drifts" into

an alarm condition Depressing reset key cures trouble temporarily but alarm condition returns. (b) Remedy - Retest junctors (Sec-

tion V, paragraph 4 0). Seize each junctor a minimum of five times. When alarm

appears, replace junctor under test.

5.0 STATION TROUBLES

5.1 General

- 5.1.1Always check ±3VDC supplies whenever a station complaint TS received.
 - Check ±3VDC fuses on TE-400 (a) systems using 320136 assemblies
 - Check ±3VDC at TP output on (b) all 130097 assemblies The 5VDC supply and alarm
 - (c) 316301 PCB includes a 3VDC

alarm lamp on TE-400 systems using the 130097 cabinet assemblies.

5.1.2 In the following repair procedures, disconnecting the station under test at

the MDF terminal and connecting a TVP is referred to as "Connect TVP.

5.2

termination.

(a) Connect TVP. (b)

No dial tone -no origination -no

TVP receives dial tone. Dial

TVP from TJP (refer to Section V). Trouble is indicated in station wiring or instrument. Locate and repair.

- Connect TVP. If no dial tone is (c)

card. Retest from TVP. Re-connect station T-R at MDF. Retest from station.

5.3 Dial tone normal - origination normal - station will not ring.

Calling party hears chopped RBT The RU relay (on line PCB) chatters when station is called (ring trip).

> (a) Connect TVP.(b) Call TVP from TJP. TVP rings. Trouble is indicated in station

wiring or instrument.

NOTE: Probable short to ground on "R" lead or "T" lead.

> TVP does not ring. Replace sta-(c) tion line card, then re-connect

station T-R to MDF. Retest from station.

5.4 No origination - termination normal

- (a) Connect TVP.
- (h) Call TVP from TJP. If TVP
- rings, check and repair station wiring or station instrument.

TVP does not ring. Replace line (c)PCB. Retest. Re-connect

station wiring. Retest.

5.5 Origination normal - no termination -BT returned to caller.

> Replace associated 316045 (a) group hunt PCB if station is a

group hunt slave. Retest.

If trouble is not corrected, con-(b) nect TVP and retest. If trouble

is no longer indicated, check station wiring and instrument.

If trouble remains, replace line (c)PCB and retest. Connect station

wiring at MDF and retest

- Wrong numbers received when dialing 5.6 from one station
 - Connect TVP Follow procedure des-cribed in paragraph 3 4, Section V,

5.6.1

but dial from TVP connected to trouble line to TVP's referred to in paragraph 3.4, Section V. Also, dial equipped feature's) digit(s).

> If wrong numbers are no longer (a)received when dialing from the

TVP connected to the trouble line, check station wiring for: high impedance .eakage, more than two ringers across like, and dial pulses from station instrument for 62.5 ±5% break. Repair as needed. Reconnect to MDF. Retest from station.

If wrong number trouble recurs, (b) retest as in Step (a). If wrong

numbers are still received, replace line card and retest. If trouble as still not corrected, repeat all procedures outlined in paragraph 3.4, Section V.

5.7 Station user complains of occasional peiods when dial tone es not available immediately but is available after station has been 'off hook.. for a few moments.

This may be normal. Explain to (a) station user that a register must be available for station to receive dial tone and that during heavy traffic periods a short watt may be necessary to receive dial tone.

(b) Follow instructions outlined in paragraph 4.0, Section V.

5.8 Station user complains of occasional periods when dial tone IS not received. "Raspy" noise (matrix blocking) IS evident "off hook." Dial tone available after short wait.

Explain to station user that a **(a)** junctor must be available to receive dial tone If one IS not available due to usage, a short waiting period will prevail. (b) Follow instructions of para-graph 4.0. Section V.

5.9 Station cannot complete feature call. Busy tone IS returned when feature call is attempted Station can complete local-tolocal calls with no wrong numbers received

Insure that station is not re-(a) stricted. If station IS restricted, inform station user that this is normal.

if station is not restricted, re-(b) place line card and retest If

trouble remains, swap regular and standby allotters If trouble IS corrected, replace standby allotter If trouble is not corrected, perform junctor test (paragraph 4 0, Section V).

5.10 Various stations complain of being "dead" when 'off hook" and remaining that way. Trouble IS intermittent

Using junctor test switch and (a) TJP, seize each junctor 15 times each Note that when only one register IS seized that dial tone IS returned on each seizure When two registers are seized, dial tone will not be returned Replace each junctor that seized two registers

Refer to paragraph 4 0, Section (b)

5.11 Various stations complain of "no ring - no answer."

5.12

5.13

5.14

- (a) Follow instructions outlined in paragraphs 3.3 and 4.0, Section v
- Various stations complain of BT being returned after dialing one or two digits.
 - Follow instructions outlined in (a) paragraphs 3.3 and 4.0, Section
- Various stations complain of weak DT, BT, RBT, and transmission.
 - Follow instructions outlined in (a) paragraph 4.0, Section V.
- Various stations complain of receiving BT when dialing feature digit(s).
 - Perform registers test (para-(a)

graph 3.4, Section V). Also dial feature digit(s) from TVP when performing registers test.

Various stations complain of failure to 5.15 reach attendant.

From TVP dial through attendant (a) trunk No. 1 and then No. 2.

Check VT level. Adjust or replace as required.

5.16 Station receives DT after dialing local station and when station answers a drop-back-to-dial tone is encountered.

> (a) If stations are equipped with key equipment, insure that all sta-

tions have tip) and ring connected to the same corresponding sides of the key

Follow instructions outlined in **(b)** paragraph 4.0, Section V.

5.17 Attendant tine - No Dial Tone

- Place junctor test switch to (a)
- OFF. Remove TJP if connected Refer to paragraph 2.1 of this (b) Section.
- Transmission is noisy and/or weak (local calls only).
 - Perform junctor test (paragraph (a) 4 0, Section V) Replace de-fective junctor.

6.0 CONSOLE TROUBLES

5.18

6.1

- Wrong numbers received when keyed or DSS
 - Replace 316054(14) and connect (a) TVP to station MDF terminals

as follows: X00, X01, X11, X22, X33, X44, X55, X66, X77, X88, and X99.

NOTE: Do not connect to an MDF terminal number corresponding to an unequipped line.

number. Re	(b) epeat fo	Seize trunk by depressing trunk button and key punch TVP line r each station number listed	6.6
adove.	(c)	If trouble is not corrected, re- place 316035(14) with original	
and replace above.	316035	(12) and repeat Steps (2) and (b)	time. Ret
	(d)	If trouble is not corrected, re-	67
Replace 316 above.	035(13)	and repeat Steps (a) and (b)	0.7
	(e)	If trouble is not corrected, in-	
for correct make, and n sary.	adjustn 10 bound	nent (i.e., good follow-through, ce). Repair or replace as neces-	6.8
6.2	Wro atter	ng numbers or BT returned when ndant attempts override.	6.9
	(a)	Repeat all steps of paragraph 6.1.	
	(b)	Replace 316062 and retest.	
6.3	Atter	ndant cannot access attendant ride.	6.10
	(a)	Replace 316062. Retest if	
	(b)	trouble remains. Disconnect all junctors, trunks, and the line PCB's Leave	
disconnected	l PCB's	s in their respective mounting	6.11
Slots.	(c)	Plug in one of the above PCB's	
through each Replace each	ch and h PCB	at a time. Complete a call attempt override from console. that cannot be overridden.	
6.4	Can wron	np-on-busy circuit terminates to ng numbers.	6.12
	(a)	Replace 316031(11) PCB No. 1	trunk only
	(b)	and retest. If trouble is not corrected, re- place 316031(11) PCB No. 3	
	(c)	and retest. If trouble is still not corrected, replace 316031(13) PCB and re-	6.13
test If tro 6.1.	ouble re	mains, perform steps of paragraph	city - no
6.5	Cam acce	np-on-busy circuit cannot be essed	
	(a)	Perform all steps of paragraph	

(b) Disconnect all trunks accessible (Table 2-3, Section II) by COB, except one Perform all steps of COB test procedure (paragraph 5 0, Section VI) with the following exceptions:

TM 11-5805-652-14

- (1) Call through one trunk.
- (2) Use only one station number.
- (c) **Repeat** for each trunk. Replace those that will not allow COB to be accessed.
- Attendant Cannot access any incoming trunk call. Replace the PCB's of the (a) 316035(2) assembly, one **at** a est after each PCB replacement. Attendant cannot answer a specific incoming trunk call. Replace trunk PCB's associated (a) with failing trunk button. Attendant cannot answer incoming call on OP1. (a) Replace 316073. Attendant cannot answer incoming call on OP2. Replace 316074 or 316099 (as (a) equipped). Attendant cannot place a trunk call on hold. (a) Replace associated trunk PCB and retest. Attendant cannot place any trunk on hold. (a) Replace 316035(11). Retest. If trouble remains, replace **(b)** 316035(12) and retest If trouble still is not corrected, (c) replace 316035(13) and retest. Incoming trunk call does not split when answered by attendant (one y) - no talk city and/or no talk extension. Replace defective associated (a) trunk assemblies. Incoming trunk calls (all) do not split when answered by attendant (no talk talk extension). Replace 316035(11) and retest. (a) If trouble does not clear, re-**(b)**
 - place 316035(12) and retest.
 - If trouble does not clear, re-(c) place 316035(13) and retest.

Attendant cannot release from a trunk.

6.14

- Replace 316035(11) and retest (a)
- If trouble remains, replace **(b)** 316035(13).

13-5

6.15 Lamps on attendant console show erratic operation.

 Replace regular originate control 316029(1).

- 6.16 Attendant cannot transfer a call from OP1 to any city trunk.
 - (a) Replace 316073 and retest.
 - (b) if trouble is it corrected, replace 316035(11) and retest.
 - (c) If trouble remains, replace 316035(12) and retest.
 - (d) If trouble is not corrected, replace 316035(13).
- 6.17 Attendant cannot transfer a call from OP2 to any city trunk.
 - (a) Replace (as equipped) 316074 or 316099 and retest.
 - (d) **Perform Steps** (b), (c), and (d) of paragraph 6.16.

6.18 Attendant cannot extend incoming trunk call into conference (conference

- assembly 316077 only).
 - (a) Replace line PCB corresponding to conference trunk accessing number. Retest.
 - (b) If trouble remains, replace 316077 conference assembly. Retest.
 - (c) if trouble remains, replace 316009 attendant trunk No 2. Retest.
- 6.19 Attendant cannot extend a station into conference (conference assembly 316077 only)
 - (a) **Insure that station is not re**stricted **from conference.**
 - (b) Perform Steps (b) and (c) of paragraph 6.18.

6.20 Console trunk button flashes 120 ipm bright and buzzer sounds approximately two times.

(a) Station has disconnected **from** call **before central** office equipment **has** This is normal if EPABX "clears down" before COE.

6.21 Console trunk button flashes 120 ipm bright and buzzer sounds Upon answer, party-in trunk announces intention of out-dialing to the city

(a) Insure that party attempting call **has** not waited longer than ten seconds after dialing trunk access digit(s) This is normal if ten seconds have elapsed.

NOTE: Also insure that central office equipment is not **slow** in providing dial tone.

(b) If party attempting call has not waited longer than ten seconds before dialing trunk access digit(s), replace board A of erratic trunk circuit. Retest.

7.0

- 7.1 Trunk allotter will not "step" (allott) and one trunk/tie line always remains allotted.
 - (a) Replace board A of **trunk/tie** line that is allotted. **Retest**.

7.2 Trunk allotter skips one or more trunk groups as it 'steps'' (allotts). Allott lamp &es not turn ''on.''

- (a) Replace trunk allotter **PCB and** retest.
- 7.3 No trunk in trunk allott group will allott.
 - (a) Remove all trunks PCB's
 - assigned to that group.(b) Replace one trunk assembly at a

time and dial trunk access digit(s), **observing that** one trunk in each group **allotts. Replace board A of** trunk assembly from **allott group in difficulty that causes** return of trouble.

- 7.4 **Station cannot** access transfer junctor (trunk calls only).
 - (a) Insure that stations attempting to use trunks with transfer
- junctor access are not restricted.
 - (b) **Repeat trunk test** for equipped trunks. See Section Vi.
- 7.5 Station can access transfer junctor but falls to transfer, or ES unable to in-

clude or exclude city party from the conversation.

- (a) Repeat Steps (a) and (b) of paragraph 7.4.
- 7.6 Station user can direct distance dial from station that IS toll restricted.
 - (a) **Perform toll restrictor test** (paragraph 4.0, Section Vi).

7.7 ANA station does not ring when console ANA key operated in ANA position and incoming trunk call IS present

(a) Replace ANA 316042 PCB and retest **If trouble IS not cor**rected, replace associated trunk assembly Retest.

7.8 UNA bell does not ring Console UNA key operated in UNA position and incoming trunk call IS present

(a) **Replace UNA 316036 PCB** and retest if trouble remains, replace associated trunk assembly and retest.

7.9 UNA call cannot be answered.

(a) Perform Step (a) of paragraph 7.8.

7 . 1 0 Regular and/or standby allotter

alarms when UNA call is answered.

(a) Replace associated line PCB of answering station.

Section XIV

CONTENTS

		Page
1.0	General	 14-1

TABLES

14-1	Total System Failure	14-1
14-2	Major Alarms	14-2
14-3	Minor Alarms	14-2
14-4	Station Troubles	14-2
14-5	Console Troubles	14-4
14-6	Miscellaneous Troubles	14-5

Section XIV

T R O U B L E D E T E C T I O N T A B L E S

1.0 GENERAL

1.1 The tables in this Section are fur- nished so they may be used as a ready reference for the following types of trouble:		 (c) Minor alarms. (d) Station troubles. (e) Console troubles (f) Miscellaneous trouble.
(a) Total system failure.(b) Major alarms.	1.2	Section XIV should be referred to as noted

Table 14-1

Total System Failure

FAULT: TOTAL SYSTEM FAILURE - Refer to Section XIII, paragraph 2.0.

INDICATION	CHECK	REPLACE/REPAIR
No alarms - complete failure.	AC input. AC circuit breakers Voltage Check Transformer Fuse. Blowers (and Switch)*	As required.
	* 130997 Assembly Only	
316069 Main Alarm and Voltage Check PCB is alarmed	DC supplies. 316070 () 122 PCB 316069 () 122 PCB Refer to Section XIII, paragraph 2.2	As required.
316301 5VDC Supply PCB 15 alarmed	316301 PCB Refer to Section XIII, paragraph 2 3	As required
Valid on 130097 Assembly only		
Stations do not ring. RBT normal No alarms	Ring Generator Fuse Refer to Section XIII, paragraph 2 4	As required
No dial tone - all stations - major alarms	316039, 316037, 316025, 316302 PCB's. Refer to Section XIII, paragraph 2 5	As required
No dial tone - all stations - minor alarms	Refer to Section XIII, paragraph 2 5	As required
No BT and/or no RBT - all stations - major alarms	Tones Supply 316037 PCB Refer to Section XIII, paragraph 2 4, Step (b)	As required
Wrong numbers - all stations complain - (Station-to-Station or Station-to-Feature)	Registers Refer to Section V, paragraph 3 4	As required
Wrong numbers - all stations complain - (Incoming DID trunk calls)	Registers Refer to Section V, paragraph 3 4 and Section XIII, paragraph 2 9	As required.
Attendant cannot answer in- coming trunk call	Console dial for off-normal switch closed	As required

Table 14-2

Major Alarms

FAULT: MAJOR ALARMS - Refer to Section XIII, paragraph 3.0.

Concernance of the local division of the loc	INDICATION	CHECK	REPLACE/RETURN
	Console Alarm Lamp on "bright." Console buzzer "on."	All cabinet PCB's for alarms. Refer to F.g. 5·3 and Section XIII, paragraph 3.0.	PCB's as required.
	PFCT stations connected directly to Central Office equipment.	As above.	As above.

Table 14-3 Minor Alarms

FAULT: MINOR ALARMS - Refer to Section XIII, paragraph 4.0.

INDICATION	CHECK	REPLACE
Console alarm lamp on Dim	All cabinet PCB's for alarms Refer to Fig. 5-1, Section V.	PCB's as required.
Fuse alarm (320136 cabinets only).	48VDC luses on DID-E&M 'Trunks/ Tic Lines	As required.
Blower alarm (320136 cabinets only).	Blower in Auxiliary Power Cabinet.	As required.
Originate Control and ' amp Generator 316029 "drifts" into alarm condition. Can be re- set, but trouble returns	Refer to Section V, paragraph 4.0.	As required.

Table 14-4Station Troubles

FAULT: STATION TROUBLES - Refer to Section XIII, paragraph 5.0

INDICATION	CHECK	REPLACE
No DT No origination No termination.	Instrument. House Cables. Line PCB Refer to Section XIII, paragraph 5 2	As required.
DT normal Origination normal Station will not ring (Ring Trip)	Instrument House Cable. Line PCB Refer to Section XIII, paragraph 5 3	As required.
No origination. Termination normal	Instrument House Cable Line PCB. Refer to Section XIII, paragraph 5 4	As required.
Origination normal No termination	Group Hunt PCB Line PCB. Refer to Section XIII, paragraph 5 5	As required.

Table 14-4 (Cont'd)

INDICATION	CHECK	REPLACE
Wrong numbers.	Instrument (Total no ringers). House Cable. Lune PCB. Register PCB. Refer to Section XIII, paragraph 5.6.	As required.
Occasionally no DT.	Check traffic. Junctors. Refer to Section XIII, paragraph 5.7.	As required.
Occasional periods of no dial tone. Matrix blocking is heard in receiver.	Traffic. Junctors. Refer to Section XIII, paragraph 5.8.	As required.
Unable to complete feature call	Restriction. Line PCB. Allotter. Refer to Section XIII, paragraph 5.9.	As required.
Intermittent periods of no DT. Clears when hook switch is bounced.	Junctor. Register. Refer to Section XIII, paragraph 5.10.	As required.
Intermittent no ring - no answer	Junctor. Register. Refer to Section XIII, paragraph 5.11.	As required.
Intermittent BT after dialing one or two digits.	Junctor. Register. R€fer to Section XIII, paragraph 5.12.	As required.
Intermittent weak DT, BT, RT, transmission.	Junctor. Refer to Section XIII, paragraph 5.13.	As required.
BT on dialing feature digits.	Registers Refer to Section XIII, paragraph 5.14.	As required
Failure to connect through to attendant.	Attendant trunks. Refer to Section XIII, paragraph 5.15.	As required.
Station drops back to dial tone after dialing local station, when local station answers (arop back to dial tone).	Junctor. Refer to Section XIII, paragraph 5.16	As required.
Attendant Line - no DT.	Junctor test switch to off position. Remove TJP. Refer to Section XIII, paragraph 5.17	As required.
Transmission is noisy or weak (local calls only)	Junctor Refer to Section XIII, paragraph 5 18.	As required.
Transmission is noisy or weak (trunk/tie lines).	Trunk/tie lines. Refer to Section V, paragraph 9 0	As required

Table 14-5

Console Troubles

FAULT CONSOLE TROUBLES - Refer to Section XIII, paragraph 6.0.

INDICATION	CHECK	REPLACE
Wrong numbens when attendant keys (DSS) station.	Attendant control PCB's. 316035(12). 316035(14). 315035(14). Key Pad. Refer to Section XIII, paragraph 6.1.	316035(12). 316035(14). 316035(13). Key Pad. As required.
Wrong numbers, or BT returned attendant attempts override. Caution: BT will be returned if station is not busy.	As above, or 316062. Refer to Section XIII, Step (b) of paragraph 6.2.	As above, or 316062.
Attendant cannot access attendant override.	316062. Refer to Section XIII, paragraph 6.3.	As required.
Camp on busy circuit terminates to wrong numbers.	316031(11). 316031(13). Refer to Section XIII, paragraph 6.4.	As required.
Camp on busy circuit cannot be accessed.	316031(2). Trunks. Refer to Section XIII, paragraph 6.5.	As required.
Attendant cannot answer any in- coming trunk.	316035(1). Refer to Section XIII, paragraph 6.6.	As required from 316035(1) assembly.
Attendant cannot answer a specific incoming trunk call.	Trunk Assembly. Refer to Section XIII, paragraph 6.7.	Trunk Assembly.
Attendant cannot answer in- coming call on OP1.	316073. Refer to Section XIII, paragraph 6.8.	316073.
Attendant cannot answer in- coming call on OP2.	316074 or 316099. Refer to Section XIII, paragraph 6.9.	316074, 316099. As equipped.
Attendant cannot place a trunk call on hold	Associated trunk assembly. Refer to Section XIII, paragraph 6.10.	Associated trunk assembly.
Attendant cannot place any trunk call on hold.	316035(11). 316035(12). 316035(13). Refer to Section XIII, paragraph 6.11.	As required.
Incoming trunk call does not split when answered by attend- ant (one trunk only) - no TC, no TE.	Associated trunk assembly. Refer to Section XIII, paragraph 6 12	Associated trunk assembly.
Incoming trunk calls (all) do not split (no TC, no TE) when	316035(11). 316035(12). 316035(13). Refer to Section XIII, paragraph 6.13.	As required.
Attendant cannot release from a trunk.	316035(11) 316035(13). Refer to Section XIII, paragraph 6 14.	As required.
Lamps on attendant's console show erratic operation.	316029. Refer to Section XIII, paragraph 6 15.	316029

Table 14-5 (Cont'd)

INDICATION	CHECK	REPLACE
Attendant cannot transfer a call from OP1 to any city trunk.	316073. 316035(11). 316035(12). 316035(13). Refer to Section XIII, paragraph 6.16.	As required.
Attendant cannot transfer a call from OP2 to any city trunk.	316074. 316099. Refer to Section XIII, paragraph 6.17.	As required.
Attendant cannot extend an in- coming trunk call into confer- ence (conference assembly 316077 only).	316040. 316077 assembly. 316099. Refer to Section XIII, paragraph 6.18.	As required.
Attendant cannot extend a station into conference (con- ference assembly 316077 only).	Station not restricted from conference. 316077 conference assembly. 316099 attendant trunk No. 2. Refer to Section XIII, paragraph 6.19.	As required.
Console trunk button flashes 120 ipm bright, buzzer sounds approximately two times.	Refer to Section XIII, paragraph 6.20.	
Console trunk button flashes 120 ipm, buzzer sounds. Upon answer, party in truck announces intention of out- dialing to the city	Trunk assembly. Refer to Section XIII, paragraph 6.21.	As required.

Table 14-6MiscellaneousTroubles

FAULT: MISCELLANEOUS TROUBLES - Refer to Section XIII, paragraph 7.0.

INDICATION	CHECK	REPLACE/REPAIR
Trunk allotter will not step (allott). One trunk/tie line always remains allotted.	Refer to Section XIII, paragraph 7.1.	Board A of trunk assem- bly, as required.
Trunk allotter skips one or more trunk groups	Trunk allotter. Refer to Section XIII, paragraph 7.2°	Trunk allotter
No trunk in trunk allott group will allott.	Refer to Section XIII, paragraph 7.3.	As required.
Station cannot access transfer junctor.	Refer to Section XIII, paragraph 7.4.	As required.
Station can access transfer junctor, but fails to transfer, and/or is unable to include/ exclude city party	Refer to Section XIII, paragraph 7.5	As required.
User can direct distance dial from station that is toll restricted.	Refer to Section XIII, paragraph 7.6	As required.
ANA station does not ring on incoming trunk call.	Refer to Section XIII, paragraph 7.7.	As required.

Table 14-6 (Cont'd)

INDICATION	CHECK	REPLACE
UNA bell does not ring on in- coming trunk call.	316036 UNA PCB. Refer to Section XIII, paragraph 7.8.	As required.
UNA call cannot be answered.	Refer to Section XIII, paragraph 7.9.	As required.
Regular and/or Standby allotter alarms when UNA call is answered.	Refer to Section XIII, paragraph 7.10.	As required.
Section XV

GLOSSARY OF TERMS

ABR1 ABR2	Attendant's busy tone return.	BUZ	-18 volts sent on BUZ lead to attend- ant's console to operate the buzzer.
AC-ALI AC-AL2	Attendant's control alarm outputs to system main alarm.	BZ	From trunks to attendant's control. At -18 volts to cause attendant's con- trol to gate out 120 inm to buzzer in
AF-AL1	Attendant's feature alarm outputs to		console.
AF-AL2	system main alarm.	BZG	Busy guard relay.
ALE	Attendant's line enable.	CBA1	Camp-on-busy alarm outputs to sys-
ALK	Attendant's line key.	CBA2 CBA3	tem main alarm.
ALL	Attendant's line lamp.	CBA4	
ALO-1 ALO-2	Inputs to attendant 's control from the systems allotters.	CBL	Conference busy lamp
ANA	Assigned night answer.	cc	Code can.
ANF	From ANA circuit to all associated	CCM1 CCM2	From code call circuit to bell or other signal unit
•	on ANF when ANA key in attendant's	CDO	Central district office.
	console is operatea.	CF	Conference.
ANK	From attendant's console to ANA cir- cuit Ground extended from ANA lead	CFF	Call forwarding feature.
	of attendant's console via ANA key to	со	Central office
	ANK lead of ANA circuit.	COB	camp-on-busy
ANM	Pulses in synchronism with the allotter pulse of its associated junctor	COM-1 COM-2	Supplies -18 volts from contacts of attendant's ALF relays in attendant's
APS	Auxiliary power relay signal from voltage check circuit. Controls appli- cation of power to the system busses		attendant's trunk circuits
AR	Attendant release	COS	Class of service May be used in con- junction with feature symbols, TA-COS, PA-COS, etc.
ARS	Attendant's reset. From attendant's reset key to attendant's override cir-	СТ	Cut-through relay
ASU AST ASH	cuit Attendant's stores units, tens, and hundreds Enables the digit stores in sequence.	DBF1 DBF2	Drop back to feature control lead used on attendant's trunk circuits to return extension to attendant's trunk during failure of extension to city trunk
ATE-1	Extends ground via attendant's trunk	DBI	Drop back inhibit.
ATE-2	key to operate attendant's AL relay in attendant's control circuit.	DIA DIB	Inputs on override circuits from trunks and junctors Assigned on a
ATK-1 ATK-2	Attendant's trunk key	DIC DID DIF	coded basis for override access of particular trunk or junctor.
ATL-1 ATL-2	Attendant's trunk lamp	DIE	Outputs from trunk and junctors to
ATR	Attendant release	DIX DIY	override inputs DIA, DIB, DIC, DID, DIE
BL.	Busy lamp	DIZ	
BLF	Busy lamp field.	DJ	Drop junctor
BT	Busy tone (600 Hz Interrupted at 60 ipm)	DJB	Drop back to junctor

DKP	Supplies -18 volts through attendant's digit buttons to the digit stores for	HNST	Hand set.
	digit storage.	IBC	Inter-board connection.
DT	Dial tone (600 Hz continuous tone - 15 volts P/P at supply).	ĸ	From trunks to trunk buttons on con- sole. Operator seizes trunk by
EIT	Extension in trunk. Ground if no ex- tension in trunk and -18 volts with extension in trunk.		to operate KR relays in trank. LU lead maintains ground to keep KR relays operated.
EMA	Extends +3 pulse from override cir- cuits to all local ju nctors and trunks to enable override.	KL	Ground extended from KL through trunk button to K lead on trunk to operate RK relays.
ЕОН	Executive override originate control.	KR	Attendant's trunk access relay.
EOS	Executive override.	LB	Connection from line circuit to
EOT	Executive override.		message registers.
ERA ERB	Code lead inputs on executive over- ride circuit from DIX. DIW, DIY, and	LGA1 LGA2	Lamp generator originate control alarm indication to system main alarm.
ERD FRF	DIZ outputs of junctors and trunks.	LGO-1	Lamp generator output (120 ipm)
ESP	Executive override.	LGO-2	Lamp generator output (60 ipm).
EV	Multiples toll restrictor code detec-	LGO3	Lamp generator output (wink).
	tors together.	LU	Lock up
EGC	Attendant's feature speech gate con- trol.	MA1 MA2	Alarm outputs from system allotter to system main alarm.
FGE-1 FGE-2	Attendant's feature speech gate con- trol.	MC	Extends ground from the executive junctor to the executive override
MA	Positive mark from feature to allotter when a feature IS accessed. Allotter then extends the terminate multiple and		circuit when the register terminate pulse appears.
	sends a DJ mark to the junctor	MJIG	Alarm inputs to system main alarm.
FMI	Feature mark one.	MJIN	Alarm inputs to system main alarm.
FMII	Feature mark two.	ОН	Originate control pulse to line cir- cuits.
GHA	Group hunt access (group hunt circuit accesses slave lines by pulsing the line circuit terminate gate over the GHA lead).	ON	Off normal (indicates attendant's stores are not set in the idle condition).
GHCR	Group hunt extends -18 volts on this lead to inhibit the trap circuit in the regular system allotter	OPT	Operators transmission path for override.
GHCS	Group hunt trap inhibit lead to standby system allotter	ORI	Operator recall inhibit Inhibits operator recall during UNA or ANA operation Also used to inhibit receil or UNA operation when DID
GHU GHT GHH	Units, tens, and hundreds leads of group hunt master numbers		trunks are used and prevents ringing time out on DID trunks.
Н	Hundreds (hundreds divit or mark)	ORU	Extends -18 volts from attendant's control via the UNA kev to the UNA
HDST	Head set		circuit to start the multivibrator
HLD	Trunk hold leads in attendant's con- tral. Depressing hold her extends	ORT	Operator's ring lead from attendant's control to console.
	ground on HLD1 and -18 volts on HLD2	OTT	Operator's tip lead from attendant's control to console.

ovo	Override voice originate.	RSC	Ground is extended on these leads
OVT	Override voice terminate.	rft SFT	from the trunk allotter to the trunk to mark the VT negative when an exten- sion with the proper class of service
PA	Public address.		dials the trunk feature digit.
PCHG	Punching.	RT	Ring tone (600 Hz interrupted at 12 ips switched on for one (1) second and
PFCT	Power failure cut-through.		off for three (3) seconds.
рн	Primary horizontal (inputs to primary matrix from line circuit VO).	RTA	From attendant's console to allotter. Extends -18 volt signal to allotter to lengthen the allotter pulse
P8	Primary second (matrix).	рт.	Crown hunt ainsuit autonda a nagativa
PU	Pulse up. From attendant's control to trunk to mark the trunk VT negative for termination.	KIA	pulse to the system allotter on RTA lead to extend allotter pulse.
PUA	Pulse up lead for ANA to trunks to mark trunk VT negative from termi- nation.	RU	From trunks to ANA or UNA circuit. Extends -18 volts when an incoming call seizes a trunk.
Q1, 2	Strapping points on ring counter in	S	Signal lead. Attendant's did pulses are sent to the trunk via the S lead.
R1, 2 S1, 2 T1, 2	er in size to accommodate larger systems.	SAT	Saturate. When at ground trunk speech gate is off.
QH	Quarternary horizontals (quaternary outlets to trunk and feature VT ter-	SBC1 SBC2	Alarm outputs from tone supplies to system main alarm.
_	nimais).	SEN	Standby enable.
R	King.	SF	Strapping field.
RC	Release city.	SG1	Speech gate leads. Used to extend
RCK	Release city key.	SG2	aftendant's speech leads throughout all attendant's override circuits.
RE	Release extension.	сц	Secondary horizontals (secondary out
REN	Regular enable. Release extendon	511	lets to junctor VT terminals for origination and to tertiary matrix horizontale for termination cross con-
nea	Release extendion.		nects.
RF	Register free.	SIT	Outputs from the standby system allotter.
RG	Ring generator (30 Hz sub-cycle,	SOM	Stop out mark.
DOG	Ring generator start (applies ground	SRK	Standby routine key.
RGS	to interrupter circuit on tone supply to	SZ	Seize relay.
	start ringing).	Т	Tens (tens digit or mark).
RIT	Outputs from the regular system allotter to all circuits requiring an	Т	Tip.
	allotter pulse or mark.	T1 R1	Tip and ring connections for power failure cut-through of selected lines
RK	Reset key.	TT1 TR1 TS1	to trunks.
ROI and ROF	Used as chain leads on UNA circuit. ROI of preceding circuit is connected to ROF of succeeding circuit.	ŘŠÎ TA	Trunk allotter.
A V 4		TΔ	Trunk allott extends ± 18 volts to
RRK	Regular routine key.	IA	trunks for allotting during -18 volt mark from TG or TK at +2 volts when allotted
RS	Register seize.		when anoued.

15-3

TB	Trunk allotter to all trunks.	TQ	Tertiary quaternary (matrix).
TCK	Talk city key.	TRA	Switches negative when trunk is seized. Remains at ground when a trunk is
TCL	Talk city lamp.		allotted.
TDP	Dial pulse lead. Trunk extends dial pulses from extension to transfer junctor over TDP.	TRAP	Connection from all line circuits to system allotters. Trap is fired when a line accesses a feature.
TE	Terminate bus.	TRB	From toll restrictor message
TEK	Talk extension key.		ing marked if a toll restriction circuit is not available.
TEL	Talk extension lamp.	TRC	From monitor to access circuit.
TEO	From attendant's control to city trunks. Supplies -18 volts from attendant's TE relay to city trunk	TDE	Pulses negative when calling party is restricted.
	for talk extension mode.	IKE	to access circuit.
TF	Toll restrictor monitor circuit free indication extended to access circuit.	TRF	Toll restrictor free indication to trunk. Restriction circuit is seized via the TRF lead.
TFA	From toll restrictor to trunk allotter.	TRG	Toll restrictor extends ground to
TG	Trunk group (trunk allotter groups (4) within a group of trunks) - pre-selects		trunk to prevent further outdialing when restricted.
TU	or allotts a trunk for seizure outgoing from EPABX.	TRH	As the extension dials, the first three digits are passed to the toll restric- tion circuit via the TRH.
тн	Trunk hold Ground sent from attend-	TRJ	Extends busy tone from toll restrictor
111	ant's control to trunk via TH lead to put trunk in hold.		to trunk.
TJ	Transfer junctor	TRK	Multiples toll restrictor code detec- tors together.
TJA	Transfer junctor access.	TRS	Toll restrictor free indication to trunk. Remains at +18 volts when a
TJH	Transfer junctor hold. Extends		monitor circuit is free.
	to hold trunk during short hang up (ten-second hold).	TSS	Trunk splitting. From trunk to at- tendant's control to extend -18 volts from trunk to TE relay in attendant's
TJR	Transfer junctor release		control.
ТК	From trunk allotter to all trunks in	TST,MAT	Test matrix.
	all trunks to pre-allott a trunk if a trunk was not allotted when the TG was applied.	TUN	Extends ground from UNA circuit to trunk, causing trunk VT to mark to -18 volts.
TL	Trunk lamp (from trunks to console	U	Units (units digit or mark).
	lamps).	UNA	Unassigned night answer
ТМ	Tens mark (terminate mark - may be from hundreds or tens counter of registers, depending on type of regis- ters and size of system). May be designated as H on some features.	URL1 URL2 URL3 URL4	Outputs from UNA and CC circuits to bell (URL1 and URL2 to UNA, URL3 and URL4 to CC)
	This IS the digit mark input for ac- cessing a feature in conjunction with FM1 or FMII mark	V	Matrix verticals (PS verticals and TQ verticals).
ТМА	Register alarm output to system main	VO	Voice originate
	alarm.	VT	Voice Terminate.

15-4

EPABL SYSTEM DESCRIPTION

I. Line to Line Call Allotter Tone Supply Junctor Register Originate Control Line Matrix Terminate Buss

II.Group Hunting Call Group Hunt

- III. Feature Call-PA Public Address Class of Service
- IV. <u>Line to Trunk Call-Dictation Trunk</u> Trunk Allotter **Dictation Trunk**
- V. <u>Line to Trunk Call City Trunk</u> City Trunk
- VI. <u>Feature Call -Attendant</u> Attendants Feature **Allotter** Attendants Feature Attendants Control city Trunk
- VII. Attendants Line Call Attendants Line

VIII. Incoming Trunk Call City Trunk

Attendants Control Camp on Busy Assigned Night Answer Unassigned Night Answer

- IX, <u>Executive Line Call</u> Executive Junctor Executive Override Applique
- X. <u>Feature Call Conference</u> Conference
- XI. <u>Feature Call Code Call</u> Code Call
- XII. <u>Attendants Override Call</u> Attendants Override



TM 11-5805-652-14 KSP501-101 KEY SYSTEM PRACTICE PART FOUR Page 1, 7-72 CHAPTER 1 K-501 SERIES KEY SERVICE UNITS Sec. 10

Figure 1-1

KSP501-101 Page 2, 7-72

TABLE OF CONTENTS

	Page No.
SECTION 1. GENERAL DESCRIPTION	3 thru 6
SECTION 2. INSTALLATION	7 thru 20
SECTION 3. CIRCUIT DIAGRAMS	21 thru 29
SECTION 4. MAINTEN ANCE SECTION 5. ASSEMBLY AND TEST Figures	30 thru 32 33 thru 55
1-1. Views of K501 Key Service Units	1, 3
1-3 KAOLA Manual Intercom Card	5
1-4. Pushbutton Panel of Key Telephone Set	6
2-1. Layout of Connecting Blocks	8
2-2. Termination of Cables on Connecting Blocks	8
2-3. CO or PBX Line Circuit Options	9
2-4. Manual Intercom Station Signaling	10
2-5. Connections for 10 TEL-TOUCH Stations	12
2-6. Connections for 19 TEL-TOUCH Stations	12
2-7. Mounting of K347 KTU	12
2-8. Addition of K216A or K216A1 KTU	13
3-1. K501 KSU Wiring Diagram	22
3-2. Interrupter Diagram	23
3-3. K207C1 KTU Diagram	24
3-4. K216A1 KTU Diagram	25
3-5. K347C KTU Diagram	26
3 - 6 K400B KTU Diagram	27
3-7. K400D KTU Diagram	28
3-8. K401A KTU Diagram	29
TARLES	
1 - 1 Ordering Information	3
2-1 Terminal Connections for Adding K207C1 KTU	11
2-2. Terminal Connections for Adding K347C KTU	12
2.3 Connections Retween Station Apparatus and	14
Connecting Blocks	14
2-4. Terminal Connections (K501 KSU)	16
2-5. Cable Termination for 66F3 Connecting Ricek	17
2.6 Connection Chart K564 K565 K2564	1,
Z=0. Connection Chart, 1307, 1303, 12307, K2507,	18
2-7. Connection Chart KB36 K831 K854	10
K2830 K2831 and K2854 Talanhana Sata	20
ix2000, ix2001 and ix2004 i cicpitone dets	-0
4-1 Replacement Parts K501 KSU	32

SECTION 1. GENERAL DESCRIPTION

KSP501-101 Page 3, 7-72

ORDERING INFORMATION

	Intercom										
Power Supply	NONE	ROTAR	PUSHBUTTON DIAL								
		9 DIAL CODES	18 DIAL CODES	10 DIAL CODES	19 DIAL CODES						
	6-Line	Capacity Key Ser	vice Units, Wall	Cabinet							
NONE	50100-000-963	50109-000-963	50118-000-963	50110-000-963 50119-000-963							
	6-Line Ca	pacity Key Servic	e Units, Floorsta	nd Cabinet							
NONE	50100-00F-963	50109-00F-963	50118-00F-963	50110-00F-963	50119-00F-963						
180125-001*	50100-0FP-963	50109-0FP-963	50118-0FP-963	50110-0FP-963	50119-0FP-963						
180125-002**	50100-FPG-963	50109-FPG-963	50118-FPG-963	50110-FPG-963	50119-FPG-963						

Key Telephone Units (KTU's)

Code	Description
K-400-00D	KTU. CO/PBX Line Card. One required per CO or PBX Line.
K-401-00A	KTU. Manual Intercom Card. One required per manual intercom circuit.
K-207-0C1	KTU, Rotary Dial Selective Intercom Circuit; 9 Dial Codes.
K-216-0A1	KTU, Transfer Circuit. Used with K-207-0C1 to add 9 Dial Codes.
K-347-00C	KTU, Pushbutton Dial Selective Intercom Circuit; 10 Dial Codes.
K-316-00A	KTU, Transfer Circuit. Used with K-347-00C to add 9 Dial Codes.

*Does not include ringing generator. (For buzzer signaling.) **Includes ringing generator. (For buzzer or ringer signaling.)



FLOORSTAND UNIT WITH POWER SUPPLY. (COVER REMOVED)

FLOORSTAND AND COVER Figure 1-1 --Continued.

TM 11-5805-652-14 KSP501-101 Page 4, 7-72

SECTION 1. GENERAL DESCRIPTION

1. INTRODUCTION

The K1A2 Key Telephone System consists of a basic prewired mounting package designated a Key Service Unit, and is provided to receive and associate "plug-in" units in verying combinations to meet subscribers requirements. The "plug-in" units, designated Key Telephone Units, are new station switching circuits which employ miniature relays and solid state components such as transistors, varistors, diodes., etc., assembled on printed circuit "plug-in" cards.

The system is designed to provide means for signaling a subscriber station associated with a Central Office (CO) or Private Branch Exchange (PBK), for holding that line and for indicating by means of visual signals whenever the line is being called, held or is busy. It also provides for originating and holding outgoing calls. On an optional basis, it provides a common talking-manual signaling intercomminicating system and/or a common talking-dial selective signaling intercommunicating s y s t e m .

2. FEATURES

The following features are provided:

- a. Flashing visual signals on incoming calls.
- b. Common audible signals for incoming calls. (Optional.)
- c. Station audible signals for incoming calls, non-bridged ringers. (Optional.)
- d. Steady visual signals while a station associated with the key telephone system makes a lie busy.
- **e.** Holding calls on a line, while another line is used.
- f. Winking or steady visual signal to indicate a held line.
- g. Removal of the hold on a line when a station associated with the key telephone system seizes the line or when the line is momentartly opened at the CO or PBX.
- h. Operation of the CO or PBX line under local power failure conditions.
- i. Release of visual and audible signals (timeout) in approximately 15 or 30 seconds, on a per-line basis, if the call is unanswered and abandoned.

3. SYSTEMS COMPONENTS

Each key telephone system consists essentially of a K501 Key Service Unit with or without a dial intercommunicating circuit, and with an elextromechanical interrupter. To which is added on the subscriber's premises, a printed circuit type "plugin" line card on the basis of 1 per CO or PBX line and (if required) 1 - printed circuit type "plug-in" manual intercommunicating card, on the basis of 1 per circuit.

The Key Service Units are available with floorstand cabinet with or without Power Supply. Power Supplies are of two types; one with and one without ringing sub-cycle. Approximate dimensions of power supplies are 4%" D x 7" H x 8¹/₄" W.

Dimensions of the Floor Stand and cover are: 13" wide, 11^{14} " deep and 27^{34} " high.

4. DESCRIPTION OF KEY TELEPHONE SYSTEM COMPONENTS

a. The K501 KSU consists of: a die-cast backboard upon which are mounted, 2 "quick connect^{**} connecting blocks, cable clamps and hooks for station cables; a mounting frame hinged and locked to the backboard, and upon which is mounted a card mounting assembly including 6 printed circuit connectors and an electro-mechanical interrupter. The interrupter is common to the entire system and provides *for* lamp flashing, lamp winking and interrupted ringing.

> The backboard is arranged so that the hinge and locking bracket may be transposed to permit the gate opening from right to left instead of left to right as factory assembled.

> The KSU is completely pre-wired between the connecting blocks, interrupter and all connectors for "plug-in" cards. It is designed for wall mounting and is providedwith a light grey fiberglass overall cover. The approximate dimensions are: 16-1/2 in. If x 13-1/2 in. W x 10-1/2 in. D.

b. The K400 KTU is the CO or PBX Line Circuit and consists of a printed circuit board upon which is assembled miniature relays, thermistors, varistors, diodes, capacitors and resistors to provide the necessary switching circuitry. It also provides easily accessible terminals to change optional features by the installer on the subscribers premises. The assembly is approximately 5-1/4 in. L x 3-1/2 in. W x 1-1/4 in. D.

SECTION 1. GENERAL DESCRIPTION

KSP501-101 Page 5, 7-72

- c. The K401A KTU is the Manual Intercommunicating Line Circuit and consists of a printed circuit board upon which is assembled a battery feed inductor, current limiting resistors and a busy lamp relay. The assembly is approximately 5-1/4 in. L x 3-1/2 in. W x 1-1/4 in. D. (See figure 1-3).
- d. The K207C1 KTU provides for a 9-station common talking, dial selective intercommunicating circuit. It is a modular type unit designed to mount within the K501 KSU, and consists of a two-bank selector and wire spring relays and other components, all of which are nounted on a metal panel and wired to a terminal panel having screw type terminals for external connections. The approximate size is 6-15/16 in. H x 5-7/32 in. W x 6-1/2 in. D.
- e. The K216A1 KTU is used in conjunction with a K207C1 KTU to expand rotary-dial selective intercom from 9 to 18 stations. Approximate size is 7"H x 3"W x 6-1/2"D.

- f. The K347C KTU provides for a 10-station common talking, pushbutton dial selective intercom circuit. It is a modular type unit designed to mount within the K501 KSU and consists of a panel and 5 plug-im cards. The approximate size is 7"H x 9-1/4"W x 8-1/2"D.
- g. The K316A plugs into the K347C KTU to expand pushbutton-dial intercom from 10 to 19 stations.

5. OPERATING INSTRUCTIONS

5.1 General

Operation of the Key Telephone System is extremely simple. The push buttons on the station telephones (See Fig. 1-4) serve to switch the talking circuit to any one line. Visual signals indicate incoming calls . . illumjnated buttons insure rapid identification of calls and busy lines, and prevent interruptions of calls in progress.





Figure 1-2. K400(D)962 Key Telephone Unit



Figure 1-3 K401(A)962 Key Telephone Unit

KSP501-101 Page 6, 7-72

SECTION 1. GENERAL DESCRIPTION

5.2 Central Office or PBX Lines

Incoming Call

Incoming calls on Central Office or PBX lines are audibly indicated by the station ringer or by a buzzer. The Key Telephone Sets equipped with signal lamps also provide a visual signal indicating which line has an incoming call waiting to be answered by illuminating (flashing lamps) the button of the as_e-related line at all stations at which the line appears.

An incoming call is answered by pressing the line pick-up button associated with the audible and/or visual signals received and picking up the telephone handset.

Outgoing Call

An outgoing call is originated by selecting an idle line, pressing the associated line pick-up button and picking up the telephone handset.

Holding

When it is **desf**ired to hold an incoming call for the purpose of obtaining information over another line, transferring an incoming calf, etc., the hold button is depressed. This causes a holding bridge to be placed across the line and keeps the lamp associated with the line being held lighted or winking so that other stations will know the line is in use. When the subscriber (or the station to which the call has been transferred) desires to pick up the line again the line pick-up button associated with that line is depressed and the hold condition is automatically released permitting the subscriber (or the person to whom the call was transferred) to converse on the line.

When the wink feature is provided, a visual hold signal in conjunction with the holding feature enables a subscriber to distinguish between a line in a normal busy condition or a line with an incoming call. This "wink" signal feature uses the same signal lamp used for line and busy signals, but has a long "on" period and a short "off" period giving the impression of a wink when the line is in a held condition

Operation With Local Power Failure

Should a local power failure occur, ail lines except the Central Office or PBX lines become inoperative. With Central Office or PBX Lines it is possible during a power failure to make an outgoing call since selection of a line and removal of the station handset permits the subscriber to originate a call regardless of the availability of local power to the Key Telephone System If, during a local power failure, the local ringing supply remains operative, the com-



Figure **1-4.** Push Button Panel of Key Telephone Set

mon audible signals, if provided, will operate during an incoming call but will follow the incoming ringing on the line rather that locking in as would be the case when power is available to the relays of the system. During a complete local power failure it is possible. on Central Office and PBX Lines, to receive incoming signals only if ringers are bridged across the lines to operate on incoming ringing current.

Disconnection

Upon completion of the conversation, returning the handset to the cradle will extinguish the signal lamp and restore the line to the idle (non-busy) condition.

5.3 Intercommunication Line

Incoming Call

An incoming call is answered by pressing the line pick-up button associated with the audible and/or visual signals received and picking up the telephone handset.

Outgoing Call - Manual Selection of Stations

An outgoing call is originated by lifting the telephone handset, pressing the line pick-up button associated with the intercom line, and operating the signaling button associated with the intercom line.

Outgoing Call - Dial Selection of Stations

An outgoing call is originated by lifting the telephone handset and depressing the line pick-up button associated with the intercom line. The desired station may then be selected by dialing the required digit or digits. For a nine station line, a single digit from "2" to "0" IS used for station identification. At the completion of dialing the buzzer or bell assigned to the called station only will ring for a period of from one to three seconds indicating to the called station fails to be answered. If the called station fails to answer the calling station can repeat the audible signal by redialing without hanging up.

TM 11-5805-652-14 KSP501-101 Page 7, 7-72

SECTION 2. INSTALLATION

1. GENERAL

The instructions contained in this section are furnished to aid and assist the installer in the installation of K1A2 Key Telephone Systems.

In preparing for the installation of a K1A2 KTS, the installer should thoroughly read the literature contained in Section 1, so that he will be familiar with the equipment and features of the system.

2. MOUNTING

The K501 Key Service Unit is designed **for** wall mounting and **should be firmly** mounted to a wall at **convenient** height from the floor and in such a position as to allow the equipment gate to swing out to the right to facilitate wiring, **inspection** and maintenance.

The connecting blocks are furnished with aluminum pressure sensitive tapes, preprinted with terminal designations. The tape is one continuous strip and should be cut at the slots on the fanning strips with the sharp edge of the R714B tool (used for terminating the station cable to the connecting block terminals) before the wires are terminated on the blocks.

Figure 2-1 shows the layout of the connecting **blocks and figure 2-3 shows the termination of cable on the blocks.**

The location of the apparatus cabinet should be determined primarily with the following considerations in mind:

- a. A centralized location to minimize cable *runs* to telephone stations.
- b. Accessibility for inspection and maintenance.
- C. Proximity to telephone stations to eliminate possible conversation distractions.
- d. Compatibility with the decor of the premises.
- e. Accessibility of a power outlet for the power supply.

3. WIRING

3.1 General

The K501 Key Service Unit is a factory interwired unit. Connections between station apparatus and the Key Service Unit are made on connecting blocks with quick connect terminals.



Figure 2-1. Layout of Connecting Blocks



Figure 2-2. Termination of Cables on Connection Blocks

3.2 Connecting Blocks

Connections between station apparatus and the connection blocks are shown in Table 2-3 for the K501 Key Service Unit.

Connect an external power source as shown in the tables when power is not provided as part of the package.

The connecting blocks are designed for use with unskinned plastic insulated wire. Insulated conductors are terminated on the clip by insertion into the clip opening, placing the R714B tool over the clip (with the cutoff blade up) and pushing firmly toward the terminal block. This action forces the wire to a SECTION 2.

INSTALLATION

TM 11-5805-652-14 KSP501-101 Page 9, 7-72

fixed position in the clip and at the same time cuts the wire insulation. The force of the blade against the base of the connecting blocks cuts-off the unused end of wire. For wires that are not to be cut off, the "bit" of the R714B tool is reversed so that the cut-off blade is not used.

4. EQUIPPING

4.1 Installation of CO or PBX Line Circuits

K400 KTU's on the basis of one KTU per line required, are plugged Into the printed circuit connectors, usually from left to right. The KTU is oriented in a vertical position with the printed circuitry to the left, inserted in the guides on the mounting frame and firmly inserted into the connector. After all units areinserted the screwsfor the retaining bar are tightened to prevent any KTU's from falling out of the card mounting assembly.

4.2 CO or PBK Line Circuit Options

- Z Short Time-out. (Used with automatic ringing co or PBK).
- W Interrupted Station Audible Signaling
- T Steady Station Audible Signaling
- V Auxiliary Common Audible Signal Circuit
- Y Winking Hold Lamp
- X Steady Hold Lamp

Each K400 is factory wired for options Z, W and Y. If other options are required by the subscriber, the installer should re-wire the option terminals on the printed circuit board to meet specific requirements. (See Figure 2-3).

4.3 Installation of Manual Intercommunicating Circult

The K401A KTU is plugged into the selected printed circuit connector in the same manner as the K400 **KTU**. Strapping of battery feed wires for the K401A KTU is required (See Table 2-3). For suggested wiring for station audible signals see Fig. 2-4.

4.4 Station Apparatus

A connection chart is provided to facilitate connection of the K564 and K565 **Key** Telephone Sets to the Connecting Blocks of the **K501** KSU. (See Table 2-6).









K400(D) KTU

5. POWER SUPPLY

Before the power supply is connected to the K-501 KSU, the power cord should be connected to a **wall service** outlet and the DC voltage to Talking (A) and Relay (B) Battery outputs measured with a DC Voltmeter.' If the voltage obtained is higher than 26 V. DC compensation should be made by moving the tap on the primary of the power supply to a higher value of AC Voltage. Most power supplies are normally tapped at 111, 117 and 123 V. AC.

SPECIAL NOTE:

Since the K400 CO or PBK Line Circuit is designed for grounded ringdown only, it is necessary that the "B" Battery Ground of the Power Supply be connected to **a** good earth ground. This connection is to be made to terminal 45(BG) on Connecting Block "C" of the K501 KSU. TM 11-5805-652-14 KSP501-101 Page 10, 7-72



Figure 2-4. Manual Intercom Station Signaling

SECTION	2.	INSTALLATION	



TABLE 2-1. Terminal Connections for Adding K207C1 KTU

TM 11-5805-62-14 KSP501-101, Page 12, 7-72



6. INSTALLATION

6.1 MOUNTING

If mounting the unit in a wall cabinet, install it behind the mounting frame using the four I 1/4" machine screws and spacers provided. If mounting in a floorstand cabinet, use the 3/8 provided. If mounting in a noorstand cabinet, use the 3/8" machine screws provided and mount the unit behind the frame. In the 501 KSU type will cabinet, it may be necessary to move the connecting blocks to the higher Position. In the 511 KSU type wall cabinet, remove the acoustic podding from inside the cover to allow clearance for the 347 KTU. Certain 511 KSU type cabinets will not permit use of the fiber spacers provided because of interference with the "T" section of the center mounting frame. In such case, use the 1" machine screws provided. Install the four 1" screws from the front of the frame approximately 6" apart so each corner of the 347 panel will rest against the ends of the screws. Install the 1 1/4" screws from the rear. (See figure 2-7)

6.2 CONNECTING

	CAUTION. When connecting, be sure that "A Battery" and "A Ground" leads are terminated on terminals is 9B and 10B respectively. Otherwise damage to the printed cards may result.
(a)	For pure TEL-TOUCH System. (Figures 2-5 and 2-6)
<u> </u>	1. On terminal block B of the K 501 Key Service
	Unit:
	- Strop from 17 to 37:
	-Connect one end of a 4-foot wire, (any color),
	to terminal 3B, spudger into main cable arm and
	bring out in long sub-arm in group of conductors
	for termination on the K 347.
	- For 10 stations, terminate the 4-foot wire on
	terminal 11A of the K 347; for 19 stations,
	terminate the wire on terminal 1A.
	2. Terminate other conductors as shown in figure 2-5 for 10 stations of figure 2-6 for 19 stations.

3. Plug "A-W', "Al", "A2", "A3", and "A4" cards, with printed circuit side to the left, into designated connectors For 19 stations, also plug in the "A5" cord and install the "A3 Auxiliary" card on the "A3" card.

(b) For mixed TEL-TOUCH and Rotary Dial Stations (Table 1)

- Install the 207 C in the normal manner for rotary dial 1. intercom. For 18 stations, install the 216 A in the normal manner also.
- 2. Mount the 347 B near the 207 C and strop between the two units OS shown in table 1. For 18 stations strop between the 347 B and the 207 C and the 216 A OS shown in table 1. Note diodes
- 3. Plug "Al", "A2", "A3" and "A4" cards into their designate connectors. For 18 stations, install the "A3 Auxiliary" card on the "A3" cord.
- 4. Connect T and R leads of rotary dial phones to 1B and 2B respectively on the 207C. Connect T and R leads of pushbutton dial phones to 1B and 2B respectively on the 347 B.

TARIE 1	CONNECTING	K-3/7R to	K_207C a	and $K_2216\Delta$
WARE	2214146614146	******	120123	NU KYZIVA



NOTE: If shorter signal is desired, remove strap from terminals C3 and C5 on A3 Cord.



Addition of a K216A or K216A1 KTU to K50109 Key Service Unit

TM 11-5805-652-14 KSP501-101 Page 14, 7-72

Table 2-3.

CONNECTIONS BETWEEN STATION APPARATUS AND CONNECTING BLOCKS

NOTES:

- 1. Terminate connector or running cables from left to right on clips as required.
- 2. Shop wiring from apparatus is terminated on this clip.
- 3. When K401 A KTU, manual intercom is required, associated BAT. A and GND. A leads are to be strapped to terms. 44 and 43 of Block C.
- 4. These spare terms. may be used for manual signaling. (See Fig. 2-4).

		BLOCK "	Å"					
PPATIIPP	LEAD	TERM.	Γ		CI	IP		
FERIORS	DEEIG.	NO.	1	2	8	4	5	8
	T	1	T	F	I	1	T	T
	R	2				11		
	A	3	11					
LINE 1	AI	4	11		Ш			
	LG	5	11					
	L	6						
	Т	7					11	
	R	8	11			11		11
LINE 2	A	9			11			11
		10			11			11
	1.	12						11
<u> </u>	T	13						
	R	14						
	A	15	11					
LINE 3	Al	16						
	LG	17	11					
	L	18						
	T	19					11	
	R	20			11			
LINE 4		21						
		22.					1	
	1.	24	E	5	E	NOTE 1		
	T	25	ĮŽ	Ž	Ž	Ž	ž	ž
	R	26						
	A	27	11				11	
LINE 5	LA I	28			11			
	LG	29						
	L	30	11					
	T	31						
	R	32	 					
LINE 6	A1	34	! ا					
	LG	35	11					
	L	36						
	T	37	11.					
	R	38						
DIAL	T	39						
SELECTIVE	R	40						
INTERCOM		41						
LINE	16	43						
	L L	44						
· · · · · · · · · · · · · · · · · · ·	в	45						
	R	46						
SPARE	В	47						
NOTE 4	R	48						
	В	49						
	R	50	16	1	18	16	16	18

SECTION 2. INSTALLATION

TM 11-5805-652-14 KSP501-101 Page 15, 7-72

BLOCK "B"							Γ	BLOCK "C"													
PEATURE LEA		LEAD	TERM.	(LU	•	Г	C	L	P	TERM.	LEAD	FEATURE	-							
F	ZATURE	DESIG.	NO.	1	8	3	1	T	2	1	NO.	DESIG.									
		B	1	ľ		I	ľ	T	T		1 2	до АВ	LINE 1	FEED							
DIAL		B 	3								3	AG	LINE 2	TERY							
		R - B	5								5	AG	LINE 3	BAT							
D		R "4" B	6				E E		a Lo		6 7	AB AG	1.717	RCON							
SELECTIVE INTERCOM STATION SIGNALS		R "5"	8				Ĩ		ן ו		8	AB AG		I.I.I.							
		R "6"	10								10	AB	LINE 5	IVIN							
		R "7"	12				μ	4	ł		12	AB	LINE 6	E							
		B R "8"	13 14								13										
		B	15				H	Ц			15										
		R	16	11	17		11	1			10	<u> </u>									
		B R "0"	17	E E	JE	3LO					17	BZ1	COMMON								
	LINE 1	B1, BZ1	19	Ĩ		Ĩ					19 20	RN ST	LEADS								
2	ILINE 2	B1. B21	21	11	ł I	11					21	LF1	CONNECTING								
Z		RI, BZ	22			Ш			i		22	LW1	EQUIPMEN	NT							
12		B1, B21	23	11		Н	1				23	LF2		·-							
		RI, BZ	24			11					24	LW2									
105V	LINE 4	B1, B21	25	11							25 26	T R	LINE 1								
N		R1, 82	27	11							27	T		11							
TAT	LINE 5	RI, BZ	28																28	R	LINE 2
	LINE 6	B1, B21 R1, B2	29 30										29 30	T R	LINE 3	X					
		CA	31								31	Т	LINEA	Ā							
	Common	CA	32	41		П				ł	32	R		Į į							
	AUDIBLE SIGNALING	CA CA	33 34				1	5			33 34	T R	LINE 5	ğ							
	CIRC UIT	CA CA	35]			ľ				35 36	T R	LINE 6								
			37	╇	Ť	ſ	1				37	LG1	LP GND.	Π							
			38	1		1		lł			38	LBI	LP. BAT								
1			39					ł			39	LG2	LP GND								
		1	40		1		1				40	LB2	LP BAT								
			41								41 42	BG BB	RLY GND RLY. BAT	PLY							
	SPARE		43			I			8	1	43	AG	TLK.GND	5							
	NOTE 4		44						2 Z	1	44	AB	TLK.BAT	E							
			45]	45	BG	RLY GND	POWI							
			47								47	RG	R G GND	1							
			48	1				$\ $			48	RB	RG±								
			49 50								49 50	RG RB	BZ GND BZ ±								

TM 11-5805-652-14 KSP501-101 Page 16, 7-72

SECTION 2. INSTALLATION

NOTES

- OALY CLIPS I, 2, 3, 4 AND 5 OF BLOCK "A" AND CLIPS 2 AND 3 OF BLOCK B ARE USEABLE IN TERMINATION OF STATION CONN OR EXTERNAL CONN. BLOCK CABLES
- 2 COMMON AUDIBLE SIGNAL TERMINATION FOR SIX CO OR PEX LINES, STRAP DESMED TERMINALS TOGETHER AND CONNECT SLT-YEL CABLE CONDUCTOR TO THE STRAPPED GROUP AND YEL-SLT CABLE CONDUCTOR TO A VACANT BI TERMINAL
- 3 DEPENDING UPON WHERE THE K401(A)962 KTU MANUAL INTERCOM CARD IS INSERTED, IT IS NECESSARY TO STRAP THE CORRESPONDING AG AND AB CUPS TO CLIPS 43 AND 44 RESP,
- 4 TO USE BUZZERS FOR INTERCOM SIGNALS AND RINGERS FOR CO LINES, CHANGE STRAPPING AS FOLLOWS: A. REMOVE BIN STRAPFROM 48 AND 50 ON BLOCK "C" B. CONNECT RING BATTERY TO TERMINAL 48 ON BLOCK "C". C. REMOVE BRN LEAD FROM TERMINAL 39 OF K207 C KTU D CONNECT AN WRE FROM TERMINAL 50 OF BLOCK "C" TO TERMINAL 398 OF K207 C KTU E CONNECT 18 V T FROM POWER SUPPLY TO TERMINAL 50 OF BLOCK "C"
- 5 to use ringers for intercom signals and buzzers for CO lines, connect 1054 to terminal so and 164 to terminal 40 on block CC

- 6 IF DIAL INTERCOM (K207 C,OR K347 C KTU) IS INCLUDED IN THE PACKAGE, THE CONDUCTORS NORMALLY USED FOR LINE 5 MIST BE CONNECTED FOR DIAL INTERCOM AS SHOWN
- 7. IF RINGERS ARE USED FOR AUDIBLE SIGNALS, CONNECT YEL-SIT TO B, OR BL, TERMINAL, AND SLT-YEL TO R, OR RI TERMINAL IF BUZZERS ARE USED FOR AUDIBLE SIGNALS, CONNECT YEL-BLU TO B OR BI TERMINAL, AND BULVEL TO R OR RI TERMINAL-FOR THE K564 AND K565 IF THE K231, OR K232 TELEPHONE IS USED, THE ORN-YEL AND YEL-ORN CONDUCTORS ARE USED FOR BUZZER CONNECTIONS
- 8 THESE CONDUCTORS ARE USED FOR ACCESS TO DIAL INTERCOM "WITH ALL PICKUP BUTTONS RELEASED"
- 9 CONNECT DIAL INTERCOM STATION AUDIBLE SIGNALING LEADS PER ASSIGNED DIGITS



SECTION 2. INSTALLATION

TM 11-5805-652-14 KSP501-101 Page 17, 7-72

NOTES.

÷

- 1- THIS FEATURE APPLICABLE TO BOTH K565()40 M AND K565()42 M 1- THIS FEATURE APPLICABLE TO BOTH KSGS()40 M AND KSGS()42 M TELEPHONES 2- THIS FEATURE APPLICABLE TO KSGS()42 M TELEPHONE ONLY 3- THESE FEATURES ARE NOT APPLICABLE TO KSG4()40 M TELEPHONE. 4- THESE FEATURES APPLICABLE TO K231()50 M TELEPHONE

	CIRCUIT		CIRCUIT	CABLE CONDUCTOR	KSOIKSU. W/O COML TERM NO	KSOIKSU W/COML TERM NO
	}ł		17			1
		2		BILL WHT	2	2
		2	14	WHT-ORN	3	3
	LINEI	Å	A	ORN - WHT	4	4
				WHT - GRM	5	5
		6		GRN - WHT	6	6
	├ ──── ┤	7	21	WHT - BRN	7	7
		å	28	BRN - WHT	8	8
LINEI LINEZ		9	24	WHT - SLT	9	9
T 10 14 AL 10 1 27 28 24	LINE 2	10	_	- 1	-	-
		11	LG	RED - BLU	H H	11
		12	2L	BLU - RED	12	12
		13	31	RED - ORN	13	13
		14	3R	ORN - RED	14	14
LINE 2 LINE 3 LINE 4		15	3A	RED - GRN	15	15
10 21 3T 38 34 LO 31 4T 48	LINE 3	16	-	- 1	-	-
	1	17	LG	RED - BRN	17	17
		18	3L	BRN - RED	18	18
25 IIIII _ 25		19	4T	RED - SLT	19	19
50		20	4R	SLT - RED	20	20
LINE 4 LINE 5		21	44	BLK - BLU	21	21
4A LG 4L ST SR 5A LG 5L	LINE 4	22	-	-		
		23	LG	BLK - ORN	23	23
		24	4L	ORN -BLK	24	24
1 1 1 1 1 1 1 1 26 1 1		25	5T	ULK - GRN	25	37 UK 39
	1	26	NC	GRN - BLK	20	38 UK 40
AUD SIGNALS BZ AUD. SIG.	LINES	27	5A	DLK - BKN	21	
1 2 3 4 LG L POLE NT RR	LINES	28			-	4 08 47
		29	Lu		29	
			<u> </u>	BLU VE		
	(3) (4)	12		VEL BUT	BLOCK	BLOCK
EXCLUDED	AUX	32	3	ORN - YE	B	A
	SIGNALS	14	Ă	YEL - ORN		-
ET EN EN EUR RIN ON ONI LI N		- 35	+	GRN - YEL	<u> </u>	
	LAMP	36	1 -	YEL - GRN	AUX CIR.	AUX CIR
	PBSIG	37	SG	BRN - YEL	GND TO AUX	APPAR.
	BZ LAMP	36	L2	YEL -BRN	TO BZ LAM	P FIELD
		39	RT	YEL SLT	CONN. BLK	CONN. BLK
	RINGER	40	ER	SLT -YEL	B	8
GGE3 COMMECTING BLOCK	(1)(3)	41	ET	VIO -BLU	(I) TO	(I) TO
	EXCLUSION	42	ER	BLU - VIO	SUCCEEDING	SUCCEEDING
	KS65 ONLY	43	EH	VIO - ORN	EXCLUDED	EXCLUDED
		44	EB	ORN-VIO	PHONES	PHONES
	(2)(3)(4) R	45	RI	GRN - VIO		
	RR	46	TI	VIO - GRN		
	38 ON	47	P3	8RN - VIO	1488 00	149 R
	SPKRPH ONI	46	P4	VIO - BRN	ADA	PTER
	1 1	49	LK	SLT - VIO	TO 55 BW	CONT UNIT
	N	50	AG	VIO -SLT		
			a			
Table 2-5 Cable Te	rmination 1	tor 66E3 (Connecting	Block		

CKT. TERM FEAT. IN		TERMINAL IN SET	NOUNTING CORDS (a)			CONNECTING BLOCK		CONNECTING	
			50-Cond.	42-Cond. 34-Cond.		NUMBER			-
	ISAD		K-565/39	K-565/30	K-564/30				
LINE	DESIG.		phones	phones	phones	SCREW TYPE	AMPHENOL	50-Cond. (a)	40-Cond. (h)
	R	IR	BLU-WHT	BLU-WHT	BLU-WHT	1-1	1	BLU-WHT	BLU
	T	IT	WHT-BLU	WHT-BLU	WHT-BLU	1-2	26	WHT-BLU	WHT
1	Al	IB	ORN-WHT	ORN-WHT	ORN-WHT	1-4	2	ORN-WHT	ORN
	<u>A</u>	ін	WHT-ORN	WHT-ORN	WHT-ORN	1-5	27	WHT-ORN	WHT
LAMP	L	11_	GRN-WHT	GRN-WHT	GRN-WHT	4-1	3	GRN-WHT	GRN
	LG	ILG	WHT-GRN	WHT-GRN	WHT-GRN	4-2	28	WHT-GRN	WHT
	R	2 R	BRN-WHT	BRN-WHT	BRN-WHT	1-6	4	BRN-WHT	BRN
	T	2T	WHT-BRN	WHT-BRN	WHT-BRN	1-7	29	WHT-BRN	WHT
2			SLT-WHT(b)	SLT-WHT(b)	SLT-WHT(b	1-9	5	SLT-WHT	BRN 🎝
	A	2H	WHT-SLT	WHT-SLT	WHT-SLT	1-10	30	WHT-SLT	SL L
LAMP	L	2L	BLU-RED	BLU-RED	BLU-RED	4-4	6	BLU-RED	WHT a
	LG	2LG	RED-BLU				31	RED-BLU	ª
	R	3R	ORN-RED	ORN-RED	ORN-RED	1-3	7	ORN-RED	BLU
	T	<u> </u>	RED-ORN	RED-ORN	RED-ORN	1-8	32	RED-ORN	RED
3			GRN-RED(b)	GRN-RED(b)	GRN-RED(b)	2-1	8	GRN-RED	YEL 🚽
	A	3H	RED-GRN	RED-GRN	RED-GRN	2-2	33	RED-GRN	ORN
LAMP	L	<u>3L</u>	BRN-RED	BRN-RED	BRN-RED	4-6	9	BRN-RED	RED
	LG	<u> </u>	RED-BRN				34	RED-BRN	
	R	4R	SLT-RED	SLT-RED	SLT-RED	2-4	10	SLT-RED	GRN
	T	4T	RED-SLT	RED-SLT	RED-SLT	2-5	35	RED-SLT	RED
A			BLU-BLK(b)	BLU-BLK(b)	BLU-BLK(b)	2-6	11	BLU-BLY	SLT 🖣
	Α	4H	BLK-BLU	BLK-BLU	BLK-BLU	2-7	36	BLK-BLU	BRN
LAMP	L	4L	ORN-BLK	ORN-BLK	ORN-BLK	4-9	12	ORN-BLK	RED
Ditter	LG	4LG	BLK-ORN				37		⁰
	R	5R	GRN-BLK	GRN-BLK	GRN-BLK	2-9	13	GRN-BLK	SLT
	T	5T	BLK-GRN	BLK-GRN	BLK-GRN	2-10	38	BLK-GRN	RED
5			BRN-BLK(b)	BRN-BLK(b)	BRN-BLK(b)	2-3	14	BRN-BLK	YEL 🛶 🖌
	Α	5H	BLK-BRN	BLK-BRN	BLK-BRN	2-8	39	BLK-BRN	BLU
LAMP	L	5L	SLT-BLK	SLT-BLK	SLT-BLK	4-3	15	SLT-BLK	BLK
	LG	5LG	BLK-SLT				40	BLK-SLT	
AUX		1	BLU-YEL	BLU-YEI			16	BLU-YEL	
510	<u></u>	2	YEL-BLU	YEL-BLU			41	YEL-BLU	
(d)		3	ORN-YEL		ORN-YEL(b)		17	ORN-YEL	
		4	YEL-ORN		YEL-ORN(b)		42	YEL-ORN	
HOLD		HL	GRN-YEL		GRN-YEL(b)		18	GRN-YEL	
LAMP		HLG	YEL-GRN		YEL-GRN(b)		43	YEL-GRN	
PB S	IG	SG	BRN-YEL	BRN-YEL	BRN-YEL	3-1	19	BRN-YEL	ORN
BZ LP		L2(c)	YEL-BRN	YEL-BRN	YEL-BRN	3-2	44	YEL-BRN	BLK
R-R1		RR	SLT-YEL	SLT-YEL	SLT-YEL	3-4	20	SLT-YEL	GRN
B-B	1	RT	YEL-SLT	YEL-SLT	YEL-SLT	3-5	45	YEL-SLT	BLK
	R	ER	BLU-VIO(e)	BLU-VIO(e)		3-6	21	BLU-VIO	BRN
KER PHONE EXCLUD	T	ET	VIO-BLU(e)	VIO-BLU(e)		3-7	46	VIO-BLU	BLK
	Al	EB	ORN-VIO(e)	ORN-VIO(e)		3-9	22	ORN-VIO	SLT
	A	EH	VIO-ORN(e)	VIO-ORN(e)		3-10	47	VIO-ORN	B! K
	RI	R	GRN-VIO(e)	GRN-VIO(e)		3-3	23	GRN-VIO	BLU
	TI	RR(c)	VIO-GRN(e)	VIO-GRN(e)		3-8	48	VIO-GRN	YEL
	P3	ON	BRN-VIO(e)	BRN-VIO(e)		4-8	24	BRN-VIO	ORN
	P4	ON1	VIO-BRN(e)	VIO-BRN(e)		4-10	49	VIO-BRN	YEL
<u>a</u>	LK	LI(c)	SLT-VIO(e)	SLT-VIO(e)	••	4-5	25	SLT-VIO	GRN
s -	AG	N	VIO-SLT(e)	VIO-SLT(e)		4-7	50	VIO-SLT	YEL
						the second second			

Table 2-6. Connection Chart, K564. K565. K2564, K2565 Telephone Sets

SECTION 2. INSTALLATION

TM 11-5805-652-14 KSP501-101 Page 19, 7-72

NOTES:

- Colors are designated, only first stripe second.
- b. Spare conductors, tage and store in telephone.
- c. Terminal on telephone network.
- d. Auxiliary term strip in telephone.
- e Exclusion and speakerphone leads not associated with these features, must be disconnected. taped and stored in telephone set when two or more sets are connected in multiple through bridging adaptors.
- f Terminals 28, 31, 34, 37 and 40 are strapped on the mounting cord connector.
- g. These terminals are not used except when the telephone is modified to install an exclusion switch.
- h. If Superior Cable Co. No. 25 x 24 ICRS "Ring Stripe" Cable is used, colors designated become a two color combination, I.E. BLU-WHT, WHT-BLU, ORN-WHT, WHT-ORN etc. The first color is the body and the second the "ring stripe".

TM 11-5805-652-14 KSP501-101, Page 20, 7-72

SECTION 2. INSTALLATION

L	LINES I THROUGH 9				LINES 10 THROUGH 19					
Tell Ser Term	Leod Color	Leas Desig	Conn or Plug Tenn	Tel Set Tem	Lead Color	Lead Desig	Conn or Plug Term			
Line 1 (Blue) Plug	81-W W-81 W-0 G-W W-G	R T L LG	1 26 27 3 28	Line 10 (Blue) Plug	8L-W W-8L W-O G-W W-O	R T A L LG	1 26 27 3 28			
Line 2 (Orange) Plug	8R-W W-8R W-5 BL-R R-BL	R T A or S L LG	4 29 30 6 31	Line II (Orange) Flug	BR-W W-BR W-S BL-R R-BL	R T A er S L LG	4 29 30 6 31			
Line 3 (Graan) Flug	0-R R-0 R-G BR-R R-2R	R T A or S L LG	7 32 33 9 34	Line 12 (Groen) Plug	O-R R-O R-G BR-R R-BR	R T A or S L LG	7 32 33 9 34			
Line 4 (livory) Plug	S-R R-S BK-BL O-BK BK-O	R T A or S L LG	10 35 36 12 37	Line 13 (Ivary) Plug	S-R R-S 8K-BL 0-8K 8K-0	R T A or S L LG	10 35 36 12 37			
Line S ((Slote) Plug	G-BK BK-G BK-BR S-BK BK-5	R T A or S L LG	13 38 39 15 40	Line 14 (Siate) Plug	G-8K 8K-G 8K-8R 5-8K 8K-5	R T A or S L LG	13 38 39 15 40			
Line 6 (Mihite) Plug	81-Y Y-81 87-84 G-Y Y-G	R T A or S L LG	16 41 14 18 43	Line 15 (White) Plug	rl-y y-bl br-bk G-y y-g	R T A or S L LG	16 41 14 18 43			
ijine 7 (Red) Mug	8R Y Y-BR BL-BK BL-V V-BL	R T A or S L LG	89 44 11 21 45	Line 16 (Rod) Flug	8R-Y Y-8R 8L-8K 8L-V V-8L	R T A or S L LG	19 44 11 21 46			
Line 8 (Slocii) Pig 1 4	0-V V-0 G-R BR-V V-DR	R T A or S L, PL, or IT LG, P3 or IR	22 47 8 24 49	Line 17 (Black) Plug	0-V V-0 G-R BR-V V-BR	R A or S L LG	22 47 8 24 49			
Line 9 (Yallow) Plug 2 5	S-V V-S S-W G-V V-G	R T A or S Lor R1 LG or T1	25 50 5 23 48	Line 18 (Yallaw) Plug	V-S S-W G-V V-G S-Y	T Aors L LG R	50 51 45 20			
10 6 15	0-w 0-Y Y-0	Ai SG, LK or Sparse BL, AG	2 117 42	Line 19 (Violet) Plup	9-5 0-47 0-7 7-0	4 A or S L LG	49 12 17 42			
AG	5-Y Y-5	or Spore R or RI B or B1	20 45	Note. Lood o	nion, loud desi ple ano consister	ignations, and g	hg/convector			

Table 2-7. Connection Chart, K830, K831, K854, K2830, K2831, and K2854 Telephone Set

to cont and connecting addie.

TM 11-5805-652-14 KSP501-101 Page 21, 7-72

SECTION 3. CIRCUIT DIAGRAMS

TM 11-5805-652-14 SECTION 3. CIRCUIT DIAGRAMS KSP501-101 Page 22, 7-72

Figure 3-1 K501 Key Service Unit Wiring Diagram

NOTES:

1. When a K401A KTU Manual Intercom IS used, Associated BAT. A and GND A leads are to be strapped to terminals 44 and 43 on Block c



SECTION 3. CIRCUIT DIAGRAMS

TM 11-5805-652-14 KSP501-101 Page 23, 7-72



Figure 3-2. Interrupter Diagram

TM 11-5805-652-14 KSP501-101 SECTION 3. CIRCUIT DIAGRAMS Page 24, 7-72 Figure 3-3. K207C1 KTU Diagram និដែន 2 68 BH-R 5 ETHOR Q.N. 3 .05UF 16e 0-0 STRAP 356 87 20 150 EM I 34 44 EN TOLAN 58 60 10 v . Mio 7A -C 87 **M**4 **\$** M4 (TO SIG. BATT 369 ICA NS I . EN96 . 238 8-12 10 _00_ 68 8x-R 338 0-1 . 550. HA 킜 12A R(2) 44 <u>a(3)</u> P-1882 550 ~ 144 R(4) 82 ~~~ 47A 15A e(S) ATM Milles P NGA 676 R(7) 2 **180** <u>R (R)</u> figa TO 20-26YOC PERSUPPLY **A(6)** <u>24 aga</u> 610 **9-**# Mg 2**9**0 300 300 3/8 220 (20 sþ 112 2:0 368 338 368 468 TO COME STAS

Figure 3-4. K316A1 KTU Diagram



TM 11-5805-652-14 KSP501-101 Page 26, 7-72



NOTES

- Whon using the K-347 B KTU, connect 18(1) and 28(R) on K-207C KTU to 48(1) and 28(R) respectively on the K-347 B KTU. Connect rotary dial phones to the K207 C and pushbutton dial phones to the K-347 B.
- Y without transfer circuit (10 dial codes)
 Z with transfer circuit (19 dial codes)

Figure 3-5. K347C KTU Diagram

SECTION 3. CIRCUIT DIAGRAMS



C.O. OR PBX LINE CIRCUIT K490B KEY TEL. UNIT

Figure 3-6. K400B KTU Diagram

TM 11-5805-652-14 KSP501-101 Page 28, 7-72

SECTION 3. CIRCUIT DIAGRAMS





TECHNICAL NOTE

An incompatibility problem may occur between 1A2 KTS equipped with 400D KTU's and certain CO's or PBX's. A lost call can result, in some cases, when the switching system resultches an established connection while the key telephone station is on hold. If the reswitching sequence opens the loop for an interval greater than that required to release the 400D KTU hold circuit, a disconnection will occur.

The release time of the 400D KTU can be extended to bridge the open intervals generated by these reswitching sequences by applying the appropriate one of two options following. (See figure 2.) Option ZC, delay hold release for electronic switching, consists of adding a 5 MF capacitor across terminals 2 and 3 of the 400D KTU option block.

Option 2D, delay hold release for affices other than electronic consists of adding a 1 MF capacitor across terminals 2 and 3 of the black.

When the existing Z option is provided with ZC or ZD option, the strap between terminals 1 and 2 of the option block should be removed and replaced with one of the expecttor pigtail leads.

Figure 3-7. K400D KTU Diagram

TM 11-5805-652-14 KSP501-101 Page 29, 7-72



Figure 3-8. K401A KTU Diagram

1 GENERAL

The Key Telephone Units utilized as modular components of the K501 Key Service Unit (KSU) have been designed to provide a high grade of service to the subscriber, thus reducing maintenance effort to a minimum. It is the purpose of this section to provide maintenance personnel with information and aids that are useful in the upkeep of the Key Service Units to a state of the highest performance efficiency.

1.1 Periodic Preventive Maintenance

The frequency of this type of maintenance is determined by the local operating company and is influenced by the environmental conditions surrounding the location of the switching apparatus for the Key Service Units on the subscriber's premises.

Essential factors involved in preventive maintenance are:

- a. A visual inspection to determine that wiring is properly dressed and that terminal screws have not become loosened, and that conductors terminated on connecting blocks have not become broken.
- b. Inspection of power supply fuses.
- C. The removal of any accumulated dust or dirt that might affect component operation. This probability is very slight since all of the KTU's are protected by an overall apparatus cabinet cover. The use of a small portable vacuum cleaner should prove usefull in the removal of dust.
- d. Lubrication- The only moving part requiring lubrication is the selector in the K207C **KTU** which requires the application of a few drops of high grade light machine or clock oil applied on the top of the arm assembly immediately underneath the retaining collar. The frequency of this operation should be such that it occurs at least every six months.
- e. Cleaning If an inspection of the selector reveals that some tarnishing or collection of dirt on the segment contacts of the selector has occurred, cleaning with an approved liquid contact cleaner such as John B. Moore, Solvent M-3 (Frigisol), applied with a brush or cloth is recommended.
- 1.2 "On the Site" Trouble Shooting and Repairs

Systematic and regular preventive maintenance schedules should substantially reduce this type of maintenance to the following phases:

- a. A visual inspection to determine possible sources of trouble such as broken wires. loose connections and burned out fuses. In many cases thoroughness in this endeavor will greatly expedite the restoration of normal service to the subscriber.
- b. A system checkout and analysis as outlined in Section 2, may pinpoint the localization of trouble.
- C. A closer analysis of the Key Telephone Unit may be performed with the aid of a Volt-Ohmmeter or other visual or audible trouble shooting device, once the trouble has been localized to a particular KTU.
- d. If tests reveal that some part or component on the K400B or K401A KTU is defective, the entire printed circuit card should be replaced by one that is in good condition and the defective card returned to the shop.

Normally, complete *relay* replacements are not made in field repairs since it has been determined that it is more economical to change out the KTU having the defective relay. However, if a 4000 type relay coil on the K227A KTU is found to be defective, a replacement coil may be installed "on the job". The armature gap may require slight readjustment; therefore, the gag should be checked (see par 4.0) and the necessary *readjustment made*.

If, in the course of the investigation, defective parts such as capacitors, resistors, rectifiers (diodes) etc. are found, they may redly be replaced on the K207C or K227A KTU's by removing the terminal panel assembly on the rear of the K'U, permitting access to the faulty component.

1.3 Shop Maintenance

Shop Maintenance is that phase of the maintenance and service program that is related to the major repairs and servicing of KTU's that are not normally performed in the field. Involved here is the replacement and adjustment of complete relays, selectors etc. on the K207C and K227A KTU's only. In order to promote efficient shop maintenance, an adequate supply of replacement KTU's should be on hand to prevent interruption of subscriber's services. In addition, spare parts may be required to be available on a determined quantity level. See Table 4.2 for list of replacement parts.

It is recommended that defective K400° and K401A KTU's be returned to ITT Telecom for repair and return in any case where the operating company does not have skilled personnel trained in printed circuit repair techniques and the necessary "desoldering" devices required for replacement of components. In the event such skills and devices are accessible, replacement parts are listed elsewhere in this section for this purpose. It should be noted that there are no adjustments required for the miniature relays on the printed circuit cards and the plastic covers for these relays should not be removed

2. MAINTENANCE AIDS

In order to provide thorough and efficient maintenance of the KSU's, the serviceman should have a thorough and workable knowledge of all the circuits involved in the three arrangements and the optional features of the system. This data is included in the manual and is located as follows:

> Functional Schematics, Sect. 3 Circuit Schematics, Sect 3

3 RELAY MAINTENANCE

Proper relay maintenance is essential to continued good performance of the switching circuits utilized in the K207 & K227A Key telephone units of the K501 KSU. This maintenance can be divided into three important phases, namely:

- 1. Periodical Inspection
- 2. Cleaning
- 3. Adjustment or Replacement

Each phase is briefly discussed to provide useful information in the performance of these *services*.

3.1 Periodic Inspection

Properly protected relays generally need little inspection, either visual or by means of gauges, as it is believed that regular routine tests will disclose abnormal relay troubles. However, it is known that dirt and dust are the natural enemies of proper and continued relay performance. Therefore, in such environmental conditions, periodic inspections may be warranted.

3.2 Cleaning

If, due to relay failure or as a result of a periodic inspection, a relay is found to be dirty, it should be properly cleaned. Superficial dust or dirt should be removed by means of a suitable soft brush and collected by means of a .inen cloth or a small portable vacuum cleaner. Adjacent relays or other apparatus should be adequately protected.

The contacts should be cleaned with a "camel hair" brush and then slightly burnished with a small piece of "bond" paper. Care should be taken to not bend, distort or damage the springs. No cleaning agent is required for relays. 3.3 Adjustment or Replacement

Wire Spring Relays, due to their unique construcion, normally require no adjustment in the field Since each group of contact springs are factory pretensioned and will normally retain these characteristics throughout its lifetime, no attempt should be made to apply readjustment to these items.

Should a spring, or set of springs, become damaged or distorted, the entire relay should be replaced and the defective relay returned to the factory through ITT Telecom Regional Offices. In the event that a contact spring has been crossed with another through misuse *or* mishandling of the relay with the cover removed, it may be replaced by the use of an orange stick Insert the orange stick under the contact spring and move the spring away from the stationary spring allowing the spring to fall into its proper position in plastic spring comb. Note: In replacing a make spring, it may be necessary to temporatialy remove the fixing spring from the contact cover. When misplaced springs are restored to their proper position, the fixing spring should be reinstalled and followed by the cover.
TM 11-5805-652-14

KSP501-101 Page 32, 7-72

TABLE 4-1REPLACEMENT PARTS LIST

UNIT NUMBER	COMPONENT	PART NJMBER	REMARKS
к501 КЭЛ	Cable Block Connecting Block Connecting Block Cable Hook Receptacle Designation Strip Connector Cand Retaine. Interrupter Cover THM Screw RHM Screw RHM Screw Hex Nut RHM Screw RHM Screw RHM Screw Stud Retaining Ring RHM Screw	86425 95634 95743 86423 95827 95647-3 95744 95745 86419 95742 86434 65817 58713 56233 66435 54855 68959 79593 79580 86855	Usea in pairs Block "A" Block "B" & "C" Gate Lock Printed Circuit Card Interrupter Conn. Biocs Mtg. 95744 Conn. Mtg. Cable Hook Mtg. 95744 Conn. Mtg. 95745 Conn. Mtg. Cable Block Mtg. Cover Lycking Cover Stud Card Retaines

TM 11-5805-652-14

Page 33

Change: PER ORDER T = 35400 DTD ______ ASSEMBLY AND FEST ______ 5 = .20 = 6.8 _______ K501 KEY SERVICE: UNIT

1. <u>SCOPE</u>

1.1 This specification covers the assembly, wiring and test for all K501 Key Service Units.

2. <u>APPLICABLE DOCUMENTS</u>

2.1 The following ITT Kellogg specifications standards, drawings, and publications form a part of this specification:

Specifications

GM Spec. 716 Inspection & Tests for Key Telephone Units GM Spec. 317 Inspection & Tests for Key Telephone Units

Drawings

190647		K501	Key	Service	Unit,	Assembly	
316100()68	K501	Key	Service	Unit,	Schemati	c
316100)77	K501	Key	Service	Unit	, Circuit	Description
316000)113	K501	Key	Service	Unit,	Terminal	-
	,		·			Connecti	ons

3. REQUIREMENTS

- 3.1 General
 - 3.1.1 Key Service Units when assembled and wired strictly in conformance with their respective drawings shall be capable of **meeting** the requirements of this specification. The inspection and testing of a completely assembled and wired **key** service unit shall assure a commercial product consistent with good workmanship, appearance and operation. Any deviation is subject to engineering department approval.

3.2 Assembly

3.2.1 The K207C Key Telephone Unit (when required) shall be assemblies to the mounting frame and located in pogion as shown on the key service unit assembly drawing. The drawing shows the rear (or terminal) side of the key telephone unit.

aver 5381-1-A

3.2.2 The key service unit cable shall be installed as show on the KSU assembly drawing. Each arm of the cable to be terminated on the K207C KTU (if required) shall be fed through its respective fanning strips from below. If the K207C KTU is not required the cable arms shall be grouped and taped to the lower part of the mounting frame, Cable arms that are to be terminated on the KSU connecting blocks may be temporarily taped to the cable hooks located at the top of the backboard.

3.3 Wiring

3.3.1 Connecting Blocks - The connecting blocks are designed for **use** with unskinned plastic insulated wire. Insulated conductors are terminated on the clips by insertion into the clip opening, placing the R714B tool over the clip (with the cutoff blade up) and pushing firmly toward the terminal block. This action forces the wire to a fixed position **in** the clip and at the **same** time cuts **the** wire insulation, The force of the blade against the base of the connecting block cuts off the unused end of **the** wire. Conductors shall be terminated on the connecting block in accordance with drawing 316100()113.

Caution

For wires that are not to be cut off (going elsewhere) the "bit" of the R714B tool shall be reversed or another tool may be used, with the blade installed inside **the tool** handle.

3.3.2 Printed Circuit Connectors - Conductors in each arm of the cable that are to be terminated on the interrupter and line circuit connectors, are of predetermined length and shall not be appreciably shortened. Approximately 5/16 inch insulation shall be stripped from the end of the conductor. The stripped end shall than be inserted into the terminal eyelet, bent back upon itself, crimped and then soldered per G: 236. Where two or more conductors are to be terminated on a specific terminal, soldering shall be done after all conductors arc installed as above. Conductors shall be terminated on the P.C. Connectors in accordance with the KSC Terminal Connection drawing 316100()113.

- 3.3.3 K207C Key Telephone Unit (when required) -Conductors in each arm of the cable are of g-determined lengths and shall not **be** appreciably shortened. Approximately 1/2 inch insulation shall be stripped from the end of each conductor and the stripped portion then inserted under the **head** of the terminal screw. Care shall be exercised that the wire falls within the terminal cup to preclude cutting of the wire when the screw is tightened. Conductors shall be terminated on the K207C KTU in accordance with the **KSU** Terminal Connection drawing 316100() 113.
- 3.3.4 Cable and Wire Dressing After the cable has been terminated on the connecting blocks, printed circuit connectors and the K207C KTU (if required) the cable shall be hand dressed neatly so as to preclude damage to the cable when the gate of the KSU is closed, Individual conductors terminated on the connecting blocks shall be depressed downward in a "U" shape while those terminated on the printed circuit connectors and the K207C KTU (if required) shall be dressed upward in a "U" shape.
- 3.4 Testing Each key service unit shall be completely tested to determine that it is properly wired and that all **features** will **function in a** normal and proper manner.
 - 3.4.1 Test Equipment Since the Key Telephone Units that might be associated with the K501 KSU are tested separately an operational test using telephones is not necessary.

Each KSU shall be tested in a manner so **as** to assure **that** continuity, **between** all points **of** the KSU as designated on **the** schematic drawing **316100()68**, is in evidence.

Test equipment to be used shall be at the discrimination of the manufacturing test equipment department.

3.4.2 Testing Unterminated Wiring - If the KSU to be tested does not utilize the K207C KTU, certain leads in the cable form are unterminated. Each KSU shall be visually inspected to determine that the proper color conductors for cable arm L, M, N, P & Q are terminated on the corresponding proper terminals on the connecting blocks, interrupter connector and printed circuit connector for circuit 6. (See Dwg. 316100()68).

Page 36

- 4. QUALITY ASSURANCE PROVISIONS
 - 4.1 Tests shall be performed in accordance with **procedures** outlined in 3.4 testing.
 - 4.2 Each Key Service Unit assembled and wired shall be 100% tested.
- 5. PREPARATION FOR DELIVERY
 - 5.1 There are **no** preparation for delivery **provisions** applicable to this specification.
- 6. NOTES

					SH	EET	REC	ORD				т ~	M 11-5805-652-14 Page 37
CHANGES	APP					ISSU	ED S	HEET	S			-	
145UE: Î						ļ	<u> </u>	Į	12-2	12-2		ŀ	TADLE "A"
35798		_					╂───		<u><u> </u></u>	<u> </u>	$\frac{2.1}{3.1}$	3	Assembly
ISSUE 4 VOIDED	L AS								4.3	4.2	4.1	4	Stocklist
ADDED CODES	0.2-			_	_								
5010, 50118				ļ		ļ				┼───			
LWG	24												
ISSUE: 5	g.11~												
T-35935		-							1		1		
RECRAWN.													
1994 E: La	1	N	TT FS										
7-21:333													
ADDED KSP. 347 70-3,-5,		1.		ASS	емв	LY,	WIR	E AN	D TE	ST I	PER (GM 7	17.
-18,-20,-23,		2.		FOR	TE	RMIN	JAL	CONI	VECT	NON	5, RE	FER	TO DWG.
-25,-28,-30,				1	31742	2					-		
-40													
155:7		3.		FOF	SCI	HEM.	ATIC	DIA	GRAI	MS, I	REFE	R T	O DWG.
1-30307 181742 WAS 316100() 120. 181736 WAS 316100() 68. 31741 WAS	Ac	4.		DIM R	ens: Efe:	IONS REN(ARI CE A	: Sho Nd I	OWN : DEN:	FOR [IFIC	ENG CATIC	inef On P	CRING VURPOSES ONLY.
ADDED 181716	5.0												
5-1-72													
ISSUE: 8	1												
	1												
		l .											
1	1												
		I											
	1												
	ĺ												
1	1	1											

TM 11-5805-652-14 Page 38

ASY.	DESCRIPTI ON	CIRCUIT DRAWING NO.	IFIG. REMARKS
190647-101-602 050100 (0 00) 963	Consists of Apparatus Mounting with Cover, Interrupter and Inter- nal Coble. Arranged for: 6 CO or PBX Line Circuits or Manual Intercom Circuits. Wired, but not equipped for Rotary or Tel-Touch Dial Selective Inter- com.	181736	Wire per 181742
190647-102-602 050109(000)963	Same as -101 except equipped for 9-station Rotary Dial Intercom	181736	Wire per 181742
190647-103-602 050110 (000) 963	Same as -101 except equipped for 10-station Tel-Touch Intercom	181736	Wire per 181742
190647-104-601 050118(000)963 190647-105-602 050119-000-963	Same as -101 except equipped for 18-station Rotary Dial Intercom Same as -101 except equipped for 19-station Tel-Touch Dial Intercom	181736	Wire per 181742
190647–10 6–6 02 050100 (00) 963 هو به	Consists of Appar .us Mounting with Interrupter and Internal Cable mounted on a floorstand with cover. Arranged for: 6 CO or PBX Line Circuits or Manual Intercom Circuits Wired, but not equipped for Rotary or Tel-Touch Dial Selective Inter- com.	181736	Wire per 181742
190647-7 050109(F)963	Same as -106 except equipped for 9-station Rotary Dial Intercom	181736	Wire per 181742
190647-8 050110 (F) 963	Same as -106 except equipped for 10-station Tel-Touch Dial Intercom	181736	Wire per 181742
190647 -9 050118 (F) 963	Same as -106 except equipped for 18-station Rotary Dial Intercom	181736	Wire per 181742
190647-10	Same as -106 except equipped for	181736	Wire per 161742

TM 11-5805-652-14 Page 39, ISSUE 1

Asv.		Circuit		
or STKL.	DESCRIPTION	Drawing No.	Fig.	REMARKS
013112. 190647-11 50100(FP)963	Consists of Apparatus Mounting with Interrupter and Internal Cable, mounted on a floorstand with cover. Equipped with Power Supply for Buzzer only audible signals. Arranged for: 6 CO or PBX Line Circuits or Manual Intercom Circuits. Wired but not equipped for Rotary or	161736		Wire per 181742
1 1	Tel-Touch Fial Selective Intercom			
190647-12 50109(FP)963	Same as -11 except equipped for 9-station Rotary Dial Intercom	181736		Wire per 161742
190647-13 5011J(FP)963	Same as -11 except equipped for 10-station Tel-Touch Dial Intercom	181736		Wire per 161742
190647-14 50115/FP1963	Same as -11 except equipped for 18-station Botary Dial Intercom	181736		Wire per ,181742
190647-15	Same as -11 except equipped for 19-station Tel-Touch Dial Intercom	181736		Wire per 181742
190647-16 50100(FPG) 963	Consists of Apparatus Mounting with Interrupter and Internal Cable, mounted on a floorstand with cover. Equipped with Power Supply for Buzzer or Ringer audible signals. Arranged for: 6 CO or PBX Line Circuits or Manual Intercom Circuits Wired but not equipped for Rotary or Tel-Touch Dial Selective Intercom	161736		Wire per 161742
190647-17	Same as -16 except equipped for 9-station	181736		Wire per 181742
190647-18	Same as -16 except equipped for 10-station	161736		Wire per 111742
190647-19 50118(FPG)96	Same as -16 except equipped for 18-station 3 Rotary Dial Intercom	181736		Wire per 161742

TABLE "A"

F - 5 0 4	TABL	Е "А"		
ASY.	DPSODTDTTON	CIRCUIT		
OR STKL.		DRAWING NO.	FIG.	ALAAAA
190647-20	Same as -16 except equipped for 19-station	181736		Wire per 181742
50119(FPG)9	3 Tel-Touch Dial Intercom			
190647-21	Same as -11 except equipped with "CAC"	181736		Wire per 181742
50100(FP)964	Power Supply			
190647-22	Same as -21 except equipped for 9-station	181736		Wire per 181742
50109(FP)964	Rotary Dial Intercom			
190647-23	Same as -21 except equipped for 10-	181736		Wire per 181742
50110(FP)964	station Tel-Touch Dial Intercom			
190647-24	Same as -21 except equipped for 18-	181736		Wire per 181742
50118(FP)964	station Rotary Dial Intercom			
190647-25	Same as -21 except equipped for 19-	181736		Wire per 181742
50119(FP)964	station Rotary Dial Intercom			
190647-26	Same as -16 except equipped with "CAC"	181736		Wire per 181742
50100(FPG)9	4 Power Supply			
190647-27	Same as -26 except equipped for 9-	181736		Wire per 18174Z
50109(FPG)9	4 station Rotary Dial Intercom			
190647-28	Same as -26 except equipped for 10-	181736		wire per 181742
50110(FPG)9	station Tel-Touch Inal Intercom	5 / 9 / 5 /		
190047-29 6011915DC10	Same as -20 except equipped for 18-	181736		wire per 181742
DO0647 20	Same as 26 except services for 10	\$ @ \$ #3 /		
1700-1-50 60110/FDC10	Same as - so except equipped for 17-	191730		wire ber 101/46
190647-31	Consists of Annaratus Mounting with	86.8726		Wire per 101742
50200157044	Interrupter. Internal Cable and Connector	808 (JQ		
	Cable mounted on a flooratand for			
	Buzzer only audible signals. Arranged			
	for:			
	6 CO or PBX Line Circuits or			
	Manual Intercom Circuits			
	Wired, but not equipped for Rotary or			
	Tel-Touch Dial Sciective Intercom			

ISSUE 1:

ISSUE 6:

F - 5 0 4	TABLE	' A "			ISSUE 6:	
ASY.		CIRCUIT				
OR STILL.	DESCRIPTION	DRAWING NO.	FIG.		REPARKS	
190647-32 50209(FP)963	Same as -31 excopt equipped for 9-station Rotary Dial Intercom	181736		Wire per	1\$1742	
190647-33 50210(FP)963	Same as -31 except equipped for 10- station Tel-Touch Dial Intercom	181736		Wire per	181742	
190647-34 50218(FP)963	Same as -31 except equipped for 18- station Rotary Dial Intercom	181736		Wire per	181742	
190647-35 50219(FP)963	Same as -31 except equipped for 19- station Tel-Touch Dial Intercom	181736		Wire per	181742	
190647-36 50200(FPG)9	Consists of Apparatus Mounting with 3 Interrupter, Internal Cable and Connector Cable, mounted on a floorstand with cover. Equipped with Power Supply for Buzzer or Ringer audible signals. Arranged for: 6 CO or PBX Line Circuits or Manual Intercom Circuits Wired, but not equipped for Rotary or Tel-Touch Dial Selective Intercom	181736		Wire per	181742	
190647-37 50209(FPG)9	Same as -36 except equipped for 9-station 3 Rotary Dial Intercom	181736		Wire per	181742	and the second second second second second second second second second second second second second second second
190647-38 50210(FPG)9	Same as -36 except equipped for 10- 3 station Tel-Touch Dial Intercom	181736		Wire per	181742	
190647-39 50218(FPG)9	Same as -36 except equipped for 18- 3 station Rotary Pial Intercom	181736		Wire per	181742	
190647-40 50219(FPG)9	Same as -36 except equipped for 19- 3 station Tel-Touch Dial Intercom	181736		Wire per	181742	rage 4





TM 11-5805-652-14

TM 11-5805-652-14 ISSUE 1 Page 44

F-506																		
		ASSEMBLIES & QUANTITIES																
ITEM	PART BASE	NO. SUB	FGC	DESCRIPTION		101	10	103	104	105	106	107	108	109	1110	1)1	11:	11
						50100()963	50109()963	50110()963	50118()963	50119()963	50100(F)963	50109(F)963	50110(F)963	50118/F)963	50119(F)963	50100(FP)963	50109(FP)963	50110(FP)963
1	86449	1		Apparatus Mounting		1	1	1	1	1								
2	86449	2		Apparatus Mounting							1	1	1	1	1	1	1	1
3	86449	3		Apparatus Mounting														
4	95742	1		Interrupter		1	1	1	1	1	1	1	1	1	1	1	1	1
5	161741	101		Cable Assy.		 1	1	1	1	1	1	1	1	1	1	1	1	1
6	161741	102	•	Cable Assy.					1	1				1	1			
7	87671	1		Cable (Power Supply)												1	1	1
8	86434	1		Cover		 1	1	1	1	1								
9	96099	1		Floorstand & Cover		 					1	1	1	1	1	1	1	1
10	<u>96101</u>	1		Power Supply		 												
11	96101	2		Powe r Supply														
12	180125			Power Supply		 												1
13	180125	2		Power Supply														
14	79400	2		RHM Screw												4	4	4
15	75006		6 01	Connector Cable														
<u>16</u>	K207	<u>C1</u>	962	Key Telephone Unit			1		1			1		1			1	
$\frac{17}{10}$	K216	<u>A1</u>	<u>962</u>	Key ⁷ elephone Unit					1					1				
18	<u>K347</u>	C	962	Key Telephone Unit				1		1			1		1			1
19	K316		<u>962</u>	Key [¬] elephone Unit		 				1					1			
20	180514			Instruction Sheet		 1	1	1	1	1	1	1	1	1	1		<u> </u>	1
21	181741	103		Cable Assy.		 			1					1				
122	151716	101		Wood Block		 						1	1	1	1		1	1
<u> 23</u>	101716	102		Wood Block		 						1	1	1	1	Ļ	1	1
<u>24</u> -	181716	103		Wood Block		 							1	1	1	니	1	1
<u>.5</u>	100773			Wood Screw		 					1	1	1	1	1			1
2:	i'SP-347			Key System Practice				1		1			11		1			1

TM 11-5805-652-14 ISSUE 1 Page 45

																		Į
																		1
																		•
																_		
								AC0	20M		FC	8 (NTIT	IFS			
				A3:	1									-				
PART	NO.			1		114	115	116	117	118	119	120	121	122	123	124	125	124
BASE	SUB	FGC	DESCRIPTION			5	5 5	5	5	5	5	5	5	5	5	Ş	σ	IJ
						;01	01	01	ŝ	<u>°</u>	2	2	2	2	01	10	2	2
			1			18	19	00	60	10	18	19	8	60	0	18(19(ĕ
						Ĥ	मि	F	नि	F	F	E	E	E	E	FI	딉	F I
						<u>d</u>	P	PC	PC	P C	P C	о Ч	6(d	6(d	P)9	éld	9(9	ñ
						6	63)9()9();9(96()9(64	64	64	64	64	96
							~	63	63	53	υ. 3	3				L		4
86449	1		Apparatus Mounting											<u> </u>		'		
86449	2		Apparatus Mounting			1	1	1	1	1	1	1		1	1	ļ		
86449	3		Apparatus Mounting											- <u>-</u> -			1	
95742	1		Interrupter			1		1	1	1	1	÷	Ļ.	1	1	ł÷	1	++
181741	101		Cable Assy.			1	<u> </u>	1	1		1	+				h	1	
181741	102		Cable Assy.			<u></u> ↓ ↓	<u></u> <u></u> + - +		1		1	1	,	1	1	1	lī -	1
87671			Cable (Power Supply)			┝╧	┝╌	1	<u>+</u>			-	<u> </u>		<u> </u>	<u> </u>		
86434	1		Cover			+-	$\vdash_{\overline{1}}$	$\overline{1}$			1	1	1	1	1	1	1	1
96099	1		Floorstand & Cover			┢╧	⊢∸	÷	÷				1	1	1	1	1	
96101			Power Supply		<u> </u>			-					- -					1
96101	$\frac{z}{1}$		Power Supply			11	ī									I—	ļ	
180125			Power Supply					1	1	1	1	1				I	<u> </u>	
79400	2		RHM Screw			4	4	4	4	4	4	4	4	4	4	4	<u> 4</u>	4
75006	<u> </u>	601	Connector Cable													Ļ	 	
K207	CI	962	Key Telephone Unit			11			1		1	 		1		ł÷		
K216	Al	962	Key Telephone Unit			1					1	I.,			 	ł	1	
长347	C	962	Key Telephone Unit			 		 				l÷.		╂──	<u> </u>		ti-	-1
K316	A	962	Key 7 elephone Unit			+	H	l-			-	÷	+	1-	1	1	1	1
180514	1		Instruction Sheet			ł÷	╉┽	ł÷	ŀ,	÷	t	اب	t	ħ	f	1	1	1
181741	<u>p 03</u>		Cable Assy.			╉╧	ł÷	1÷	t	Ħ	h	f	亡	ti	ti-	1	1	
181716	101		Wood Block	·		╉╬	片	15	Ħ	t 🕆	忭	17	1	1	t î	1	1	
181716	102	1	Wood Block	1		+	╉	Ħ	h	†	ti	t i	1 î	T	1	1	1	
181716	103		Wood Block	·}		1†	11	11	i	Ī	Ī	1	<u>L</u> i	1	1	1	Ļ	1
180/13		1	Key System Practice_	t			Li	i_				Li		L	<u>L</u>	J		
	PART BASE 86449 86449 86449 95742 181741 181741 87671 86434 96099 96101 180125 79400 75006 K207 K216 K207 K216 K207 K216 K347 K316 180514 181716 181716 181716 181716 181716	PART NO. BASE SUB 86449 1 86449 1 86449 2 86449 3 95742 1 181741 101 181741 102 87671 1 86434 1 96099 1 96101 1 96102 1 180125 2 79400 2 75006 2 75006 2 75006 2 75006 2 75006 2 75006 2 75006 2 75006 1 180125 1 180514 1 181716 101 181716 101 181716 101 180773 1 180773 1	PART NO. BASE SUB FGC 86449 1 86449 1 86449 2 86449 3 95742 1 1 1 181741 101 1 1 181741 102 87671 1 86434 1 102 1 87671 1 1 1 86434 1 102 1 87671 1 1 1 96099 1 - 1 96101 1 - 1 96102 - 1 - 96103 2 - - 180125 2 - - 79400 2 - - 75006 601 601 - K207 C1 962 - - 180716 A 962 - - 181716 103 - - - 181716 103 - - - <td>STOCKLISTPART NO.BASESUB FGCDESCRIPTIONBASESUB FGCDESCRIPTION864491Apparatus Mounting864492Apparatus Mounting864493Apparatus Mounting864493Apparatus Mounting957421Interrupter181741101Cable Assy.181741102Cable Assy.181741102Cable Assy.876711Cable (Power Supply)864341Cover960991Floorstand & Cover961012Power Supply1801252Power Supply1801252Power Supply1801252Power Supply1801251Power Supply1801252Power Supply1801251Instruction CableK207C1962Key Telephone UnitK316A962Key Telephone UnitInstruction Sheet181716101Wood Block181716102Wood Block181716103Wood ScrewKSP-347Key System Practice</td> <td>STOCKLIST PART NO. DESCRIPTION BASE SUB FGC DESCRIPTION 86449 1 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 1 Interrupter 181741 101 Cable Assy. 181741 102 Cable Assy. 181741 102 Cable Assy. 86434 1 Cover 96099 1 Floorstand & Cover 96101 2 Power Supply 96102 1 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180514 1 Instruction Sheet 181716 103 Ca</td> <td>STOCKLIST BASE SUB FGC DESCRIPTION 86449 1 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 1 Interrupter 86449 3 Apparatus Mounting, 86449 1 Interrupter 101 Cable Assy. </td> <td>STOCKLIST PART NO. DESCRIPTION 114 BASE SUB FGC DESCRIPTION 114 B6449 1 Apparatus Mounting, Gr State State 86449 2 Apparatus Mounting, B6449 3 Apparatus Mounting, Gr 1 86449 3 Apparatus Mounting, B6449 1 Iterrupter 1 181741 101 Cable Assy. 1 1 181741 102 Cable Assy. 1 181741 102 Cable Assy. 1 187671 1 Cable (Power Supply) 1 86434 1 Cover 9 96101 2 Power Supply 1 180125 2 Power Supply 1 180125 2 RiM Screw 4 75006 601 Connector Cable 7 K207 C1 962 Key Telephone Unit 1 K316 A) 962 Key Telephone Unit <t< td=""><td>STOCKLIST PART NO. DESCRIPTION 114 115 BASE SUB FGC DESCRIPTION 114 115 G G G G G BASE SUB FGC DESCRIPTION 114 115 G G G G G G G G G G G B6449 1 Apparatus Mounting 1 1 1 86449 2 Apparatus Mounting 1 1 1 86449 3 Apparatus Mounting 1 1 1 86449 1 Interrupter 1 1 1 181741 102 Cable Assy. 1 1 1 86434 1 Cover - - - - 96099 1 Floorstand & Cover - - - - 96101 2 Power Supply - - - - -</td><td>STOCKLIST ASS PART NO. DESCRIPTION 114 115 11(BASE SUB F6C DESCRIPTION 114 115 11(WIG O D O D BASE SUB F6C DESCRIPTION 114 115 11(Wig Vig Vig Vig O D O <tho< th=""> O <tho< td=""><td>STOCKLIST ASSEM PART NO. DESCRIPTION 114 115 116 117 BASE SUB FGC DESCRIPTION 0</td><td>STOCKLIST ASSEMBLIA BASE SUB FGC DESCRIPTION 114 115 11 (117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 117 118 117 110 110 110 110 110 110 110 110 110</td><td>STOCKLIST ASSEMBLIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 1111 111 1111</td><td>STOCKLIST ASSEMBLIES B C PART NO. I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C With State <thstate< th=""> State State<</thstate<></td><td>STOCKLIST ASSEMBLIES & UAI PART NO. 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 00 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES BOUNNTIT PART NO. DESCRIPTION 114 113 116 117 118 119 126 121 122 122 122 BASE SUB FGC DESCRIPTION 114 113 116 117 118 119 126 121 120 120 121 120 B6449 1 Apparatus Mounting, 50 90 90 90 90 90 90 90 90 90 90 90 90 90</td><td>STOCKLIST ASSEMBLIES & OUANTITIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 122 123 B6449 Apparatus Mounting, SUB FGC COLSPAN= 100 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 110 00 00 10 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. 114 115 114 117 116 114 112 112 112 112 112 112 112 112 112</td></tho<></tho<></td></t<></td>	STOCKLISTPART NO.BASESUB FGCDESCRIPTIONBASESUB FGCDESCRIPTION864491Apparatus Mounting864492Apparatus Mounting864493Apparatus Mounting864493Apparatus Mounting957421Interrupter181741101Cable Assy.181741102Cable Assy.181741102Cable Assy.876711Cable (Power Supply)864341Cover960991Floorstand & Cover961012Power Supply1801252Power Supply1801252Power Supply1801252Power Supply1801251Power Supply1801252Power Supply1801251Instruction CableK207C1962Key Telephone UnitK316A962Key Telephone UnitInstruction Sheet181716101Wood Block181716102Wood Block181716103Wood ScrewKSP-347Key System Practice	STOCKLIST PART NO. DESCRIPTION BASE SUB FGC DESCRIPTION 86449 1 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 1 Interrupter 181741 101 Cable Assy. 181741 102 Cable Assy. 181741 102 Cable Assy. 86434 1 Cover 96099 1 Floorstand & Cover 96101 2 Power Supply 96102 1 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180125 2 Power Supply 180514 1 Instruction Sheet 181716 103 Ca	STOCKLIST BASE SUB FGC DESCRIPTION 86449 1 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 2 Apparatus Mounting, 86449 3 Apparatus Mounting, 86449 1 Interrupter 86449 3 Apparatus Mounting, 86449 1 Interrupter 101 Cable Assy.	STOCKLIST PART NO. DESCRIPTION 114 BASE SUB FGC DESCRIPTION 114 B6449 1 Apparatus Mounting, Gr State State 86449 2 Apparatus Mounting, B6449 3 Apparatus Mounting, Gr 1 86449 3 Apparatus Mounting, B6449 1 Iterrupter 1 181741 101 Cable Assy. 1 1 181741 102 Cable Assy. 1 181741 102 Cable Assy. 1 187671 1 Cable (Power Supply) 1 86434 1 Cover 9 96101 2 Power Supply 1 180125 2 Power Supply 1 180125 2 RiM Screw 4 75006 601 Connector Cable 7 K207 C1 962 Key Telephone Unit 1 K316 A) 962 Key Telephone Unit <t< td=""><td>STOCKLIST PART NO. DESCRIPTION 114 115 BASE SUB FGC DESCRIPTION 114 115 G G G G G BASE SUB FGC DESCRIPTION 114 115 G G G G G G G G G G G B6449 1 Apparatus Mounting 1 1 1 86449 2 Apparatus Mounting 1 1 1 86449 3 Apparatus Mounting 1 1 1 86449 1 Interrupter 1 1 1 181741 102 Cable Assy. 1 1 1 86434 1 Cover - - - - 96099 1 Floorstand & Cover - - - - 96101 2 Power Supply - - - - -</td><td>STOCKLIST ASS PART NO. DESCRIPTION 114 115 11(BASE SUB F6C DESCRIPTION 114 115 11(WIG O D O D BASE SUB F6C DESCRIPTION 114 115 11(Wig Vig Vig Vig O D O <tho< th=""> O <tho< td=""><td>STOCKLIST ASSEM PART NO. DESCRIPTION 114 115 116 117 BASE SUB FGC DESCRIPTION 0</td><td>STOCKLIST ASSEMBLIA BASE SUB FGC DESCRIPTION 114 115 11 (117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 117 118 117 110 110 110 110 110 110 110 110 110</td><td>STOCKLIST ASSEMBLIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 1111 111 1111</td><td>STOCKLIST ASSEMBLIES B C PART NO. I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C With State <thstate< th=""> State State<</thstate<></td><td>STOCKLIST ASSEMBLIES & UAI PART NO. 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 00 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES BOUNNTIT PART NO. DESCRIPTION 114 113 116 117 118 119 126 121 122 122 122 BASE SUB FGC DESCRIPTION 114 113 116 117 118 119 126 121 120 120 121 120 B6449 1 Apparatus Mounting, 50 90 90 90 90 90 90 90 90 90 90 90 90 90</td><td>STOCKLIST ASSEMBLIES & OUANTITIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 122 123 B6449 Apparatus Mounting, SUB FGC COLSPAN= 100 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 110 00 00 10 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. 114 115 114 117 116 114 112 112 112 112 112 112 112 112 112</td></tho<></tho<></td></t<>	STOCKLIST PART NO. DESCRIPTION 114 115 BASE SUB FGC DESCRIPTION 114 115 G G G G G BASE SUB FGC DESCRIPTION 114 115 G G G G G G G G G G G B6449 1 Apparatus Mounting 1 1 1 86449 2 Apparatus Mounting 1 1 1 86449 3 Apparatus Mounting 1 1 1 86449 1 Interrupter 1 1 1 181741 102 Cable Assy. 1 1 1 86434 1 Cover - - - - 96099 1 Floorstand & Cover - - - - 96101 2 Power Supply - - - - -	STOCKLIST ASS PART NO. DESCRIPTION 114 115 11(BASE SUB F6C DESCRIPTION 114 115 11(WIG O D O D BASE SUB F6C DESCRIPTION 114 115 11(Wig Vig Vig Vig O D O <tho< th=""> O <tho< td=""><td>STOCKLIST ASSEM PART NO. DESCRIPTION 114 115 116 117 BASE SUB FGC DESCRIPTION 0</td><td>STOCKLIST ASSEMBLIA BASE SUB FGC DESCRIPTION 114 115 11 (117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 117 118 117 110 110 110 110 110 110 110 110 110</td><td>STOCKLIST ASSEMBLIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 1111 111 1111</td><td>STOCKLIST ASSEMBLIES B C PART NO. I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C With State <thstate< th=""> State State<</thstate<></td><td>STOCKLIST ASSEMBLIES & UAI PART NO. 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 00 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES BOUNNTIT PART NO. DESCRIPTION 114 113 116 117 118 119 126 121 122 122 122 BASE SUB FGC DESCRIPTION 114 113 116 117 118 119 126 121 120 120 121 120 B6449 1 Apparatus Mounting, 50 90 90 90 90 90 90 90 90 90 90 90 90 90</td><td>STOCKLIST ASSEMBLIES & OUANTITIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 122 123 B6449 Apparatus Mounting, SUB FGC COLSPAN= 100 00 00 00 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 110 00 00 10 00 00 00 00 00 00 00 00 00</td><td>STOCKLIST ASSEMBLIES & OUANTITIES PART NO. 114 115 114 117 116 114 112 112 112 112 112 112 112 112 112</td></tho<></tho<>	STOCKLIST ASSEM PART NO. DESCRIPTION 114 115 116 117 BASE SUB FGC DESCRIPTION 0	STOCKLIST ASSEMBLIA BASE SUB FGC DESCRIPTION 114 115 11 (117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 118 116 117 117 118 117 110 110 110 110 110 110 110 110 110	STOCKLIST ASSEMBLIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 BASE SUB FGC DESCRIPTION 114 115 116 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 118 117 1111 111 1111	STOCKLIST ASSEMBLIES B C PART NO. I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C BASE SUB FGC DESCRIPTION I14 I15 II (I17 II8 II9 I2C With State State <thstate< th=""> State State<</thstate<>	STOCKLIST ASSEMBLIES & UAI PART NO. 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 00 00 00 00 00 00 00 00 00 00 00 00 00	STOCKLIST ASSEMBLIES BOUNNTIT PART NO. DESCRIPTION 114 113 116 117 118 119 126 121 122 122 122 BASE SUB FGC DESCRIPTION 114 113 116 117 118 119 126 121 120 120 121 120 B6449 1 Apparatus Mounting, 50 90 90 90 90 90 90 90 90 90 90 90 90 90	STOCKLIST ASSEMBLIES & OUANTITIES BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 121 122 123 BASE SUB FGC DESCRIPTION 114 115 116 117 118 119 120 122 123 B6449 Apparatus Mounting, SUB FGC COLSPAN= 100 00 00 00 00 00 00 00 00 00 00 00 00	STOCKLIST ASSEMBLIES & OUANTITIES PART NO. DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 112 122 122 122 122 BASE SUB FGC DESCRIPTION 114 115 116 117 116 117 116 117 116 117 110 00 00 10 00 00 00 00 00 00 00 00 00	STOCKLIST ASSEMBLIES & OUANTITIES PART NO. 114 115 114 117 116 114 112 112 112 112 112 112 112 112 112

506		•																	
·				STOCKLIST					AS	SEN	BL	ES	8	QUA	NTI	TIES	}		
	PART	NO.															—	Γ	Γ
ITEM	BASE	SUB	FGC	DESCRIPTION		۱۹ ۱۹	27	128	129	130	131	132	133	134	135	136	137	138	13
							50109(FPG)964	50110(FPC)964	50118(FPC)964	50119(FPC)964	5C200(FP)963	50209(FP)963	50210(FP)963	50218(FP)963	50219(FP)963	50200(FPC)963	50209(FPC)963	50210(FPC)963	512181FPC1963
1	86449	1		Apparatus Mounting	 	+	-									<u> </u>	<u> </u>		<u> </u>
2	86449	Z		Apparatus Mounting		<u> </u>	1	1	1	1	-								
3	86449	3		Apparatus Mounting	 				-		1	1	1	1	1	1	1	1	1
4	95742	1		Interrupter	 		ī		1	1	ī	1		1	1	1	$\frac{1}{1}$	1	ΙŤ
5	181741	101		Cable Assy.	 1	+	ī	1	1	$\frac{1}{1}$	1	1	1	ī	ī	1	1	1	1
6	1917.11	02		Cable Assy.	 				ī	<u>1</u>	-	-	-	1	i	·	-	<u> </u>	1÷
7	87671	1		Cable (Power Supply)	 		1	1	ī	1	- 1 - I	1	1	i	$\frac{1}{1}$	1	1	1	1
8	86434	1-T		Cover	 					<u> </u>	-			-		·	<u> </u>		1÷
9	96099	$\frac{1}{1}$		Floorstand & Cover	 		1	1	1	1	1	1	1	1	1	1	1	1	$\overline{1}$
10	96101	$\frac{1}{1}$		Power Supply	 		<u> </u>		<u> </u>		-				-		-		F
11	96101	2		Power Supply	 	-+	$\overline{1}$	1	1	1									\vdash
12	180125	1-1		Power Supply	 <u> </u>		<u> </u>		- <u>-</u>	<u> </u>	1	1	1	1	1				
13	180125	2		Power Supply	 											1		7-	h
14	79400	2		RHM Screw	 	-	4	4	4	4	4	4	4	4	4	4	4	<u>.</u>	Â
15	75006		601	Connector Cable		-+	<u> </u>				Ť	1	1	1	1	<u> </u>	1	1	Ť
16	K207	CI	962	Key Telephone Unit	 	-+	1		1			<u>i</u>	-	i	<u> </u>	<u> </u>	1	<u> </u>	fi
17	K216	Al	962	Key Telephone Unit	 		I		1					i l		-	-		Ť
18	K347	C	962	Key Telephone Unit	 		-	1	- - -	1			1	-	1	—		1	÷
19	K316	A	962	Key Telephone Unit	 					1					i			-	-
20	180514	1		Instruction Sheet	 	-1	1	1	1	i	1	1	1		1	1	1	1	1
21	181741	03		Cable Assy.	 	-+	i	ī	ī	ī	1	i	i	i	i	1	i	$\frac{1}{1}$	f
22	181716	101		Wood Block	·		1	1	1	1	ì	1	1	i	ī		1	1	ĥ
23	181716	102		Wood Block	 	-+	$\frac{1}{1}$	1	1	1	1	1	1	1	1			-i-	T
24	181716	103		Wood Block	 		<u>i</u>	i	÷1	T	1	1	Ť	1	H	τ	-i-	÷-	1
2.5	180773	1		Wood Screw	 			÷	Ť	i	$\overline{1}$	<u>i</u>	$\overline{1}$	†	Η	-	-i-	i i	Ηİ
21.	KSP-347			Key System Practice	 · · · · · ·		-	i		ī			î l	Ť,	Ť	-		-i-	ب

- - -- - -

F-506															
				STOCKLIST			Γ	A	SSEN	BLI	ES 8	QUA	NTI	TIES	;
	PART	NO.			I	1						T	T		
ITEM	BASE	SUB	FGC	DESCRIPTION			140								L
							50219(FPG)963								
1	86449	1		Apparatus Mounting		1	1-					ł			
2	86449	2		· Apparatus Mounting											
3	86449	3		Apparatus Mounting	1	1	1								
4	95742	1		Interrupter			1								ł
5	151741	1 01		Cable Assy.			1								İ.
6	181741	102		Cable Assy.	1		1								
7	87671	1		Cable (Power Supply)			1				1			1	
8	86434	1		Cover		1									
9	96099	1		Floorstand & Cover	1		1						1		ł
10	96101	1		Power Supply											l I
11	96101	2		Power Supply											
12	180125	1		Power Supply											
13	180125	2		Power Supply			1								
14	79400	2		RHM Screw			4				1	1			
15	75006		601	Connector Cable			1				ļ				
16	K207	CI	962	Key Telephone Unit											ĺ
17	K 2 16	Al	962	Key Telephone Unit		1									ļ
18	K347	С	962	Key Telephone Unit		T	1								l I
19	K316	Α	962	Key Telephone Unit	I	1	1				1				
20	180514	1		Instruction Sheet			1								İ
21	181741	103		Cable Assembly			1					1			
22	181716	101		Wood Block	1		1								
23	181716	102		Wood Block	I	1	1				ł				ĺ
24	181716	103		Wood Block		1	11				1				
25	180773			Wood Screw			1					1			1
21	KSP-347			Key System Practice			1								



FIG 1 K501 KEY SERVICE UNIT



NOTES:

- I. ORLY CLIPS 1, 2, 3, 4 AND 5 OF BLOCK "A" AND CLIPS 2 AND 3 OF BLOCK 8 ARE USEABLE IN TERMINATION OF STATION GONN OR EXTERNAL CONN. BLOCK CABLES.
- 2. COMMON AUDIBLE SIGNAL TERMINATION FOR SIX CO OR POX LINES, STRAP DESIRED TERMINALS TOGETHER AND CONNECT SLT-YEL CASLE CONDUCTOR TO THE STRAPPED GROUP AND YEL-SLT CASLE CONDUCTOR TO A VACANT B1 TERMINAL.
- 5. DEPENDING UPON WHERE THE K40LAJ962 KTU MANUAL INTERCOM CARD IS INSERTED, IT IS NECESSARY TO STRAP THE CORRESPONDING AG AND AB CLIPS TO CLIPS 43 AND 44 RESR,
- 4. TO USE BUZZERS FOR INTERCOM SIGNALS AND RINGERS FOR CO LINES, CHANGE STRAPPING AS FOLLOWS: A. REMOVE BRN STRAP FROM 48 AND 50 ON BLOCK "C" B. CONNECT RING BATTERY TO TERMINAL 48 ON BLOCK "C", C. REMOVE BRN LEAD FROM TERMINAL 398 OF BLOCK "C", O. CONNECT A WIRE FROM TERMINAL 50 OF BLOCK "C" TO TERMINAL 398 OF K207 C KTU E. CONNECT 18 V.2 FROM POWER SUPPLY TO TERMINAL 50 OF BLOCK "C"
- 5. TO USE RINGERS FOR INTERCOM SIGNALS AND BUZZERS FOR CO LINES, CONVECT 105V2 TO TERMINAL 50 AND 18V2 TO TERMINAL 48 ON BLOCK C

- 6. IF DIAL INTERCOM (K207 C, OR K347 C KTU) IS INCLUDED IN THE PACKAGE, THE CONDUCTORS NORMALLY USED FOR LINE 5 MUST BE CONNECTED FOR DIAL INTERCOM AS SHOWN.
- 7. IF RINGERS ARE USED FOR AUDIBLE SIGNALS, CONNECT YEL-SLT TO B, OR B1. TERMINAL, AND SLT-YEL TO R, OR R1 TERMINAL IF BUZZERS ARE USED FOR AUDIBLE SIGNALS, CONNECT YEL-BLU TO B OR B1 TERMINAL, AND BLU-YEL TO R OR R1 TERMINAL-FOR THE K564 AND K565 IF THE K231, OR K232 TELEPHONE IS USED, THE ORN-YEL AND YEL-ORN CONDUCTORS ARE USED FOR BUZZER CONNECTIONS.
- @ THESE CONDUCTORS ARE USED FOR ACCESS TO DIAL INTERCOM "WITH ALL PICKUP BUTTONS RELEASED".
- CONNECT DIAL INTERCOM STATION AUDIBLE SIGNALING LEADS PER ASSIGNED DIGITS. 9.



a				<u>بر</u>
nk)	K501 KEY SERVICE UNIT TERMINAL CONNECTIONS	SHEET OF 2	180514	- 4

N	эт	ES									
1 -	•	THIS	FEATURE	APPLICABLE	то	вотн	K565()40 M	AND	K565 () 42 M

- TELEPHONES 2- THIS FEATURE APPLICABLE TO K565 () 42 M TELEPHONE ONLY
- 3- THESE FEATURES ARE NOT APPLICABLE TO K564()40M TELEPHONE 4- THESE FEATURES APPLICABLE TO K231() 50 M TELEPHONE

	CIRCUIT		CIRCUIT	CABLE CONDUCTOR COLOR	K50 KS'I W/O COML TERM NO	K50IKSU W/COML TERM NO
		· 1	IT	WHT - BLU	· · · · · · · · · · · · · · · · · · ·	I
		2	IR	BLU - WHT	, 2	2
		, 3	IA	WHT ORN	3	3
	LINE	• 4	AI	ORN - WHT	' 4	4
		5	LG	WHT - GRN	5	5
		6	IL	GRN - WHT	6	6
			21	WHT - BRN	· - 7	7
		Ŕ	28	BRN - WHT	8	8
LINE I LINE 2		· .	24	WHT - SLT	9	9
	LINE 2			-	-	-
IT IR IA AI LG IL 2T 2R 2A			16	PED - BLU	11	11
		11	21		12	12
			· ···· · · · ···	BED - ABN	13	13 -
		15	31	ADD - ORN	14	14
		14	34	ORN - RED	. 15	15
	LINE 3	15	5 A	RED - GRN	. 15	-
G 2L 3T 3R 3A LG 3L 4T 4R	LINE 5	15	_		- 17	.7
* * * * * * * * *		17	Lo	RED - BRN	17	19
		. 18	. 3L	BRN - RED		10 .
	25	19	4T	RED - SLT	19	19
	50	20	4R	SLT - RED	20	20
LINE 4 LINE 5		21	· 4A	8LK - 8LU	· 21	, 21
LG 4L 5T 5R 5A LG 5L	LINE 4	22	-		-	-
		. 23	LG	BLK - ORN	23	23
		24	4L	ORN - BLK	. 24	24
	ac	25	5T	BLK - GRN	25	37 OR 39
	20	26	5R	GRN – BLK	26	38 OR 40
AUD SIGNALS AUD SIG		27	5A	BLK - BRN	27	-
BZ ALC I PR D PT PP	LINE 5	28	·	·	-	·
		29	LG	BLK - SLT	29	41 OR 43
		30	5L	SLT -BLK	30	42 OR 44
		31	5	BLU - YEL	•••••	• • •
	(3) (4)	32	6	YEL - BLU	BLOCK	BLOCK
EXCLUDED	AUX	33	3	ORN - YEL	B	B
	SIGNALS	34	4	YEL - ORN	• -	:
ET ER EH EB R RR ON ONI LI N	Ĩ I I I	35		GRN - YEL		•
		36	_	YEL - GRN	AUX CIR	AUX CIR
	DB SIG	- 37		BRN - YEI	GND TO AUX	
• • • • • • • • •		* '10 "		VEL - BON	TO BZ LAN	PFIFLO
	DE LAMP	30		TYFI -SIT	CONN BLK	CONN BLK
	RINGER	35			A B	R
66 E 3 CONNECTING BLOCK		. 40	י די		'UN TO '	in To
	(1)(3)	41			SUCCEEDING	SUCCEEDING
	EXCLUSION	42	ER			EVCLUDED
	I K565 ONLY	43	EH		EXCLUDED	EXCLUDED

43

44

45

46

47

48

49

50

-- !

•

•

K565 ONLY

(2)(3)(4) R

SPKRPH. ONI

3 B

RR

ON

LI

N

EB

RI

ТΙ

Ρ3

P4

LK

AG

ORN-VIO

GRN - VIO

VIO - GRN

BRN - VIO VIO - BRN

SLT - VIO

VIO - SLT

PHONES PHONES

CABLE THRU

TO 55 BW CONT UNIT

1488 OR 1498 ADAPTER

ÎT IR IA 1

LINE 2 LG 2L 3T + 1 +

١

44

ET ER EH





NOTES I REDUCE ABOVE TO APPROX DIMENSIONS. 2 LETTEPING TO BE BLOCK 3 LABEL TO HAVE ADHESIVE BACKING.

TOLER	NOTED	WIP.ING	LABEL			
FRACTION	<u>+</u>	EAK-I & EBK-I				
ANGLE	±	DESIGN R3	DWN RS 12-11-0			
SUPERSEDI	ES	1	C 191183			





KEY SYSTEM PRACTICE Operating Company Series KSP512-100 Page 1, 1-71

CHAPTER 2 K-512 SERIES KEY SERVICE UNITS



Overall Dimensions with Cover, Approximately 17" High, 26" Wide and 10" Deep

Figure 1. Basic K512 Key Service Unit, Cover Removed includes:

K584B or K584C KTU, 13-Line Panel.

190478, Interrupter.

180180, Equipment Cabinet with Cover.

Connecting Blocks.

Internal Wiring.

KSP512-100 Page 2, 1-71

K-512 SERIES KEY SERVICE UNITS

1.0 IDENTIFICATION

The K-512 series of Key Service Units are 13line capacity, factory packaged and pre-wired K1A2 systems Each unit consists of on apparatus mounting to accept 13 plug-in KTU's, an electro-mechanical interrupter, connecting blocks, internal wiring and a fiberglass cover. Ten different packages are offered as follows:

- (1) K51200-00-963 KSU, Basic unit, wired for but not equipped with rotary or T-T dial selective intercom.
- (2) K51200-0PG-963 KSU Same as item (1) except includes 180715-001 power supply assembly.
- (3) K51209-00-963 KSU, Same as Item (1) except equipped with K-207C1 rotary dial intercom selector (9 dial codes.)
- (4) K51209-OPG-963 KSU Same as item
 (I) except equipped with K-207C1 rotary dial intercom selector and 180715-001 power supply. (9 dial codes)
- (5) K51210-00-963 KSU. Same item (1) except equipped with K-347C T-T dial intercom selector. (10 dial codes.)
- (6) K51210-OPG-963 KSU. Same as item (2) except equipped with K-347C T-T dial intercom selector and 180715-001 power supply (10 dial codes.)
- (7) K51218-000-963 KSU. Same as item (1) except equipped with K-207C1 rotary dial intercom selector and K-216A1 intercom transfer circuit. (18 dial codes,)
- (8) K51218-OPG-963 KSU. Some as item (1) except equipped with K-207C1 rotary dial intercom selector, K-216A1 intercom transfer circuit and 180715-001 power supply. (18 dial codes.)
- (9) K51219-000-963 KSU. Some as item (I) except equipped with K-347C T-T dial intercom selector and K-316A intercom transfer circuit.. (19 T-T dial codes.)
- (10) K51219-OPG-963 KSU Same as item (1) except equipped with K-347C T-T dial intercom selector, K-316A intercom transfer circuit, and 180715-001 power supply.
 (19 T-T dial codes.)

Approximate overall dimensions of the K512 KSU's are 17" high X 26" wide X 10" deep

2.0 APPLICATION

The K512 series Key Service Units are used to provide Key Telephone Systems, with or without dial intercommunication, in applications where more than six CO or PBX lines are required Each K512 KSU utilizes a K584 type panel which has thirteen 18-pin connectors for installation of K400, K400D or K401A Key Telephone Units. Each K400B or K400D KTU is a printed circuit cord and provides the necessary facilities for pickup and hold on one CO or PBX line. Each K401A KTU provides o manual intercom circuit.

Handle plug-in KTUs carefully to avoid damage to printed wiring and delicate components.

3.0 INSTALLATION

The K512 series Key Service Units are wall mounted Incoming CO or PBX lines and Station or Distribution Cables are terminated on Connecting Blacks A, Band C as illustrated an Drawing 180532-101-120, per notes I, 2 and 3 thereon A ten conductor cable having 22 AWG wire and one pair of 18 AWG wire (for 10 VAC connections) is recommended far power and is terminated on Block C. The K400 or K401 type KTUs are installed in the appropriate connectors OS required. Dial Intercom Leads, (T and R) and (L and LG) are terminated on Block D and Audible Signal leads (B and R) ore terminated on Block E. Intercom signaling digits are assigned depending upon the type K512 KSU being installed.

A rotary dial-selective intercom circuit for 9 or 18 dialing codes may be added to the K51200 KSU by installing o K-207C1 KTU for 9 stations; or by installing a K-207C1 KTU and a K-216A1 KTU for 18 stations. A T-T dial-selective intercom circuit for 10 or 19 dialing codes may be added to the K51200 KSU by installing a K-347C KTU for 10 stations, or by installing a K-347C plus a K-316A KTU for 19 stations.

The connecting blocks in the KSU provide only five station connection points far each CO or PBX line coming into the system. In the likely event that more that five stations are to be connected to \circ line, a secondary distribution point must be provided or some stations must be connected in multiple.

CAUTIONS

- (I) Minimum size power supply recommended is Lorain RT3B or equivalent.
- (2) A maximum of 200 lamps (51A) can be supplied from the system.
- (3) A maximum of 50 lamps (51A) can be supplied from any one of the four interrupter contacts. The four interrupter contacts supply four line groups: 1-3, 4-6, 7-9, and 10-13.
- (4) A maximum of 20 lamps (51A) can be supplied from any one line.



Figure 2. Terminal Connection for K512 type KSU

TM 11-5805-652-14 K S P 5 1 2 - 1 0 0 Page 4, 71-1



KEY SYSTEM PRACTICE Operating Company Series

Figure 3. K584B Power Connections and Fuse Designation.





Figure 4. K584B Internal Wiring

T M 11-5805-652-14

KSP512-100-Page 6, 1-71



Figure 5. K584C Power Connection and Fuse Designation





Figure 7. Schematic, K207C1 KTU

KSP512-100 Page 9, 1-71



Figure 8. Schematic, K216A1 KTU



Figure 9. Block Diagram, K347C KTU

KSP512-100 Page 11, 1-71

5.0 MAINTENANCE

Maintenance of the K512 type KSU is primarily limited to o check for broken wiring at the connecting blocks on the backboard and at the KTU terminals.

A further check should be made to determine the status of fuses on the external power supply and the fuse panel of the K584 KTU. In each case, the current rating of the fuse is stomped on the panel OS well OS on the fuse.

Full use of the KTU schematics listed in Section 4.0 should be utilized for trouble shooting these units. Field repair of printed circuit cords used in the K584 and K347C Key Telephone Units is not recommended, defective cords should be returned to the factory for repair or replacement

6.0 REPLACEMENT PARTS

KSU OR KTU		DESCRIPTION
K512 KSU	079601-101 190478-101 180532-101 095634-101 095743-101 095647-108 095647-109	Cover Assembly Interrupter Cable Assembly Connecting Block (A, B, &C) Connecting Block (D & E) Designation Strip (Block A, B & C) Designation Strip (Block D&E)
К584В КТU К584С КТU	095744-101 190477-101 180051-101 180051-102 180051-103 108851-104 190473-104 190473-107 180049-101 180586-101	P.C. Card Connector Connector (Interrupter) Fuse, 1, 2 A Fuse, 1–1/3 A Fuse, 2A Fuse, 5A Connector _able) Connector _able) Connector [(5848 Program Plug) Plug, Program (K5848) Plug, Program (K584C)
K207 C KTU	079677-103 096055-101 180175-110 180175-105 180175-106 079700-106 190175-102	Selector Network Relay (A) Relay (B & T) Relay (C) Capacitor
K216 A KTU	180175-103 180175-112 083777-102	Relay (RL) Relay (TR1) Diade (CR1)
K347 C KTU	180296-101 1980299-101 180252-101 180253-101 180254-101 180256-101	P.C. Card Connector "A00" P.C. Card "A1" P.C. Card "A2" P.C. Card "A2" P.C. Card "A3" P.C. Card "A4" P.C. Card
K316 A KTU	180255-101 180275-101	"A-3X" P.C. Card "A-5" P.C. Card



NOTES I-REDUCE ABOVE TO APPROX. DIMENSIONS 2-LETTERING TO BE BLOCK 3-LABEL TO HAVE ASHESIVE BACKING

	 TOLERANCES UNLESS NOTED	WIRING L	ABEL
		EAK-4	
	ANGLE ±	DESIGN P.E.	DWN P.E.
	SUPERSEDES:		C 191
l. 3			



KEY SYSTEM PRACTICE

CHAPTER 3

K-584B AND K-584C KEY SERVICE UNITS

TM 11-5805-652-14

KSP 584-00B KSP 584-00C Page 1, 8-70



KSP584-00B KSP584-00C Page 2, 8-70

K-584B and K-584C 13-CELL CARD MOUNTING PANEL IDENTIFICATION AND INSTALLATION

1.0 IDENTIFICATION

1.1 PURPOSE OF EQUIPMENT

The K-584 type panels are designed for use at loge centralized installations of K-1A2 Key Telephone System arrangements Each panel will accept thirteen K-400 type or K-401A plug-in KTU's and one 190478 interrupter. Approximate dimensions ore 23" W X 4" H X 4 1/2" D. The panels mount on any frame structure designed for 23" mounting plates.

The K-584C supersedes the K-584B, (discontinued June 1970)

1.2 DESCRIPTION

All line and station wiring is factory wired from the connectors to three SO-contact quick-connect plugs numbered 1, 2 and 3. Power supply wiring, interpanel strapping and miscellaneous circuits ore field connected to the screw terminal field.

Lamp fusing is divided into lamp flash (LF), lamp wink (LW) and lamp steady (LS). For lamp functions, the panels ore divided into three groups of three lines each and one group of four lines. No more than 50 lamps con be supplied from any group, or the capacity of the interrupter contacts will be exceeded. Fusing and terminal assignments of the K-584C panel differs extensively from the K-584B panel. *(See Table 1)

Optional programs allow rearrangement of lamp distribution and fusing within or external to the panel. These options ore selected as follows.

- K-5848 Three 34-contact connecton lettered A, B and C serve as receptacles for the single program option plug provided
- K-584C- One 18-contact connector serves OS o receptacle for the combined Program A/Program C plug.

PROGRAM "A"

When the K-584B panel Is used alone, the program plug is placed in receptacle A. When the K-584C panel Is used alone, the program plug is inserted so that "Program A" may be mod from the plug top. Under this arrangement, fusing for on average of 17 lamps per line circuit is provided not to exceed 50 lamps per interrupter contact.

PROGRAM "B" (Not provided on K-584C)

With the program plug in receptacle B of the K-584B panel and the 10-volt ac input to the Interrupter changed to ground, the entire output of the interrupter is used to synchronously drive auxiliary (slave) relay equipment. All lamp flash and lamp wink functions, including those of the master panel, are served from auxiliary relay contacts.

PROGRAM "C"

With the program plug in receptacle C in the K-584B panel, or Inserted so "Program C" may be read from the plug top in the K-584C panel, half the output of the interrupter, (LF1, LF2, LW1, and LW2, leads), Is used to power on average of eight lamps per line within the panel. The remaining interrupter lads, (LF3, LF4, LW3, and LW4) may be used to power up to 100 lamps, In succeeding panels not equipped with on Interrupter.

CAUTION

REMOVE POWER CORD FROM OUTLET BEFORE CHANGING PROGRAMS TO PRECLUDE POSSIBILITY OF BLOWING FUSES.

KSP584-00B KSP584-00C Page 3, 8-70

3.0 INSTALLATION

3.1 MOUNTING

The K-584B and K-584C panels ore arranged to mount on 22-3/8 inch mounting centers such as provided by the following.

- (a) Any standard commercial 23 inch rock.
- 79600 apparatus mounting. (Will accomodate three K-584 panels if center horizontal bar (b) is removed.)
- 180180 apparatus cabinet. (Will accomodate three K-584 panels if center horizontal bar (c) is removed.)
- 96053-1 floor type apparatus cabinet (Ambox) Will accomodate a maximum of six K-584 (d) panels.
- 96053-2 floor type apparatus cabinet (Ambox). Will accomodate o maximum of ten K-584 panels. (e)

3.2 CONNECTIONS

LINE AND STATION CONNECTIONS (a)

Only three station or key cables can

be connected directly to a 584-type panel, so o master distribution point is normally required. Connections from the panels to the distribution point can be made by one 75-pair cable terminated at one end with three SO-contact connectors or by three 25-pair cable each terminated at one end by o SO-contact connector. Recommended cables ore:

- (1)
- (2) (3) (4)
- 75006()601, 6-feet long, 75 pair, three connectors 25025()601, 25-feet long, 25 pair, one connector 25050()601, 50-feet long, 25 pair, one connector. 25100()601, 100-feet long, 25 pair, one connector

The connector cable(s) must be brought in from the left rear of the K-584B and C panels for connection to plugs 1, 2 and 3.

(b) POWER CONNECTIONS

Power connections for the K-584B and C panels are made to the screw terminal field Be sure each fuse in the panel is the correct rating specified for the circuit and that lamp limitations of the interrupter ore not exceeded.

The K-584B and C panels may be intermixed in key system arrangements (See figure 4.)

TARIF 1	FUSE	DISTRIBUTI	ON OF	K-584R	and K-584C Panels
IADLU I.	LOOP	DISTRIDUT		N-J04D	and \mathbf{N} -JOHC I and \mathbf{N}

}	1			584B			584C			
]	PROGRAM A		PROGRAM B		PROGRAM C		PROGRAM A		PROGRAM C	
GROUP	FUSE	CONN SERVED	FUSE	CONN SERVED	FUSE	CONN SERVED	FUSE	CONN SERVED	FUSE	CONN
LFI	14	1-3	22	1-3	14	1-6	18	1-3	18	1-6
LF2	16	7-9	24	7-9	16	7-13	17	7-9	17	7-13
LF3	18	4-6	26	4-6	18		16	4-6	16	
LF4	20	10-13	28	10-13	20		15	10-13	15	
LW1	13	1-3	21	1-3	13	1-6	13	1-3	13	1-6
LW2	15	7-9	23	7-9	15	7-13	14	7-9	14	7-13
LW3	17	4-6	25	4-6	17		11	4-6	11	
LW.	19	10-13	27	10-13	19		12	10-13	12	
LST	9	1-3	9	1-3	9	1-6	10	1-3	10	1-6
LS2	10	7-9	10	7-9	10	7-13	8	7-9	8	7-13
LS3	11	4-6	11	4-6	11		9	4-6	9	
LS4	12	10-13	12	10-13	12		Γ,	10-13	7	

TM 11-5805-652-14

KSP584-00B KSP584-00C Page 4, 8-70

TABLE II FUSE ASSIGNMENT 584B 584C FUSE CAPACITY CIRCUIT CAPACITY CIRCUIT 1 10V ac or dc 5A 1-1/3 A A Bot. LW1, LW2 LF1, LF2 2 5A 10V ac or dc 1-1/3 A 8 Bat. LW3, LW4 LF3, LF4 3 **Busy Tone** 1-1/3 A B Bat. 1/2A Interrupter Motor Supply (ac or dc) 4 1-1/3 A A Bot. 1/2A 5 1-1/3 A A Bat 5A 10V ac or dc Lamp Wink 3&4 Lamp Flash 3&4 6 1-1/3 A A Bot. 5A 10V ac or dc LW1, LW2 LF1, LF2 1-1/3 A 1-1/3 A 7 B Bat 2A 2A 10V ac or de LS4 LS2 8 B Bot. 9 2A 10V ac or dc LS1 4 LS3 ĩÔ LS2 LSI LW3 LW4 LS3 11 12 L54 LWI LWI LW2 14 LFI LW2 15 LF4 LF3 16 LF2 17 LW3 LF2 LF3 LW4 18 2A 10V ac or de LF1 1/2A Interrupter Motor 19 1/2A 1/2A 1/2A Supply (ac or dc) Busy Tone AT 105V ac (óA) 105V ac (RN) 20 21 LF4 LWI 22 23 LFI 1/2A LW2 24 LF2 25 LW3 26 LF3 LW4 10V ac or dc LF4 AT, 105V ac (6A) 27 28 2A 29 1/24 30 1/2A 105V ac (RN)

TABLE III. LAMP DISTRIBUTION

CAPACITY	USED ALONE PROGRAM A	AS FIRST PANEL E/W INTERRUPTER PROGRAM B (584B ONLY)	AS FIRST PANEL E/W INTERRUPTER PROGRAM C	AS SECOND PANEL WITHOUT INTERRUPTER PROGRAM C
LAMPS	200	200•	100	100
AVERAGE LAMPS PER LINE	17	17•	0	8

* Entire output of interrupter used to drive outiliary relays. Long flash and long wink functions are served from an external source.
T M 11-5805-652-14 K S P - 584-00B K S P - 584-00C Page 5, 8-70

BLOCK C 4-TYPE CONNECTING BLOCK terminal On K 584 KTU TERMENAL ON K 584 KTU BLOCK A 66-TYPE CONNI BLOCK TERMINAL ON K SH4 KTU BLOCK 8 64-TWE CONNECTING BLOCK CINCUIT LEAD LEAD CROAT LEAD DESIG TERMAI- COLOR TERMAI- CON-NAL CLIP G-W BINDER NAL NECTOR TERMI- CUP BL-W SINDER NAL NECTOR TERMI- COLOR TERMI- CON-NAL CLIP O-W BINDER NAL NECTOR 28 1 27 2 28 3 27 CO T MEX R
 H-WO
 O
 W-H
 W-H

 H-WO
 O
 W-H
 W-H
 W-H

 W-WO
 G
 W-H
 W-H
 W-H
 W-H

 W-WO
 G
 W-H
 W-H
 W-H
 W-H
 W-H

 W-WO
 G-W-H
 R-G
 R-H
 R-G
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 R-H
 24 CO T FIEX R 26 **3**8 T 27 TA AI IG IG LINE
 17
 4
 17
 17

 17
 4
 30
 31
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 7
 33
 6
 32
 33
 10
 33
 33
 10
 33
 33
 10
 33
 10
 33
 10
 33
 10
 33
 10
 33
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10
 10< 4 30 5 31 6 32 7 33 8 34 9 34 9 35 RG 10 RC 10 CO T 11 RBX R 12 T 13 11 12 13 14 15 12 UNE 12
 A
 13

 A1
 14

 IG
 17

 IG
 17

 IG
 17

 IG
 17

 IG
 17

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 18

 IG
 <t STA LG Al LG L RG L. L III RG 19 RC 20 PAX R 22 R 20 21 21 22 23 24 25 26 27 10 36 CO T PRX R CO T RK R R A 2 12 30 13 39 UNE 13 1**1941** 3 ž. 14 10 RBF 52 -1 3 37 77 79 40 4 STA AG 495R4 10 (DEAD-ENDED AND STORED BEHIND KTU) CO. 1 K-584B ₹ 100-201 100-201 100-201 44 45 44 47 48 47 48 50 74 UNE 3 ONLY 22 5 22 22 22 22

TABLE IV. LINE AND STATION CONNECTIONS TO K-584B or C PANEL

KEY SYSTEM PRACTICE

KSP584-00B KSP584-00C Page 6, 8-70



Figure 4. Block diagram showing system arrangements.

KSP854-00B KSP584-00C Page 7, 8-70 -24V BAT (8) **()** 41 POWER SUPPLY GRD (B) 42 -24V BAT (A) 34 POWER SUPPLY (NOTE 1) GRD (A) 7) 35 . 10 VAC or DC \mathcal{T} 43 LAMP POWER SUPPLY (7) 36 FOR MAX 100 LAMPS CKTS 1-3 AND 7-9 GRD 7) 44 10 VAC or DC 7) 45 LAMP POWER SUPPLY FOR MAX 100 LAMPS **(**) 38 CKTS 4-6 AND 10-13 GRD **(7)** 46 MB Ø 13 (7) 33 INTERRUPTER MOTOR SUPPLY MG \mathcal{T} 12 18 VAC, 105 VAC or DC **(**) 31 GRD AUDIBLE SIGNAL POWER SUPPLY **(**) 25 13 VAC, 105 OR DC D 26 **(**) 30 15 **(**) 29 AT, 105 VAC **()** 10 TO 105 VAC SUPPLY GRD (B) <u>™</u> GRD RO \bigcirc 9 BZ OR BT Ø 27 TO BUSY TONE SUPPLY GRD 28 Ú) TO K6A KTS BZ1 or BT1 11 Ø (NOTE 2) ST (A or TO) NOTES: 14 (/ LF 1. Provide only when using KTU's that 2 (/ require -24V Bat (A), such as K-401A

 Ground supplied to ST lead from connecting circuit must be ground associated with the supply used to drive the interrupter motor.

KTU. Connection as shown serves all

connectors on panel.

Figure 5. K-584B Panel Equipped With Interrupter (Panel not used to control other panels) PROGRAM PLUG IN RECEPTACLE A

KSP584-00B KSP584-00C Page 8, 8-70











PROGRAM A

TM 11-5805-652-14 KSP 584-00B, KSP 584-00C, Page 11, 8-70



KSP 584-00B, KSP 584-00C, Page 12, 8-70



Figure 11. Typical Functional Layout of K-584C or B (Showing Line Circuit 1 Only)

T M 1 1 - 5 8 0 5 - 6 5 2 - 1 4 K S P 5 8 4 - 0 0 B K S P 5 8 4 - 0 0 C P a g e 1 2 , 10 - 7 0



Figure 12. Power Connections and Fuse Designation, KSP-584B

TM 11-5805-652-14 KSP 584-00B KSP 584-00C Page 14, 10-70



Figure 13. Internal Wiring, K-584B

KSP 584-008

K S P 5 8 4 - 0 0 C

Page 15, 10-70



Figure 14. Power Connections and Fuel Designations.

KSP 584-008 KSP 584-00C Page 16, 10-70



Figure 15. Internal Wiring, K-584C

KSP 584-00B KSP 584-00C Page 17, 10-70

4.0 MAINTENANCE OF K-584 TYPE PANELS

Maintenance should be limited to tracing of wriring trouble, fuse replacement, and replacement of improperly operating KTU's

5.0 REPLACEMENT PARTS

ITEM	PART NO.
Fuse, 1/2 ampere	180051-001
Fun, 1- 1/2 ampere	180051-002
Fun, 2 ampere	180051-003
Fun, 5 ampere	180051-004
Interrupter	190478-001
Plug, program (584B)	180049-001
Plug, program (584C)	180586-001
Socket miterrunter	190477-001

KSP400-00D Page 1, l-x!

KEY SYSTEM PRACTICE Operating Company Series

CHAPTER 4

K400(D)962 KEY TELEPHONE UNIT (KTU), CENTRAL OFFICE OR PBX LINE CIRCUIT FOR K-1A2 KEY TELEPHONE SYSTEM



Figure 1, K-400D Key Telephone Unit; Component Side (left) and Printed Circuit Side.

1. GENERAL

The K400(D)962 KTU is o new and improved plug-in CO or PBX line circuit. It is mechanically and electrically interchangeable with K400B type unit in the K501 and K512 series KSUs. Operationally the K400D KTU sore fully compatible with K400B KTUs in any K1A2 system installation.

Unlike the K400B KTU, the K400D provides means on the option block/handle for field addition of a pigtailed resistor to reduce the short time-out "Z" option interval to a shorter period, depending upon thee value of resistor added. Table I provides resistance required for decreasing short time-out interval.

Circuitwise the K400D KTU performs more satisfactorily in all types of field operating circumstances than does the K400B. For example, it is the only unit recommended for use behind certain type PBX's. The K-400D KTU has been designed to overcome, circumvent, and be unaffected by limiting operating conditions.

The switching of elements in the 'hold" and "ring-up" bridges is such that no switchook flash distortion will adversely affect (or be misinterpreted by) connecting switch trains.

"Ring-up" and "hold" functions are accomplished by the use of o split winding "dry-reed" relay. The ringing detector, a combined grounded-metallic circuit is not normally subject to false ring of longitudinally induced foreign potentials. Ringing bridge impedance is approximately equal to one high impedance ringer. A maximum of three ringers may be connected in combination across the line ahead of or behind the K400D KTU. A non A-lead type station, dialing ahead of the line circuit, could ring up the circuit falsely if longitudinal voltages were present, however, this will not occur unless such voltages are above 35 volts rms. The K400D KTU is normally impervious to induced foreign potentials on the CO line, transversely up to approximately 24 volts rms or longitudinally up to approximately 90 volts rms

2. INSTALLATION

2.1 PLUGGING IN

K400D KTUs are installed one KTU per line required. The units plug into the connectors provided in the key service unit with the printed circuit side of the card to the left. Be sure the KTU is plugged in firmly. After all units are inserted, tighten the retaining bar screws to prevent the units from falling out of the cord mounting assembly.

KSP400-00D Page 2, 1-70

2.2 CIRCUIT OPTIONS

Options provided on the K-400D KTU ore as follows:

- Ζ-Short Time-Out. (Used with automatic Z - Short Thile-Out. (Used with automatic ringing CO or PBX.)
 W- Interrupted Station Audible Signaling
 T - Steady Station Audible Signaling
 V - Auxiliary Common Audible Signal Circuit
 Y - Winking Hold Lamp

- X Steady Hold Lamp

Each K400D is factory wired for options Z, Wand Y. If other options are required by the subscriber, the installer must re-wire the option terminals on the printed circuit board as required. (Figure 2.)



Figure 2. K400(D) KTU, Circuit Options





3. OPERATION

3.1 INCOMING SIGNAL

(a) Idle Circuit Condition (Figure 3.)

In the idle circuit condition all relays ore in the unoperated state and transistors Q2 and Q3 are off. Transistor Q1 is held on by current to its base through the resistor network formed by resistors RT1, RT2, R16, R8, R11 and the B and C relay coils (b) Ringing Current Applied (Figure 4.)

Ringing voltage is usually applied across the line with the tip (T) side grounded. Ringing current then flows through the series connected primary and secondary windings of relay L, resistor R2 and capacitor C3, causing relay L to operate on each half-cycle of ringing current. Ringing current also flows through capacitor C2 and resistor R18 to the cathode of zener diode CR8. The negative (-) side of capacitor CT is normally maintained at about -16 volts de. Negative (-) half-cycles of ringing causes CR8 to conduct in the forward direction so that the cathode of diode CR6 is at approximately -24.4 volts dc. CR6 is thus forward biased and capacitor CT charges through resistor R3 toward o negative (-) 24 volt dc potential.





KSP400-00D Page 4, 1-70

3. OPERATION

continued

On positive half-cycles of ringing current, zener diode CR8 breaks down and conducts so that the cathode of diode CR6 is at about 0 volts potential. The negative (-) end of capacitor CT is somewhere between -16 and -18 volts, so diode CR6 is reversed biased and capacitor CT does not charge. However, capacitor CT does have a dial discharge path through resistors RT1 and RT2 combined with resistors R16, R11 and the windings of relays Band C. The charge lost by capacitor CT during this interval is much less than that gained during the negative (-) half-cycle. After approximately 0.5 second, p sufficent number of cycles of ringing current will have charged capacitor CT to about -18 volts, the base voltage of transistor Q1, which causes Q1 to "turn-off". Because of the symmetry of the detector circuit, its operation will be the some when ringing voltage is applied across the line with the ring (R) side grounded. In this case, ringing current flows to the cathode of zener diode CR8 through capacitor C5 and resistor R17.

When transistor Q1 **turn.off", its collector voltage rises and transistor O2 "turns on'; zener diode CR7 breaks down and causes transistorr Q3 to also "turn-on" to operate relay B. Relay C does not operate at this time since resistor R11 limits the current through its winding to less than its operate value. Relay B, operated, connects ground to the ST lead: the L lead to the LF lead and interrupted (option W) or steady (option TT ringing current or ground (option V) to the RC lead for audible signal control. Transistor Q1 remains off and transistors Q2 and Q3 on until either the call is answered or "timedout".

An unimpeded path for ringers in telephone sets is provided when bridged *ringing* for o particular line is required.

3.2 TIME-OUT OF RING UP CIRCUIT (Figure 5.)

(a) When Z Option is Not Provided

At the instant transistor Q1 "turns-off" the voltage at the (-) end of capacitor CT is about -18 volts, When the B relay operates, the voltage divider formed by resistors R14 and R15 is switched into the circuit and the voltage at the (+) end of CI drops from 9 volts to about -6 volts. Consequently, the (-) end of capacitor CI drops from -18 to about -24 volts. On subsequent half-cycles of ringing, diode CR6 will be reverse biased as before. Operation of relay B caused the anode of diode CR5 to be connected back to -24 volts through resistor R 8, transistor Q3 and diode CR4. The cathode of diode CR5 is connected to the base of transistor QI, which is at about -24 volts so diode CR5 does not conduct. The discharge path for capacitor CT is thus only through resistors RT1 and RT2 to ground. On negative half cycles of ringing, diode CR6 will conduct slightly to restore the charge lost by capacitor CT during the previous positive half-cycle. In this way, the change on capacitor CT which determines the duration of the time-out remains constant once the circuit has operated regardless of the duration of the ringing cycle.

When the first cycle of ringing has ceased, the (-) end of capacitor CT begins to charge toward ground through resistors RT1 and RT2. Transistor Q1 remains off until the voltage at the (-) end of capacitor CT reaches -18 volts at which time transistor Q1 "turns-on" and its collector voltage drops. This causes transistor Q2, zener diode CR7 and transistor Q3 to "turn-off" and relay B is released to restore the circuit to normal or idle condition. The time required for relay B to release after the lost cycle of ringing is approximately 30 seconds.

(b) When Z option is provided

This arrangement functions in a manner similar to that described in 3.2 (a) with the exception that resistor RT2 is short circuited, thereby, lowering the resistance through which the capacitor CT discharger. This results in o shorter time out period. On incoming calls where one cycle of ringing is received the the out is approximately 11 seconds. Subsequent cycles of ringing received before relay B releases resists the time out circuit to approximately 10 seconds. Any incoming call, for example, which is signaled by machine ringing will time out in approximately 10 seconds after the call is abandoned by the calling party.

(c) Provisions For Reduced Time Out

In cases where o shorter time out than that obtained with the Z option is required, this can be obtained by shunting he RT1 resistor with on appropriate resistor. The time out desired as a fraction of the original time out, TO, can be obtained by using the appropriate resistor, selected from the table following

NOTE: Where the stration of machine ringing is one second, we time out shall not be reduced below 50 per cent of the original time out. TABLE 1. RESISTANCE REQUIRED FOR VARIOUS TIME-OUT INTERVAL:

TIME OUT DESIRED	RESISTANCE (Megohms)
3/4 of original time-out	1.2
2/3 of original time-out	0.75
1/2 of original time-out	0.39
1/3 of original time-out	0.20
1/4 of original time-out	0.13

NOTE: If duration of machine ringing is one second, the time-out shall not be reduced below 1/2 of original time-out.



KSP400-00D 1 PRACTICE Page 6, 1-70

3.3 ANSWERING AN INCOMING CALL (Figure 6.)

An incoming call is answered by operating the pickup key associated with the line being rung and going off hook. The station is then connected across the line through the hookswitch and key contacts, and ringing is tripped at the Central Office or PBX. Ground is also connected through the hookswitch and key contact to the A lead, operating relay A which shunts terminals 1 and 3 of relay 1 preventing it from operating on line current and connects -24 volts to the winding of relay C, causing it to operate. Relay C operated disconnects the (-) end of capacitor CT from the base circuit of transistor Q1 and connects resistor R6 across capacitor CT causing it to discharge.

Figure 6. K400(D) KTU, Answering an Incoming Call; Paragraph 3.3

Transistor Q1 "turns-on" releasing relay B. Relay C operated, also removes the center tap of the ring up bridge from the rest of the circuit and shorts zencr diode CR8 thereby preventing the induction of noise into the talking path; it also disconnects the secondary of relay L and eliminates the shunting effect on the line of the secondary winding in series with resistor R2 and capacitor C3. Relays A and C operated establish the talking path to the telephone, connect the lamp "L" lead to +10 volts and open the RC lead to discontinue local audible signaling.

3.4 OUTGOING CALL

The procedure for making an outgoing call is the same OS that for answering an incoming call except that transistors Q2 and Q3 are normally off and *relay* B is released.



KSP400-00D Page 7, 1-70

3.5 HOLDING (Figure 7)

(a) Placing o Line On Hold

A busy line can be placed on hold by operating the bold key on the telephone set. When the hold key is depressed, ground is disconnected from the "A" Id allowing relay A to release. The A relay contacts shunting the relay L primary open and since the station T and R leads have not yet been disconnected from the line, relay L operates on line current. Operation of relay L causes the base of transistor Q1 to be connected through resistor R4, R3, diode CR6 and the operated relay C contacts to -24 volts. The voltage at terminal 2 of the L relay drops to nearly -24 volts causing transistor Q1 to turn off and transistors Q2 and Q3 thereby turn on. Transistor Q3 will have turned on about 2 milliseconds after relay A releases and a hold path is thereby provided for relay C through resistor R11, transistor Q3 and diode CR4 to -24 volts. Finally relay B operates through transistor Q3. Relays B and C operated connects the hold resistor R1 in series with the primary of relay L across the CO line, connects the LG lead to the ST lead to start the interrupter, connects the L lead to the LW lead (Y option) or lo +10 V (X option). When the hold key is released the telephone push button is released disconnecting the telephone from the line. line current through the L relay and resistor R1 maintains the CO line on hold until released.



TM 11-5805-652-14 KSP400-00D Page 8, 1-70

(b) Release of the Holding Bridge by o Station

Any station of the key telephone system that sizes the line by operating the associated pickup key and going "off-hook" will cause the A relay to operate and shunt the primary of relay L which thereby releases. Transistor Q1 then "turns-on" and transistors Q2 and Q3 "turn-off" releasing relay B. Relay C is held operated by relay A. The circuit is thus restored to the busy state.

(c) Release of the Holding Bridge from the Central Office or PBX

In the event o held party "abandons", the line circuit may be released from the connecting switching equipment by providing a momentary interruption of the line circuit. This causes the L relay to release. Transistor Q1 thereby "turns-on", restoring the circuit to the idle state.

(d) Release of the Holding Bridge from the Central Office or PBX, Battery Reversal

If the voltage across the line is reversed when the line is in the hold condition, the line circuit hold will be released. Reversal of the line current causes the L relay to release and then reoperate. When the L relay releases, transistor Q1 "turns-on" and transistors G2 and Q3 "turn-of" and relays B and C release.

3.6 DISCONNECTION

When all stations go on hook, the A lead is disconnected from ground causing relay A to release. Release of relay A opens the holding path relay C, which, in turn releases. In this way, the circuit is restored to the idle state.

3.7 OPERATION WITH LOCAL POWER FAILURE

During periods when the local dc supply is Inoperative it is possible to originate outgoing calls. When the station goes off hook, connection to the line is metallic. The primary and secondary of the L relay are connected in series with resistor R2 and capacitor C3 but this has o negligible effect on the talking circuits. If ringers are bridged across the line in the telephone set, incoming calls ore signaled in the usual way although visual and common audible signals an inoperative.

TECHNICAL NOTE

An incompatibility problem may occur between 1A2 KTS equipped with 400D KTU's and certain CO's or PBX's. A lost call can result, in some cases, when the switching system reswitches an established connection while the key telephone station is on hold. If the reswitching sequence opens the loop for an interval greater than that required to release the 400D KTU hold circuit, a disconnection will occur.

The release time of the 400D KTU can be extended to bridge the open intervals generated by these reswitching sequences by applying the appropriate one of two options following. (See figure 2.)

Option ZC, delay hold release far electronic switch-!*, consists of adding o 5 MF capacitor across terminals 2 and 3 of the 400D KTU option block.

Option ZD, delay hold release for officer other than electronic consists of adding a 1 MF capacitor across terminals 2 and 3 of the block.

When the existing 2 option is provided with ZC or ZD option, the strap between terminals 1 and 2 of the option black should be removed and replaced with one of the capacitor pigtail leads.

TIMM-2 Page 350.01 Date. 10-70

CHAPTER 5

CORINTHIAN MULTI-KEY TELEPHONES ROTARY DIAL (K2230 SERIES) MULTI-KEY TELEPHONE



Figure 1A. K-234 CORINTHIAN Telephone

CORINTHIAN* MULTI-KEY TELEPHONES,

ROTARY DIAL (K-230 SERIES) AND

PUSHBUTTON DIAL (K-2230 SERIES).

1. GENERAL INFORMATION

These telephones are used in conjunction with o key telephone system to provide access to o maximum of eleven lines which may be central office trunks. PBX, private, and intercom lines. The lower left hand (red) button is used to place o hold on one or more outside lines.

Ordering information is given in Table 1.

Replaceable parts are listed in Table II.

2. INSTALLATION

Mounting cords are terminated with quick-connect plugs. Installation consists basically of plugging the mounting cord into appropriately pm-wired connecting boxes of the key telephone system. Special instructions (if required) and a wiring diagram are Packed with each telephone.

2.1 HANDSFREE (SPEAKERPHONE) APPLICATIONS

All Corinthians of current manufacture are equipped to operate with handsfree (speakerphone) equipment. Corinthians with new codes, (K-234, 235, 236, 2234, 2235, 2236), hove speakerphone leads connected within the telephone. If used without speakerphone, disconnect, insulate and store following leads:

speakerphone, disconnecr, insulate the store following leads: SLATE-VIOLET wire, (pin 16, receptacle E) from L1 on network VIOLET-SLATE wire, (pin 17, teceptacle E) from G on network vIOLET-GREEN wire, (pin 18, receptacle E) from RR on network GREEN-v VIOLET wire, (mounting cord) from terminal #6 on terminal board.

Corinthians with old odes, (K231, 232, 233, 2231, 2232, 2233) as shipped with speakerphone lea a disconnected,



Figure 1B. K-2235 CORINTHIAN Telephone

2.2. ALL-BUTTONS-UP INTERCOM

All-buttons-up intercom is available on all Corinthian telephoner except K-233 and K-2233. If expansion kit 180113 is added to there units, the all-buttons-up intercom is automatically added.

TABLE 1. ORDERING INFORMATION (See also Table III)

Series Code		Description						
A	K-23l**()42M	TELEPHONE, Multi-key;						
B	K-234**()42M*	50-conductor cord with plug.						
A	K-232**()42M	TELEPHONE, Multi-key;						
B	K-235**()42M*	Rotary Dial; 11-line capacity; 80-conductor cord with plug.						
A	K-233**()42M	TELEPHONE, Multi-key;						
B	K-236**()42M*	80-conductor cord with plug.						
A	K-2232**()42M	TELEPHONE, Multi-Key;						
B	K-2234**()42M*	50-conductor cord with plug.						
A	K-2232**()42M	TELEPHONE, Multi-key						
В	K-2235**()42M*	80-conductor cord with plug.						
A	K-2233**()42M	TELEPHONE, Multi-key,						
B	K-2236**()42M*	80-conductor cord with plug.						
** SUBSTITUTE COLOR CODE AS FOLLOWS:								
00-Black, 05-Green; 13-Light Beige; 15-White								

Recommended units.

 TM
 11-5805-652-14

 TIMM -2
 Page 350.02

 Item: "CORINTHIAN"
 Date: 5-67



Figure 2. "CORINTHIAN" Telephone, Exploded View

					TIMM-2					Page: 350.03 Date: 10-70						
FIG		INDEX NO.	PART NUMBER	Indented items are included in the NAME, Description part under which they are indented)			Q	UA	NTI	TY	US	ED	ON	:	_	٦
			TABLE II. RE	PLACEABLE PARTS LIST, "CORINTHIAN" MULTI-KEY TELEPHONES	231	232	233	234	235	236	2231	2232	2233	2234	2235	2236
2		1 1A	3800(G)450 3600(G)450	DIAL, Rotary DIAL, Pushbutton	<u> </u>	1	1	<u>י</u>	1	1	-	- 1	- 1	;	7	- 1
1	F	2	65**(C2)410		H	H	H-	H	H	₽ <u></u>	H		H	붜	⊢	
	┢	3	145E(6A)4/U		┢	ŀ.	ŀ-	ŀ	ł÷	ŀ	-	-	-		H	Ė
		4 4 4	869**(15)650 870**(15)650 874**(15)650 875**(15)650	MOUNTING CORD ASSEMBLY, 80-CONDUCTOR MOUNTING CORD ASSEMBLY, 50-CONDUCTOR MOUNTING CORD ASSEMBLY, 80-CONDUCTOR MOUNTING CORD ASSEMBLY, 50-CONDUCTOR	1				- 1 -	-	-				- - 1 -	-
	ł	5	88527-**	HOUSING, for Rotary Dial	1	$\overline{1}$	1	1	1	1	-	-	-	•	_	-
		•	180167-**	FACEPLATE (Not Shown)	1	1	1	1	1	۱	-	-	•	-	-	-
		5	180143-** 88126-2	HOUSING, for Pushbutton Dial SCREW, Housing	4	4	4	4	4	4	4	4	4	4	4	4
	ł	'6 6	88528-##	BASE ASSEMBLY, Rotary Dial BASE ASSEMBLY, Rotary Dial	1	1	1	1	1	1	-	-	-	-	-	-
		J	88105-1	BRACKET, Cradle Hook	Ţ.	ţ,	1.	1	1	1	İ.	ţ	ţ,	1	1	i,
			95944-2 88532-1	RIVEI BRACKET, Rotory Dial (LH)	ti	Ħ	ľ	Ħ	ti	1	-	-	-	-	-	-
ł			88532-2 88531-1	BRACKET, Rotary Dial (RH) BRACKET, Pushbutton Dial	<u> </u>	-	-		<u> </u>	1	-2	2	2	-2	2	2
			88535-1 88533-1	BRACKET, for Key Connectors BRACKET, Key (BH Rep)	R	ľ	ľ	Ľ		ľ			1		1	h
			88545-1	BRACKET, Key (LH Rear)	li	h	li	i	i	li.	li	i	i	i	i	i
			180117-1	BAR, Tie	1	1	li.	[]	11	11	1	1	1	1	1	ľ
			88180-1		Ħ	ħ	ħ	ti	Ħ	h	h	H	t	1	i	h
			180221-1	SCREW	2	12	2	2	2	2	2	2	2	2	2	5
			87483-1	FOOT	5	5	5	5	5	5	5	5	5	5	5	5
			95943-2	RIVET, (Fool)	5	5	5	5	5	5	5	5	5	5	5	<mark>۶</mark>
		7	598(X)740	KEY AND CONNECTOR ASSEMBLY	1	13	2		1	,	-	1	1	-	12	6
			67042-3	SCREW, (Connector to Bracket)	_	2	2	Ŀ	2	2	-	2	2	-	2	2
		8	599(X)740	KEY AND CONNECTOR ASSEMBLY	1	1	2	1	1		2	11		1	1	Ľ
1			67042-3	SCREW, (Key to Bracket)	2	2	2		2	2	2	2	2	2	2	6
		9	180562-1	PIVOT BAR ASSEMBLY (Adjustable)	1-	1	-	1-	ŢŢ	-	F	Ţ	-	•	1	F
		10	190329-1 190354-1	SPRING Pivot Bar SCREW, Shoulder (Pivot Bar)		H	-		1	-		;		-	6	-
		11	88546-1	CONNECTOR AND WIRE ASSEMBLY	ŀ	1	ŀ	1	1	1	1	1	1	1	1	1
		12	88540-1	SCREW, Connector Mounting TERMINAL BOARD ASSEMBLY	† 2	fi	1 ²	†i	ħ	17	f-	17	Ĩ	2	Ĩ	F
		13	75576-4 8853A-1	SCREW BRACKET, Terminal Board (LH)	2	12	2	1	2	2	2	2	2	2	2	ĥ
		14	88536-2	BRACKET, Terminal Board (RH)	li	li	h	li	li	li	į	li	li	i	i	Ŀ
			/33/0-4	544547, (8/3644 TO 1010)	ľ	ľ	Ľ	₽	ľ	 ²	Ľ	Ľ	Ľ	4	ŕ	ř-
		15	88240-1	CRADLE HOOK ASSEMBLY	Ľ	R	R.	[!	11	[]	Ľ	11	[]	1	R	ľ
		17	73538-8	RING, Retaining	12	2	2		2	2	2	2	2	2	2	k
		18	75307-4	SPRING, Cradle Hook Return (75307-3 will work)	li	ĥ	þ	1	1Ī	1	1	Ĩ	I	1	1	ŀ.
		19	88114-2	SPRING NEST ASSEMBLY SPACER, Source Next	R	R	Ľ	Ľ	R	R	Ľ	11	11		Ľ.	ľ
			88127-2	SCREW	2	2	Ż	2	2	2	Ż	2	2	2	2	ķ
		21	78825-5	HANGER, Cable	ŀ	ŀ	ŀ	Ŀ	<u> </u>	ŀ	ŀ	ŀ	!	1	!	ŗ
			80127-2	SCREW	1	h	K	1;	H	K	6	[;	[]	1	ĥ.	6
	- 1	22	84270-2	CARD, Number (For Key Strips)	Τ	2	ħ	Γ	2	Ť	P.	2	Ţ	ŗ	2	Ŀ
		23 23	57513-1 75415-1	CARD, Number	1;	1	:		1;	1	Ľ	11	11	1	Ľ	Ľ
		24	87514-1	RETAINER, Number Card	Ŀ	Ŀ	Ŀ	Ŀ	1.	Ŀ	h	1	h	1	1	h
		25 25	180114-1		1	1	1	[;	Γ	1	F.	Γ	;	1	Ľ	F.
			95857-1	TIE, Wire	2	2	2	2	2	2	2	2	2	2	2	2

;;

COLOR: 00-Black; 05-Green; 13-Light Beige; 15-White COLOR: 21-Black; 25-Green; 33-Light Beige; 35-White

TIMM-2 Page: 350.04 Date: 10-70

3. EXPANSION FROM 5-LINE TO II-LINE CAPACITY.

The K-233, K-236, K-2233 and K-2236 5-line Corinthians may be expanded to 11-line capacity by adding kit number 180113 which Includes the additional key assembly. The K-231 and K-2231 S-line Corinthians may be exploded to 11-line capacity by adding kit number 180113 and replacing the 50-conductor mounting card with type B69 80-conductor mounting card. (See item 4, table II).

The K-234, and K-2234 S-line Corinthians may be expanded to II-line capacity by adding kit number 180113 and replacing the SO-conductor mounting cord with type 874 80-cond mounting cord. (See item 4, table II.)

4. EXPLANATION OF CODE SERIES IN TABLE 1.

Series A codes of Table I are early code numbers whereas series B codes ore latter codes.' The two series are identical except far the following differences in wiring:

(1) HANDSFREE CONNECTIONS

The latter series of Corinthians (B) have all leads associated with speakerphone (handsfree) operation connected, whereas in the early series, these leads are not connected. (See paragraph 2.1.)

(2) ALL-BUTTONS-UP INTERCOM CONNECTIONS

The early (A) series utilizes pins number 5, B, and II an line plug "A" far "all-buttons+ intercom whereas the latter (B) series uses pins number 16, 41, and 47 of line plug "A" far 6-button sets and pins number 20, U and 45 of line plug "B" far 12-button sets.

The latter series is mare universally applicable and is recommended far new installations. Far <u>additions to</u> present installations now using early series <u>Corinthians</u> and <u>using all--buttons-up intercom</u>, use early series <u>Corinthians</u>. The A and B series are compatible with each other if the all-buttons-up intercom feature is not used and the associated leads ore disconnected at the telephone connecting boxes.

NUMBER CARDS FOR KEY STRIP. Type (2) DISASSEMBLY AND REASSEMBLY 5. numbers on the side of the card with the backing paper. Secure the card to a sheet of paper with tape so the edge of the card fines up with the edge of the paper. Type the first number about 1/16 inch below the HOUSING GROUP (Figure 2) 5.1 TAPE a. REMOVAL AND DISASSEMBLY OF HOUSING GROUP, lower edge of the backing paper. Allowing two lines for much number, triple space Loosen the four cabinet lock screws and (1) 287lift Housing off. between nun&en. Peel the backing paper from each end of the card and adhere to the Escutcheon assembly in the recess provided. 4471 (2) To remove Escutcheon Assembly (25). press the four friction latch fingers of the Escutcheon up through up through the holes in the housing. (To remove Escutcheon Assembly **INSTALL HOUSING. Fit the Housing** (3) 287carefully over the dial and push buttons. Secure with the four screws or each comer. with housing in place, pry outward at the 4472 upper edge.) To remove the Number Card (23) and Retainer (24), use a straightened paper clip or similar device. Insert it in the notch at the right end of the Retainer, and (3) 287-4473 pry the Retainer and Number Card out. TRIPLE SPACE 287-4474 NUMBER CARD FOR KEY STRIP EDGE OF PAPER PX-23 BACKIN G PAPER HOLD TAPE Figure 3. Removal of Number Card and Retainer 6. ASSEMBLY AND INSTALLATION OF HOUSING GROUP 5.2 MOUNTING CORD (Item 14)

(1) To install the Number Card (23), place the Card in the recess of the Hausing. Grasp the Retainer (24) near its center and insert one and in place in the Housing. Bend the opposite and down with the other hand, and release the center hold. (See Figure 4.)



Figure 4. Installation of Number Card and Retainer

. REMOVAL OF MOUNTING CORD

- Xemave the screw and washer that secure the Cable Hanger (21) to the Base.
- (2) Unplug the Mounting Cord Plugs from the connectors in the phone.
- (3) Remove the Wire Ties as necessary and remove the Mounting Cord and the Cable Hanger.
- 6. INSTALLATION OF MOUNTING CORD
 - Plug the "C" plug of the Mounting Card to the 599 Key Connector. (Lower Connector on 232 and 1232.)
 - (2) Plug the "J" plug to the 598 Key Connector.
 - (3) Plug the "E" plug to the Terminal Board Connector.
 - (4) Secure the Mounting Cord to the Base with the Cable Hanger, Washer and Screw.
 - (5) Install the Wire Ties to prevent wires from interfering with other components.

TIMM-2	Page: 360.06
em: "CORINTHIAN"	Dŏte: 5-67

- 5.3 KEY ASSEMBLY
 - a. LAMP REPLACEMENT (Housing Removed). Pull off the brass frame. Use long nose pliers and pull out the foam rubber pocked behind the contacts. Replace lamp and repack the foam rubber. Install the brass frame.
 - b. REMOVAL OF KEY AND CONNECTOR ASSEMBLY
 - (1) Remove the two screws that secure the connector to its bracket.
 - (2) Remove the two screws that secure the Key Assembly to its brackets.

NOTE: The lower connector (232 and 1232 phones) is wired to the 599 Key Assembly.

- c. INSTALLATION OF KEY AND CONNECTOR ASSEMBLY
 - (1) Place the Key Assembly and its Connector loosely in places on their respective brackets. Install the four mounting screws and tighten.

NOTE: When replacing a key assembly on a 12-button "CORINTHIAN", remove and discard the Latch Bar Return Spring and Retaining Pin – indicated by "See Note 1" on figure 2.

5.4 PIVOT BAR GROUP (Items 9 and 10)

- a. REMOVAL OF PIVOT BAR (Item 9)
 - (1) Use round now pliers and remove Spring (10)
 - (2) Remove the Shoulder Screw that secures the Pivot Bar (9) to the outer Key Mounting Bracket. Slip the Pivot Bar dawn and inward until it clean the stud of the inner Key Bracket.
- b. INSTALLATION OF PIVOT BAR (Itom 9)
 - Place the inner end of the Pivot Bar over the stud of the inner Key Bracket. Engage the slots of the Pivot Bar with the notchos in the Latch Bars of the Key Assemblies.
 - (2) Install the Shoulder Screw through the hole in the outer Key Bracket and thread it into the Pivot Bar.
 - (3) Attach spring (9).
 - (4) Test operation of the puch buttoms to see that both key assemblies function as one unit.

- 5.5 DIAL (Items 1 and 1A)
 - a. REMOVAL OF DIAL
 - (1) Loosen the dial mounting screws and lift dial from bracket.
 - (2) Disconnect leads and remove dial.
 - b. INSTALLATION OF DIAL
 - (1) Refer to the appropriate circuit label and connect dial lad, as indicated.
 - (2) Place Dial in the brackets so the small holes in the dial flanges sat over the punched bosses of the mounting bracket. Tighten the mounting screws.

5.6 TERMINAL BOARD AND CONNECTOR GROUP (Items 11, 12, 13, 14)

NOTE: If this group is being removed temporarily, do not disconnect leads between the connector (11) and the terminal board (12), but remove both components as a unit.

- a. REMOVAL OF TERMINAL AND CONNECTOR GROUP
 - (1) Refer to theppropriate circuit label and disconnect leads to Network, Hookewitch, and Dbl.
 - (2) Remove the two screws that attach the Connector and the two screws that attach the Terminal Board.
- b. INSTALLATION OF TERMINAL BOARD AND CONNECTOR GROUP
 - Install the Terminal Baard and the Connector and secure with two screws each.
 - (2) Refer to the appropriate circuit label and connect leads.

TIMM-2 Page: 350.07 Item: "CORINTHIAN" Date: 5-67

5.7 RINGER (Item 3)

a. REMOVAL OF RINGER

- (1) Disconnect red lead from "K" terminal on Network.
- (2) Disconnect black lead from "4" **terminal** on Terminal Board.
- (3) Pull slate and slatea-red leads loose from base. (Taped down)
- (4) If Dial and Terminal Board have not been removed, loosen them from their mounting brackets and move them as necessary to gain access to the Ringer Mounting Screws.
- b. INSTALLATION OF RINGER
 - (1) Secure Ringer to Base with the two mounting screws.
 - (2) Tape the slate and state-red leads to the base. (Be sure their terminals are taped or otherwise insulated.)
 - (3) Connect black lard to "4" terminal on Terminal Board.
 - (4) Connect red lead to "K" terminal on Network.
 - (5) Install Dial and Terminal Board.
- 5.8 HANDSET ASSEMBLY (Item 2)
 - a. **REMOVAL OF HANDSET**. (Housing Removed) **Disconnect the leads and** pull the cord hook off the finger in the base.
 - b. INSTALLATION OF HANDSET. Refer to the aparopriate circuit label and connect the handset leads. Secure the cord to the base by slipping the cord hook over the finger in the base.

5.9 <u>CRADLE HOOK AND SWITCH GROUP</u> (Items 15 thru 20)

- a. REMOVAL OF CRADLE HOOK AND SWITCH GROUP
 - (1) Use round nose pliers and remove Spring (18).
 - (2) Remove one Retaining Ring (17) and remove Pin (16).
 - (3) Work the Cradle Hook (15) up and out from the Spring Nest Assembly (19).
 - (4) Disconnect the Spring Nest leads.
 - (5) Remove the two screws that secure them to the Bose, and remove the Spring Nest Assembly (19) and Spacer (20). If necessary, remove the Spacer from the leads.
- b. INSTALLATION OF CRADLE HOOK AND SWITCH GROUP
 - (1) Insert Spring Nest leads through rectangular holes in Spacer (19). (Refer to figure 2).
 - (2) **Position the Spacer and Spring Nest over** their mounting holes in the base. (Pull leads toward center of Base.) Secure with the two screws.
 - (3) Refer to the appropriate circuit label and connect Spring Nest leads.
 - (4) Position the Cradle Hook (15) so its mounting holes line up with hales in the Bracket. Actuating arm of the Cradle Hook must engage the mating arm of the *ring Nest Assembly. Install Pin (16) and Retaining Ring (17).
 - (5) **UK round nose** pliers and install Spring (18).

T M 11-5805-652-14 T I M M - 2 Page 350.11 D a t e D a t e 10-70



TM 11-5805-652-14 TIMM-2 Page 350.12 Date 10-70

	333 TYDE TELEDUONE	CIRCUIT	
WIRING DIAGRAM	NOTES (CONT)		P CAUP SCALL AS CONVERS ON OF ALL AN A
ACVR BAARD TYPE RINAR COMMECTIONS 	5 BUZZER COMMECT BUZZER LEADS TO TERMS EZ B BZ/ ON TERMINAL BOARD B ALL EF POSITIONS ARE ARRANDED IN THE SMOP AS PICKUP POSITIONS TO CONVERT A PICKUP REV LOCKINGI TO SIGNALING INDN-LOCKING PENDOVET THE PLUNCER SERVE FROM REV POSITION INDVLED AND MARE	9 WHEN CONVERTED FIELD FIELS ARE USED FOR SIGNALING ON ANY REF UNIT THE SMORT INFOLLED ASSOCIATED MINI HERE CONVERTED REVS SHALL BE CONRECTED TO THE 5 OR 56 TERMINAL OF THE REF UNIT INVOLVED THE 5 TERMINAL SHALL BE STRAPPED TO THE 56 TERMINAL	20 00 56404 111 4 40 05 40 41 400 41 400 41 400 41 40 56 400 1100 401 1 40 56 400 1100 401 1 40 56 40 1100 400 4
	CONNECTION CHANGE PER TABLE A	IO & SPEAKERPHONE CONNECTIONS	· ┝┛┝╌╌╞╴╶╴┥╴╶╴┇╹╡╹╪┻┿╹╞╹╞╺
	T WHEN USING COMMON SIGNALING FOR SEVERAL PRIVATE OR INTERCOM	II & SPARE CONDUCTORS INDIVIDUALLY TAPED AND STORED	
	CONVERTED TO NON LOCKING AND IS USED AS THE SIGNALING KET	12 IN 142 SYSTEM TO PROVIDE INTERCOM WITH ALL BUTTONS RELEASED	「白色山」「「「「「」」」」「「「「」」」」」」」
ALL ()	B WHEN MORE THAN 5 PRIVATE AND INTERCOM LINES WITH COMMON SIGNAL	MUST BE CONNECTED TO SYSTEM TO PROVIDE T R AND A LEAD	
	HEY ARE RED THE S & SC ARE STRAPPED TOGETHER	COMMECTIONS	
			·
		وليستال محالي المستحد والمحاجب والمحاجب والمحاجب والمحاجب والمحاج والمحاجب وا	
		· 57	
		NOS	
		A 27 27 27 WH A 5 OF 51 27	
	25 yig 54	13. 13 GRN-4017 RIOR A . 5.	
i ant can t 24 a	24, 24	1 48P	2 CTN 1
TYEL BAN B. CALSES	HOLD 21, 23	= = = = = = =	¥
	40 40	ZADY	9
	50, 50	POS 20 20 mm1-00m T 20	
	2 0 GRN YEL 18	4 4 BH4 WHT	
1 1 1 2 2 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1	+	4 4 BL/ PED POPL 6	Ύτε κ
		1 39 39 PD Ru P2 0P LG 3	TEL 7 24
			B ME BU D. C. YL. B. AVESS 24
	ا همینده السب المحمد المحم	3 40	
	POS 26 26 WH' BLU 1 26	N S NO NO MED-OMM 12,	2 82 - 3 - 1 - 3 - 3 - 9 - 9 - 1 - 8
			3mh v C
NORMAL PULSE bas last a SET NOTE :	3 3 GRN-enft 1 3	5.5 SHW RED P30PL 9.	
	38 38 ##1 GRM LG 28.	AUP	
	Ban wet a 4		
*** **** ·	2.10	a marine a	<u>12 YEL BAN 23 23 E BAN 8 44</u>
CIRCUIT SCHEMAT C	(1)5 (1 <u>32</u> <u>32</u> <u>41</u> <u>9</u>	56
	29 29 WHT SL A 5 09 5 30	4 13, 33 BLK BLU A.5 OF 5 36	5. 0 26, 26
	AMP BL. RED L 6	AND	
8 - Sau - Sa	2 19 19 PED BLL LG 3	The second second second second second second second second second second second second second second second s	2 4 413 28 29 684-413 9
		· · · · · · · · · · · · · · · · ·	2 7 X 54 112 50 54 110 - 0
	3 RD		V S 32 32 V S S 7
	4 5 8 80 SED		3 6 F .
• • • • · · · · · · · · · · · · · · · ·	NED 200 A.S ON 5.33.	1	A
000^}	AMP	ANP	
	والسياد <u>معالي المرجم من معمم من المرجم المرجم المرجم المرجم المرجم المرجم المرجم المرجم المرجم المرجم المرجم ا</u>	·····	ALC
	a series a series a series and the series of a series		42,
2350.0			2
i [™] ™ i	BLK BLL_AS 18 5_96	E	1
nomin_4 (* }	AMP 55		
· - h	2 9 GPH BLK 4 3	9. 4 49 B.4	35, 12, 5 ^m B.
	1 m f	10 ⁻⁰ - 0 ⁻⁰ - 0 ⁻⁰	- 56 35 GRN BLS
w1165	PCC	1	40. 40 BLK GRN
WHEN HANDSET IS REMOVED	But Bob AS B. 19		TEAM DAL REPERINGE
HODESWITCH CONTACTS (& c) WARE DEFORE (a b) BREARS AND d e)	• + + + +	5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	80482 STRAP
THE LINES HAVE CONSISTED FOR COMMON COLD IN THE - HE RECTO	AMP		
OF AN LINE BY CONNEL ING ME CUN E LEAD ASSOL & ED WITH RINA AND	ALP A	179 I 74 IE	
BOW P FARS TO THE DES RED MOL AT HE EQUIPMENT			
A ADD IN FEATURE FOR KEA SYS EN CONNECT OF PADE T PREDIFAU FROM PIS, ON SY E ED TO OF PRI TERMINA.			· · · ·
a 5 at the Busy Laws	16		· · · · · · · · · · · · · · · · · · ·
RANSEER 5 ATE BRINK I SET HLEAD FROM TERM IG TO ON NETWIRK B	alico e c.		
51243 CONVE - ODE NSO C H3111 STREEMEEN FRANS S B CLAS SHOWN	**33 # 4 *F		·

TM 11-5805-652-14 TIMM-2 Page 350.13 D a t e 1 0 - 7 0



TM 11-5805-652-14 TIMM-2 Page 350.14 Date 10-70



TM 11-5805-652-14 TIMM-2 Page: 21 Date: 10-70



Diagram, 1231 Multi-Key Telephone (5-Line TEL-TOUCH "CORINTHIANS"



Diagram 1231 Multi-Key Telephone (11-Line TEL-TOUCH "CORINTHIAN")

TM 11-5805-652-14 TIMM-2 Page: 530.23 Date: 10-70



TM 11-5805-652-14 TIMM-2 Page: 350.24 Date: 10-70


TIMM-2 Page: 350.25

Date 10-70



TM 11-5805-652-14 TIMM-2 Page: 350.26

Date: 10-70





VOICE SWITCHING CIRCUIT

2.06 Voice switching is the automatic trans-

fer from receiving to transmitting condition. This circuit is controlled by speech from the speakerphone user. The level of speech necessary to cause switching is predetermined.

2.07 If possible, place audible signals away from the transmitter unit where there

will be no interference with the voice switching feature. It may be nece-sary to lower the volume of the audible signal or install audible signal using the cutoff feature of the NE-55 type control unit. Use the leads to common signal control and common ringer or buzzer circuit for this cutoff feature.

LOUDSPEAKER ADJUSTMENT

- 2.08 An alternate loudspeaker connection is provided in each NE-35 type control unit to compensate for a strongly reverberant room condition.
- 2.09 This condition may cause voice switching while receiving calls. The effect of voice switching causes portions of the incoming speech to be chopped off.

2.10 To compensate for this condition, move the loudspeaker lead from terminal 33 to terminal 24 on the NE-55A control unit. Move lead from terminal 29 to terminal 30 on the NE-55QB unit. This places a resistor in series with one loudspeaker lead.

TEST CALL

- 2.11 When all connections have been made.
- (a) Place a call to test desk.

Page 2

- (b) Adjust the loudspeaker volume to a moderately loud listening level by turning the volume control on the transmitter clockwise. Position volume control halfway between lowest and highest level.
- (c) Have the test center repeat the question, "In what suburb does Joe live?" several times.
- (d) If choppiness is detected in the sentence, particularly in the first b in suburb and the t in what, change one loudspeaker lead according to 2.10.
- (e) Repeat this test at a high listening level by turning the volume control to the extreme clockwise position.
- (f) When there is no chappiness, the room adjustment is satisfactory.

3. MAINTENANCE

- 3.01 Table A provides maintenance procedures for troubles which may be encountered.
- 3.02 After working on equipment, test the complete system as specified in 2.11.

RADIO INTERPERENCE

3.03 Where radio interference is experienced in the telephone set only, a suppressor may be installed. See procedure on radio signal suppression in telephone sets.

3.04 Miere radio interformes is experienced with as NE-667.W transmitter, solder an WS-13014, " capicitae from torminal B to terminal D of the printed - 2009 board or ground the transmitter unit case.

PART FIVE

SPEAKERPHONE SYSTEMS (USFI WITH AN/FDC-37(A)1 AN/FTC-37(V)2)

CHAPTER 1

SPEAKERPHONE SYSTEMS-NE-3 TYPE

INSTALLATION AND MAINTENANCE

1. GENERAL

1.01 For connection information between speakerphone components and telephone sets, refer to appropriate procedures, 2.03 A standard 120-volt as receptable is required for the operators of the ME-20138 transformer. This receptable must be on a concust that as not controlled by a swarth, Power outlet as furnished and mantaned by the customers.

2. INSTALLATION

2.01 Amaid installing apparatus with plastic covers or parts (control units, "ransmitters, etc.) in locations where the tempetature in the immidiate virinity encode 140 digenes F. 2.04 The length of wire between the transformer and the control unit about met council 100 feet of standard anothe wire,

TRANSHITTER AND LOUDSPEAKER

BE-55 TYPE CONTROL WRIT

2.02 Locate the central unit to more than 100 feet from the other components, Nount the control unit on an insulated corface (wooden backboord, etc), NETT: There about the second second technologies of a second technologies of a second





PERCENCE IN CONSIGN

VOICE SWITCHING CIRCUIT

2.06 Voice switching is **the** automatic transfer from receiving to transmitting condition. This circuit **is** controlled by speech from the speakerphone user. The level of speech necessary to cause switching is predetermined.

2.07 If possible, place audible signals away from the transmitter unit where there will **be no** interference with **the** voice switching feature. It may be necessary to lower the volume of **the audible** signal or install audible signal using the cutoff feature of the NE-55 type control unit. Use the leads to **common** signal control and common ringer or buzzer circuit for this cutoff feature.

LOUDSPEAKER ADJUSTMENT

2.08 An alternate loudspeaker connection is provided in each NE-55 type control unit to compensate for a strongly reverberant room condition.

2.09 This condition may cause voice switching while receiving calls. The effect of voice switching causes portions of the incoming speech to be chopped off.

2.10 To compensate for this condition, move RADIO INTERFERENCE the loudspeaker lead from terminal 33
to terminal 24 on the NE-55A control unit.
Move lead from terminal 29 to terminal 30 on the NE-55QB unit. This places a resistor in series with one loudspeaker lead.
3.03 Where radio inte in the telephone may be installed. Se signal suppression

TEST CALL

- 2.11 When all connections have been made:
- (a) Place a call to test desk.

- (b) Adjust the loudspeaker volume to a moderately loud listening level by turning the volume control on the transmitter clockwise. Position volume control halfway between lowest and highest. level.
- (c) Have the test center repeat the question, "In what suburb does Joe live?" several times.

(d) If choppiness is detected in the sentence, particularly in the first b in suburb and the t in what, change one loud-speaker lead according to 2.10.

- (e) Repeat this test at a high listening level by turning the volume control to the extreme clockwise position,
- (f) When there is no choppiness, the room **adjustment** is satisfactory.

3. MAINTENANCE

- 3.01 Table A provides maintenance procedures for troubles which may be encountered.
- 3.02 After working on equipment, test the complete system as specified in 2.11.

3.03 Where radio interference is experienced in the telephone set only, **a** suppressor may be installed. See procedure on radio signal suppression in telephone sets.

3.04 Where radio interference is experienced with an NE-667AN transmitter, solder an NS-13814, L7 capacitor from terminal B to terminal D of the printed wiring board or ground the transmitter unit case.

Trouble Indication	Probable Cause	Check
On key does not light	No power or open wiring	Power supply outlet with a neon lamp voltage tester or equivalent or check LK lead
Lamp does not light but set works	Light burnt out	Lamp in transmitter
Lamp lights but does not stay lit when ON button is released	Loose connection in local wiring	Contacts 10M of K1 relay or OFF key contacts in transmitter
Amplification seems defi- cient in receiver or trans- mitter circuit or both	Faulty control unit	Replace control unit
Set transmits but does not receive	Loose or open wiring to loudspeaker	SP1 and SP3 leads Loudspeaker
Set receives but does not transmit	Local wiring open Defective transmitter	<i>M</i> lead Transmitter
No variation in volume when control is rotated	Open wiring Defective volume control Defective control unit	PI lead Check volume control Replace control unit
Dial heard over loudspeaker	Dial Wiring	For proper dial P3 and P4 leads

TABLE A TROUBLES AND PROBABLE CAUSES

3.05 Where radio interference is experienced with an NE-666A or NE-666QB transmitter, place an NS-13014, L7 capacitor across the terminals of the NE-AC1 or NE-AB2 transmitter unit.

4. GENERAL

Purpose Of Circuit

4.01 This circuit provides distant taiking and loudspeaker receiving, as well as circuitry for connection to special telephone sets to provide for a combination of regular and distant talking telephone circuits, together with facilities for switching between two types of telephone circuits. The speakerphone control unit is a voice-switched, fully transistorized station set designed to give transmit and receive volume performance comparable to that obtained from an NE-500 type telephone set on typical loop. Special singleline telephone sets are provided as part of this circuit, but the equipment may also be, used with special key telephone sets of connecting circuits to provide for use with line and signaling circuits of key telephone system-NE-1A, -1A1 or -64.

5. CIRCUIT DESCRIPTION

OR IGENATING OR ANSWERING A CALL

Nandset Operation

5.01 To arraginate ar answer a call, the bandset us hifted from its mountany, thus

operating he swatchhook assembly in the telephone set. The operated swatchhook assembly supplaces ground to the A lead to operate the A relay of the law carcuit, af connected to key telephone system Ni-IAI or to the ML load to operate the station busy lamp carcuit, af provided. The swatchhook assembly also completes the talking carcuit from the tig- and rang of the lame through the celephone are equipment and, an sequence, removes the short carcuit from the receiver carcuit.

Speakerphone Operation

5.02 To originate of answer a call, the nonlocking ON Loy, Fig. 2, as depressed. Thas charges a ground on the S load to operate the KI relay in the control mat. The ki relay locks up through contact 108, the OFF has and the break contacts of the unuscrated solution hank around, the handset and he an the manteand the constance the helden constant of the ki mailing where the approximation is used. The 4.4 problem assessment of the supported the technical sold of the ft and fit leads, through the transits all the collecture settland, with presence, down ! the energy of and an elevel to the time to the the and the parameter provid over the \$4 total, the bould bank constants of the persention constitutes, and the at the kit rates to the A load to the line surveys, if committed to be to faithing evenue Wollel. It also impossion a ground to wher 🐘 bread an also exceptions been there are every of preventional, and such all the competer, the Harms are offer the antiperson terms are in the settle of the settle of the where the approximation of the test of the state હીંક દાક હત્વક ,

Bialling with dangert descend from dealering

5.03 The dial on the tablestan and is approximated in the tablest matter the tablest in the tablest matter at a spectra and the tablest matter with the tablest matter at the tablest tablest and the tablest tablest and the tablest tablest and tablest tablest and tablest tablest and tablest tablest at the tablest tablest at the tablest tablest at tablest at tablest and tablest tablest at tables

Page 4

during votors dialong. The (47) and (48) loods are connected for M(28))4 use. It the completion of dialong, the handoot corrupt is restared to the tasking, condition.

Disting with the Speakerphone Energiess

5.04 After the ON her an fag, 9 hes been

appearance and the signed tamp tighted, the dual to operated in the usual manner, orsupport that the handless is not differed from its mountaing. The off-morenal contacts connected to the PS and Pd tools about circuit the input to the receives angleties in the control unit. More a MCHTMM dust is used, the tames resonanted with each digit, are beard. The other pair of off-morenal contacts perform to uniful function at this time. In the completion of dusting, the entary dust about circuit is removed and the application circuit is reationed and the application circuit is re-

NUMBING From Nandoot to Renatorphore Operation

5.05 To sharpe to the spinoredisor struct when this, ry with the handson, the M ton of Tag. 2 is operated Operation of the M ton of Tag. 2 is operated Operation of the M ton operation the to since which the Tag. 3 entry when the handson is in the monttage tot this remain it is remember to beid the M ton the handson it is remember to beid the M ton the handson it is remember to beid the M ton the handson is a remember to be to be the to the the theritant is replaced. Then the handson is required through the basis contact of the word through is spinored by a period the hard of the word through the basis contacts of the word through the spin-tot the restance is required the spin-the operated by

Balifikang Yang Sangkangkana ku Kandani. Operataki

The release of the K1 relay opens the tip and ring at the speakerphone control unit, extinguighes the signal lamp in Fig. 2, and restores the speakerphone control unit to the idle condition. The operated switchlock assembly connects the talking circuit of the handset as described in 4.01.

Terminating a Conversation When Using the Speakerphone

5.07 Operation of the OFF key in Fig. 2 breaks the lockup circuit of the K1 relay which releases and restores the speaker-phone circuit to the idle condition.

6. DESCRIPTION OF TRANSISTORIZED CONTROL UN IT

6.01 The control unit is designed to work with connecting pairs up to 100 feet in length from the other units which connect to it. The NE-666 type transmitter contains a microphone, a volume control potentiometer, and the ON-OFF control buttons. It also contains a 3-stage amplifier so that the transmitter output will override noise interference which may be picked up in long connecting cords and cable pairs. The NE-630 series telephone sets incorporate an NE-667AN transmitter which plugs into the last module. The loudspeaker has a relatively high impedance voice coil to reduce the losses caused by the resistance of the cords and cable pairs.

CIRCUIT OPERATION (SEE FIGS 4 AND 8)

 6.02 The control unit consists of five plugin amplifiers which utilize transistors
 in the common emitter configuration. Various
 types of negative feedback are applied to
 stabilize ac gains against transistor purameter variations and to control both the input
 and output impedances. DC bias circuits are included for stabilizing operating currents and voltages.

6.03 The microphone amplifier, AM, consists of three stages designed to work between a 300-ohm transmitter unit and a 4.000-ohm transformer load impedance. A shunt feedback connection is used on the output in order to make the output impedance low. This reduces noise interference on long cord and cable pairs between the transmitter and the subscriber set and also reduces the power supply hum being passed to the control vario-losser. A shunt feedback connection is used on the input for the convenience of hawing one side of the microphone terminals grounded. Me resulting low input impedance is built up by means of a 300-ohm resistor to match the nominal microphone impedance and to equalize its response characteristic.

6.04 The receive amplifier, AR, also has three stages and was designed to work between a 5,000-ohm transformer secondary and a 27-ohm loudspeaker load. To provide a power output of 100 milliwatts without requiring a heat sink, an NE-16A diffused silicon NPN transistor is used for the output in place of the NE-12 type used in the other amplifier stages. Shunt feedback produces a low output impedance so that the low frequency loudspeaker reasonance will be dampened. Series feedback is applied to the input to increase the input impedance so that a shunt resistor on the transformer secondary can effectively control the impedance on the primary side which terminates the receive vario-losser.

6.05 The transmit amplifier, AT, consists of two stages designed to work from a

5,000-ohm transformer secondary to a hybrid coil winding which presents a 4,000-ohm load impedance and was a 5,000-ohm resistor shunted

across it. Series feedback on the output raises the output impedance to such a high value that the 5,000-ohm resistor effectively terminates the hybrid coil winding. The input also employs series feedback so that a 5,000-olm resistor across the secondary of the input transformer effectively terminates it for the transmit vario-looser.

6.06 The switchguard amplifier, 4G, has one transistor employing cautter degencration, causing both input and cutput impedance to be high. Since the input is bridged across the loudspeaker, the source impedance is very much smaller than the input impedance of the amplifier and the gain is nearly independent of transistor parameters. The load impedance is the primary of a 2,000- to 10,000-ohm output transformer which connects to a full wave rectifier and the load of the time constant circuit plus the control vario-losser. Since this is essentially a bridging amplifier, the power gain can be taken as the ratio of power into its 2,000-ohm load impedance versus the power into the 27-ohm loudspraker impedance, which is 10 dB.

6.07 The transmit control and noise threshold amplifiers. AC and AN, are similar in configuration and consists of two stages, each with shunt feedback on the outputs and series feedback on the inputs. The high input impedance of the control amplifier, shunted by the base biasing resistors of the first stage, provides a stable load impedance for the control vario-losser. In the noise threshold amplifier, the high input impedance is desirable is order to minimize the bridging effect on the output of AM.

7. VARIO-LOSSERS

7.01 The transmit varue-losser, TVL, is a single, balanced, series type. Together

with its input pad, it has an insertion loss of 500 ohms impedance and 69 dB when no control current is flowing. This drops to 33 dB at currents of 1 ma or higher. The pad keeps the speech levels low enough so that nonlinearity up to the overload point of AM is held to 1 dB-or less for all values of dc control current.

7.02 The receive vario-losser, RML, is a 2-

stage balanced shunt type. At zero control current its insertion loss, including its input pad, between 10,000-ohm impedances is 21 dB. At 1 ma and above, its loss increases to 63 dB. The pad keeps non-linearity within 1 dB up to the over-load point of AR for all values of control current.

7.03 The control vario-losser, CML, is essentially a voltage divider having one

variable resistor. The voltage insertion loss between the output of AM and the input of AC is 11 dB with no direct current coming from the switchguard rectifier. At maximum current of 1.5 ma, the insertion loss is 54 dB.

8. POWER RELAY

8.01 An NE-2012B plug-in transformer or a CT. 20-22B Wulti-Tap transformer supplies

18 volts at full load to the control unit. The low-voltage alternating current supplied to the subscriber set is rectified by a full wave germanium diode bridge and filtered with a pi-shaped LC network. The inductance of the filter is provided by the 10-okm winding of the power control KI relay. A typica? power line voltage of 117 volts nets a dc output of 15 volts at 100 ma. Of this current, 35 mm is used to light the 10-volt OA lamp through a voltage-dropping resistor and 65 mm is supplied to the electronic circuits.



Fig. 5 -NE-667AN Transmitter Unit



Fig. 6 NE-668QO Transmitter



Fig, 7- Station Busy Lamp Control Circuit





Figure 6- Station Busy Lamp Control CKT





Figure 7- Plug-in Power Transform NE-20120



Figure 8- Clock Diagram of Control Unit

CHAPTER 2

NE-3 TYPE SPEAKERPHONE SYSTEM

CONTROL UNITS

AND TRANSMITTER UNITS

CONNECTIONS

1. CERERAL

runpennets in the W-1 type speakerphone - the 20019 seriou, 47980 cm.

1.02 For connections to the various to lephone sets see the other WP's on the 20230 冰炉管白炉油

1.03 For information on the W-3 type openious phone assteau and in conjunction with 1.01 This procedure shows connections between the Ausineus Interphone Nortes, are Mile's in



Fig. 1- Modifications of NE-55A Control Unit for Use with DIGI TONE Dial Telephone Set.



NETE 7. UNE FOR SIG. C. OR SUMMERSION IN THE RETAIN ON AL TRUMPING SETS THAT ARE DALLED FROM 46 'OR HEAT'S CONSTITUT.

Fig. 2- NE584 Control Unit



Page 3



	90998 (I	fin televisi (never) (never) Andreniste (nev.
	460AE 8-	dae fo fie correas. Instanting, fiel 40-dolar femelikaring antikari an 1980 fo forski fiel al-don goringa (dit?).
	4697E 3	00.007 USE 007 00000000 00000 00000 00000 10.00000 000 00 00 000 00000, 00 1. 2077 00.000 00 00 00 00 000 000 0000 0000
	400976 sl.	Aussistianais advises and in a cadas, and caracter Tables values and in an architecture (builds attention is notice
NIA Mi-AAN	16071E &	-IF ANDARED ADALARE IN ADSIDUE FROM THE COMP- REGISTION, COMPARED BY COMP TO TEMPORE ADD
Junes as	31 1998 - 19	complet to distribut bits textingul sets. They also classes to to the able discriming.
-)TO SPIC MORP	46976 3.	- 155 MAR BORG, GLASS GEPARTSSIDE WITH GEREF Dirb, MELEPARRE SETS Dati dati Gerefe film Mar 166 1860; GRADITICA.
	46976.48	COMPLET LOCA COORS
1	489A 0	(1999) () () () () () () () () () () () () ()
10		
10		
10 10, 90 1000 11	ф и	diana ang sur o tunip ana tuk
40) 40(et n et n	dicines gauge also de famile ausau fade dicines danss s gaude
	به روی ۱۰ داده ۱۰ والوی	idi lanka anuka 4517 di 490146 alikub 40.06 idi lanka alikisi (di lanka) inisi aliki (di lanka) alimitiki
	фа ала фе	idi cantos galagos filta for familio acianos fondos idi idades fondes fonden ididadas (idi - i da canta i da andaris
	به ۱۹ ۱۹ ۱۹ ۵۵ ۱۹	idi cantes gauges filser dr Panifier andras padas idi cantes fondiscos galada antika, kilo-i gananto-idi addats
	به ۹۹ ۱۹ ۵۵ ۱۹ ۵۵ ۱۹ ۹۹ ۹	idi canto galge filo e de antes antes antes di cintes duassos galada antes 400+1 anto 10+1 ab addas
	n (*) n da n da n (*) n (*) n (*)	Dicinico galega (ilio) di Panifer all'ale filma Dicinico dialisi (ilio) Dicine, alle i alectris (ilio) Dicine, alle i alectris (ilio)
	н (†) н Ца э фр () на са	Branco golgo (ilia) & fantie alian fantie Branco golgo (ilian) Branc (ilian alian) do 10005
	н (9 н Ал а ар а ар а ар а ар а ар а ар а ар а а	Branco golgo (1947 & 4907), albao 4006 Branco golgo (1948) Branc (1944 albai), albailt
40) 400-55 - 41 400-55 - 41 400-55 - 41 400-55 - 41 400-55 - 41 400-45 - 45 400-45 400-45 - 45 400-450-45 400-4500-45	н Ф а (р а) а) а) а) а) а) а) а) а) а) а) а) а)	(DIGNED GROUP (FDIP & PONTE) GIBAD PONE (DIGNES DISSIS (FORE) (DIGNE (DISSIS) (DISSI)
40, 3 41, 3 <t< td=""><td>н фо и ла фо от фо от фо от от фо от от от от от от от от от от от от от</td><td>(Dicanto gauge (ibir & famile ama dafe (Dicante) datisti (dicate) (dicate) datisti (dicate) (dicate) datisti (dicate)</td></t<>	н фо и ла фо от фо от фо от от фо от от от от от от от от от от от от от	(Dicanto gauge (ibir & famile ama dafe (Dicante) datisti (dicate) (dicate) datisti (dicate) (dicate) datisti (dicate)
Φ0 40 4 400-10 5 5 6 500 7 5 8 500 9 500 9 500 9 500 9 500 9 500 9 500 9 500 9 500 100 500 100 500 100 500 100 500 100 500 100 500 100 500 100 500 100 500	ар (Ф) ад н ар (Ф) ар (Ф) ар (Ф) н н	() () () () () () () () () () () () () (
100 100 <td>99 44 483 8 399 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 49 49 49 49 49 49 49 49 49 49 49 49 49</td> <td>(B) (2014): Gold (B) (G) (B) (G) (G) (G) (G) (G) (G) (G) (G) (G) (G</td>	99 44 483 8 399 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 499 9 49 49 49 49 49 49 49 49 49 49 49 49 49	(B) (2014): Gold (B) (G) (B) (G) (G) (G) (G) (G) (G) (G) (G) (G) (G

8.02 The plug-in transformer. the rectifier, and the filter are connected at all

times to the ac power. Closing the ON button in the transmitter completes the dc **power sup**ply circuit between the relay winding and the second filter capacitor in.pallel with the load. The circuit locks up through the relay make contact, the OFF button break contact of the transmitter, ^{and} a switch-hook assembly break contact in the telephone act.

 REASON FOR REISSUE: To add information on NE-3B speakerphone system: To correct code numbers.



The Color Codes in Brackets are for the TLQB Cord in Brackets

Figure 2- Key and Transmitter Circuit



Figure 3- NE-567AN Transmitter Unit



Fig. 8- Auxiliary Relay Circuitfor Use With NE-758A or NE-755A PBX

Page 9 9 Pages



APPENDIX A

REFERENCES

Following is a list of	references applicable to the AN/FTC-37(V)1 and the AN/FTC-37(V)2.
AR 55-38	REPORTING OF Transportation Discrepancies in Shipments.
AR 380-5	Department of the Amy Information Security Program.
AR 700-58	Reporting of Packaging and Handling Deficiencies.
DA PAM 310-4	Index of Technical Manuals, Technical Bulletins, Supply Manuals (Types 7, 8, and 9), Supply Bulletins, and Lubrication Orders.
DA Pam 310-7	US Army Equipment Index of Modification Work Orders.
MIL-P-116	Preservation, Packaging, Methods of.
SB 38-100	Preservation, Packaging, Packing, and Marking Materiels, Supplies, and Equipment Used by the Army.
TB SIG 291	Safety Measures To Be Observed When Installing and Using Whip Antennas, Field Type Masts, Towers, Antennas, and Metal Poles That Are Used with Communication, Radar, and Direction Finder Equip- ment.
TM 11-5805-652-24P	Organizational, Direct Support and General Support Maintenance Re- pair Parts and Special Tools Lists (Including Depot Maintenance Re- pair Parts and Special Tools) for Central Office, Telephone, Electronic AN/FTC-37(V)1 (FSN 5805-155-8029), AN/FTC-37(V)2 (FSN 5805- 134-8419), and Associated Equipment.
TM 38-750	The Army Maintenance Management System (TAMMS).
TM 740-90-1	Administrative Storage of Equipment.

APPENDIX B MAINTENANCE ALLOCATION

Section I. INTRODUCTION

B-1. General

This appendix provides a summary of the maintenance operations covered in the equipment literature. It authorizes categories of maintenance for specific maintenance functions on repairable items and components and the tools and equipment required to perform each function. This appendix may be used as an aid in planning maintenance operations.

B-2. Maintenance Functions

Maintenance functions will be limited to and defined as follows:

a. Inspect. To determine the serviceability of an item by comparing its physical, mechanical, and/or electrical characteristics with established standards through examination.

b. Test. To verify serviceability and to detect incipient failure by measuring the mechanical or electrical characteristics of an item and comparing those characteristics with prescribed standards.

c. Service. Operations required periodically to keep an item in proper operating condition, i.e., to clean, preserve, drain, paint, or to replenish fuel/lubricants/hydraulic fluids or compressed air supplies. d. Adjust. Maintain within prescribed limits

d. Adjust. Maintain within prescribed limits by bringing into proper or exact position, or by actting the operating characteristics to the specified parameters.

 Align. To adjust specified variable elements of an item to about optimum or desired performance.

f. Calibrate. To determine and cause corrections to be made or to be adjusted on instruments or test measuring and diagnostic equipment used in precision measurement. Consists of the comparison of two instruments, one of which is a certified standard of known accuracy, to detect and adjust any discrepancy in the accuracy of the instrument being compared.

g. Install. The act of emplacing, seating, or fizing into position an item, part, module (component or assembly) in a manner to allow the proper functioning of the equipment/system.

h. Replace. The act of substituting a serviceable like-type part, subassembly, module (component or assembly) in a manner to allow the proper functioning of an equipment/system.

i. Repair. The application of maintenance services (inspect, test, service, adjust, align, calibrate, replace) or other maintenance actions (welding, grinding, riveting, straightening, facing, remachining, or resurfacing) to restore serviceability to an item by correcting specific damage, fault, malfunction, or failure in a part, subassembly, module/component/assembly, end item or system.

j. Overhaul. That maintenance effort (service/action) necessary to restore an item to a completely serviceable/operational condition as prescribed by maintenance standards (e.g., DMWR) in pertinent technical manuals. Overhaul is normally the highest degree of maintenance performed by the Army. Overhaul does not normally return an item to like-new condition.

k. Rebuild. Consists of those services/actions necessary for the restoration of unserviceable equipment to a like-new condition in accordance with original manufacturing standards. Rebuild is the highest degree of materiel maintenance applied to Army equipment. The rebuild operation includes the act of returning to zero those age measurements (hours, miles, etc.) considered in classifying Army equipment/ components.

l. Symbols. The uppercase letter placed in the appropriate column indicates the lowest level at which that particular maintenance function is to be performed.

B-3. Explanation of Format

a. Group Number. Column 1 lists group numbers, the purpose of which is to match components, assemblies, subassemblies and modules with the next higher assembly. b. Functional *Group*. Column 2 lists the next higher assembly group and the item names of components, assemblies, subassemblies and modules within the group for which maintenance is authorized.

c. Maintenance Functions. Column 3 lists the twelve maintenance functions defined in B-2 above. Each maintenance function required for an item is specified by the symbol among those listed in d below which indicates the level responsible for the required maintenance. Under this symbol is listed an appropriate work measurement time value determined as indicated in e below.

d. Use of Symbols. The following symbols are used to prescribe work function res**ponsibility:**

C-Operator/Crew O-Organization F-Direct Support H-General Support D-Depot

e. Work Measurement Time. The active repair time required to perform the maintenance function is included directly below the symbol identifying the category of maintenance. The skill levels used to obtain the measurement times approximate those found in typical TOE units. Active repair time is the average aggregate time required to restore an item (subassembly, assembly, component, module, end item or system) to a serviceable condition under typical field operating conditions. This time includes preparation time, fault isolation/diagnostic time, and QA/QC time in addition to the time required to perform specific maintenance functions identified for the tasks authorized in the maintenance allocation chart. This time is expressed in man-hours and carried to one decimal place (tenths of hours).

f. Tools *and* Test Equipment. This column is used to specify, by code, those tools and test equipment required to perform the designated function.

g. Remarks. Self-explanatory.

B-4. Explanation of Format of Table 1, Tool and Test Equipment Requirements

The columns in Table I, Tool and Test Equipment Requirements, are as follows:

a. Tools and Equipment. The numbers in this column coincide with the numbers used in the tools and equipment column of the applicable tool for the maintenance function.

b. Maintenance Category. The codes in this column indicate the maintenance category normally allocated the facility.

c. Nomenclature. This column lists tools, test, and maintenance equipment required to perform the maintenance functions.

d. Federal Stock Number. This column lists the Federal stock number of the specific tool or test equipment.

e. Tool Number. Not used.

(i) GROUP	(2) FUNCTIONAL GROUP				M	AINTE	(3) NANCE	E FUN	CTION	s			(4) TOOLS AND	(5) REMARKS
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	REBUILD	EQUIPMENT	
А	AN/FGC-37(V)1 AND AN/FGC-37(V)2	0 1.5	н 2.0	0 0.5									1 2 thru 15 1	Replace fuses, lamps
A1	LINE CRT AND MATRIX SWITCHING GROUP OA 8721/FTC 37(V).	0			H 1.0					ң 5.5			2 thru 11 2 thru 15 1	
	OA_2658/FTC-37(V)		H 1.5	0 0.3	н 0.8								2 thru 15 1 2 thru 11	Replace Cures, lamps
A1a	CIRCUTT CARD ASSEMBLIES		н 0.5					н 9,1		н 4.1			2 thru 15 2 thru 8, 11 2	
A1b	PRI-SEC MATRIX ASSEMBLIES		H						н 0.1	₩ 1.0			2 2,3,4,11 2 thrs 8, 11	
								₩ 0,1	н 0.1	1°0 Jį			ر≩¤مو⊄و ج ¢	
Alc	TERTIARY QUARTERNARY MATRIX ASSEMBLIES		H 0.5					н о.1	₩ 0.1				2 there 8, 12 2 2	

SECTION II. MAINTENANCE ALLOCATION CHART

ROUP	(2) FUNCTIONAL GROUP				M	AINTE	NANCE	FUN	CTION	5			(4) TOOLS AND	(5) REMARKS
JMBER	Component Assembly Nomenclature	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	REBUILD	EQUIPMENT	
Ald	FUSE PAREL ASSEMBLY		H 0,2					H 0.1		H 1.0			2,3,4,11 2,3 2	
Ale	MODULE SUB-ASSEMBLY		H 0.5					H 0.5	н 0.1	H 0.4			2 2,3 2,3,4 2,3,4	
Alf	Power Supply Assemblies		н 0.5					H 0.7	H 0.7	H 1.3			₽ ₽ _₽ ֈ _ց ե ₽ _₽ յյրե ₽	
Alg	NEGISTER ASSEMBLIES		н 0.5					11 0.1	H 1.0	8 2.0			2,3,4 2,3,4 2 Uhru 8, 11 2	
Alj	NESC. PANEL ASSENCETES		# 0.5	o 0.2				æ	0.1	1.0 1			ت \$ there \$, 11 \$	Nacjilianse Same

SECTION II. MAINTENANCE. ALLOCATION CHART

TM 11-5805-652-14

(I) GROUP	(2) FUNCTIONAL GROUP				м	AINTE	(3) NANC	E FUI	NCTIO	NS			(4) TOOL 5, AND	(5)
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	REBUILD	EQUIPMENT	REMARKS
Ali	SUB-RACK ASSEMBLIES	н 0.3						н 0.5	н 0.5 н 0.7	н 1.2 н			2 2,3,4 1 2 2 2,3,4	
A1j	CABINET ASSEMULTES	0 0.3		0 0.3						1.3			1	
A2	LINE CRT AND SWITCHING GROUP QA -8659/FTC -37(V)	0 1.0	H 1.5	0 0.3						к 0.5			1 1 2 thru 15 1	Replace Tures, Langue
A2a	CIRCUIT CARD ASSEMBLIES		н 0.5		н 0.8			18 0.1	R	H 4.1			2 thru 11 2 thru 15 2 thru 8, 11 2 2	
A2b	Pomer Supply Assembling		18 0.5					н 0.3	0.1	R 1.0			2 tànu ê, 11 2,3,4 2	

SECTION II. MAINTENANCE ALLOCATION CHART

B-5

-

(i) GROUP	(2) FUNCTIONAL GROUP				M		(3) NANCE	FUNC	TIONS	6			(4) TOOLS AND	(5) REMARKS
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	צבפחורם	EQUIPMENT	
A2c	MISC PANEL ASSEMBLY		Н 0.5	0 0.2				н 0.3	н 0.5	н 1.2			2 2,3,4 2,3,4 1 2	Replace fuses, lamps
A2d	SUB-RACK ASSEMBLIES	н 0.3						н 0.5	о.5 н 0.7	н 1.2			2,34 1 2 2	
A3	POWER SUPPLY GROUP OP -83/FTC -37(V)	0 U.5	н 0.5	0 0.3						H 1.3			?,3,4 1 2 thru 8 1	Replace fuses, lamp
A3a	MESC PANEL ASSEMBLY		н 0.5	0	н 0.3			Н 0.3	H 0.5	H 1.8 H 1.2			2 thru 8 2 thru 8 2,3,4 1 2 2 2,3,4	Replace fuses

SECTION II. MAINTENANCE ALLOCATION CHART

088A-PE 1935-72

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $								(3)							
$A = \frac{1}{1} $	(I) GROUP	(2) FUNCTIONAL GROUP				<u></u>	AINTE	NANCE	FUN	CTION	5			(4) TOOLS AND	(5) REMARKS
A 4 POMER SUPPLY GROUP 0.5 H 0.5 H 0.5 H 0.3 H 0.3 H 1 2 thru 8 Replace fuses, lamps A4 MESC PAREL ASSBULY 0.5 H 0.5 H 0.2 H 0.3 H 1.8 2 thru 8 2 thru 8 2 thru 8 A5 CONTROL-MORTTOR C-9263/PTC 0.3 H 0.2 I I H 1.8 2 thru 8 2 thru 8 2 thru 8 2 thru 8 2 thru 8 2 thru 8 2 thru 9	NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	REBUILD	EQUIPMENT	
A4aRESC PAREL ASSEMBLY $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 3 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 \\ 0 & 2 \\ 0 & 2 \\ 0 & 2 \end{bmatrix}$ $\begin{bmatrix} 1 & 3 \\ 0 & 2 $	A 4	Power Supply Group op -71/FTC -37(V)	0 0.5	н 0.5	0									1 2 thru 8 1	Replace fuses, lamps
A5 CONTROL-MONITOR C-9263/PTC 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 ,	A4a	MISC PAREL ASSENDLY		H	0.3	н 0.3					н 1.8			2 thru 8 2 thru 8	
A5 CONTROL-MORITOR C-9263/FTC $0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\$				0.5	0.2 0				¥ 0.3	н				1 2 2	Replace fuse, lamps
A5a REY ASSEMBLIES $\left(\begin{array}{c} 0\\ 0.2\\ 0.2\\ 0.2\\ 0.2\\ 0.2\\ 0.2\\ 0.2\\ 0$	A5	CONTROL-MONITOR C-9263/FTC	0 0.3	н 0.5						0.5	H 1.2			2,3,4 1 2,3,4,9,11	
A5b CIRCUIT CARE ASSEMBLY H 0.5 H 0.5 H 0.5 H 1.0 H 1.	A5a	KEY ASSIGNLIES			0.2					н 1.5	н 1.5			ا 2 2,3,4,9,11	Heplace lamps/caps
A5b CIRCUIT CARL ASSEMBLY H 0.5					0.2				н 0.3	H 0.5	н			5"3"7"11 5 5	Skylace ladja
N 0.1	A5b	CIRCUIT CARE ASSEMBLY		н 0.5					и 0.1		1.0			? thana ô, li 2	

SECTION II. MAINTENANCE ALLOCATION CHART

(I) GROUP	(2) FUNCTIONAL GROUP				м	AINTE	(3) NA NCE	FUN	CTION	5			(4) TOOLS AND	(5) REMARKS
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	BULLD	EQUIPMENT	
A5c	PANEL ASSEMBLY		н 0.3						н 0.1	Н 1.0			2 2 thra e, .: 2,3	
A6	CONTROL MONITOR C-8958/FTC	0						н оц	Н 0.4	н с.8			2 233	
		0.3	н 0.5	0 0.`					H 1.5	H			2,3,4,9,11 1 2 2,3,4,9,11	Seplace lamps/caps
Аба	CIRCUIT CARD ASSEMBLIES		н 0.5					н 0.1	H 0.1				2 thra 8, 11 2 2 2 thra 8, 11	
B B1	ASSOCTATED EQUIPT CONTROL UNIT ASSEMBLY	0.3	H 0,8							: . 0			s Again fr	
B1a	CINCULF CARD ASSEMBLIES		н 0.5							∦1 ≆,≶			.> there t≻ .> there ê, tt	

SECTION II. MAINTENANCE ALLOCATION CHART

T M 1 1 - 5 8 0 5 - 8 5 2 - 1 4

(I) GROUP	(2) FUNCTIONAL GROUP				M	AINTE	(J) NANCI	FUN	CTION	s			(4) TOOLS AND	(5) REMARKS
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	REBUILD	EQUIPMENT	
1								H 0.1	н 0,1				5 5	
В2	TRANSMITTER		н							н 1.0			2 thru 8, 11 2,3,4,11	
B2a	MENTINE CAED ASSUMETS		0.3							н 0.6			2,3,4,11	
DZa	CINCUL GAR ASSERDLY		" 0.5					н 0.3	н				2,3,4,11 2 2	
									0.5	н 1-3			2,3,4,11	
В3	KEY SERVICE UNITS		H 0.5							н 0.8			2,3,12 thru 15 2,3	
B3a	POWER SUPPLY		н 0.3	0									2,3,4 1	Seplacy Subes
				v.)				н 0.3	N 0.5				2	
B3b	KEY "ELEPHONE UNITS		R		1				-	1.0			2,3,6 2,3,6	
			0.2					н 0.1	N 0.1				2 2	
									v.,	м 1.0			2 thru 8, 11	

SECTION II. MAINTENANCE ALLOCATION CHART

B - 9

(i) GROUP	(2) FUNCTIONAL GROUP				M	INTE	(3) NANCE	FUNC	TION	3			(4) TOOLS AND	(5) REMARKS
NUMBER	COMPONENT ASSEMBLY NOMENCLATURE	INSPECT	TEST	SERVICE	ADJUST	ALIGN	CALIBRATE	INSTALL	REPLACE	REPAIR	OVERHAUL	ULIUSIA	EQUIPMENT	
B4	LOUDSPEAKER	0 0.3	H 0.3										1 2,3,11	
B5	TRANSFORMER, FOWER (STEP-DOWN)		H 0.3					H 0.1	м				2,3	
B6	TELEPHONE SET, 11 GET		H 0.3						0.1	н 0.5			2,3,4,11 2,3,4,11	
		n an an an an an an an an an an an an an												

SECTION II. MAINTENANCE ALLOCATION CHART

TABLE I. TOOL AND TEST EQUIPMENT REQUIREMENTS

TOOLS AND EQUIPMENT	MAINTENANCE CATEGORY	NOMENCLATURE	FEDERAL STOCK TOOL NUMBER NUMBER
1	0	TOOL KIT, ELECTRONIC EQUIPMENT TK-101/G	5180-064-5178
2	н	TOOL KIT, ELECTRONIC EQUIPMENT TK-105/G	5180-610-8177
3	H,D	MULTINETER AN/USM-210	6625-019-0815
4	H,D	OSCILLOSCOPE HP-1208	6625-860-2400
5	H,D	CARD EXTENDER, ITT NO. 130587-1	
6	H,D	CARD EXTENDER, ITT NO. 130587-2	
7	H,D	CARD EXTENDER, ITT NO. 130587-3	
8	H,D	CARD EXTENDER, ITT NO. 130680-1	
9	H,D	HAND TEST TELEPHONE, AUTOMATIC ELEC. L-9066-CF	
10	H,D	CORD AND PLUG ASSY, AUTOMATIC ELEC D-543142-A	
11	H,D	AUDIO FREQUENCY GENERATOR TS 4210	6625-669-0228
12	Ħ	WIRE WRAP TOOL, GARDNER-DENVER P/N 14H-1C	5120-978-3493
13	H	BIT, GARDMER-DENVER P/N 500131	5120-919-3478
14	H	SLEEVE, GARDWER-DENVER P/N 18840	5130-987-7057
15	H	WIRE UNWRAP TOOL, GARDNER DENVER P/N A-31478	5120-897-7518

B - 1 1

By Order of the Secretary of the Army:

Official:

VERNE L. BOWERS Major General, United States Army The Adjutant General

Distribution:

ì

Active Army CNGB (1) **TSG** (1) USAMB (2) USAARENBD (10) USASA (1) **TRADOC**(2)**AMC (1)** MICOM (1) TECOM (2) ABADCOM (2) ARADCOM Rgn (1) OS Maj Comd (2) USACC (2) HISA (Ft Monmouth) (18) Armies (1) USASESS (5) Svc College (1) NG: None USAR: None For explanation of abbreviations used, see AR 310-50. FRED C. WEYAND General, United States Army Acting Chief of Staff

Fort Huachuca (5) WSMR(1) Fort Carson (5) **USAERDAA**(1) **USAERDAW** (1) Army Dep (1) except LBAD (10) SAAD (15) TOAD (14) ATAD (10) USA Dep (1) Sig See USA Dep (2) Sig Dep (2) Sig FLDMS (1) Ft Richardson (ECOM Ofe) (1) Units org under fol TOE: 29-134 (1) 29-136 (1)

END

DATE 10-18-82




